UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-Q

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF X 1934

For the quarterly period ended September 30, 2021

OR

TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

> For the transition period from _____ to

> > Commission File No. 000-50028

WYNN RESORTS, LIMITED

(Exact name of registrant as specified in its charter)

Nevada

(State or other jurisdiction of incorporation or organization)

46-0484987 (I.R.S. Employer Identification No.)

3131 Las Vegas Boulevard South - Las Vegas, Nevada 89109 (Address of principal executive offices) (Zip Code)

(702) 770-7555

(Registrant's telephone number, including area code)

N/A

(Former name, former address and former fiscal year, if changed since last report)

Securities registered pursuant to Section 12(b) of the Act:

Title of each class	Trading Symbol(s)	Name of each exchange on which registered
Common stock, par value \$0.01	WYNN	Nasdaq Global Select Market

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days: Yes ⊠ No □

Indicate by check mark whether the registrant has submitted electronically every Interactive Data File required to be submitted pursuant to Rule 405 of Regulation S-T (§232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit such files). Yes ⊠ No □

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company, or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," "smaller reporting company," and "emerging growth Exchange company" in Rule 12b-2 of the Act. Large accelerated X Accelerated filer filer Non-accelerated filer Smaller reporting company

Emerging growth company

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. \Box

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes 🗆 No 🗵 Indicate the number of shares outstanding of each of the issuer's classes of common stock, as of the latest practicable date.

Class Outstanding at October 26, 2021

Common stock, par value \$0.01

115,657,896

WYNN RESORTS, LIMITED AND SUBSIDIARIES FORM 10-Q INDEX

Part I.	Financial Information	
<u>Item 1.</u>	Financial Statements	
	Condensed Consolidated Balance Sheets - September 30, 2021 (unaudited) and December 31, 2020	<u>3</u>
	Condensed Consolidated Statements of Operations (unaudited) - Three and Nine Months Ended September 30, 2021 and 2020	<u>4</u>
	<u>Condensed Consolidated Statements of Comprehensive Loss (unaudited) - Three and Nine Months Ended September</u> <u>30, 2021 and 2020</u>	<u>5</u>
	<u>Condensed Consolidated Statements of Stockholders' Equity (Deficit) (unaudited) - Three and Nine Months Ended</u> <u>September 30, 2021 and 2020</u>	<u>6</u>
	Condensed Consolidated Statements of Cash Flows (unaudited) - Nine Months Ended September 30, 2021 and 2020	
	Notes to Condensed Consolidated Financial Statements (unaudited)	<u>8</u> 9
<u>Item 2.</u>	Management's Discussion and Analysis of Financial Condition and Results of Operations	<u>25</u>
<u>Item 3.</u>	Quantitative and Qualitative Disclosures About Market Risk	<u>48</u>
<u>Item 4.</u>	Controls and Procedures	<u>49</u>
Part II.	Other Information	
<u>Item 1.</u>	Legal Proceedings	<u>50</u>
<u>Item 1A.</u>	Risk Factors	<u>50</u>
<u>Item 2.</u>	Unregistered Sales of Equity Securities and Use of Proceeds	<u>50</u>
<u>Item 5.</u>	Other Information	<u>50</u>
<u>Item 6.</u>	<u>Exhibits</u>	<u>51</u>
<u>Signature</u>		<u>52</u>

Part I. FINANCIAL INFORMATION Item 1. Financial Statements

WYNN RESORTS, LIMITED AND SUBSIDIARIES CONDENSED CONSOLIDATED BALANCE SHEETS (in thousands, except share data)

		September 30, 2021		December 31, 2020
		(unaudited)		
ASSETS				
Current assets:				
Cash and cash equivalents	\$	2,475,757	\$	3,482,032
Restricted cash		3,618		_
Accounts receivable, net of allowance for credit losses of \$94,309 and \$100,329		221,933		200,158
Inventories		64,043		66,285
Prepaid expenses and other		90,535		64,672
Total current assets		2,855,886		3,813,147
Property and equipment, net		8,859,157		9,196,644
Restricted cash		3,779		4,352
Goodwill and intangible assets, net		286,124		278,195
Operating lease assets		378,344		398,594
Other assets		224,405		178,615
Total assets	\$	12,607,695	\$	13,869,547
LIABILITIES AND STOCKHOLDERS' DEFICIT				
Current liabilities:				
Accounts and construction payables	\$	154,663	\$	148,478
Customer deposits		466,004		646,856
Gaming taxes payable		51,908		66,346
Accrued compensation and benefits		180,504		126,846
Accrued interest		143,341		136,421
Current portion of long-term debt		50,000		596,408
Other accrued liabilities		240,172		159,533
Total current liabilities		1,286,592	-	1,880,888
Long-term debt		11,693,785		12,469,362
Long-term operating lease liabilities		117,156		123,124
Other long-term liabilities		102,751		133,490
Total liabilities		13,200,284		14,606,864
Commitments and contingencies (Note 15)				
Stockholders' deficit:				
Preferred stock, par value \$0.01; 40,000,000 shares authorized; zero shares issued and outstanding				_
Common stock, par value \$0.01; 400,000,000 shares authorized; 131,361,484 and 123,482,836 shares issued; 115,658,818 and 107,888,336 shares outstanding, respectively		1,314		1,235
Treasury stock, at cost; 15,702,666 and 15,594,500 shares, respectively		(1,433,535)		(1,422,531)
Additional paid-in capital		3,485,759		2,598,115
Accumulated other comprehensive income		4,119		3,604
Accumulated deficit		(2,110,895)		(1,532,420)
Total Wynn Resorts, Limited stockholders' deficit		(53,238)		(351,997
Noncontrolling interests		(539,351)		(385,320)
Total stockholders' deficit		(592,589)		(737,317)
Total liabilities and stockholders' deficit	\$	12,607,695	\$	13,869,547
	Ψ	12,007,000	Ψ	10,000,047

The accompanying notes are an integral part of these condensed consolidated financial statements.

WYNN RESORTS, LIMITED AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS (in thousands, except per share data) (unaudited)

		Three Mo Septer		Nine Mor Septer	
		2021	2020	2021	2020
Operating revenues:					
Casino	\$	496,264	\$ 201,872	\$ 1,615,228	\$ 782,074
Rooms		173,817	61,144	387,772	231,240
Food and beverage		217,501	76,586	435,152	250,007
Entertainment, retail and other		107,062	 30,850	 283,287	 146,545
Total operating revenues		994,644	 370,452	 2,721,439	1,409,866
Operating expenses:					
Casino		315,316	160,861	1,048,897	734,689
Rooms		52,100	35,940	136,187	139,787
Food and beverage		163,655	76,536	354,709	314,335
Entertainment, retail and other		156,490	13,370	310,871	75,823
General and administrative		197,350	160,896	574,669	547,305
Provision for credit losses		(347)	11,588	7,461	60,548
Pre-opening		1,333	877	5,455	5,614
Depreciation and amortization		177,110	183,486	545,538	541,498
Property charges and other		15,301	9,905	 26,569	 43,701
Total operating expenses		1,078,308	653,459	 3,010,356	 2,463,300
Operating loss		(83,664)	 (283,007)	 (288,917)	 (1,053,434)
Other income (expense):					
Interest income		507	2,033	2,131	13,969
Interest expense, net of amounts capitalized		(150,325)	(145,142)	(453,601)	(407,187)
Change in derivatives fair value		1,176	4,675	6,557	(14,279)
Loss on extinguishment of debt		(738)	(3,139)	(2,060)	(4,601)
Other		(11,784)	412	(17,324)	12,980
Other income (expense), net		(161,164)	(141,161)	 (464,297)	 (399,118)
Loss before income taxes		(244,828)	 (424,168)	 (753,214)	 (1,452,552)
Provision for income taxes		(1,155)	(407,365)	(2,345)	(564,103)
Net loss		(245,983)	 (831,533)	 (755,559)	 (2,016,655)
Less: net loss attributable to noncontrolling interests		79,734	73,391	176,963	218,912
Net loss attributable to Wynn Resorts, Limited	\$	(166,249)	\$ (758,142)	\$ (578,596)	\$ (1,797,743)
Basic and diluted net loss per common share:			 		
Net loss attributable to Wynn Resorts, Limited:					
Basic	\$	(1.45)	\$ (7.10)	\$ (5.10)	\$ (16.85)
Diluted	\$	(1.45)	(7.10)	(5.10)	(16.85)
Weighted average common shares outstanding:	-	(()	()	()
Basic		114,655	106,783	113,420	106,720
Diluted		114,655	106,783	113,420	106,720
Dividends declared per common share	\$		\$ 	\$ 	\$ 1.00

The accompanying notes are an integral part of these condensed consolidated financial statements.

WYNN RESORTS, LIMITED AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF COMPREHENSIVE LOSS (in thousands)

(unaudited)

	Three Mor Septen		Nine Mon Septem	
	2021	2020	 2021	2020
Net loss	\$ (245,983)	\$ (831,533)	\$ (755,559)	\$ (2,016,655)
Other comprehensive (loss) income:				
Foreign currency translation adjustments, before and after tax	(2,863)	14	879	1,082
Total comprehensive loss	(248,846)	(831,519)	(754,680)	(2,015,573)
Less: comprehensive loss attributable to noncontrolling interests	80,423	73,387	176,599	218,611
Comprehensive loss attributable to Wynn Resorts, Limited	\$ (168,423)	\$ (758,132)	\$ (578,081)	\$ (1,796,962)

The accompanying notes are an integral part of these condensed consolidated financial statements.

WYNN RESORTS, LIMITED AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF STOCKHOLDERS' EQUITY (DEFICIT) (in thousands, except share data) (unaudited)

					For t	he 🛛	Three Months End	ed S	September 30, 20	021				
	Common	stoc	k											
	Shares outstanding		Par value	Treasury stock	Additional paid-in capital		Accumulated other comprehensive income	1	Accumulated deficit		Total Wynn Resorts, Ltd. stockholders' equity (deficit)	Noncontrolling interests	st	Total ockholders' deficit
Balances, July 1, 2021	115,683,983	\$	1,313	\$ (1,427,094)	\$ 3,466,908	\$	6,293	\$	(1,944,668)	\$	102,752	\$ (456,591)	\$	(353,839)
Net loss	_		—	_	—		—		(166,249)		(166,249)	(79,734)		(245,983)
Currency translation adjustment	_		—	_	—		(2,174)		_		(2,174)	(689)		(2,863)
Issuance of restricted stock	48,435		1	_	(1)		_		_		—	_		_
Cancellation of restricted stock	(5,224)		_		—		—		22		22	4		26
Shares repurchased by the Company and held as treasury shares	(68,376)		_	(6,441)	_		_		_		(6,441)	_		(6,441)
Distribution to noncontrolling interest	_		—		—		—		_		_	(5,364)		(5,364)
Stock-based compensation	_		_		18,852		_		_		18,852	3,023		21,875
Balances, September 30, 2021	115,658,818	\$	1,314	\$ (1,433,535)	\$ 3,485,759	\$	4,119	\$	(2,110,895)	\$	(53,238)	\$ (539,351)	\$	(592,589)

					For t	he	Three Months End	led \$	September 30, 20)20				
-	Common	stoc	k											
	Shares outstanding		Par value	Treasury stock	Additional paid-in capital		Accumulated other comprehensive loss		Retained earnings (accumulated deficit)		Total Wynn Resorts, Ltd. stockholders' equity (deficit)		Noncontrolling interests	Total ockholders' iity (deficit)
Balances, July 1, 2020	107,869,865	\$	1,234	\$ (1,419,435)	\$ 2,543,718	5	\$ (908)	\$	(505,090)	\$	619,519	5	6 (345,321)	\$ 274,198
Net loss	_		_	_	_		_		(758,142)		(758,142)		(73,391)	(831,533)
Currency translation adjustment	—		—	—	—		10		—		10		4	14
Issuance of restricted stock	182,087		2	_	(3)		_		—		(1)		_	(1)
Cancellation of restricted stock	(164,633)		(2)	_	2		_		—		_		—	—
Shares repurchased by the Company and held as treasury shares	(15,021)		_	(1,100)	_		_		_		(1,100)		_	(1,100)
Cash dividends declared	_		—				_		285		285		15	300
Distribution to noncontrolling interest	_		_	_	_		—		_		—		(998)	(998)
Stock-based compensation	—		—		10,726		—		—		10,726		1,763	12,489
Balances, September 30, 2020	107,872,298	\$	1,234	\$ (1,420,535)	\$ 2,554,443	\$	\$ (898)	\$	(1,262,947)	\$	(128,703)	9	6 (417,928)	\$ (546,631)

The accompanying notes are an integral part of these condensed consolidated financial statements.

WYNN RESORTS, LIMITED AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF STOCKHOLDERS' EQUITY (DEFICIT) (continued) (in thousands, except share data) (unaudited)

		For the Nine Months Ended September 30, 2021														
	Common	stock														
	Shares outstanding	Par value	-	Treasury stock	1	Additional paid-in capital		Accumulated other comprehensive income	A	Accumulated deficit		Total Wynn Resorts, Ltd. stockholders' equity (deficit)		ontrolling terests	sto	Total ckholders' deficit
Balances, January 1, 2021	107,888,336	\$ 1,235	\$	(1,422,531)	\$	2,598,115	\$	3,604	\$	(1,532,420)	\$	(351,997)	\$	(385,320)	\$	(737,317)
Net loss	—	_		_		_		_		(578,596)		(578,596)		(176,963)		(755,559)
Currency translation adjustment	—	—		_		—		515		_		515		364		879
Issuance of common stock, net of \$17.7 million underwriter discounts, commissions and other expenses	7,475,000	75		_		841,821		_		_		841,896		_		841,896
Issuance of restricted stock	428,406	4		_		5,898		_		_		5,902		370		6,272
Cancellation of restricted stock	(24,758)	_		_		_		_		121		121		19		140
Shares repurchased by the Company and held as treasury shares	(108,166)	_		(11,004)		_		_		_		(11,004)		_		(11,004)
Distribution to noncontrolling interest	_	_		_		_		_		—		_		(11,843)		(11,843)
Subsidiary equity issuance	_	_		_		(20,211)		_		_		(20,211)		25,371		5,160
Stock-based compensation	_	_		_		60,136				—		60,136		8,651		68,787
Balances, September 30, 2021	115,658,818	\$ 1,314	\$	(1,433,535)	\$	3,485,759	\$	4,119	\$	(2,110,895)	\$	(53,238)	\$	(539,351)	\$	(592,589)

					Fort	the	e Nine Months Ende	ed S	eptember 30, 20	20			
	Common	stoc	k										
	Shares outstanding		Par value	Treasury stock	Additional paid-in capital		Accumulated other comprehensive loss	(Retained earnings (accumulated deficit)		Total Wynn Resorts, Ltd. stockholders' equity (deficit)	Noncontrolling interests	Total tockholders' juity (deficit)
Balances, January 1, 2020	107,363,943	\$	1,228	\$ (1,410,998)	\$ 2,512,676	5	\$ (1,679)	\$	641,818	\$	1,743,045	\$ (201,573)	\$ 1,541,472
Net loss	_		—	_	_		_		(1,797,743)		(1,797,743)	(218,912)	(2,016,655)
Currency translation adjustment	_		_	_			781		_		781	301	1,082
Issuance of restricted stock	843,102		8	_	6,700		_		—		6,708	818	7,526
Cancellation of restricted stock	(237,680)		(2)		2		—		_		—	—	—
Shares repurchased by the Company and held as treasury shares	(97,067)		_	(9,537)	_		_		_		(9,537)	141	(9,396)
Cash dividends declared	—		_				—		(107,022)		(107,022)	45	(106,977)
Distribution to noncontrolling interest	_		_	_	_		_		_		_	(1,996)	(1,996)
Stock-based compensation	—		—	—	35,065		_		—		35,065	3,248	38,313
Balances, September 30, 2020	107,872,298	\$	1,234	\$ (1,420,535)	\$ 2,554,443	5	\$ (898)	\$	(1,262,947)	\$	(128,703)	\$ (417,928)	\$ (546,631)

The accompanying notes are an integral part of these condensed consolidated financial statements.

WYNN RESORTS, LIMITED AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (in thousands) (unaudited)

	Nine Months E	Nine Months Ended September 30,							
	2021		2020						
Cash flows from operating activities:									
Net loss	\$ (755,559)\$	(2,016,655						
Adjustments to reconcile net loss to net cash used in operating activities:									
Depreciation and amortization	545,538		541,498						
Deferred income taxes	(527		562,262						
Stock-based compensation expense	75,033		40,50						
Amortization of debt issuance costs	19,891		21,770						
Loss on extinguishment of debt	2,060		4,60						
Provision for credit losses	7,461		60,548						
Change in derivatives fair value	(6,557)	14,27						
Property charges and other	43,893		30,72						
Increase (decrease) in cash from changes in:									
Receivables, net	(29,729)	105,879						
Inventories, prepaid expenses and other	(38,377)	15,92						
Customer deposits	(178,959)	45,852						
Accounts payable and accrued expenses	99,000		(208,258						
Net cash used in operating activities	(216,832	,	(781,079						
Cash flows from investing activities:		-	· · · ·						
Capital expenditures, net of construction payables and retention	(213,088)	(261,093						
Purchase of intangible and other assets	(19,741		_						
Proceeds from sale of assets and other	3,689		3,733						
Net cash used in investing activities	(229,140	- <u></u>	(257,360						
Cash flows from financing activities:			(-)						
Proceeds from issuance of long-term debt	1,141,026		3,919,372						
Repayments of long-term debt	(2,477,690		(1,540,609						
Proceeds from issuance of Wynn Resorts, Limited common stock	841,896		(_,_ ,_ ,, ,, ,, ,, ,, ,, ,, ,, ,, ,, ,,						
Repurchase of common stock	(11,004		(9,532						
Finance lease payments	(11,709		(2,194						
Proceeds from exercise of stock options	(11,705)	(2,15						
Dividends paid	(932		(108,282						
Distribution to noncontrolling interest	(11,843		(1,996						
Proceeds from issuance of subsidiary common stock	4,662		(1,550						
Payments for debt financing costs	(29,975		(20,350						
Net cash (used in) provided by financing activities			2,236,47						
	(555,569								
Effect of exchange rate on cash, cash equivalents and restricted cash	(1,689	<u> </u>	3,412						
Cash, cash equivalents and restricted cash:	(4,000,000		4 0.04 45						
(Decrease) increase in cash, cash equivalents and restricted cash	(1,003,230		1,201,45						
Balance, beginning of period	3,486,384		2,358,292						
Balance, end of period	\$ 2,483,154	\$	3,559,744						
Supplemental cash flow disclosures:									
Cash paid for interest, net of amounts capitalized	\$ 426,462	\$	313,69						
Liability settled with shares of common stock	\$ 6,272		6,72						
Accounts and construction payables related to property and equipment	\$ 70,844		85,58						
Other liabilities related to intangible assets	\$ 12,335		12,880						
Finance lease liabilities arising from obtaining finance lease assets	\$ 7,423		42,432						

The accompanying notes are an integral part of these condensed consolidated financial statements.

Note 1 - Organization

Organization

Wynn Resorts, Limited, a Nevada corporation (together with its subsidiaries, "Wynn Resorts" or the "Company"), is a designer, developer, and operator of integrated resorts featuring luxury hotel rooms, high-end retail space, an array of dining and entertainment options, meeting and convention facilities, and gaming.

In the Macau Special Administrative Region of the People's Republic of China ("Macau"), the Company owns approximately 72% of Wynn Macau, Limited ("WML"), which includes the operations of the Wynn Palace and Wynn Macau resorts. The Company refers to Wynn Palace and Wynn Macau as its Macau Operations. In Las Vegas, Nevada, the Company operates and, with the exception of certain retail space, owns 100% of Wynn Las Vegas. Additionally, the Company is a 50.1% owner and managing member of a joint venture that owns and leases certain retail space at Wynn Las Vegas (the "Retail Joint Venture"). The Company refers to Wynn Las Vegas, Encore, an expansion at Wynn Las Vegas, and the Retail Joint Venture as its Las Vegas Operations. On June 23, 2019, the Company opened Encore Boston Harbor, an integrated resort in Everett, Massachusetts, that is owned 100% by the Company. In October 2020, Wynn Interactive Ltd. ("Wynn Interactive") was formed through the merger of the Company's U.S. online sports betting and gaming business, social casino business, and Wynn Resorts' strategic partner, BetBull Limited ("BetBull"). Wynn Resorts holds an approximately 74% interest in, and consolidates, Wynn Interactive. The results of Wynn Interactive's operations are presented within Corporate and other in the accompanying condensed consolidated financial statements, except where otherwise noted.

Recent Developments Related to COVID-19

Since the outbreak of COVID-19 in early 2020, steps have been taken by various countries, including those in which the Company operates, to advise citizens to avoid non-essential travel, to restrict inbound international travel, to implement closures of non-essential operations, and to implement quarantines and lockdowns to contain the spread of the virus. As part of the immediate response to the initial outbreak of COVID-19, each of the Company's properties was subject to partial or full closure for varying lengths of time during 2020, and each has since reopened. Several vaccines have been granted authorization in numerous countries and are being rolled out to citizens based on availability and priority of need. There can be no assurance as to when a sufficient number of individuals will be vaccinated, or when travel restrictions may be fully lifted thereafter.

Macau Operations

Visitation to Macau has fallen significantly since the outbreak of COVID-19, driven by the strong deterrent effect of the COVID-19 pandemic on travel and social activities, quarantine measures put in place in Macau and elsewhere, travel and entry restrictions and conditions in Macau, the PRC, Hong Kong and Taiwan involving COVID-19 testing, among other things, and the suspension or reduced accessibility of transportation to and from Macau. Beginning in June 2020, certain restrictions and conditions have eased to allow for some visitation to Macau as certain regions recover from the COVID-19 pandemic. Quarantine-free travel, subject to COVID-19 safeguards such as testing and the usual visa requirements, has been reintroduced between Macau and most areas and cities within the PRC, and in September 2020, PRC authorities fully resumed the IVS exit visa program, which permits individual PRC citizens from nearly 50 PRC cities to travel to Macau for tourism purposes. Given the evolving conditions created by and in response to the COVID-19 pandemic, measures that have been lifted may be reintroduced if there are adverse developments in the COVID-19 situation in Macau and other regions with access to Macau, and the Company is currently unable to determine when protective measures and the suspension of certain offerings in effect at our Macau Operations will be lifted. Given the uncertainty around the extent and timing of the potential future spread or mitigation of COVID-19 and around the imposition or relaxation of protective measures, management cannot reasonably estimate the impact to the Company's future results of operations, cash flows, or financial condition.

Las Vegas Operations and Encore Boston Harbor

In response to the COVID-19 outbreak, the Company's Las Vegas Operations and Encore Boston Harbor each implemented certain COVID-19 specific protective measures, such as limiting the number of seats per table game, slot machine spacing, temperature checks, mask protection, and suspension of certain entertainment and nightlife offerings. Over the course of the nine months ended September 30, 2021, the Company's Las Vegas Operations and Encore Boston Harbor have each incrementally resumed full operations, including reopening gaming areas to 100% of capacity and restoring seven-day-per-week hotel operations, as permitted by governmental authorities and in response to increased customer demand. Given the

evolving conditions created by and in response to the COVID-19 pandemic, measures that have been lifted may be reintroduced if there are adverse developments in the COVID-19 situation, and management cannot reasonably estimate the impact of such developments to the Company's future results of operations, cash flows, or financial condition.

Liquidity

As of September 30, 2021, the Company had total cash and cash equivalents, excluding restricted cash, of \$2.48 billion, and had access to \$834.2 million of available borrowing capacity from the WRF Revolving Facility and \$413.1 million of available borrowing capacity from the WM Cayman II Revolver (as defined and discussed further in Note 7, "Long-Term Debt"). The Company has suspended its dividend program. Given the Company's liquidity position as of September 30, 2021, the Company believes it is able to support continuing operations and respond to the current COVID-19 pandemic challenges.

Macau Gaming Concession

The term of the Company's concession agreement with the Macau government ends on June 26, 2022. If the term of this concession agreement is not extended or renewed or is not replaced by a new gaming concession, all of the Company's gaming operations and related equipment in Macau will be automatically transferred to the Macau government without compensation on that date and the Company will cease to generate gaming revenues from its Macau Operations. In addition, under the indentures governing the Company's \$4.7 billion aggregate principal amount of WML Senior Notes and the facility agreement governing the WM Cayman II Revolver, upon the occurrence of any event after which the Company does not own or manage casino or gaming areas or operate casino games of fortune and chance in Macau in substantially the same manner as of the issue date of the respective senior notes or the date of the facility agreement, for a period of 10 consecutive days or more in the case of the WML Senior Notes or a period of 30 consecutive days or more in the case of the WML Cayman II Revolver, and such event has a material adverse effect on the financial condition, business, properties or results of operations of WML and its subsidiaries, taken as a whole, holders of the WML Senior Notes can require the Company to repurchase all or any part of the WML Senior Notes at par, plus any accrued and unpaid interest (the "Special Put Option"), and any amounts owed under the WM Cayman II Revolver may become immediately due and payable (the "Property Mandatory Prepayment Event").

The Company is monitoring developments with respect to the Macau government's concession renewal or extension process, and at this time believes that its concession will be renewed or extended beyond June 26, 2022. The failure to extend or renew the Company's concession or obtain a new concession and the resulting ability of the WML Senior Note holders to exercise the Special Put Option and triggering of the Property Mandatory Prepayment Event would have a material adverse effect on the Company's business, financial condition, results of operations, and cash flows.

Business Combination Agreement

On May 10, 2021, Wynn Interactive entered into a business combination agreement (the "Business Combination Agreement") with Austerlitz Acquisition Corporation I, a Cayman Islands exempted company ("Austerlitz I"), and Wave Merger Sub Limited, an exempted company limited by shares incorporated in Bermuda and a direct, wholly owned subsidiary of Austerlitz I ("Merger Sub"). The Business Combination Agreement provides for, among other things, the consummation of the following transactions: (i) Austerlitz I will transfer by way of continuation from the Cayman Islands to Bermuda and change its name to "Wynn Interactive, Limited"; and (ii) Merger Sub will merge with and into Wynn Interactive (the "Merger"), with Wynn Interactive being the surviving company of the Merger and direct, wholly owned subsidiary of Austerlitz I. Upon closing of the transaction, assuming no share redemptions by the public stockholders of Austerlitz I, the Company is expected to retain an approximately 58% equity interest (and approximately 72% voting interest) in Wynn Interactive. Closing of the proposed business combination is subject to approval by Austerlitz I's stockholders, gaming regulatory approval and other customary closing conditions.

Note 2 - Basis of Presentation and Significant Accounting Policies

Basis of Presentation

The accompanying condensed consolidated financial statements have been prepared by the Company pursuant to the rules and regulations of the Securities and Exchange Commission ("SEC"). Certain information and footnote disclosures normally included in financial statements prepared in accordance with U.S. generally accepted accounting principles ("GAAP") have been condensed or omitted pursuant to such rules and regulations, although the Company believes that the disclosures herein

are adequate to make the information presented not misleading. In the opinion of management, the accompanying condensed consolidated financial statements reflect all adjustments, which are of a normal recurring nature, necessary to a fair presentation of the results for the interim periods presented. The results for the three and nine months ended September 30, 2021 are not necessarily indicative of results to be expected for the full fiscal year. These condensed consolidated financial statements should be read in conjunction with the consolidated financial statements and notes thereto in the Company's Annual Report on Form 10-K for the year ended December 31, 2020.

Principles of Consolidation

The accompanying condensed consolidated financial statements include the accounts of the Company, its majority-owned subsidiaries and entities the Company identifies as variable interest entities ("VIEs") of which the Company is determined to be the primary beneficiary. For information on the Company's VIEs, see Note 16, "Retail Joint Venture." All significant intercompany accounts and transactions have been eliminated. Certain amounts in the condensed consolidated financial statements for the first quarter of 2021 have been reclassified to be consistent with the current quarter presentation. These reclassifications had no effect on the previously reported net loss or operating loss.

Use of Estimates

The preparation of consolidated financial statements in conformity with GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates. Significant estimates and assumptions reflected in the financial statements relate to and include, but are not limited to, inputs into the Company's estimated allowance for credit losses, estimates regarding the useful lives and recoverability of the cost of long-lived assets, fair value estimates of intangible assets and their estimated useful lives, and litigation and contingency estimates.

Gaming Taxes

The Company is subject to taxes based on gross gaming revenues in the jurisdictions in which it operates, subject to applicable jurisdictional adjustments. These gaming taxes are recorded as casino expenses in the accompanying Condensed Consolidated Statements of Operations. These taxes totaled \$188.7 million and \$66.1 million for the three months ended September 30, 2021 and 2020, respectively, and \$637.9 million and \$332.5 million for the nine months ended September 30, 2021 and 2020, respectively.

Recently Issued Accounting Standards

In March 2020, the FASB issued ASU No. 2020-04, "Reference Rate Reform (Topic 848): Facilitation of the Effects of Reference Rate Reform on Financial Reporting" ("ASU 2020-04"). ASU 2020-04 provides optional expedients and exceptions for applying GAAP to contracts, hedging relationships, and other transactions affected by reference rate reform if certain criteria are met. In response to the concerns about structural risks of interbank offered rates and, particularly, the risk of cessation of the London Interbank Offered Rate (referred to as "LIBOR"), regulators in several jurisdictions around the world have undertaken reference rate reform initiatives to identify alternative reference rates that are more observable or transaction-based and less susceptible to manipulation. ASU 2020-04 also provides companies with optional guidance to ease the potential accounting burden associated with transitioning away from reference rates that are expected to be discontinued. ASU 2020-04 can be adopted no later than December 1, 2022 with early adoption permitted. The Company is currently assessing the impact the adoption of the new guidance will have on its consolidated financial statements.



Note 3 - Cash, Cash Equivalents and Restricted Cash

Cash, cash equivalents and restricted cash consisted of the following (in thousands):

	S	eptember 30, 2021	December 31, 2020
Cash and cash equivalents:			
Cash (1)	\$	2,054,901	\$ 2,501,452
Cash equivalents (2)		420,856	980,580
Total cash and cash equivalents		2,475,757	 3,482,032
Restricted cash (3)		7,397	4,352
Total cash, cash equivalents and restricted cash	\$	2,483,154	\$ 3,486,384

(1) Cash consists of cash on hand and bank deposits.
 (2) Cash equivalents consist of bank time deposits and money market funds.
 (3) Restricted cash consists of cash subject to certain contractual restrictions, cash collateral associated with obligations and cash held in a trust in accordance with WML's share award plan.

Note 4 - Receivables, net

Accounts Receivable and Credit Risk

Receivables, net consisted of the following (in thousands):

	ptember 30, 021	ecember 31, 2020
Casino	\$ 217,788	\$ 207,823
Hotel	23,593	7,075
Other	74,861	85,589
	 316,242	300,487
Less: allowance for credit losses	(94,309)	(100,329)
	\$ 221,933	\$ 200,158

As of September 30, 2021 and December 31, 2020, approximately 72.8% and 77.3%, respectively, of the Company's markers were due from customers residing outside the United States, primarily in Asia. Business or economic conditions or other significant events in the countries in which the Company's customers reside could affect the collectability of such receivables.

The Company's allowance for casino credit losses was 42.2% and 47.2% of gross casino receivables as of September 30, 2021 and December 31, 2020, respectively. Although the Company believes that its allowance is adequate, it is possible the estimated amounts of cash collections with respect to receivables could change. The Company's allowance for credit losses from its hotel and other receivables is not material.

The following table shows the movement in the Company's allowance for credit losses recognized for receivables that occurred during the period (in thousands):

	Sej	otember 30, 2021	Sep	tember 30, 2020
Balance at beginning of year	\$	100,329	\$	39,317
Provision for credit losses		7,461		60,548
Write-offs		(14,022)		(1,233)
Recoveries of receivables previously written off		736		228
Effect of exchange rate		(195)		129
Balance at end of period	\$	94,309	\$	98,989



Note 5 - Property and Equipment, net

Property and equipment, net consisted of the following (in thousands):

	September 30, 2021	December 31, 2020
Buildings and improvements	\$ 9,794,863	\$ 9,758,846
Land and improvements	1,277,046	1,265,510
Furniture, fixtures and equipment	3,099,055	3,093,481
Airplanes	110,623	110,623
Construction in progress	194,912	136,390
	14,476,499	 14,364,850
Less: accumulated depreciation	(5,617,342)	(5,168,206)
	\$ 8,859,157	\$ 9,196,644

As of September 30, 2021 and December 31, 2020, construction in progress consisted primarily of costs capitalized for various capital enhancements at the Company's properties, including the Wynn Las Vegas room remodel.

Depreciation expense for the three months ended September 30, 2021 and 2020 was \$169.7 million and \$177.3 million, respectively, and depreciation expense for the nine months ended September 30, 2021 and 2020 was \$523.6 million and \$522.4 million, respectively.

Note 6 - Goodwill

The following table shows the movement in the Company's goodwill balance that occurred during the nine-month period (in thousands):

Balance as of January 1, 2021	\$ 144,094
Foreign currency translation	(4,167)
Balance as of September 30, 2021	\$ 139,927

Note 7 - Long-Term Debt

Long-term debt consisted of the following (in thousands):

	September 30, 2021	December 31, 2020
Macau Related:		
Wynn Macau Credit Facilities (1):		
Wynn Macau Term Loan, due 2022	\$ —	\$ 1,268,106
Wynn Macau Revolver, due 2022	_	407,443
WM Cayman II Revolver, due 2025 (2)	1,089,143	—
WML 4 7/8% Senior Notes, due 2024	600,000	600,000
WML 5 1/2% Senior Notes, due 2026	1,000,000	1,000,000
WML 5 1/2% Senior Notes, due 2027	750,000	750,000
WML 5 5/8% Senior Notes, due 2028	1,350,000	1,350,000
WML 5 1/8% Senior Notes, due 2029	1,000,000	1,000,000
U.S. and Corporate Related:		
WRF Credit Facilities (3):		
WRF Term Loan, due 2024	900,000	937,500
WRF Revolver, due 2024	_	716,000
WLV 4 1/4% Senior Notes, due 2023	500,000	500,000
WLV 5 1/2% Senior Notes, due 2025	1,780,000	1,780,000
WLV 5 1/4% Senior Notes, due 2027	880,000	880,000
WRF 7 3/4% Senior Notes, due 2025	600,000	600,000
WRF 5 1/8% Senior Notes, due 2029	750,000	750,000
Retail Term Loan, due 2025 (4)	615,000	615,000
	11,814,143	13,154,049
Less: Unamortized debt issuance costs and original issue discounts and premium, net	(70,358)	(88,279)
	11,743,785	13,065,770
Less: Current portion of long-term debt	(50,000)	(596,408)
Total long-term debt, net of current portion	\$ 11,693,785	\$ 12,469,362

(1) In September 2021, the Company prepaid the aggregate amount of \$1.26 billion of borrowings outstanding under Wynn Macau Credit Facilities.

(2) The borrowings under the WM Cayman II Revolver bear interest at LIBOR or HIBOR plus a margin of 1.875% to 2.875% per annum based on WM Cayman II's leverage ratio on a consolidated basis. Approximately \$226.5 million and \$862.6 million of the WM Cayman II Revolver bears interest at a rate of LIBOR plus 2.625% per year and HIBOR plus 2.625% per year, respectively. As of September 30, 2021, the weighted average interest rate was approximately 2.69%. As of September 30, 2021, the available borrowing capacity under the WM Cayman II Revolver was \$413.1 million.

(3) The WRF Credit Facilities bear interest at a rate of LIBOR plus 1.75% per year. As of September 30, 2021, the weighted average interest rate was approximately 1.84%. Additionally, as of September 30, 2021, the available borrowing capacity under the WRF Revolver was \$834.2 million, net of \$15.8 million in outstanding letters of credit.

(4) The Retail Term Loan bears interest at a rate of LIBOR plus 1.70% per year. As of September 30, 2021, the effective interest rate was 2.70%.

WM Cayman II Revolver

On September 16, 2021, WM Cayman Holdings Limited II, an indirect wholly owned subsidiary of WML, as borrower ("WM Cayman II") and WML as guarantor, each an indirect subsidiary of Wynn Resorts, entered into a facility agreement with, among others, Bank of China Limited, Macau Branch as agent and a syndicate of lenders (the "Facility Agreement"), pursuant to which the lenders will make available in an aggregate amount of \$1.50 billion equivalent revolving unsecured credit facility consisting of one tranche in an amount of \$312.5 million and one tranche in an amount of HK\$9.26 billion (approximately \$1.19 billion) to WM Cayman II (the "WM Cayman II Revolver"). WM Cayman II has the ability to upsize the total WM Cayman II Revolver by an additional \$1.00 billion equivalent under the Facility Agreement and related agreements upon the satisfaction of various conditions.



The final maturity of all outstanding loans under the WM Cayman II Revolver is September 16, 2025 (or if September 16, 2025 is not a business day, the next business day in the relevant calendar month), by which time any outstanding borrowings from the WM Cayman II Revolver must be repaid.

Borrowings of \$1.09 billion under the WM Cayman II Revolver, along with \$200.0 million of cash, were used to facilitate the prepayment of the outstanding \$1.26 billion of borrowings under the Wynn Macau Credit Facilities, and to pay related fees and expenses totaling \$29.1 million, of which \$28.5 million was recorded as debt issuance costs within the Condensed Consolidated Balance Sheet. The Company recognized this transaction primarily as a modification of existing debt with the related unamortized debt issuance costs reallocated to the WM Cayman II Revolver. For those components of debt that were deemed extinguished, the Company recognized a loss on extinguishment of debt of \$0.7 million.

The Facility Agreement contains representations, warranties, covenants and events of default customary for similar financings. The Facility Agreement also contains certain mandatory prepayment provisions relating to the loss or termination of the Company's gaming operations or concession contracts in Macau.

Debt Covenant Compliance

As of September 30, 2021, management believes the Company was in compliance with all debt covenants.

Fair Value of Long-Term Debt

The estimated fair value of the Company's long-term debt as of September 30, 2021 and December 31, 2020, was approximately \$11.68 billion and \$13.35 billion, respectively, compared to its carrying value, excluding debt issuance costs and original issue discount and premium, of \$11.81 billion and \$13.15 billion, respectively. The estimated fair value of the Company's long-term debt is based on recent trades, if available, and indicative pricing from market information (Level 2 inputs).

Note 8 - Stockholders' Deficit

Equity Offering

On February 11, 2021, the Company completed a registered public offering of 7,475,000 newly issued shares of its common stock, par value \$0.01 per share, at a price of \$115.00 per share for proceeds of \$841.9 million, net of \$17.7 million in underwriting discounts, commissions, and other expenses. The Company used \$716.0 million of the proceeds from the equity offering to repay the then outstanding borrowings under the WRF Revolver, and used the remaining net proceeds for general corporate purposes.

Dividends

During the first quarter of 2020, the Company paid a cash dividend of \$1.00 per share, and recorded \$107.5 million as a reduction of retained earnings from cash dividends declared.

On May 6, 2020, the Company announced that it had suspended its quarterly dividend program due to the financial impact of the COVID-19 pandemic.

Noncontrolling Interests

On April 16, 2021, Wynn Interactive issued a pre-emptive rights notice to its shareholders in connection with the proposed creation and issuance of new Class A shares. Upon the consummation of the share issuance in May 2021, Wynn Interactive issued 3,229 new Class A shares to noncontrolling interest holders in exchange for aggregate proceeds of \$4.7 million.

The WML board of directors concluded not to recommend the payment of a dividend with respect to either of the years ended December 31, 2020 or 2019 due to the financial impact of the COVID-19 pandemic. As such, WML paid no dividends during 2020 or the nine months ended September 30, 2021.

During the nine months ended September 30, 2021 and 2020, the Retail Joint Venture made aggregate distributions of approximately \$11.8 million and \$2.0 million, respectively, to its noncontrolling interest holder. For more information on the Retail Joint Venture, see Note 16, "Retail Joint Venture".

Note 9 - Fair Value Measurements

The following tables present assets and liabilities carried at fair value (in thousands):

			Fair Value Measurements Using:				
	Se	ptember 30, 2021		Quoted Market Prices in Active Markets (Level 1)		Other Observable Inputs (Level 2)	Unobservable Inputs (Level 3)
Assets:							
Cash equivalents	\$	420,856	\$	—	\$	420,856	—
Restricted cash	\$	7,397	\$	5,672	\$	1,725	—
Liabilities:							
Interest rate collar	\$	10,351		_	\$	10,351	—
			Fair Value Measurements Using:				
				Fai	r Val	ue Measurements Usir	ıg:
	D	ecember 31, 2020		Fai Quoted Market Prices in Active Markets (Level 1)	r Val	ue Measurements Usir Other Observable Inputs (Level 2)	ng: Unobservable Inputs (Level 3)
Assets:	D	ecember 31, 2020		Quoted Market Prices in Active Markets	r Val	Other Observable Inputs	Unobservable Inputs
Assets: Cash equivalents	D \$	ecember 31, 2020 980,580	\$	Quoted Market Prices in Active Markets	r Val	Other Observable Inputs	Unobservable Inputs
		2020	\$ \$	Quoted Market Prices in Active Markets (Level 1)		Other Observable Inputs (Level 2)	Unobservable Inputs
Cash equivalents Restricted cash	\$	2020 980,580		Quoted Market Prices in Active Markets (Level 1) 504,980	\$	Other Observable Inputs (Level 2) 475,600	Unobservable Inputs
Cash equivalents	\$	2020 980,580		Quoted Market Prices in Active Markets (Level 1) 504,980	\$	Other Observable Inputs (Level 2) 475,600	Unobservable Inputs

Note 10 - Customer Contract Liabilities

In providing goods and services to its customers, there is often a timing difference between the Company receiving cash and the Company recording revenue for providing services or holding events.



The Company's primary liabilities associated with customer contracts are as follows (in thousands):

S	September 30, 2021	December 31, 2020In	crease / (decrease)S	eptember 30, 2020	December 31, 2019In	crease / (decrease)
asino outstanding chips and front money deposits \$1) 380,442\$	596,463\$	(216,021\$)	836,370\$	769,053\$	67,317
dvance room deposits and ticket sales (2)	62,993	29,224	33,769	28,885	49,834	(20,949)
ther gaming-related liabilities (3)	21,217	7,882	13,335	6,212	13,970	(7,758)
oyalty program and related liabilities (4)	33,440	22,736	10,704	23,362	21,148	2,214
\$	498,092\$	656,305	(158,213)	894,829	854,005	40,824

(1) Casino outstanding chips generally represent amounts owed to gaming promoters and customers for chips in their possession, and casino front money deposits represent funds deposited by customers before gaming play occurs. These amounts are included in customer deposits on the Condensed Consolidated Balance Sheets and may be recognized as revenue or redeemed for cash in the future.

(2) Advance room deposits and ticket sales represent cash received in advance for goods or services to be provided in the future. These amounts are included in customer deposits on the Condensed Consolidated Balance Sheets and will be recognized as revenue when the goods or services are provided or the events are held. Decreases in this balance generally represent the recognition of revenue and increases in the balance represent additional deposits made by customers. The deposits are expected to primarily be recognized as revenue within one year.

(3) Other gaming-related liabilities generally represent unpaid wagers primarily in the form of unredeemed slot, race and sportsbook tickets or wagers for future sporting events. The amounts are included in other accrued liabilities on the Condensed Consolidated Balance Sheets.

(4) Loyalty program and related liabilities represent the deferral of revenue until the loyalty points or other complimentaries are redeemed. The amounts are included in other accrued liabilities on the Condensed Consolidated Balance Sheets and are expected to be recognized as revenue within one year of being earned by customers.

Note 11 - Stock-Based Compensation

The total compensation cost for stock-based compensation plans was recorded as follows (in thousands):

	Three Months Ended September 30,			Nine Months Ended September 30,			
		2021		2020	2021		2020
Casino (1)	\$	3,423	\$	2,496	\$ 11,000	\$	4,796
Rooms		351		403	1,260		1,098
Food and beverage		737		868	2,709		2,171
Entertainment, retail and other (2)		7,071		102	16,292		259
General and administrative (3)		13,998		6,184	43,772		32,177
Total stock-based compensation expense		25,580		10,053	75,033		40,501
Total stock-based compensation capitalized		2,242		659	4,271		1,362
Total stock-based compensation costs	\$	27,822	\$	10,712	\$ 79,304	\$	41,863

(1) For the nine months ended September 30, 2020, reflects the reversal of \$3.3 million of compensation cost previously recognized for awards forfeited in connection with the departure of an employee.

(2) For the nine months ended September 30, 2021, reflects compensation cost of \$2.7 million recognized in connection with the vesting of restricted stock performance awards. (3) For the nine months ended September 30, 2020, reflects compensation cost of \$4.4 million recognized in connection with the vesting of restricted stock performance awards.

Note 12 - Income Taxes

The Company recorded an income tax expense of \$1.2 million and \$407.4 million for the three months ended September 30, 2021 and 2020, respectively and an income tax expense of \$2.3 million and \$564.1 million for the nine months ended September 30, 2021 and 2020, respectively. The 2021 income tax expense primarily related to an increase in valuation allowance for U.S. deferred tax assets and to the Macau dividend tax agreement that provides for an annual payment of MOP 12.8 million (approximately \$1.6 million) as complementary tax otherwise due by stockholders of Wynn Macau SA. The 2020 income tax expense primarily related to the increase in the valuation allowance for U.S. foreign tax credits.

In March 2021, the Company received an extension of its Macau dividend tax agreement, providing for a payment of MOP 12.8 million (approximately \$1.6 million) for 2021 and MOP 6.3 million (approximately \$0.8 million) for the period ending June 26, 2022, the expiration date of the gaming concession agreement.

The Company records valuation allowances on certain of its U.S. and foreign deferred tax assets. During the third quarter of 2020, the Company concluded it could no longer rely on forecasted future taxable income in assessing a valuation allowance on its deferred tax assets. This conclusion was reached due to cumulative operating losses incurred by the Company and tax legislation that reduced future sources of taxable income. As of September 30, 2021, the Company continues to rely solely on the reversal of net taxable temporary differences in assessing a need for a valuation allowance.

In April 2020, Wynn Macau SA received an extension of the exemption from Macau's 12% Complementary Tax on casino gaming profits earned from January 1, 2021 to June 26, 2022, the expiration date of the gaming concession agreement.

For the three and nine months ended September 30, 2021 and 2020, the Company did not have any casino gaming profits exempt from the Macau Complementary Tax. The Company's non-gaming profits remain subject to the Macau Complementary Tax and its casino winnings remain subject to the Macau special gaming tax and other levies in accordance with its concession agreement.

In March 2021, the Financial Services Bureau concluded its review of the 2017 and 2018 Macau income tax returns of Palo with no changes.

Note 13 - Earnings Per Share

Basic earnings per share ("EPS") is computed by dividing net loss attributable to Wynn Resorts by the weighted average number of common shares outstanding during the period. Diluted EPS is computed by dividing net loss attributable to Wynn Resorts by the weighted average number of common shares outstanding during the period increased to include the number of additional shares of common stock that would have been outstanding if the potential dilutive securities had been issued, to the extent such impact is not anti-dilutive. Potentially dilutive securities include outstanding stock options and unvested restricted stock.

The weighted average number of common and common equivalent shares used in the calculation of basic and diluted EPS consisted of the following (in thousands, except per share amounts):

	Three Months Ended September 30,			Nine Months Ended September 30,			
	2	021		2020	 2021		2020
Numerator:							
Net loss attributable to Wynn Resorts, Limited	\$	(166,249)	\$	(758,142)	\$ (578,596)	\$	(1,797,743)
Denominator:							
Weighted average common shares outstanding		114,655		106,783	113,420		106,720
Potential dilutive effect of stock options, nonvested, and performance nonvested shares		_		_	_		_
Weighted average common and common equivalent shares outstanding		114,655		106,783	 113,420		106,720
Net loss attributable to Wynn Resorts, Limited per common share, basic	\$	(1.45)	\$	(7.10)	\$ (5.10)	\$	(16.85)
Net loss attributable to Wynn Resorts, Limited per common share, diluted	\$	(1.45)	\$	(7.10)	\$ (5.10)	\$	(16.85)
Anti-dilutive stock options, nonvested, and performance nonvested shares excluded from the calculation of diluted net income per share		954		1,104	 954		1,104



Note 14 - Leases

Lessor Arrangements

The following table presents the minimum and contingent operating lease income for the periods presented (in thousands):

	Three Months Ended September 30,			Nine Months Ended September 30,			
	 2021		2020		2021		2020
Minimum rental income (1)	\$ 27,719	\$	(2,430)	\$	75,676	\$	57,293
Contingent rental income	19,349		15,280		75,464		23,798
Total rental income	\$ 47,068	\$	12,850	\$	151,140	\$	81,091

(1) For the three and nine months ended September 30, 2020, reflects the impact of rent concessions provided to tenants.

Note 15 - Commitments and Contingencies

Litigation

In addition to the actions noted below, the Company and its affiliates are involved in litigation arising in the normal course of business. In the opinion of management, such litigation is not expected to have a material effect on the Company's financial condition, results of operations, and cash flows.

Massachusetts Gaming License Related Actions

On September 17, 2014, the Massachusetts Gaming Commission ("MGC") designated Wynn MA the award winner of the Greater Boston (Region A) gaming license (the "Boston area license"). On November 7, 2014, the gaming license became effective.

Revere Action

On October 16, 2014, the City of Revere, the host community to the unsuccessful bidder for the Boston area license, the International Brotherhood of Electrical Workers, Local 103, and several individuals, filed a complaint against the MGC and its gaming commissioners in Suffolk Superior Court in Boston, Massachusetts (the "Revere Action"). Mohegan Sun ("Mohegan"), the other applicant for the Boston area license, joined the lawsuit and challenged the MGC's award of the Boston area license. On December 3, 2015, the court granted the MGC's motion to dismiss the claims asserted in the Revere Action and the court dismissed all claims except Mohegan's claim alleging procedural error by the MGC in granting the license to Wynn MA. The plaintiffs appealed. After multiple appeals and cross appeals, only two claims remained: (1) individual plaintiffs' claim for violation of the open meeting laws; and (2) Mohegan's claim for procedural error. On July 12, 2019, the Suffolk Superior Court granted the MGC's motion for summary judgment and dismissed the open meeting law claim, leaving only Mohegan's procedural claim for procedural error.

On August 2, 2019, Mohegan filed a motion to file a second amended complaint, to add new claims related to the MGC's allegedly inadequate 2013 investigation. On October 15, 2019, the court granted Mohegan's motion to amend and allowed it to file a second amended intervenor's complaint.

Wynn MA is not a party to and is not named in the Revere Action.

Derivative Litigation

A number of stockholder derivative actions were filed in state and federal court located in Clark County, Nevada against certain current and former members of the Company's Board of Directors and, in some cases, the Company's current and former officers. Each of the complaints alleged, among other things, breach of fiduciary duties in failing to detect, prevent and remedy alleged inappropriate personal conduct by Stephen A. Wynn in the workplace.

The actions filed in the Eighth Judicial District Court of Clark County, Nevada were consolidated as *In re Wynn Resorts*, *Ltd. Derivative Litigation* ("State Derivative Case").

On June 3, 2019, a separate stockholder derivative action was filed in the Eighth Judicial District Court of Clark County, Nevada alleging substantially similar causes of action as the State Derivative Case with the additional allegation that various of the Company's attorneys committed professional malpractice, and certain current and former executives also breached fiduciary duties and aided and abetted the breach of fiduciary duties, in connection with the alleged inappropriate personal conduct by Stephen A. Wynn in the workplace. This case was consolidated in September 2019 into the State Derivative Case.

On November 27, 2019, the State Derivative Case parties agreed to terms of a settlement agreement. The court approved the settlement agreement on February 12, 2020, and entered a written order approving the settlement on March 10, 2020. Following the Nevada Supreme Court's dismissal of the only appeal, the settlement agreement became effective and final. Following the dismissal, the Company received net proceeds of \$30.2 million, which has been recognized as a reduction of general and administrative expense within the Condensed Consolidated Statements of Operations for the nine months ended September 30, 2020.

In 2018, several actions filed in the United States District Court, District of Nevada were consolidated as *In re Wynn Resorts, Ltd. Derivative Litigation* ("Federal Derivative Case"), which also claim corporate waste and violation of Section 14(a) of the Exchange Act. In June 2018, the Company filed a motion to dismiss and a motion to stay pending resolution of the Securities Action (described below). On March 29, 2019, the Court granted the Company's request for a stay. On March 25, 2020, the parties stipulated to dismiss the Federal Derivative Case given the approved settlement in the State Derivative Case.

On March 25, 2019, a separate stockholder derivative action was filed in the United States District Court, District of Nevada alleging similar causes of action as the Federal Derivative Case with the additional allegation that the Board of Directors improperly refused the stockholder's demand to commence litigation against the officers and directors of the Company. On April 30, 2020, the Company filed a motion for summary judgment, seeking dismissal of the claims given the approved settlement in the State Derivative Case. On January 12, 2021, the court granted the Company's motion for summary judgment of this action and denied the stockholder's request to vacate the parties' stipulation to dismiss the Federal Derivative Case. On February 11, 2021, the stockholder filed a notice of appeal to the United States Court of Appeals for the Ninth Circuit. On May 12, 2021, the parties stipulated to dismiss the appeal.

Each of the actions sought to recover for the Company unspecified damages, including restitution and disgorgement of profits, and also sought to recover attorneys' fees, costs and related expenses for the plaintiff.

Securities Action

On February 20, 2018, a putative securities class action was filed against the Company and certain current and former officers of the Company in the United States District Court, Southern District of New York (which was subsequently transferred to the United States District Court, District of Newada) by John V. Ferris and Joann M. Ferris on behalf of all persons who purchased the Company's common stock between February 28, 2014 and January 25, 2018. The complaint alleges, among other things, certain violations of federal securities laws and seeks to recover unspecified damages as well as attorneys' fees, costs and related expenses for the plaintiffs. On April 15, 2019, the Company filed a motion to dismiss, which the court granted on May 27, 2020, with leave to amend. On July 1, 2020, the plaintiffs filed an amended complaint. On August 14, 2020, the Company filed a motion to dismiss the amended complaint. On July 28, 2021, the court granted in part, and denied in part, the Company's motion to dismiss the amended complaint, dismissing certain of plaintiffs' claims, including all claims against Mr. Billings and the individual directors, and allowing other claims to proceed against the Company and several of the Company's current and former executive officers, including Mr. Maddox, Stephen A. Wynn, Kimmarie Sinatra, and Steven Cootey.

The defendants in these actions will vigorously defend against the claims pleaded against them. These actions are in preliminary stages and management has determined that based on proceedings to date, it is currently unable to determine the probability of the outcome of these actions or the range of reasonably possible loss, if any.

Federal Investigation

From time to time, the Company receives regulatory inquiries about compliance with anti-money laundering laws. The Company received requests for information from the U.S. Attorney's Office for the Southern District of California relating to its anti-money laundering policies and procedures, and in the first half of 2020, received two grand jury subpoenas regarding various transactions at Wynn Las Vegas relating to certain patrons and agents who reside or operate in foreign jurisdictions. The Company continues to cooperate with the U.S. Attorney's Office in its investigation, which remains ongoing. Because no charges or claims have been brought, the Company is unable to predict the outcome of the investigation, the extent of the materiality of the outcome, or reasonably estimate the possible range of loss, if any, which could be associated with the resolution of any possible charges or claims that may be brought against the Company.

Note 16 - Retail Joint Venture

As of September 30, 2021 and December 31, 2020, the Retail Joint Venture had total assets of \$98.2 million and \$96.3 million, respectively, and total liabilities of \$626.6 million and \$633.5 million, respectively. As of September 30, 2021 and December 31, 2020, the Retail Joint Venture's liabilities included long-term debt of \$612.8 million and \$612.3 million, respectively, net of debt issuance costs, related to the outstanding borrowings under the Retail Term Loan.

Note 17 - Segment Information

The Company reviews the results of operations for each of its operating segments, and identifies reportable segments based upon factors such as geography, regulatory environment, and the Company's organizational and management reporting structure. Wynn Macau and Encore, an expansion at Wynn Macau, are managed as a single integrated resort and have been aggregated as one reportable segment ("Wynn Macau"). Wynn Palace is presented as a separate reportable segment and is combined with Wynn Macau for geographical presentation. Other Macau primarily represents the assets for the Company's Macau holding company. Wynn Las Vegas, Encore, an expansion at Wynn Las Vegas, and the Retail Joint Venture are managed as a single integrated as one reportable segment ("Las Vegas Operations"). Encore Boston Harbor is presented as one reportable segment. The results of Wynn Interactive are presented within Corporate and other.

The following tables present the Company's segment information (in thousands):

		Three Mor Septen	nths End iber 30,	led				ths Ended ber 30,	
		2021		2020		2021		2020	
Operating revenues									
Macau Operations:									
Wynn Palace									
Casino	\$	134,064	\$	12,301	\$	532,040	\$	208,449	
Rooms		15,639		4,506		53,534		26,647	
Food and beverage		10,952		6,856		36,429		24,385	
Entertainment, retail and other (1)		20,668		(7,962)		67,017		24,451	
		181,323		15,701		689,020		283,932	
Wynn Macau									
Casino		98,264		27,154		379,610		213,758	
Rooms		10,896		4,938		39,025		23,480	
Food and beverage		7,628		5,606		23,620		18,821	
Entertainment, retail and other (1)		13,874		13,670		52,086		36,686	
		130,662		51,368		494,341		292,745	
Total Macau Operations		311,985		67,069		1,183,361		576,677	
Las Vegas Operations:									
		112,575		65,694		305,253		161,354	
Casino Rooms		112,575		44,961		266,250		161,354	
Food and beverage		132,704		44,901 55,043		333,390		103,419	
Entertainment, retail and other (1)		50,269		20,999		104,892		73,520	
Total Las Vegas Operations		476,003		186,697		1,009,785		575,407	
Encore Boston Harbor:									
Casino		151,361		96,723		398,325		198,513	
Rooms		14,578		6,739		28,963		17,694	
Food and beverage		18,466		9,081		41,713		29,687	
Entertainment, retail and other (1)		7,809		4,143		18,544		11,888	
Total Encore Boston Harbor		192,214		116,686		487,545		257,782	
Corporate and other:									
Entertainment, retail and other		14,442				40,748			
Total Corporate and other		14,442		—		40,748		_	
Total operating revenues	\$	994,644	\$	370,452	\$	2,721,439	\$	1,409,866	
ivia operating revenues	Ψ	557,044	Ψ	570,-52	Ψ	2,721,700	Ψ	1,705,000	

		Three Months Ended September 30,		Nine Months E Septembe	
		2021	2020	2021	2020
djusted Property EBITDA (2)					
Macau Operations:					
Wynn Palace	\$	12,112\$	(77,64 75)	93,03 \$	(178,379)
Wynn Macau		(1,939)	(34,457)	28,703	(97,895)
Total Macau Operations		10,173	(112,104)	121,739	(276,274)
Las Vegas Operations		183,416	20,258	344,719	(77,383)
Encore Boston Harbor		64,565	25,986	141,844	(40,429)
Corporate and other		(103,593)	—	(187,961)	—
Total	_	154,561	(65,860)	420,341	(394,086)
ther operating expenses					
Pre-opening		1,333	877	5,455	5,614
Depreciation and amortization		177,110	183,486	545,538	541,498
Property charges and other		15,301	9,905	26,569	43,701
Corporate expenses and other (3)		18,901	12,826	56,663	28,034
Stock-based compensation		25,580	10,053	75,033	40,501
Total other operating expenses		238,225	217,147	709,258	659,348
perating loss		(83,664)	(283,007)	(288,917)	(1,053,434)
ther non-operating income and expenses	_				i
Interest income		507	2,033	2,131	13,969
Interest expense, net of amounts capitalized		(150,325)	(145,142)	(453,601)	(407,187)
Change in derivatives fair value		1,176	4,675	6,557	(14,279)
Loss on extinguishment of debt		(738)	(3,139)	(2,060)	(4,601)
Other		(11,784)	412	(17,324)	12,980
Total other non-operating income and expenses	_	(161,164)	(141,161)	(464,297)	(399,118)
oss before income taxes		(244,828)	(424,168)	(753,214)	(1,452,552)
Provision for income taxes		(1,155)	(407,365)	(2,345)	(564,103)
et loss	_	(245,983)	(831,533)	(755,559)	(2,016,655)
Net loss attributable to noncontrolling interests		79,734	73,391	176,963	218,912
et loss attributable to Wynn Resorts, Limited	\$	(166,249)	(758,142)	(578,596)	(1,797,743)

(1) Includes lease revenue accounted for under lease accounting guidance. For more information on leases, see Note 14, "Leases".

(2) "Adjusted Property EBITDA" is net loss before interest, income taxes, depreciation and amortization, pre-opening expenses, property charges and other, management and license fees, corporate expenses and other (including intercompany golf course, meeting and convention, and water rights leases), stock-based compensation, change in derivatives fair value, loss on extinguishment of debt, and other non-operating income and expenses. Adjusted Property EBITDA is presented exclusively as a supplemental disclosure because management believes that it is widely used to measure the performance, and as a basis for valuation, of gaming companies. Management uses Adjusted Property EBITDA as a measure of the operating performance of its segments and to compare the operating performance of its properties with those of its competitors, as well as a basis for determining certain incentive compensation. We also present Adjusted Property EBITDA because it is used by some investors to measure a company's ability to incur and service debt, make capital expenditures and meet working capital requirements. Gaming companies have historically reported EBITDA as a supplement to GAAP. In order to view the operations of their casinos on a more stand-alone basis, gaming companies, including us, have historically excluded from their EBITDA calculations preopening expenses, property charges, corporate expenses and stock-based compensation, that do not relate to the management of specific casino properties. However, Adjusted Property EBITDA should not be considered as an alternative to operating income as an indicator of our performance, as an alternative to cash flows from operating activities as a measure of liquidity, or as an alternative to any other measure determined in accordance with GAAP. Unlike net income, Adjusted Property EBITDA does not include depreciation or interest expense and therefore does not reflect current or future capital expenditures or the cost of capital. We have significant uses of cash flows, including capital expenditures, interest payments, debt principal repayments, income taxes and other non-recurring charges, which are not reflected in Adjusted Property EBITDA. Also, our calculation of Adjusted Property EBITDA may be different from the calculation methods used by other companies and, therefore, comparability may be limited. (3) For the nine months ended September 30, 2020, included \$30.2 million net gain recorded in relation to a derivative litigation settlement.

	September 30, 2021	December 31, 2020
ssets		
Iacau Operations:		
Wynn Palace	\$ 3,178,848\$	3,393,790
Wynn Macau	905,443	1,202,709
Other Macau	1,240,994	2,026,098
Total Macau Operations	5,325,285	6,622,597
as Vegas Operations	3,038,127	2,992,870
ncore Boston Harbor	2,236,837	2,300,016
orporate and other	2,007,446	1,954,064
Total	\$ 12,607,695	13,869,547

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

The following discussion should be read in conjunction with, and is qualified in its entirety by, the condensed consolidated financial statements and the notes thereto included elsewhere in this Form 10-Q and the consolidated financial statements appearing in our annual report on Form 10-K for the year ended December 31, 2020. Unless the context otherwise requires, all references herein to the "Company," "we," "us," or "our," or similar terms, refer to Wynn Resorts, Limited, a Nevada corporation, and its consolidated subsidiaries. This discussion and analysis contains forward-looking statements. Please refer to the section below entitled "Forward-Looking Statements."

Overview

We are a designer, developer, and operator of integrated resorts featuring luxury hotel rooms, high-end retail space, an array of dining and entertainment options, meeting and convention facilities, and gaming, all supported by an unparalleled focus on our guests, our people, and our community. Through our approximately 72% ownership of Wynn Macau, Limited ("WML"), we operate two integrated resorts in the Macau Special Administrative Region of the People's Republic of China ("Macau"), Wynn Palace and Wynn Macau (collectively, our "Macau Operations"). In Las Vegas, Nevada, we operate and, with the exception of certain retail space, own 100% of Wynn Las Vegas. Additionally, we are a 50.1% owner and managing member of a joint venture that owns and leases certain retail space at Wynn Las Vegas (the "Retail Joint Venture"). We refer to Wynn Las Vegas, Encore, an expansion at Wynn Las Vegas, and the Retail Joint Venture as our Las Vegas Operations. On June 23, 2019, we opened Encore Boston Harbor, an integrated resort in Everett, Massachusetts. In October 2020, Wynn Interactive Ltd. ("Wynn Interactive") was formed through the merger of our U.S. online sports betting and gaming business, social casino business, and our strategic partner, BetBull Limited ("BetBull"). Wynn Resorts holds an approximately 74% interest in, and consolidates, Wynn Interactive. The results of Wynn Interactive are presented within Corporate and other.

Recent Developments Related to COVID-19

Since the outbreak of COVID-19 in early 2020, steps have been taken by various countries, including those in which the Company operates, to advise citizens to avoid non-essential travel, to restrict inbound international travel, to implement closures of non-essential operations, and to implement quarantines and lockdowns to contain the spread of the virus. As part of the immediate response to the initial outbreak of COVID-19, each of the Company's properties was subject to partial or full closure for varying lengths of time during 2020. Several vaccines have been granted authorization in numerous countries and are being rolled out to citizens based on availability and priority of need. There can be no assurance as to when a sufficient number of individuals will be vaccinated, or when travel restrictions may be fully lifted thereafter.

Macau Operations

Visitation to Macau has fallen significantly since the outbreak of COVID-19, driven by the strong deterrent effect of the COVID-19 pandemic on travel and social activities, quarantine measures put in place in Macau and elsewhere, travel and entry restrictions and conditions in Macau, the PRC, Hong Kong and Taiwan involving COVID-19 testing, among other things, and the suspension or reduced accessibility of transportation to and from Macau. Beginning in June 2020, certain restrictions and conditions have eased to allow for visitation to Macau as certain regions recover from the COVID-19 pandemic. Quarantine-free travel, subject to COVID-19 safeguards such as testing and the usual visa requirements, has been reintroduced between Macau and most areas and cities within the PRC, and in September 2020, PRC authorities fully resumed the IVS exit visa program, which permits individual PRC citizens from nearly 50 PRC cities to travel to Macau for tourism purposes. Total visitation from PRC to Macau increased meaningfully in the nine months ended September 30, 2021 compared to the same period in 2020. Total visitation decreased 75.6% compared to the same period in 2019. Given the evolving conditions created by and in response to the COVID-19 pandemic, measures that have been lifted may be reintroduced if there are adverse developments in the COVID-19 situation in Macau and other regions with access to Macau, and the Company is currently unable to determine when protective measures and the suspension of certain offerings in effect at our Macau Operations will be lifted. Given the uncertainty around the extent and timing of the potential future spread or mitigation of COVID-19 and around the imposition or relaxation of protective measures, management cannot reasonably estimate the impact to the Company's future results of operations, cash flows, or financial condition.



Las Vegas Operations and Encore Boston Harbor

In response to the COVID-19 outbreak, the Company's Las Vegas Operations and Encore Boston Harbor each implemented certain COVID-19 specific protective measures, such as limiting the number of seats per table game, slot machine spacing, temperature checks, mask protection, and suspension of certain entertainment and nightlife offerings. Over the course of the nine months ended September 30, 2021, the Company's Las Vegas Operations and Encore Boston Harbor have each incrementally resumed full operations, including reopening gaming areas to 100% of capacity and restoring seven-day-per-week hotel operations, as permitted by governmental authorities and in response to increased customer demand. Given the evolving conditions created by and in response to the COVID-19 pandemic, measures that have been lifted may be reintroduced if there are adverse developments in the COVID-19 situation, and management cannot reasonably estimate the impact of such developments to the Company's future results of operations, cash flows, or financial condition.

Macau Gaming Concession

The term of the Company's concession agreement with the Macau government ends on June 26, 2022. If the term of this concession agreement is not extended or renewed or is not replaced by a new gaming concession, all of the Company's gaming operations and related equipment in Macau will be automatically transferred to the Macau government without compensation on that date and the Company will cease to generate gaming revenues from its Macau Operations. In addition, under the indentures governing the Company's \$4.7 billion aggregate principal amount of WML Senior Notes and the facility agreement governing the WM Cayman II Revolver, upon the occurrence of any event after which the Company does not own or manage casino or gaming areas or operate casino games of fortune and chance in Macau in substantially the same manner as of the issue date of the respective senior notes or the date of the facility agreement, for a period of 10 consecutive days or more in the case of the WML Senior Notes or a period of 30 consecutive days or more in the case of the WML Cayman II Revolver, and such event has a material adverse effect on the financial condition, business, properties or results of operations of WML and its subsidiaries, taken as a whole, holders of the WML Senior Notes can require the Company to repurchase all or any part of the WML Senior Notes at par, plus any accrued and unpaid interest (the "Special Put Option"), and any amounts owed under the WM Cayman II Revolver may become immediately due and payable (the "Property Mandatory Prepayment Event").

The Company is monitoring developments with respect to the Macau government's concession renewal or extension process, and at this time believes that its concession will be renewed or extended beyond June 26, 2022. The failure to extend or renew the Company's concession or obtain a new concession and the resulting ability of the WML Senior Note holders to exercise the Special Put Option and triggering of the Property Mandatory Prepayment Event would have a material adverse effect on the Company's business, financial condition, results of operations, and cash flows.

Key Operating Measures

Certain key operating measures specific to the gaming industry are included in our discussion of our operational performance for the periods for which the Condensed Consolidated Statements of Operations are presented. These key operating measures are presented as supplemental disclosures because management and/or certain investors use these measures to better understand period-over-period fluctuations in our casino and hotel operating revenues. These key operating measures are defined below:

- Table drop in mass market for our Macau Operations is the amount of cash that is deposited in a gaming table's drop box plus cash chips purchased at the casino cage.
- Table drop for our Las Vegas Operations is the amount of cash and net markers issued that are deposited in a gaming table's drop box.
- Table drop for Encore Boston Harbor is the amount of cash and gross markers issued that are deposited in a gaming table's drop box.
- Rolling chips are non-negotiable identifiable chips that are used to track turnover for purposes of calculating incentives within our Macau Operations' VIP program.
- Turnover is the sum of all losing rolling chip wagers within our Macau Operations' VIP program.
- Table games win is the amount of table drop or turnover that is retained and recorded as casino revenues. Table games win is before discounts, commissions and the allocation of casino revenues to rooms, food and beverage and other revenues for services provided to casino customers on a complimentary basis. Table games win does not include poker rake.



- Slot machine win is the amount of handle (representing the total amount wagered) that is retained by us and is recorded as casino revenues. Slot machine win is after adjustment for progressive accruals and free play, but before discounts and the allocation of casino revenues to rooms, food and beverage and other revenues for services provided to casino customers on a complimentary basis.
- Poker rake is the portion of cash wagered by patrons in our poker rooms that is retained by the casino as a service fee, after adjustment for progressive accruals, but before the allocation of casino revenues to rooms, food and beverage and other revenues for services provided to casino customers on a complimentary basis. Poker tables are not included in our measure of average number of table games.
- Average daily rate ("ADR") is calculated by dividing total room revenues, including complimentaries (less service charges, if any), by total rooms occupied.
- Revenue per available room ("REVPAR") is calculated by dividing total room revenues, including complimentaries (less service charges, if any), by total rooms available.
- Occupancy is calculated by dividing total occupied rooms, including complimentary rooms, by the total rooms available.

Below is a discussion of the methodologies used to calculate win percentages at our resorts.

In our VIP operations in Macau, customers primarily purchase rolling chips from the casino cage and can only use them to make wagers. Winning wagers are paid in cash chips. The loss of the rolling chips in the VIP operations is recorded as turnover and provides a base for calculating VIP win percentage. It is customary in Macau to measure VIP play using this rolling chip method. We expect our win as a percentage of turnover from these operations to be within the range of 2.7% to 3.0%.

In our mass market operations in Macau, customers may purchase cash chips at either the gaming tables or at the casino cage. The measurements from our VIP and mass market operations are not comparable as the measurement method used in our mass market operations tracks the initial purchase of chips at the table and at the casino cage, while the measurement method from our VIP operations tracks the sum of all losing wagers. Accordingly, the base measurement from the VIP operations is much larger than the base measurement from the mass market operations. As a result, the expected win percentage with the same amount of gaming win is lower in the VIP operations when compared to the mass market operations.

In Las Vegas, customers purchase chips at the gaming tables in exchange for cash and markers. Customers may then redeem markers at the gaming tables or at the casino cage. The cash and markers, net of redemptions, used to purchase chips are deposited in the gaming table's drop box. This is the base of measurement that we use for calculating win percentage. Each type of table game has its own theoretical win percentage. Our expected table games win percentage is 22% to 26%.

At Encore Boston Harbor, customers purchase chips at the gaming tables in exchange for cash and markers. Customers may then redeem markers only at the casino cage. The cash and gross markers used to purchase chips are deposited in the gaming table's drop box. This is the base of measurement that we use for calculating win percentage. Each type of table game has its own theoretical win percentage. Our expected table games win percentage is 18% to 22%.

Results of Operations

Summary of third quarter 2021 results

	Three Mor Septen				_	Nine Mon Septen			
	2021	2020	Increase/ (Decrease)	Percent Change		2021	2020	Increase/ (Decrease)	Percent Change
Operating revenues	\$ 994,644	\$ 370,452	\$ 624,192	168.5	\$	2,721,439	\$ 1,409,866	\$ 1,311,573	93.0
Net loss attributable to Wynn Resorts, Limited	(166,249)	(758,142)	591,893	78.1		(578,596)	(1,797,743)	1,219,147	67.8
Diluted net loss per share	(1.45)	(7.10)	5.65	79.6		(5.10)	(16.85)	11.75	69.7
Adjusted Property EBITDA (1)	154,561	(65,860)	220,421	NM		420,341	(394,086)	814,427	NM
NM - Not meaningful.									

The following table summarizes our financial results for the periods presented (in thousands, except per share data):

(1) See Item 1—"Financial Statements," Note 17, "Segment Information," for a reconciliation of Adjusted Property EBITDA to net loss attributable to Wynn Resorts, Limited.

The increase in operating revenues for the three months ended September 30, 2021 was primarily driven by increases of \$165.6 million, \$79.3 million, \$289.3 million, and \$75.5 million from Wynn Palace, Wynn Macau, our Las Vegas Operations, and Encore Boston Harbor, respectively, as a result of increased gaming volumes across our properties as well as an increase in hotel occupancy at our Las Vegas Operations.

The decrease in net loss attributable to Wynn Resorts, Limited for the three months ended September 30, 2021 was primarily related to increased operating revenues at our integrated resort properties, partially offset by increased operating expenses.

The increase in Adjusted Property EBITDA for the three months ended September 30, 2021 was primarily driven by increased operating revenues at our integrated resort properties. Adjusted Property EBITDA increased \$89.8 million, \$32.5 million, \$163.2 million, and \$38.6 million at Wynn Palace, Wynn Macau, our Las Vegas Operations, and Encore Boston Harbor, respectively, and decreased \$103.6 million at Corporate and other. In addition, Adjusted Property EBITDA for the third quarter of 2021 includes the reversal of \$8.6 million and \$8.1 million of previously accrued operating expenses at Wynn Palace and Wynn Macau, respectively, recorded in relation to a change in estimate.

Financial results for the three months ended September 30, 2021 compared to the three months ended September 30, 2020.

Operating revenues

The following table presents our operating revenues (in thousands):

	1	Three Months En	ded Sep	otember 30,			
		2021		2020	Increase/ (Decrease)	Percent Change	
Operating revenues							
Macau Operations:							
Wynn Palace	\$	181,323	\$	15,701	\$ 165,622	1,054.9	
Wynn Macau		130,662		51,368	79,294	154.4	
Total Macau Operations		311,985	-	67,069	 244,916	365.2	
Las Vegas Operations		476,003		186,697	289,306	155.0	
Encore Boston Harbor		192,214		116,686	75,528	64.7	
Corporate and other		14,442			14,442	NM	
	\$	994,644	\$	370,452	\$ 624,192	168.5	

NM - Not meaningful.

The following table presents our casino and non-casino operating revenues (in thousands):

	Th	ree Months En	ded Sej	ptember 30,			
	2021 2020			Increase/ (Decrease)		Percent Change	
Operating revenues	-						
Casino revenues	\$	496,264	\$	201,872	\$	294,392	145.8
Non-casino revenues:							
Rooms		173,817		61,144		112,673	184.3
Food and beverage		217,501		76,586		140,915	184.0
Entertainment, retail and other		107,062		30,850		76,212	247.0
Total non-casino revenues		498,380		168,580		329,800	195.6
	\$	994,644	\$	370,452	\$	624,192	168.5

Casino revenues for the three months ended September 30, 2021 were 49.9% of operating revenues, compared to 54.5% for the same period of 2020. Non-casino revenues for the three months ended September 30, 2021 were 50.1% of operating revenues, compared to 45.5% for the same period of 2020.

Casino revenues

Casino revenues increased primarily due to increased table drop, table games win and slot machine win at our Las Vegas Operations and Encore Boston Harbor respectively, and increased VIP turnover and table games win and mass market table drop, table games win and slot machine win at our Macau Operations. Encore Boston Harbor was closed to the public from March 15, 2020 until July 10, 2020.

The table below sets forth our casino revenues and associated key operating measures (dollars in thousands, except for win per unit per day):



		Three Months E	nded Sep	otember 30,	_		
		2021		2020		Increase/ (Decrease)	Percent Change
Macau Operations:							
Wynn Palace:							
Total casino revenues	\$	134,064	\$	12,301	\$	121,763	989.9
VIP:							
Average number of table games		89		102		(13)	(12.7)
VIP turnover	\$	1,234,733	\$	311,676	\$	923,057	296.2
VIP table games win	\$	54,943	\$	3,244	\$	51,699	1,593.7
VIP win as a % of turnover		4.45 %	ó	1.04 %	ó	3.41	
Table games win per unit per day	\$	6,691	\$	347	\$	6,344	1,828.2
Mass market:							
Average number of table games		231		223		8	3.6
Table drop	\$	508,779	\$	86,347	\$	422,432	489.2
Table games win	\$	110,820	\$	19,015	\$	91,805	482.8
Table games win %		21.8 %	ó	22.0 %	ó	(0.2)	
Table games win per unit per day	\$	5,223	\$	929	\$	4,294	462.2
Average number of slot machines		712		569		143	25.1
Slot machine handle	\$	327,017	\$	78,580	\$	248,437	316.2
Slot machine win	\$	11,538	\$	3,995	\$	7,543	188.8
Slot machine win per unit per day	\$	176	\$	76	\$	100	131.6
Wynn Macau:							
Total casino revenues	\$	98,264	\$	27,154	\$	71,110	261.9
VIP:							
Average number of table games		75		91		(16)	(17.6)
VIP turnover	\$	1,335,694	\$	498,519	\$	837,175	167.9
VIP table games win	\$	32,602	\$	19,679	\$	12,923	65.7
VIP win as a % of turnover		2.44 %	ó	3.95 %	ó	(1.51)	
Table games win per unit per day	\$	4,704	\$	2,351	\$	2,353	100.1
Mass market:							
Average number of table games		238		240		(2)	(0.8)
Table drop	\$	441,899	\$	133,006	\$	308,893	232.2
Table games win	\$	87,132	\$	24,898	\$	62,234	250.0
Table games win %		19.7 %	ó	18.7 %	ó	1.0	
Table games win per unit per day	\$	3,972	\$	1,128	\$	2,844	252.1
Average number of slot machines		574		472		102	21.6
Slot machine handle	\$	200,543	\$	87,988	\$	112,555	127.9
Slot machine win	\$	9,142	\$	3,072	\$	6,070	197.6
Slot machine win per unit per day	\$	173	\$	71	\$	102	143.7
	•						

	Three Months E	nded Sep				
	2021		2020		Increase/ (Decrease)	Percent Change
-						
\$	112,575	\$	65,694	\$	46,881	71.4
	224		223		1	0.4
\$	507,188	\$	324,939	\$	182,249	56.1
\$	110,265	\$	65,611	\$	44,654	68.1
	21.7 %	Ď	20.2 %)	1.5	
\$	5,354	\$	3,191	\$	2,163	67.8
	1,746		1,738		8	0.5
\$	1,156,858	\$	739,291	\$	417,567	56.5
\$	80,303	\$	48,267	\$	32,036	66.4
\$	500	\$	302	\$	198	65.6
\$	2,910	\$	10	\$	2,900	NM
\$	151,361		96,723	\$	54,638	56.5
	181		186		(5)	(2.7)
\$	350,145	\$	217,797	\$	132,348	60.8
\$	74,818	\$	47,528	\$	27,290	57.4
	21.4 %	, D	21.8 %)	(0.4)	
\$	4,498	\$	3,079	\$	1,419	46.1
	2,734		1,877		857	45.7
\$	1,196,299	\$	813,435	\$	382,864	47.1
\$	98,816	\$	63,458	\$	35,358	55.7
\$	393	\$	407	\$	(14)	(3.4)
	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	2021 \$ 112,575 224 \$ \$ 507,188 \$ 507,188 \$ 507,188 \$ 507,188 \$ 507,188 \$ 507,188 \$ 5354 \$ 1,0265 21.7 % \$ \$ 5,354 \$ 1,746 \$ 1,156,858 \$ 80,303 \$ 500 \$ 2,910 \$ 151,361 \$ 151,361 \$ 350,145 \$ 74,818 21.4 % 2,734 \$ 1,196,299 \$ 98,816	2021 \$ 112,575 \$ \$ 507,188 \$ \$ 507,188 \$ \$ 507,188 \$ \$ 507,188 \$ \$ 507,188 \$ \$ 507,188 \$ \$ 507,188 \$ \$ 507,188 \$ \$ 507,188 \$ \$ 51,10,265 \$ \$ 5,354 \$ \$ 1,156,858 \$ \$ 1,156,858 \$ \$ 80,303 \$ \$ 500 \$ \$ 2,910 \$ \$ 151,361 \$ \$ 151,361 \$ \$ 350,145 \$ \$ 350,145 \$ \$ 74,818 \$ \$ 4,498 \$ \$ 1,196,299 \$ \$ <td< td=""><td>$\begin{array}{c c c c c c c c c c c c c c c c c c c$</td><td>20212020\$112,575\$65,694\$$224$223224223\$507,188\$324,939\$\$507,188\$324,939\$\$10,265\$65,611\$$21.7$%20.2%20.2%\$5,354\$3,191\$$1,746$1,7381$1,746$1,738\$\$1,156,858\$739,291\$\$\$3003\$48,267\$\$\$500\$302\$\$\$500\$302\$\$\$2,910\$10\$\$\$500\$302\$\$\$\$96,723\$\$\$\$21,7797\$\$\$\$\$21,7797\$\$\$\$\$3,079\$\$\$\$\$3,079\$<</td><td>2021$2020$Increase/ (Decrease)\$112,575\$$65,694$\$$46,881$$224$$223$1\$507,188\$$324,939$\$$182,249$\$507,188\$$324,939$\$$182,249$\$507,188\$$324,939$\$$182,249$\$507,188\$$324,939$\$$182,249$\$10,265\$$65,611$\$$44,654$$21.7$%$20.2$%1.5\$5,354\$$3,191$\$\$5,354\$$3,191$\$$1,746$$1,738$8\$1,156,858\$$739,291$\$\$1,156,858\$$739,291$\$\$1,156,858\$$739,291$\$\$80,303\$$48,267$\$\$80,303\$$48,267$\$\$80,303\$$48,267$\$\$80,303\$$48,267$\$\$80,303\$$48,267$\$\$2,910\$10\$\$2,910\$10\$\$181186(5)\$350,145\$$217,797$\$\$132,348\$$47,528$\$\$74,818\$$47,528$\$\$4,498\$$3,079$\$\$1,196,299\$$813,435$\$\$</td></td<>	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	20212020\$112,575\$65,694\$ 224 223224223\$507,188\$324,939\$\$507,188\$324,939\$\$10,265\$65,611\$ 21.7 %20.2%20.2%\$5,354\$3,191\$ $1,746$ 1,7381 $1,746$ 1,738\$\$1,156,858\$739,291\$\$\$3003\$48,267\$\$\$500\$302\$\$\$500\$302\$\$\$2,910\$10\$\$\$500\$302\$\$\$\$96,723\$\$\$\$21,7797\$\$\$\$\$21,7797\$\$\$\$\$3,079\$\$\$\$\$3,079\$<	2021 2020 Increase/ (Decrease)\$112,575\$ $65,694$ \$ $46,881$ 224 223 1\$507,188\$ $324,939$ \$ $182,249$ \$507,188\$ $324,939$ \$ $182,249$ \$507,188\$ $324,939$ \$ $182,249$ \$507,188\$ $324,939$ \$ $182,249$ \$10,265\$ $65,611$ \$ $44,654$ 21.7 % 20.2 %1.5\$5,354\$ $3,191$ \$\$5,354\$ $3,191$ \$ $1,746$ $1,738$ 8\$1,156,858\$ $739,291$ \$\$1,156,858\$ $739,291$ \$\$1,156,858\$ $739,291$ \$\$80,303\$ $48,267$ \$\$80,303\$ $48,267$ \$\$80,303\$ $48,267$ \$\$80,303\$ $48,267$ \$\$80,303\$ $48,267$ \$\$2,910\$10\$\$2,910\$10\$\$181186(5)\$350,145\$ $217,797$ \$\$132,348\$ $47,528$ \$\$74,818\$ $47,528$ \$\$4,498\$ $3,079$ \$\$1,196,299\$ $813,435$ \$\$

NM - Not meaningful. Note: In response to the initial outbreak of COVID-19 in early 2020, each of our properties was subject to partial or full closure for varying lengths of time during 2020, and each has since reopened with certain COVID-19 specific protective measures in place.

Non-casino revenues

The table below sets forth our room revenues and associated key operating measures:

	 Three Months En	ded Se				
	2021		2020		Increase/ (Decrease)	Percent Change
Macau Operations:						
Wynn Palace:						
Total room revenues (dollars in thousands)	\$ 15,639	\$	4,506	\$	11,133	247.1
Occupancy	51.9 %		11.6 %		40.3	
ADR	\$ 187	\$	225	\$	(38)	(16.9)
REVPAR	\$ 97	\$	26	\$	71	273.1
Wynn Macau:						
Total room revenues (dollars in thousands)	\$ 10,896	\$	4,938	\$	5,958	120.7
Occupancy	51.3 %		16.6 %		34.7	
ADR	\$ 211	\$	291	\$	(80)	(27.5)
REVPAR	\$ 108	\$	48	\$	60	125.0
Las Vegas Operations:						
Total room revenues (dollars in thousands)	\$ 132,705	\$	44,961	\$	87,744	195.2
Occupancy	83.0 %		39.2 %		43.8	
ADR	\$ 392	\$	269	\$	123	45.7
REVPAR	\$ 326	\$	105	\$	221	210.5
Encore Boston Harbor (1):						
Total room revenues (dollars in thousands)	\$ 14,578	\$	6,739	\$	7,839	116.3
Occupancy	87.8 %		72.7 %		15.1	
ADR	\$ 351	\$	294	\$	57	19.4
REVPAR	\$ 308	\$	214	\$	94	43.9

(1) Encore Boston Harbor room statistics have been computed based on 69 days and 47 days of operation in the three months ended September 30, 2021 and 2020, respectively, representing the number of nights hotel rooms were offered for sale to the public.

Room revenues increased \$112.7 million, primarily due to higher occupancy at our Las Vegas Operations, our Macau Operations and Encore Boston Harbor. The hotel tower at Encore Boston Harbor was closed in connection to a COVID-19 related state directive from November 6, 2020 through January 25, 2021, when it reopened its hotel tower on a Thursday through Sunday weekly schedule. Encore Boston Harbor reopened its hotel tower to seven days per week as of September 1, 2021.

Food and beverage revenues increased \$140.9 million, primarily driven by increased covers at our Las Vegas Operations and our Macau Operations.

Entertainment, retail and other revenues increased \$76.2 million, primarily due to an increase in visitation at our Las Vegas Operations and our Macau Operations.

Operating expenses

The table below presents operating expenses (in thousands):

	Three Months Ended September 30,						
	2021			2020		Increase/ (Decrease)	Percent Change
Operating expenses:							
Casino	\$	315,316	\$	160,861	\$	154,455	96.0
Rooms		52,100		35,940		16,160	45.0
Food and beverage		163,655		76,536		87,119	113.8
Entertainment, retail and other		156,490		13,370		143,120	1,070.5
General and administrative		197,350		160,896		36,454	22.7
Provision for credit losses		(347)		11,588		(11,935)	(103.0)
Pre-opening		1,333		877		456	52.0
Depreciation and amortization		177,110		183,486		(6,376)	(3.5)
Property charges and other		15,301		9,905		5,396	54.5
Total operating expenses	\$	1,078,308	\$	653,459	\$	424,849	65.0

Total operating expenses increased \$424.8 million compared to the third quarter of 2020, primarily due to increased casino, room, food and beverage, entertainment, retail and other, and general and administrative expenses associated with increased levels of business as our properties recover from the effects of the COVID-19 pandemic, partially offset by decreased provision for credit losses.

Casino expenses increased \$69.7 million, \$39.3 million, \$20.0 million, and \$25.4 million at Wynn Palace, Wynn Macau, our Las Vegas Operations, and Encore Boston Harbor, respectively. These increases were primarily due to increased gaming tax expense driven by the increase in casino revenues at each property.

Room expenses increased \$13.5 million and \$2.2 million at our Las Vegas Operations and Wynn Palace. These increases were primarily a result of higher operating costs related to the increase in occupancy.

Food and beverage expenses increased \$76.1 million and \$7.4 million at our Las Vegas Operations and Encore Boston Harbor, respectively. The increases were primarily a result of higher operating costs related to the increase in food and beverage revenues at each property as well as higher nightlife entertainment costs at our Las Vegas Operations.

Entertainment, retail and other expenses increased primarily due to marketing expenses incurred by Wynn Interactive.

General and administrative expenses increased primarily due to increases of \$7.5 million, \$6.1 million, \$5.4 million, and \$5.3 million at Wynn Palace, Wynn Macau, our Las Vegas Operations, and Encore Boston Harbor, respectively. These increases were primarily attributable to increased payroll, operating costs, and general and administrative expenses.

Provision for credit losses decreased \$8.8 million and \$3.1 million at Wynn Palace and our Las Vegas Operations, respectively. The decreases were primarily due to the impact of historical collection patterns and expectations of current and future collection trends, as well as the specific review of customer accounts, on our estimated credit loss for the respective periods.

Our property charges and other expenses for the quarter ended September 30, 2021 consisted primarily of advocacy-related expenses of \$12.5 million at Wynn Interactive. Our property charges and other expenses for the quarter ended September 30, 2020 consisted primarily of asset abandonments of \$6.2 million and \$1.1 million at Encore Boston Harbor and our Las Vegas Operations, respectively.

Interest expense, net of capitalized interest

The following table summarizes information related to interest expense (dollars in thousands):

	Three Months E	nded Sej			
	 2021 2020			Increase/ Decrease)	Percent Change
Interest expense					
Interest cost, including amortization of debt issuance costs and original issue discount and premium	\$ 150,325	\$	145,142	\$ 5,183	3.6
Weighted average total debt balance	\$ 12,044,272	\$	13,110,457		
Weighted average interest rate	4.99 %)	4.42 %		

Interest costs increased primarily due to an increase in the weighted average interest rate.

Other non-operating income and expenses

We incurred a foreign currency remeasurement loss of \$11.8 million and a gain of \$0.4 million for the three months ended September 30, 2021 and 2020, respectively. The impact of the exchange rate fluctuation of the Macau pataca, in relation to the U.S. dollar, on the remeasurements of U.S. dollar denominated debt and other obligations from our Macau-related entities drove the variability between periods.

We recorded a gain of \$1.2 million and \$4.7 million for the three months ended September 30, 2021 and 2020, respectively, from change in derivatives fair value.

We recorded a \$0.7 million loss on extinguishment of debt for the three months ended September 30, 2021 related to the prepayment of the Wynn Macau Credit Facilities. We recorded a \$3.1 million loss on extinguishment of debt for the three months ended September 30, 2020 related to the partial prepayment of the Wynn Macau Term Loan.

Income taxes

We recorded an income tax expense of \$1.2 million and \$407.4 million for the three months ended September 30, 2021 and 2020, respectively. The 2021 income tax expense primarily related to an increase in valuation allowance for U.S. deferred tax assets and the Macau dividend tax agreement that provides for an annual payment of MOP 12.8 million (approximately \$1.6 million) as complementary tax otherwise due by stockholders of Wynn Macau SA. The 2020 income tax expense primarily related to the increase in the valuation allowances for U.S foreign tax credits, U.S. loss carryforwards and other U.S. deferred tax assets.

Net loss attributable to noncontrolling interests

Net loss attributable to noncontrolling interests was \$79.7 million and \$73.4 million for the three months ended September 30, 2021 and 2020, respectively. These amounts are primarily related to the noncontrolling interests' share of net loss from WML.

Financial results for the nine months ended September 30, 2021 compared to the nine months ended September 30, 2020.

Operating revenues

The following table presents our operating revenues (in thousands):

	 Nine Months Ended September 30,					
	2021		2020		Increase/ (Decrease)	Percent Change
Operating revenues						
Macau Operations:						
Wynn Palace	\$ 689,020	\$	283,932	\$	405,088	142.7
Wynn Macau	494,341		292,745		201,596	68.9
Total Macau Operations	 1,183,361		576,677		606,684	105.2
Las Vegas Operations	1,009,785		575,407		434,378	75.5
Encore Boston Harbor	487,545		257,782		229,763	89.1
Corporate and other	40,748		—		40,748	NM
	\$ 2,721,439	\$	1,409,866	\$	1,311,573	93.0

NM - Not meaningful.

The following table presents casino and non-casino operating revenues (in thousands):

	ľ	Nine Months End	led Sept	ember 30,		
		2021		2020	Increase/ (Decrease)	Percent Change
Operating revenues						
Casino revenues	\$	1,615,228	\$	782,074	\$ 833,154	106.5
Non-casino revenues:						
Rooms		387,772		231,240	156,532	67.7
Food and beverage		435,152		250,007	185,145	74.1
Entertainment, retail and other		283,287		146,545	136,742	93.3
Total non-casino revenues		1,106,211		627,792	 478,419	76.2
	\$	2,721,439	\$	1,409,866	\$ 1,311,573	93.0

Casino revenues for the nine months ended September 30, 2021 were 59.4% of operating revenues, compared to 55.5% for the same period of 2020. Non-casino revenues for the nine months ended September 30, 2021 were 40.6% of operating revenues, compared to 44.5% for the same period of 2020.



Casino revenues

Casino revenues increased primarily due to increased table drop, table games win and slot machine win at our Las Vegas Operations and Encore Boston Harbor, respectively, and increased VIP table games win and mass market table drop and table games win at our Macau Operations. Our Las Vegas Operations were closed to the public from March 17, 2020 until June 4, 2020. Encore Boston Harbor was closed to the public from March 15, 2020 until July 10, 2020. Our casino operations in Macau were closed for a 15-day period in February 2020. The table below sets forth our casino revenues and associated key operating measures (dollars in thousands, except for win per unit per day):

	 Nine Months Er	nded Sej				
	2021		2020	Inc	crease/(Decrease)	Percent Change
Macau Operations (1):						
Wynn Palace:						
Total casino revenues	\$ 532,040	\$	208,449	\$	323,591	155.2
VIP:						
Average number of table games	96		97		(1)	(1.0)
VIP turnover	\$ 5,246,296	\$	6,823,955	\$	(1,577,659)	(23.1)
VIP table games win	\$ 222,968	\$	113,007	\$	109,961	97.3
VIP win as a % of turnover	4.25 %)	1.66 %		2.59	
Table games win per unit per day	\$ 8,548	\$	4,458	\$	4,090	91.7
Mass market:						
Average number of table games	227		209		18	8.6
Table drop	\$ 1,823,792	\$	583,599	\$	1,240,193	212.5
Table games win	\$ 406,016	\$	156,897	\$	249,119	158.8
Table games win %	22.3 %)	26.9 %		(4.6)	
Table games win per unit per day	\$ 6,555	\$	2,889	\$	3,666	126.9
Average number of slot machines	708		586		122	20.8
Slot machine handle	\$ 1,107,058	\$	542,708	\$	564,350	104.0
Slot machine win	\$ 44,553	\$	24,796	\$	19,757	79.7
Slot machine win per unit per day	\$ 230	\$	163	\$	67	41.1
Wynn Macau:						
Total casino revenues	\$ 379,610	\$	213,758	\$	165,852	77.6
VIP:						
Average number of table games	83		88		(5)	(5.7)
VIP turnover	\$ 4,629,987	\$	4,069,809	\$	560,178	13.8
VIP table games win	\$ 130,624	\$	130,143	\$	481	0.4
VIP win as a % of turnover	2.82 %)	3.20 %		(0.38)	
Table games win per unit per day	\$ 5,745	\$	5,692	\$	53	0.9
Mass market:						
Average number of table games	239		219		20	9.1
Table drop	\$ 1,703,189	\$	752,058	\$	951,131	126.5
Table games win	\$ 321,236	\$	146,231	\$	175,005	119.7
Table games win %	18.9 %)	19.4 %		(0.5)	
Table games win per unit per day	\$ 4,914	\$	2,564	\$	2,350	91.7
Average number of slot machines	583		509		74	14.5
Slot machine handle	\$ 802,337	\$	516,537	\$	285,800	55.3
Slot machine win	\$ 28,573	\$	18,993	\$	9,580	50.4
Slot machine win per unit per day	\$ 179	\$	144	\$	35	24.3
Poker rake	\$ _	\$	2,083	\$	(2,083)	(100.0)

	 Nine Months Ended September 30,					
	2021 2020		Inc	crease/(Decrease)	Percent Change	
Las Vegas Operations (2):				_		
Total casino revenues	\$ 305,253	\$	161,354	\$	143,899	89.2
Average number of table games	205		228		(23)	(10.1)
Table drop	\$ 1,258,733	\$	830,745	\$	427,988	51.5
Table games win	\$ 285,939	\$	166,196	\$	119,743	72.0
Table games win %	22.7 %	ó	20.0 %	6	2.7	
Table games win per unit per day	\$ 5,117	\$	3,711	\$	1,406	37.9
Average number of slot machines	1,670		1,751		(81)	(4.6)
Slot machine handle	\$ 3,063,267	\$	1,650,517	\$	1,412,750	85.6
Slot machine win	\$ 209,682	\$	112,464	\$	97,218	86.4
Slot machine win per unit per day	\$ 460	\$	328	\$	132	40.2
Poker rake	\$ 8,704	\$	2,185	\$	6,519	298.4
Encore Boston Harbor (3):						
Total casino revenues	\$ 398,325	\$	198,513	\$	199,812	100.7
Average number of table games	192		174		18	10.3
Table drop	\$ 890,777	\$	493,428	\$	397,349	80.5
Table games win	\$ 189,070	\$	104,814	\$	84,256	80.4
Table games win %	21.2 %	ó	21.2 %	6	_	
Table games win per unit per day	\$ 3,613	\$	3,839	\$	(226)	(5.9)
Average number of slot machines	2,268		2,330		(62)	(2.7)
Slot machine handle	\$ 3,204,272	\$	1,581,174	\$	1,623,098	102.7
Slot machine win	\$ 263,197	\$	122,906	\$	140,291	114.1
Slot machine win per unit per day	\$ 425	\$	336	\$	89	26.5
Poker rake	\$ 	\$	5,105	\$	(5,105)	(100.0)

In response to the initial outbreak of COVID-19 in early 2020, each of our properties was subject to partial or full closure for varying lengths of time during 2020, and each has since reopened (1) Our casino operations in Macau were closed for a 15-day period in February 2020 and resumed operations on a reduced basis on February 20, 2020.

(2) Our Las Vegas Operations closed on March 17, 2020 and reopened on June 4, 2020. On October 19, 2020, Encore at Wynn Las Vegas adjusted its operating schedule to five days/four nights each week due to reduced customer demand levels. This adjusted operating schedule remained in effect through the first quarter of 2021, and on April 8, 2021, Encore at Wynn Las Vegas resumed full operations.

(3) Encore Boston Harbor closed on March 15, 2020 and reopened on July 10, 2020. In addition, on November 6, 2020, Encore Boston Harbor temporarily suspended hotel operations and overnight casino operations pursuant to a state directive limiting the operating hours of certain businesses, including restaurants and casinos. On January 25, 2021, the limitations on operating hours were lifted, and Encore Boston Harbor restored 24-hour casino operations and reopened its hotel tower on a Thursday through Sunday weekly schedule. The property reopened its hotel tower to seven days per week as of September 1, 2021.

Non-casino revenues

The table below sets forth our room revenues and associated key operating measures:

	Nine Months Ended September 30,						
	 2021		2020	Increase/(Decrease)		Percent Change	
Macau Operations:							
Wynn Palace:							
Total room revenues (dollars in thousands)	\$ 53,534	\$	26,647	\$	26,887	100.9	
Occupancy	61.1 %)	19.3 %		41.8		
ADR	\$ 181	\$	283	\$	(102)	(36.0)	
REVPAR	\$ 111	\$	54	\$	57	105.6	
Wynn Macau:							
Total room revenues (dollars in thousands)	\$ 39,025	\$	23,480	\$	15,545	66.2	
Occupancy	60.0 %)	24.4 %		35.6		
ADR	\$ 217	\$	316	\$	(99)	(31.3)	
REVPAR	\$ 130	\$	77	\$	53	68.8	
Las Vegas Operations (1):							
Total room revenues (dollars in thousands)	\$ 266,250	\$	163,419	\$	102,831	62.9	
Occupancy	63.4 %)	55.8 %		7.6		
ADR	\$ 360	\$	323	\$	37	11.5	
REVPAR	\$ 228	\$	180	\$	48	26.7	
Encore Boston Harbor (2) (3):							
Total room revenues (dollars in thousands)	\$ 28,964	\$	17,694	\$	11,270	63.7	
Occupancy	84.1 %)	74.6 %		9.5		
ADR	\$ 320	\$	293	\$	27	9.2	
REVPAR	\$ 269	\$	218	\$	51	23.4	

 (1) Wynn Las Vegas closed on March 17, 2020 and reopened on June 4, 2020.
 (2) Encore Boston Harbor closed on March 15, 2020 and reopened on July 10, 2020.
 (3) Encore Boston Harbor room statistics have been computed based on 158 days and 121 days of operation in the nine months ended September 30, 2021 and 2020, respectively, representing the number of nights hotel rooms were offered for sale to the public. The property reopened its hotel tower to seven days per week as of September 1, 2021.

Room revenues increased \$156.5 million, primarily due to higher occupancy at our properties and the closure of our Las Vegas Operations from March 17, 2020 until June 4, 2020, resulting from the adverse effects of the COVID-19 pandemic.

Food and beverage revenues increased \$185.1 million, primarily due to increased covers at our restaurants and an increase in nightlife offerings at our Las Vegas Operations as a result of ongoing recovery from the effects of COVID-19.

Entertainment, retail and other revenues increased \$136.7 million, primarily due to an increase in visitation to our Macau Operations and our Las Vegas Operations as a result of ongoing recovery from the effects of COVID-19.

Operating expenses

The table below presents operating expenses (in thousands):

	Nine Monuis Ended September 50,				
	 2021	2020		Increase/ (Decrease)	Percent Change
Operating expenses:					
Casino	\$ 1,048,897	\$ 734,689	\$	314,208	42.8
Rooms	136,187	139,787		(3,600)	(2.6)
Food and beverage	354,709	314,335		40,374	12.8
Entertainment, retail and other	310,871	75,823		235,048	310.0
General and administrative	574,669	547,305		27,364	5.0
Provision for credit losses	7,461	60,548		(53,087)	(87.7)
Pre-opening	5,455	5,614		(159)	(2.8)
Depreciation and amortization	545,538	541,498		4,040	0.7
Property charges and other	26,569	43,701		(17,132)	(39.2)
Total operating expenses	\$ 3,010,356	\$ 2,463,300	\$	547,056	22.2
			_		

Nine Months Ended Sentember 20

Total operating expenses increased \$547.1 million compared to the nine months ended September 30, 2020, primarily due to increased casino, food and beverage, entertainment, retail and other, and general and administrative expenses, partially offset by decreased provision for credit losses and property charges and other expenses.

Casino expenses increased \$157.1 million, \$84.2 million, \$17.4 million, and \$55.4 million at Wynn Palace, Wynn Macau, our Las Vegas Operations, and Encore Boston Harbor, respectively. These increases were primarily due to increased gaming tax expense driven by the increase in casino revenues at each property.

Room expenses decreased \$4.8 million at Encore Boston Harbor. The decrease was primarily a result of lower payroll and operating costs.

Food and beverage expenses increased \$47.1 million at our Las Vegas Operations. The increase was primarily a result of higher operating costs related to the increase in food and beverage revenues as well as higher nightlife entertainment costs.

Entertainment, retail and other expenses increased primarily due to marketing expenses incurred by Wynn Interactive. The increase was partially offset by a decrease in entertainment, retail and other expenses of \$7.9 million at our Las Vegas Operations primarily as a result of lower operating costs.

General and administrative expenses increased primarily due to an increase in corporate and other general and administrative expenses of \$36.3 million, primarily due to a credit of \$30.2 million for the net proceeds of a derivative action settlement recognized during the nine months ended September 30, 2020. In addition, general and administrative expenses increased \$7.0 million at Encore Boston Harbor primarily due to the closure of our operations from March 15, 2020 until July 10, 2020.

The provision for credit losses decreased \$23.5 million, \$24.0 million, and \$3.4 million at our Las Vegas Operations, Wynn Palace, and Wynn Macau, respectively. The decreases were primarily due to the impact of historical collection patterns and expectations of current and future collection trends, as well as the specific review of customer accounts, on our estimated credit loss for the respective periods.

For the nine months ended September 30, 2021 and 2020, pre-opening expenses totaled \$5.5 million and \$5.6 million, which primarily related to restaurant remodels at our Las Vegas Operations.

Depreciation and amortization increased \$5.9 million at Encore Boston Harbor primary due to equipment placed in service during the nine months ended September 30, 2021.



Our property charges and other expenses for the nine months ended September 30, 2021 consisted primarily of advocacy- related expenses of \$12.5 million at Wynn Interactive and asset abandonments of \$4.3 million and \$4.1 million at our Las Vegas Operations and Wynn Palace, respectively. Our property charges and other expenses for the nine months ended September 30, 2020 consisted primarily of asset disposals and abandonments of \$23.4 million and \$9.7 million at Wynn Palace and Encore Boston Harbor, respectively.

Interest expense, net of capitalized interest

The following table summarizes information related to interest expense (dollars in thousands):

	Nine Months En	ded Se	ptember 30,			
	2021		2020		Increase/ (Decrease)	Percent Change
Interest expense						
Interest cost, including amortization of debt issuance costs and original issue discount and premium	\$ 453,601	\$	408,439	\$	45,162	11.1
Capitalized interest	—		(1,252)		(1,252)	(100.0)
	\$ 453,601	\$	407,187	\$	46,414	11.4
Weighted average total debt balance	\$ 12,257,424	\$	12,050,631			
Weighted average interest rate	4.93 %		4.51 %			

Interest costs increased primarily due to an increase in the weighted average debt balance and the weighted average interest rate. Capitalized interest decreased due to the completion of the meeting and convention expansion in February 2020.

Other non-operating income and expenses

We incurred a foreign currency remeasurement loss of \$17.3 million and a gain of \$13.0 million for the nine months ended September 30, 2021 and 2020, respectively. The impact of the exchange rate fluctuation of the Macau pataca, in relation to the U.S. dollar, on the remeasurements of U.S. dollar denominated debt and other obligations from our Macau-related entities drove the variability between periods.

We recorded a gain of \$6.6 million and a loss of \$14.3 million for the nine months ended September 30, 2021 and 2020, respectively, from change in derivatives fair value.

We recorded a \$2.1 million loss on extinguishment of debt for the nine months ended September 30, 2021 related to the partial prepayment of the Wynn Macau Term Loan and the prepayment of the Wynn Macau Credit Facilities. We recorded a \$4.6 million loss on extinguishment of debt for the nine months ended September 30, 2020 primarily related to the partial prepayment of the Wynn Macau Term Loan.

Income taxes

We recorded an income tax expense of \$2.3 million and \$564.1 million for the nine months ended September 30, 2021 and 2020, respectively. The 2021 income tax expense primarily related to an increase in valuation allowance for U.S. deferred tax assets and the Macau dividend tax agreement that provides for an annual payment of MOP 12.8 million (approximately \$1.6 million) as complementary tax otherwise due by stockholders of Wynn Macau SA. The 2020 income tax expense primarily related to the increase in the valuation allowances for U.S foreign tax credits, U.S. loss carryforwards and other U.S. deferred tax assets.

Net loss attributable to noncontrolling interests

Net loss attributable to noncontrolling interests was \$177.0 million and \$218.9 million for the nine months ended September 30, 2021 and 2020, respectively. These amounts are primarily related to the noncontrolling interests' share of net loss from WML.

Adjusted Property EBITDA

We use Adjusted Property EBITDA to manage the operating results of our segments. Adjusted Property EBITDA is net loss before interest, income taxes, depreciation and amortization, pre-opening expenses, property charges and other, management and license fees, corporate expenses and other (including intercompany golf course, meeting and convention, and water rights leases), stock-based compensation, change in derivatives fair value, loss on extinguishment of debt, and other non-operating income and expenses. Adjusted Property EBITDA is presented exclusively as a supplemental disclosure because management believes that it is widely used to measure the performance, and as a basis for valuation, of gaming companies. Management uses Adjusted Property EBITDA as a measure of the operating performance of its segments and to compare the operating performance of its properties with those of its competitors, as well as a basis for determining certain incentive compensation. We also present Adjusted Property EBITDA because it is used by some investors to measure a company's ability to incur and service debt, make capital expenditures and meet working capital requirements. Gaming companies have historically reported EBITDA as a supplement to GAAP. In order to view the operations of their casinos on a more stand-alone basis, gaming companies, including us, have historically excluded from their EBITDA calculations preopening expenses, property charges, corporate expenses and stock-based compensation, that do not relate to the management of specific casino properties. However, Adjusted Property EBITDA should not be considered as an alternative to operating income as an indicator of our performance, as an alternative to cash flows from operating activities as a measure of liquidity, or as an alternative to any other measure determined in accordance with GAAP. Unlike net income, Adjusted Property EBITDA does not include depreciation or interest expense and therefore does not reflect current or future capital expenditures or the cost of capital. We have significant uses of cash flows, including capital expenditures, interest payments, debt principal repayments, income taxes and other non-recurring charges, which are not reflected in Adjusted Property EBITDA. Also, our calculation of Adjusted Property EBITDA may be different from the calculation methods used by other companies and, therefore, comparability may be limited.

The following table summarizes Adjusted Property EBITDA (in thousands) for Wynn Palace, Wynn Macau, Las Vegas Operations, and Encore Boston Harbor as reviewed by management and summarized in Item 1—"Notes to Condensed Consolidated Financial Statements," Note 17, "Segment Information." That footnote also presents a reconciliation of Adjusted Property EBITDA to net loss attributable to Wynn Resorts, Limited.

	Three Mo Septer	nths E nber 3			Nine Months Ended September 30,							
	2021		2020	Increase/ (Decrease)	Percent Change		2021		2020		Increase/ (Decrease)	Percent Change
Wynn Palace	\$ 12,112	\$	(77,647)	\$ 89,759	NM	\$	93,036	\$	(178,379)	\$	271,415	NM
Wynn Macau	(1,939)		(34,457)	32,518	94.4		28,703		(97,895)		126,598	NM
Las Vegas Operations	183,416		20,258	163,158	805.4		344,719		(77,383)		422,102	NM
Encore Boston Harbor	64,565		25,986	38,579	148.5		141,844		(40,429)		182,273	NM
Corporate and other	(103,593)		—	(103,593)	NM		(187,961)		_		(187,961)	NM

NM - Not meaningful.

Adjusted Property EBITDA at Wynn Palace increased \$89.8 million and \$271.4 million for the three and nine months ended September 30, 2021, respectively, primarily due to an increase in operating revenues, partially offset by an increase in operating expenses. Our casino operations at Wynn Palace were closed for a 15-day period in February 2020.

Adjusted Property EBITDA at Wynn Macau increased \$32.5 million and \$126.6 million for the three and nine months ended September 30, 2021, respectively, primarily due to an increase in operating revenues, partially offset by an increase in operating expenses. Our casino operations at Wynn Macau were closed for a 15-day period in February 2020.

Adjusted Property EBITDA at our Las Vegas Operations increased \$163.2 million and \$422.1 million for the three and nine months ended September 30, 2021, respectively, primarily due to an increase in operating revenues, partially offset by an increase in operating expenses. Our Las Vegas Operations closed to the public on March 17, 2020, and reopened on June 4, 2020 on a reduced basis.

Adjusted Property EBITDA at Encore Boston Harbor increased \$38.6 million and \$182.3 million for the three and nine months ended September 30, 2021, respectively, primarily due to an increase in operating revenues, partially offset by an increase in operating expenses. Encore Boston Harbor closed to the public on March 15, 2020 and reopened on July 10, 2020 on a reduced basis.

Adjusted Property EBITDA at Corporate and other was \$(103.6) million and \$(188.0) million for the three and nine months ended September 30, 2021, respectively, primarily due to ramp up of Wynn Interactive operations.

Refer to the discussions above regarding the specific details of our results of operations.

Liquidity and Capital Resources

Our cash flows were as follows (in thousands):

	Nine Months End	led Septe	mber 30,
Cash Flows - Summary	2021		2020
Cash flows from operating activities	\$ (216,832)	\$	(781,079)
Cash flows from investing activities:			
Capital expenditures, net of construction payables and retention	(213,088)		(261,093)
Purchase of intangible and other assets	(19,741)		—
Proceeds from sale of assets and other	3,689		3,733
Net cash used in investing activities	(229,140)		(257,360)
Cash flows from financing activities:			
Proceeds from issuance of long-term debt	1,141,026		3,919,377
Repayments of long-term debt	(2,477,690)		(1,540,609)
Proceeds from issuance of Wynn Resorts, Limited common stock	841,896		—
Repurchase of common stock	(11,004)		(9,537)
Finance lease payments	(11,709)		(2,194)
Proceeds from exercise of stock options			70
Dividends paid	(932)		(108,282)
Distribution to noncontrolling interest	(11,843)		(1,996)
Proceeds from issuance of subsidiary common stock	4,662		—
Payments for debt financing costs	(29,975)		(20,350)
Net cash (used in) provided by financing activities	 (555,569)		2,236,479
Effect of exchange rate on cash, cash equivalents and restricted cash	(1,689)		3,412
(Decrease) increase in cash, cash equivalents and restricted cash	\$ (1,003,230)	\$	1,201,452

Operating Activities

Our operating cash flows primarily consist of operating income (excluding depreciation and amortization and other non-cash charges), interest paid and earned, and changes in working capital accounts such as receivables, inventories, prepaid expenses, and payables. Our table games play is a mix of cash play and credit play, while our slot machine play is conducted primarily on a cash basis. A significant portion of our table games revenue is attributable to the play of a limited number of premium international customers who gamble on credit. The ability to collect these gaming receivables may impact our operating cash flow for the period. Our rooms, food and beverage, and entertainment, retail and other revenue is conducted on a cash and credit basis. Accordingly, operating cash flows will be impacted by changes in operating income and accounts receivable, net.

During the nine months ended September 30, 2021, the decrease in net cash used in operating activities was primarily due to increased operating revenues, partially offset by an increase in operating expenses and changes in working capital accounts. During the nine months ended September 30, 2020, the decrease in net cash provided by operations was primarily due to the adverse effects of the COVID-19 pandemic on the results of our operations.

Investing Activities

Our investing activities primarily consist of project capital expenditures and maintenance capital expenditures associated with maintaining and continually refining our world-class integrated resort properties.



During the nine months ended September 30, 2021, we incurred capital expenditures of \$118.7 million at our Las Vegas Operations primarily related to the Wynn Las Vegas room remodel, and \$33.4 million at Encore Boston Harbor, \$24.4 million at Wynn Palace, and \$18.3 million at Wynn Macau primarily related to maintenance capital expenditures.

During the nine months ended September 30, 2020, we incurred capital expenditures of \$56.2 million at Encore Boston Harbor primarily for the payment of construction retention and other payables related to its construction, \$81.5 million at our Las Vegas Operations for restaurant remodels and maintenance capital expenditures, \$41.3 million for the construction of the additional meeting and convention space at Wynn Las Vegas, and \$38.5 million and \$42.1 million at Wynn Palace and Wynn Macau, respectively, primarily related to maintenance capital expenditures.

Financing Activities

During the nine months ended September 30, 2021, we received proceeds of \$841.9 million from our February 2021 equity offering and used \$716.0 million of the proceeds from the equity offering to repay the outstanding borrowings under the WRF Revolver. We also paid \$464.7 million of outstanding principal owed under the Wynn Macau Term Loan and prepaid the outstanding \$1.26 billion of borrowings under the Wynn Macau Credit Facilities along with related financing costs, using proceeds from the borrowing of \$1.09 billion under the WM Cayman II Revolver along with \$200.0 million of cash. In addition, we made quarterly amortization payments under the WRF Term Loan totaling \$37.5 million.

During the nine months ended September 30, 2020, we issued \$1.0 billion aggregate principal amount of WML 5 1/2% Senior Notes due 2026, issued \$600.0 million aggregate principal amount of WML 5 5/8% Senior Notes due 2028, issued \$600.0 million aggregate principal amount of WRF 7 3/4% Senior Notes due 2025, borrowed \$400.0 million, net of amounts repaid, under the Wynn Macau Revolver, borrowed \$816.0 million under the WRF Revolver, paid \$1.0 billion of outstanding principal owed under the Wynn Macau Term Loan, and made quarterly amortization payments under the WRF Term Loan totaling \$37.5 million. In addition, we used cash of \$108.3 million for the payment of dividends.

Capital Resources

The COVID-19 pandemic has impacted and will continue to impact, materially, our business, financial condition and results of operations. While we believe our strong liquidity position will enable us to fund our current obligations for the foreseeable future, COVID-19 has resulted in significant disruption, which has had and will continue to have a negative impact on our operating income and could have a negative impact on our ability to access capital in the future. We continue to monitor the rapidly evolving situation and guidance from international and domestic authorities.

The following table summarizes our unrestricted cash and cash equivalents and available revolver borrowing capacity under the Company as of September 30, 2021 (in thousands): 10.1

T (**L** C **L**

	1018	Equivalents	ке	Capacity
Wynn Resorts (Macau) S.A. and subsidiaries	\$	184,546	\$	—
Wynn Macau, Limited and subsidiaries (1)		1,203,074		413,130
Wynn Resorts Finance, LLC (2)		327,438		834,195
Wynn Resorts, Limited and other		760,700		
Total	\$	2,475,758	\$	1,247,325

(1) Excluding Wynn Resorts (Macau) S.A. and subsidiaries. (2) Excluding Wynn Macau, Limited and subsidiaries.

Wynn Resorts (Macau) S.A. and subsidiaries. Wynn Resorts (Macau) S.A. ("Wynn Macau SA") generates cash from our Macau Operations and may utilize proceeds from the WM Cayman II Revolver (discussed further below) to fund short term working capital requirements as needed. We expect to use this cash to fund working capital and capital expenditure requirements at our Macau Operations. In addition, we may use this cash to make distributions to WML.

In January 2021, Wynn Macau SA prepaid approximately \$412.5 million of the term loan outstanding under the Wynn Macau Credit Facilities using proceeds from WML senior notes issuances.

In September 2021, borrowings of \$1.09 billion under the WM Cayman II Revolver, along with \$200.0 million of cash, were used to facilitate the prepayment of the outstanding \$1.26 billion of borrowings under the Wynn Macau Credit Facilities and to pay related fees and expenses.

The Company is currently designing the second phase of Wynn Palace. We do not expect to incur significant capital expenditures related to the construction of this project in 2021.

Wynn Macau, Limited and subsidiaries. Wynn Macau, Limited primarily generates cash through distributions from Wynn Macau SA and utilizes available borrowing capacity under the WM Cayman II Revolver as needed. We expect to use this cash to service our existing WML Senior Notes, pay dividends to shareholders of WML (of which we own approximately 72%), and fund working capital requirements at WML.

On September 16, 2021, WM Cayman Holdings Limited II ("WM Cayman II"), an indirect wholly owned subsidiary of WML, entered into an unsecured revolving credit facility agreement (the "Facility Agreement") in an aggregate principal amount of \$1.50 billion consisting of one tranche in an amount of \$312.5 million and one tranche in an amount of HK\$9.26 billion (approximately \$1.19 billion). WM Cayman II has the ability to upsize the total revolving credit facility by an additional \$1.00 billion equivalent under the Facility Agreement and related agreements upon the satisfaction of various conditions.

The borrowings under the WM Cayman II Revolver bear interest at LIBOR or HIBOR plus a margin of 1.875% to 2.875% per annum based on WM Cayman II's leverage ratio on a consolidated basis. The final maturity of all outstanding loans under the Revolving Facility is September 16, 2025.

WML paid no dividends during 2020 or the nine months ended September 30, 2021. The WML board of directors will be continuously monitoring the situation and market conditions in Macau and Greater China and may consider a special dividend in the future when such conditions have stabilized.

If our portion of our cash and cash equivalents were repatriated to the U.S. on September 30, 2021, it would be subject to minimal U.S. taxes in the year of repatriation.

Wynn Resorts Finance, LLC and subsidiaries. Wynn Resorts Finance, LLC ("WRF" or "Wynn Resorts Finance") generates cash from distributions from its subsidiaries, which include our Macau Operations, Wynn Las Vegas, and Encore Boston Harbor, and contributions from Wynn Resorts, as required. In addition, WRF may utilize its available revolving borrowing capacity as needed. We expect to use this cash to service our WRF Credit Facilities, 2025 WRF Notes, 2029 WRF Notes, and WLV Notes, and to fund working capital and capital expenditure requirements as needed.

WRF is a holding company and, as a result, its ability to pay dividends to Wynn Resorts is dependent on WRF receiving distributions from its subsidiaries, which include WML, Wynn Las Vegas, LLC, and Wynn MA, LLC (the owner and operator of Encore Boston Harbor). The WRF Credit Agreement contains customary negative and financial covenants, including, but not limited to, covenants that restrict WRF's ability to pay dividends or distributions and incur additional indebtedness.

Wynn Las Vegas is currently undergoing its planned room remodel, which we temporarily postponed during 2020. We expect to incur between \$110 million and \$130 million of remaining project costs related to this remodel, which we expect to complete during the second quarter of 2022.

The Company repaid \$716.0 million of the outstanding borrowings under the WRF Revolver in February 2021, using proceeds from the February 2021 equity offering described below.

Wynn Resorts, Limited and other subsidiaries. Wynn Resorts, Limited is a holding company and, as a result, our ability to pay dividends is dependent on our ability to obtain funds and our subsidiaries' ability to provide funds to us. Wynn Resorts, Limited and other primarily generates cash from royalty and management agreements with our resorts, dividends and distributions from our subsidiaries, and the operations of the Retail Joint Venture of which we own 50.1%. We expect to use this cash to service our Retail Term Loan, to fund working capital needs of Wynn Interactive, and for general corporate purposes.

On February 11, 2021, the Company completed a registered public offering of 7,475,000 newly issued shares of its common stock, par value \$0.01 per share, at a price of \$115.00 per share for proceeds of \$841.9 million, net of \$17.7 million in

underwriting discounts, commissions and other expenses. The Company used \$716.0 million of the net proceeds from this equity offering to repay the outstanding borrowings under the WRF revolver in February 2021, and intends to use the remaining net proceeds for general corporate purposes.

On May 10, 2021, Wynn Interactive entered into a business combination agreement (the "Business Combination Agreement") with Austerlitz Acquisition Corporation I, a Cayman Islands exempted company ("Austerlitz I"), and Wave Merger Sub Limited, an exempted company limited by shares incorporated in Bermuda and a direct, wholly owned subsidiary of Austerlitz I ("Merger Sub"). The Business Combination Agreement provides for, among other things, the consummation of the following transactions: (i) Austerlitz I will transfer by way of continuation from the Cayman Islands to Bermuda and change its name to "Wynn Interactive, Limited"; and (ii) Merger Sub will merge with and into Wynn Interactive (the "Merger"), with Wynn Interactive being the surviving company of the Merger and direct, wholly owned subsidiary of Austerlitz I. Upon closing of the transaction, assuming no share redemptions by the public stockholders of Austerlitz I, the Company is expected to retain approximately 58% equity interest (and approximately 72% voting interest) in Wynn Interactive. We expect the assets of the combined business to include approximately \$634.0 million of net cash currently held by Austerlitz I. Closing of the proposed business combination is subject to approval by Austerlitz I's stockholders, gaming regulatory approval and other customary closing conditions.

Other Factors Affecting Liquidity

We may refinance all or a portion of our indebtedness on or before maturity. We cannot assure you that we will be able to refinance any of the indebtedness on acceptable terms or at all.

Legal proceedings in which we are involved also may impact our liquidity. No assurance can be provided as to the outcome of such proceedings. In addition, litigation inherently involves significant costs. For information regarding legal proceedings, see Note 15, "Commitments and Contingencies."

Our Board of Directors has authorized an equity repurchase program of up to \$1.0 billion. Under the equity repurchase program, we may repurchase the Company's outstanding shares from time to time through open market purchases, in privately negotiated transactions, and under plans complying with Rules 10b5-1 and 10b-18 under the Exchange Act. As of September 30, 2021, we had \$800.1 million in repurchase authority remaining under the program. We did not repurchase shares under the repurchase program during the nine months ended September 30, 2021.

We have in the past repurchased, and in the future, we may periodically consider repurchasing our outstanding notes for cash. The amount of any notes to be repurchased, as well as the timing of any repurchases, will be based on business, market and other conditions and factors, including price, contractual requirements or consents, and capital availability.

New business developments or other unforeseen events, including related to COVID-19, may occur, resulting in the need to raise additional funds. We continue to explore opportunities to develop additional gaming or related businesses in domestic and international markets. There can be no assurances regarding the business prospects with respect to any other opportunity. Any new development may require us to obtain additional financing. We may decide to conduct any such development through Wynn Resorts, Limited or through subsidiaries separate from the Las Vegas, Boston or Macau-related entities.

Off-Balance Sheet Arrangements

We have not entered into any transactions with special purpose entities nor do we engage in any derivatives except for an interest rate collar associated with our Retail Term Loan. We do not have any retained or contingent interest in assets transferred to an unconsolidated entity. As of September 30, 2021, we had outstanding letters of credit totaling \$15.8 million.

Critical Accounting Policies and Estimates

A description of our critical accounting policies is included in Item 7 of our Annual Report on Form 10-K for the year ended December 31, 2020. There have been no significant changes to these policies for the nine months ended September 30, 2021.

Recently Adopted Accounting Standards and Accounting Standards Issued But Not Yet Adopted

See related disclosure in Note 2, "Basis of Presentation and Significant Accounting Policies."

Forward-Looking Statements

We make forward-looking statements in this Quarterly Report on Form 10-Q based upon the beliefs and assumptions of our management and on information currently available to us. Forward-looking statements include, but are not limited to, information about our business strategy, development activities, competition and possible or assumed future results of operations, throughout this report and are often preceded by, followed by or include the words "may," "will," "should," "could," "believe," "expect," "anticipate," "estimate," "intend," "plan," "continue" or the negative of these terms or similar expressions.

Forward-looking statements are subject to a number of risks and uncertainties that could cause actual results to differ materially from those we express in these forward-looking statements, including the risks and uncertainties in Item 1A — "Risk Factors" of our Annual Report on Form 10-K for the year ended December 31, 2020 and other factors we describe from time to time in our periodic filings with the Securities and Exchange Commission ("SEC"), such as:

- extensive regulation of our business and the cost of compliance or failure to comply with applicable laws and regulations;
- pending or future claims and legal proceedings, regulatory or enforcement actions or probity investigations;
- our ability to maintain our gaming licenses and concessions, including the renewal or extension of the concession in Macau that expires on June 26, 2022;
- our dependence on key employees;
- general global political and economic conditions, in the U.S. and China (including the Chinese government's ongoing anti-corruption campaign), which may impact levels of travel, leisure, and consumer spending;
- restrictions or conditions on visitation by citizens of PRC to Macau;
- the impact on the travel and leisure industry from factors such as an outbreak of an infectious disease, public incidents of violence, riots, demonstrations, extreme weather patterns or natural disasters, military conflicts, civil unrest, and any future security alerts and/or terrorist attacks;
- doing business in foreign locations such as Macau;
- our ability to maintain our customer relationships and collect and enforce gaming receivables;
- our relationships with Macau gaming promoters;
- our dependence on a limited number of resorts and locations for all of our cash flow and our subsidiaries' ability to pay us dividends and distributions;
- competition in the casino/hotel and resort industries and actions taken by our competitors, including new development and construction activities of competitors;
- factors affecting the development and success of new gaming and resort properties (including limited labor resources, government labor and gaming policies and transportation infrastructure in Macau; and cost increases, environmental regulation, and our ability to secure necessary permits and approvals in Everett, Massachusetts);
- construction risks (including disputes with and defaults by contractors and subcontractors; construction, equipment or staffing problems; shortages of materials or skilled labor; environment, health and safety issues; and unanticipated cost increases);
- legalization and growth of gaming in other jurisdictions;
- any violations by us of the anti-money laundering laws or Foreign Corrupt Practices Act;
- adverse incidents or adverse publicity concerning our resorts or our corporate responsibilities;
- changes in gaming laws or regulations;
- changes in federal, foreign, or state tax laws or the administration of such laws;
- continued compliance with all provisions in our debt agreements;
- conditions precedent to funding under our credit facilities;
- leverage and debt service (including sensitivity to fluctuations in interest rates);
- cybersecurity risk, including cyber and physical security breaches, system failure, computer viruses, and negligent or intentional misuse by customers, company employees, or employees of third-party vendors;
- our ability to protect our intellectual property rights; and
- our current and future insurance coverage levels.

Further information on potential factors that could affect our financial condition, results of operations and business are included in this report and our other filings with the SEC. You should not place undue reliance on any forward-looking

statements, which are based only on information available to us at the time this statement is made. We undertake no obligation to update or revise any forward-looking statement, whether as a result of new information, future developments or otherwise.

Item 3. Quantitative and Qualitative Disclosures About Market Risk

Market risk is the risk of loss arising from adverse changes in market rates and prices, such as interest rates, foreign currency exchange rates and commodity prices.

Interest Rate Risks

One of our primary exposures to market risk is interest rate risk associated with our debt facilities that bear interest based on floating rates. We attempt to manage interest rate risk by managing the mix of long-term fixed rate borrowings and variable rate borrowings, supplemented by hedging activities as believed by us to be appropriate. We cannot assure you that these risk management strategies will have the desired effect, and interest rate fluctuations could have a negative impact on our results of operations.

Interest Rate Sensitivity

As of September 30, 2021, approximately 78% of our long-term debt was based on fixed rates. Based on our borrowings as of September 30, 2021, an assumed 100 basis point change in the variable rates would cause our annual interest expense to change by \$20.4 million.

In order to mitigate exposure to interest rate fluctuations on the Retail Term Loan, the Company entered into a five year interest rate collar with a notional value of \$615.0 million. The interest rate collar establishes a range whereby the Company will pay the counterparty if one-month LIBOR falls below the established floor rate of 1.00%, and the counterparty will pay the Company if one-month LIBOR exceeds the ceiling rate of 3.75%.

Foreign Currency Risks

We expect most of the revenues and expenses for any casino that we operate in Macau will be denominated in Hong Kong dollars or Macau patacas; however, a significant portion of our Wynn Macau, Limited debt is denominated in U.S. dollars. Fluctuations in the exchange rates resulting in weakening of the Macau pataca or the Hong Kong dollar in relation to the U.S. dollar could have materially adverse effects on our results, financial condition and ability to service debt. Based on our balances as of September 30, 2021, an assumed 1% change in the U.S. dollar/Hong Kong dollar exchange rate would cause a foreign currency transaction gain/loss of \$37.7 million.

Item 4. Controls and Procedures

Disclosure Controls and Procedures

The Company's management, with the participation of the Company's Chief Executive Officer ("CEO") and Chief Financial Officer ("CFO"), has evaluated the effectiveness of the Company's disclosure controls and procedures (as such term is defined in Rules 13a-15(e) and 15d-15(e) under the Securities Exchange Act of 1934, as amended (the "Exchange Act")) as of the end of the period covered by this report. In designing and evaluating the disclosure controls and procedures, management recognized that any controls and procedures, no matter how well designed and operated, can only provide reasonable assurance of achieving the desired control objectives and management is required to apply its judgment in evaluating the cost-benefit relationship of possible controls and procedures. Based on such evaluation, the Company's CEO and CFO have concluded that, as of the period covered by this report, the Company's disclosure controls and procedures were effective, at the reasonable assurance level, in recording, processing, summarizing and reporting, on a timely basis, information required to be disclosed by the Company in the reports that it files or submits under the Exchange Act and were effective in ensuring that information required to be disclosed by the Company in the reports that it files or submits under the Exchange Act and were effective in ensuring that information required to be disclosed by the Company's CEO and CFO, as appropriate to allow timely decisions regarding required disclosure.

Changes in Internal Control Over Financial Reporting

There were no changes in our internal control over financial reporting (as such term is defined in Rules 13a-15(f) and 15d-15(f) under the Exchange Act) during the quarter to which this report relates that materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

Part II. OTHER INFORMATION

Item 1. Legal Proceedings

We are occasionally party to lawsuits. As with all litigation, no assurance can be provided as to the outcome of such matters and we note that litigation inherently involves significant costs. For information regarding the Company's legal proceedings see Item 1—"Notes to Condensed Consolidated Financial Statements," Note 15, "Commitments and Contingencies" of Part I in this Quarterly Report on Form 10-Q.

Item 1A. Risk Factors

A description of our risk factors can be found in Item 1A, Part I of our Annual Report on Form 10-K for the year ended December 31, 2020. There were no material changes to those risk factors during the nine months ended September 30, 2021.

Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

Issuer Purchases of Equity Securities

The following table summarizes the share repurchases in satisfaction of tax withholding obligations on vested restricted stock during the quarter ended September 30, 2021:

For the Month Ended	Number of Shares Repurchased	Weighted Average Price Paid Per Share	Approximate Dollar Value of Repurchased Shares (in thousands)
July 31, 2021	12,417	\$ 100.83	\$ 1,252,049
August 31, 2021	54,378	\$ 92.45	\$ 5,027,142
September 30, 2021	1,581	\$ 102.42	\$ 161,926

None of the foregoing repurchases that occurred during the three months ended September 30, 2021 were part of the Company's publicly announced repurchase program. As of September 30, 2021, we had \$800.1 million in repurchase authority under the program.

Item 5. Other Information

None.



Item 6. Exhibits

(a) Exhibits

Description

Exhibit

NO.	Description
3.1	Third Amended and Restated Articles of Incorporation of the Registrant. (Incorporated by reference from the Quarterly Report on Form 10-Q filed by the Registrant on May 8, 2015.)

- 3.2 <u>Ninth Amended and Restated Bylaws of the Registrant. (Incorporated by reference from the Annual Report on Form 10-K filed by the Registrant on February 28, 2020.)</u>
- *10.1 Facility Agreement, dated as of September 16, 2021, by and among WM Cayman Holdings Limited II, as borrower, Wynn Macau, Limited, as guarantor, and Bank of China Limited, Macau Branch, as agent and a syndicate of lenders.
- *10.2 Transition Agreement, dated November 9, 2021, by and between Wynn Resorts, Limited and Matt Maddox.
- *10.3 <u>Employment Agreement, dated November 9, 2021, by and between Wynn Resorts, Limited and Craig S. Billings.</u>
- *31.1 Certification of Chief Executive Officer of Periodic Report Pursuant to Rule 13a 14(a) and Rule 15d 14(a).
- *31.2 Certification of Chief Financial Officer of Periodic Report Pursuant to Rule 13a 14(a) and Rule 15d 14(a).
- *32 Certification of CEO and CFO Pursuant to 19 U.S.C. Section 1350.
- 101 The following material from Wynn Resorts, Limited's Quarterly Report on Form 10-Q, formatted in Inline XBRL (Inline Extensible Business Reporting Language): (i) the Condensed Consolidated Balance Sheets as of September 30, 2021 and December 31, 2020; (ii) the Condensed Consolidated Statements of Operations for the three and nine months ended September 30, 2021 and 2020; (iii) the Condensed Consolidated Statements of Comprehensive Loss for the three and nine months ended September 30, 2021 and 2020; (iv) the Condensed Consolidated Statements of Stockholders' Equity (Deficit) for the three and nine months ended September 30, 2021 and 2020; (iv) the Condensed Consolidated Statements of Cash Flows for the nine months ended September 30, 2021 and 2020; (v) the Condensed Consolidated Statements of Cash Flows for the nine months ended September 30, 2021 and 2020; and (vi) Notes to Condensed Consolidated Financial Statements. The instance document does not appear in the Interactive Data File because its XBRL tags are embedded within the Inline XBRL document.
- 104 Cover Page Interactive Data File The cover page XBRL tags are embedded within the Inline XBRL document.

Wynn Resorts, Limited agrees to furnish to the U.S. Securities and Exchange Commission, upon request, a copy of each agreement with respect to longterm debt not filed herewith in reliance upon the exemption from filing applicable to any series of debt which does not exceed 10% of the total consolidated assets of the company.

* Filed herewith.

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

Dated: November 9, 2021

WYNN RESORTS, LIMITED

By: /s/ Craig S. Billings Craig S. Billings Chief Financial Officer (Principal Financial and Accounting Officer)

FACILITY AGREEMENT

dated 16 September 2021

for

WM CAYMAN HOLDINGS LIMITED II

arranged by

BANK OF CHINA LIMITED, MACAU BRANCH

with BANK OF CHINA LIMITED, MACAU BRANCH acting as Agent

Legal counsel to the Obligors:

Legal counsel to the Finance Parties:

KIRKLAND & ELLIS

CLIFFORD CHANCE

CONTENTS

Cla	use	
1.	Definitions and Interpretation	<u>1</u>
2.	The Facilities	<u>37</u>
3.	Purpose	<u>40</u>
4.	Conditions of Utilisation	<u>41</u>
5.	Utilisation	<u>43</u>
6.	Repayment	<u>45</u>
7.	Prepayment and Cancellation	<u>46</u>
8.	Interest	<u>51</u>
9.	Interest Periods	<u>53</u>
10.	Changes to the Calculation of Interest	<u>53</u>
11.	Fees	<u>55</u>
12.	Tax Gross-up and Indemnities	<u>57</u>
13.	Increased Costs	<u>61</u>
14.	Other Indemnities	<u>63</u>
15.	Mitigation by the Lenders	<u>65</u>
16.	Costs and Expenses	<u>65</u>
17.	Guarantee and Indemnity	<u>67</u>
18.	Representations	<u>70</u>
19.	Information Undertakings	<u>77</u>
20.	Financial Covenants	<u>81</u>
21.	General Undertakings	<u>87</u>
22.	Events of Default	<u>98</u>
23.	Changes to the Lenders	<u>103</u>
24.	Changes to the Obligors	<u>108</u>
25.	Role of the Agent and the Arranger	<u>109</u>
26.	Conduct of Business by the Finance Parties	<u>118</u>
27.	Sharing Among the Finance Parties	<u>119</u>
78	Daymont Machanics	171

<u>2</u> 0,	i ayıncın iviccianics	161
29.	Set-off	<u>124</u>
30.	Notices	<u>125</u>
31.	Calculations and Certificates	<u>127</u>
32.	Partial Invalidity	<u>127</u>
33.	Remedies and Waivers	<u>127</u>
34.	Amendments and Waivers	<u>128</u>
35.	Confidential Information	<u>136</u>

36.	Confidentiality of Funding Rates and Reference Bank Quotations	<u>141</u>
37.	Contractual Recognition of Bail-in	<u>143</u>
38.	Non-Recourse Liability	<u>145</u>
39.	Counterparts	<u>145</u>
40.	USA PATRIOT Act	<u>146</u>
41.	Acknowledgement Regarding any Supported QFCs	<u>146</u>
42.	Beneficial Ownership	<u>147</u>
43.	Governing Law	<u>147</u>
44.	Enforcement	<u>148</u>
Sche	dule 1 The Original Lenders	<u>150</u>
Sche	dule 2 Conditions Precedent	<u>151</u>
Sche	dule 3 Utilisation Request	<u>153</u>
Sche	dule 4 Form of Transfer Certificate	<u>155</u>
Sche	dule 5 Form of Assignment Agreement	<u>157</u>
Sche	dule 6 Form of Compliance Certificate	<u>160</u>
Sche	dule 7 Timetables	<u>161</u>
Sche	dule 8 Form of Increase Confirmation	<u>162</u>

THIS AGREEMENT is dated <u>16 September</u> 2021 and made between:

- (1) **WYNN MACAU, LIMITED**, an exempted company incorporated in the Cayman Islands with limited liability with registered office at c/o Maples Corporate Services Limited, PO Box 309, Ugland House, South Church Street, George Town, Grand Cayman KY1-1104, Cayman Islands with registration number 230109 (the "**Company**");
- (2) **WM CAYMAN HOLDINGS LIMITED II**, an exempted company incorporated in the Cayman Islands with limited liability with registered office at c/o Maples Corporate Services Limited, PO Box 309, Ugland House, South Church Street, George Town, Grand Cayman KY1-1104, Cayman Islands with registration number 230468 (the "**Borrower**");
- (3) BANK OF CHINA LIMITED, MACAU BRANCH as mandated lead arranger (the "Arranger");
- (4) **THE FINANCIAL INSTITUTIONS** listed in Schedule 1 (*The Original Lenders*) as lenders (the "**Original Lenders**"); and
- (5) BANK OF CHINA LIMITED, MACAU BRANCH as agent of the other Finance Parties (the "Agent").

IT IS AGREED as follows:

SECTION 1 INTERPRETATION

1. DEFINITIONS AND INTERPRETATION

1.1 **Definitions**

In this Agreement:

"**2019 Indenture**" means the indenture dated 17 December 2019 between the Company and Deutsche Bank Trust Company Americas, as trustee.

"Acceleration Event" means an Event of Default in respect of which the Agent has taken any action pursuant to paragraph (b) or (c) of Clause 22.14 (*Acceleration*).

"Affiliate" means, in relation to any person, a Subsidiary of that person or a Holding Company of that person or any other Subsidiary of that Holding Company.

"**Annual Financial Statements**" means the financial statements for a Fiscal Year delivered pursuant to paragraph (a) of Clause 19.1 (*Financial statements*).

"Anti-Bribery Laws" has the meaning given to that term in Clause 21.13 (FCPA).

"Anti-Terrorism Laws" means the Executive Order, the Bank Secrecy Act (31 U.S.C. §§ 5311 et seq.), the Money Laundering Control Act of 1986 (18 U.S.C. §§ 1956 et seq.), the USA Patriot Act, the International Emergency Economic Powers

Act (50 U.S.C. §§ 1701 et seq.), the Trading with the Enemy Act (50 U.S.C. App. §§ 1 et seq.), any other applicable law or regulation administered by the Office of Foreign Assets Control of the U.S. Department of the Treasury or other Governmental Authority, in each case for the purpose of controlling or preventing terrorism and related activities and any law enacted in the United States of America after the date of this Agreement for such purposes.

"**Approved Corporate Administrative Fees**" means, for any Fiscal Year, such amount of Corporate Administrative Fees, when added to all other Corporate Administrative Fees paid by or on behalf of Wynn Resorts Macau during such Fiscal Year, does not exceed 50 per cent. of the costs incurred by Wynn Resorts and certain of its affiliates during such Fiscal Year in relation to its participation in respect of certain corporate overheads with the Company and its subsidiaries plus fees related to design, marketing and employment services provided by Wynn Resorts and certain of its affiliates.

"**Approved IP Fees**" means the IP Fees as set out in the IP Agreement but without regard to any amendment, variation or supplement that would have the effect of increasing such IP Fees pursuant to the terms of the IP Agreement.

"Assignment Agreement" means an agreement substantially in the form set out in Schedule 5 (*Form of Assignment Agreement*) or any other form agreed between the relevant assignor and assignee.

"Auditors" means Ernst & Young LLP or such other firm of independent accountants of international recognised standing as may be appointed by the Borrower.

"Authorisation" means an authorisation, consent, approval, resolution, licence, exemption, filing, notarisation or registration.

"**Availability Period**" means the period from and including the date of this Agreement to and including the date falling one Month before the Termination Date.

"Available Commitment" means, in relation to a Facility, a Lender's Commitment under that Facility minus:

- (a) the amount of its participation in any outstanding Loans under that Facility; and
- (b) in relation to any proposed Utilisation, the amount of its participation in any Loans that are due to be made under that Facility on or before the proposed Utilisation Date,

other than that Lender's participation in any Loans under that Facility that are due to be repaid or prepaid on or before the proposed Utilisation Date.

"**Available Facility**" means, in relation to a Facility, the aggregate for the time being of each Lender's Available Commitment in respect of that Facility.

"Base Currency" means US dollars.

"**Basel II**" means International Convergence of Capital Measurement and Capital Standards, a Revised Framework" published by the Basel Committee on Banking Supervision in June 2004.

"Board of Directors" means:

- (a) with respect to a corporation, the board of directors of the corporation;
- (b) with respect to a limited partnership, the board of directors of the general partner of the partnership; and
- (c) with respect to any other Person (other than a natural person), the board or committee of such Person serving a similar function.

"Borrower Group" means the Borrower and its Subsidiaries for the time being and

"**Borrower Group Member**" shall mean any member of the Borrower Group.

"Break Costs" means the amount (if any) by which:

(a) the interest (excluding any portion thereof representing the Margin) which a Lender should have received for the period from the date of receipt of all or any part of its participation in a Loan or Unpaid Sum to the last day of the current Interest Period in respect of that Loan or Unpaid Sum, had the principal amount or Unpaid Sum received been paid on the last day of that Interest Period;

exceeds:

(b) the amount which that Lender would be able to obtain by placing an amount equal to the principal amount or Unpaid Sum received by it on deposit with a leading bank for a period starting on the Business Day following receipt or recovery and ending on the last day of the current Interest Period.

"**Business Day**" means a day (other than a Saturday or Sunday) on which banks are open for general business in the Macau SAR, Singapore and, in relation to any day for payments or purchase of either HK dollars or US dollars, both Hong Kong and New York City.

"**Capital Stock**" means any and all shares, interests, participations or other equivalents (however designated) of capital stock of a corporation, any and all classes of membership interests in a limited liability company, any and all classes of partnership interests in a partnership, any and all equivalent ownership interests in a Person and any and all warrants, rights or options to purchase any of the foregoing.

"Code" means the US Internal Revenue Code of 1986.

"Commitment" means a Facility A Commitment or a Facility B Commitment.

"**Common Terms Agreement**" means the common terms agreement between, among others, Wynn Resorts Macau and Bank of China Limited, Macau Branch as the

intercreditor agent originally dated 14 September 2004 and most recently amended and restated by the Common Terms Agreement Sixth Amendment Agreement dated 21 December 2018 between, among others, Wynn Resorts Macau and Bank of China Limited, Macau Branch as the intercreditor agent (as further amended and/or restated from time to time).

"**Company's Agent**" means the Borrower, appointed to act on behalf of the Company in relation to the Finance Documents pursuant to Clause 2.4 (*The Company's Agent*).

"**Compliance Certificate**" means a certificate substantially in the form set out in Schedule 6 (*Form of Compliance Certificate*).

"**Concession Contract**" means the concession contract dated 24 June 2002 between the Macau SAR and Wynn Resorts Macau for the operation of games of chance and other games in casinos in the Macau SAR (including any extension or renewal thereof to the extent permitted pursuant to Clause 21.24 (*Amendments to documents*)).

"**Concession Extension**" means the granting of an extension or renewal by the Macau SAR of the concession granted to Wynn Resorts Macau for the operation of games of chance and other games in casinos in the Macau SAR pursuant to the Concession Contract.

"**Confidential Information**" means all information relating to the Borrower, any Obligor, the Group, the Finance Documents or a Facility of which a Finance Party becomes aware in its capacity as, or for the purpose of becoming, a Finance Party or which is received by a Finance Party in relation to, or for the purpose of becoming a Finance Party under, the Finance Documents or a Facility from either:

- (a) any Group Member or any of its advisers; or
- (b) another Finance Party, if the information was obtained by that Finance Party directly or indirectly from any Group Member or any of its advisers,

in whatever form, and includes information given orally and any document, electronic file or any other way of representing or recording information which contains or is derived or copied from such information but excludes:

- (i) information that:
 - (A) is or becomes public information other than as a direct or indirect result of any breach by that Finance Party of Clause 35 (*Confidential Information*); or
 - (B) is identified in writing at the time of delivery as non-confidential by any Group Member or any of its advisers; or
 - (C) is known by that Finance Party before the date the information is disclosed to it in accordance with paragraph (a) or (b) above or is lawfully obtained by that Finance Party after that date, from a source which is, as far as that Finance Party is aware,

unconnected with the Group and which, in either case, as far as that Finance Party is aware, has not been obtained in breach of, and is not otherwise subject to, any obligation of confidentiality; or

- (D) is independently developed by a Finance Party without that Finance Party breaching Clause 35 *(Confidential Information)*; and
- (ii) any Funding Rate or Reference Bank Quotation.

"**Confidentiality Undertaking**" means a confidentiality undertaking substantially in a recommended form of the LMA or in any other form agreed between the Borrower and the Agent.

"**Corporate Administrative Fees**" means any fees payable by Wynn Resorts Macau to any Corporate Services Provider pursuant to the Corporate Administrative Fees Agreements.

"**Corporate Administrative Fees Agreements**" means the framework agreements disclosed in the Company's filings with HKSE (prior to the date of this Agreement) regarding corporate services, design services, employment services and marketing services and any other agreements which may be needed in the future by and among Wynn Resorts Macau, the Company, Wynn Resorts and certain of its subsidiaries in respect of the abovementioned services.

"**Corporate Services Provider**" means Wynn Resorts and certain of its affiliates in their capacity as a party to the Corporate Administrative Fees Agreements.

"**Cotai Property**" means the luxury hotel resort, retail and entertainment complex and casino facilities known as "Wynn Palace" designed, developed, constructed, operated and maintained on land leased under the Cotai Land Concession Contract.

"Cotai Site" means the land described in the Cotai Land Concession Contract.

"**Cotai Land Concession Contract**" means the land concession contract agreed to by and between Palo, Wynn Resorts Macau and the Macau SAR dated 2 May 2012 pursuant to Dispatch number 16/2012 (including any extension or renewal thereof to the extent permitted pursuant to Clause 21.24 (*Amendments to documents*)).

"**Cotai Site Easements**" the easements appurtenant, easements in gross, licence agreements and other rights running for the benefit of Palo and/or appurtenant to the Cotai Site.

"**Default**" means an Event of Default or any event or circumstance specified in Clause 22 (*Events of Default*) which would (with the expiry of a grace period, the giving of notice, the making of any determination under the Finance Documents or any combination of any of the foregoing) be an Event of Default.

"**Defaulting Lender**" means any Lender:

- (a) which has failed to make its participation in a Loan available (or has notified the Agent or the Borrower (which has notified the Agent) that it will not make its participation in a Loan available) by the Utilisation Date of that Loan in accordance with Clause 5.4 (*Lenders' participation*);
- (b) which has otherwise rescinded or repudiated a Finance Document; or
- (c) with respect to which an Insolvency Event has occurred and is continuing,

unless, in the case of paragraph (a) above:

- (i) its failure to pay is caused by:
 - (A) administrative or technical error; or
 - (B) a Disruption Event, and

payment is made within three Business Days of its due date; or

(ii) the Lender is disputing in good faith whether it is contractually obliged to make the payment in question.

"**Designated Jurisdiction**" means, at any time, any country, region or territory to the extent that such country, region or territory (or portion thereof) itself is the subject of any Sanctions.

"Designated Person" means a Person:

- (a) listed in the annex to, or otherwise subject to the provisions of, the Executive Order;
- (b) named as a "Specially Designated National and Blocked Person" on the most current list published by Office of Foreign Assets Control of the U.S. Department of the Treasury at its official website or any replacement website or other replacement official publication of such list; or
- (c) with which any Lender is prohibited from dealing or otherwise engaging in any transaction by any Anti-Terrorism Law or who or which is the subject of any Sanctions **provided that** such Person is named as a "prohibited person", "prohibited entity" or equivalent or, as the case may be, who or which is named as a Person that is the subject of any Sanctions (in each case) on any list published by any Governmental Authority (including, without limitation, the authorities referred to in the definition of "Sanctions" in this Clause 1.1) pursuant to or otherwise in connection with any Anti-Terrorism Laws or any Sanctions.

"**Disrupted Lender**" has the meaning given to that term in Clause 10.2 (*Market disruption*).

"Disruption Event" means either or both of:

- (a) a material disruption to those payment or communications systems or to those financial markets which are, in each case, required to operate in order for payments to be made in connection with the Facilities (or otherwise in order for the transactions contemplated by the Finance Documents to be carried out) which disruption is not caused by, and is beyond the control of, any of the Parties; or
- (b) the occurrence of any other event which results in a disruption (of a technical or systems-related nature) to the treasury or payments operations of a Party preventing that, or any other Party:
 - (i) from performing its payment obligations under the Finance Documents; or
 - (ii) from communicating with other Parties in accordance with the terms of the Finance Documents,

and which (in either such case) is not caused by, and is beyond the control of, the Party whose operations are disrupted.

"**Eligible Institution**" means any Lender or other person selected by the Borrower and which, in each case, is not a Group Member.

"Environment" means land, including any natural or man-made structures; water; and air.

"**Environmental Claim**" means any formal claim by any Person as a result of or in connection with any material violation of Environmental Law which claim could reasonably be expected to give rise to any remedy or penalty (whether interim or final) or liability for any Group Member or any Finance Party (in its capacity as such in the transactions contemplated by the Finance Documents).

"Environmental Law" means any law or regulation of the Macau SAR with regard to:

- (a) harm to the health of humans; or
- (b) the pollution or protection of the Environment.

"**Environmental Licence**" means any material permit, licence, approval, registration, notification, exemption or any other authorisation required under any Environmental Law.

"Equity Injection" means the cash proceeds of:

(a) fully paid ordinary shares or non-redeemable preference shares in the Borrower, or fully paid redeemable shares in the Borrower with a redemption date at least six Months after the Termination Date, which are (in each case) issued to WML Corp and/or the Company, in each case, for cash; and/or

(b) Subordinated Loans.

"Event of Default" means any event or circumstance specified as such in Clause 22 (Events of Default).

"Excluded Lender" means any Defaulting Lender or Non-Consenting Lender.

"**Executive Director**" means Ms. Linda Chen and any successors approved by the government of the Macau SAR to hold a de minimis economic interest in Wynn Resorts Macau solely for the purposes of complying with applicable Legal Requirements in the Macau SAR.

"**Executive Order**" means Executive Order 13224 on Terrorist Financing: Blocking Property and Prohibiting Transactions With Persons Who Commit, Threaten To Commit, Or Support Terrorism issued September 23, 2001, as amended by Order 13268.

"**Existing Facilities**" means the Term Loan Facilities and the Revolving Credit Facilities (as each such term is defined in the Common Terms Agreement).

"Facility" means Facility A or Facility B.

"**Facility A**" means the USD revolving loan facility made available under this Agreement as described in Clause 2 (*The Facilities*).

"Facility A Commitment" means:

- (a) in relation to an Original Lender, the amount set opposite its name under the heading "Facility A Commitment" in Schedule 1 (*The Original Parties*) and the amount of any other Facility A Commitment transferred to it under this Agreement or assumed by it in accordance with Clause 2.2 (*Increase*); and
- (b) in relation to any other Lender, the amount of any Facility A Commitment transferred to it under this Agreement or assumed by it in accordance with Clause 2.2 (*Increase*),

to the extent not cancelled, reduced or transferred by it under this Agreement.

"**Facility A Loan**" means a loan made or to be made under Facility A or the principal amount outstanding for the time being of that loan.

"**Facility B**" means the HKD revolving loan facility made available under this Agreement as described in Clause 2 (*The Facilities*).

"Facility B Commitment" means:

(a) in relation to an Original Lender, the amount set opposite its name under the heading "Facility B Commitment" in Schedule 1 (*The Original Lenders*) and the amount of any other Facility B Commitment transferred to it under this Agreement or assumed by it in accordance with Clause 2.2 (*Increase*); and

(b) in relation to any other Lender, the amount of any Facility B Commitment transferred to it under this Agreement or assumed by it in accordance with Clause 2.2 (*Increase*),

to the extent not cancelled, reduced or transferred by it under this Agreement.

"**Facility B Loan**" means a loan made or to be made under Facility B or the principal amount outstanding for the time being of that loan.

"**Facility Office**" means the office or offices notified by a Lender to the Agent in writing on or before the date it becomes a Lender (or, following that date, by not less than five Business Days' written notice) as the office or offices through which it will perform its obligations under this Agreement.

"FATCA" means:

- (a) sections 1471 to 1474 of the Code or any associated regulations;
- (b) any treaty, law or regulation of any other jurisdiction, or relating to an intergovernmental agreement between the US and any other jurisdiction, which (in either case) facilitates the implementation of any law or regulation referred to in paragraph (a) above; or
- (c) any agreement pursuant to the implementation of any treaty, law or regulation referred to in paragraph (a) or (b) above with the US Internal Revenue Service, the US government or any governmental or taxation authority in any other jurisdiction.

"FATCA Application Date" means:

- (a) in relation to a "withholdable payment" described in section 1473(1)(A)(i) of the Code (which relates to payments of interest and certain other payments from sources within the US), 1 July 2014; or
- (b) in relation to a "passthru payment" described in section 1471(d)(7) of the Code not falling within paragraph (a) above, the first date from which such payment may become subject to a deduction or withholding required by FATCA.

"FATCA Deduction" means a deduction or withholding from a payment under a Finance Document required by FATCA.

"FATCA Exempt Party" means a Party that is entitled to receive payments free from any FATCA Deduction.

"**Fee Letter**" means any letter or letters dated on or about the date of this Agreement between the Arranger and the Borrower (or the Agent and the Borrower) setting out any of the fees referred to in Clause 11 (*Fees*).

"**Finance Document**" means this Agreement, the Subordination Agreement, any Fee Letter, any Utilisation Request and any other document designated as such by the Agent and the Borrower.

"Finance Party" means the Agent, the Arranger or a Lender.

"Financial Indebtedness" means any indebtedness for or in respect of:

- (a) moneys borrowed;
- (b) any amount raised by acceptance under any acceptance credit facility or dematerialised equivalent;
- (c) any amount raised pursuant to any note purchase facility or the issue of bonds, notes, debentures, loan stock or any similar instrument (but, in each case, excluding Trade Instruments);
- (d) the amount of any liability in respect of any lease or hire purchase contract which would, in accordance with GAAP, be treated as a balance sheet liability (other than any liability in respect of a lease or hire purchase contract which would, in accordance with GAAP in force prior to 1 January 2019, have been treated as an operating lease);
- (e) receivables sold or discounted, but only to the extent of the recourse to the relevant Borrower Group Member;
- (f) any amount raised under any other transaction (including any forward sale or purchase agreement) of a type not referred to in any other paragraph of this definition which is classified as a borrowing under GAAP;
- (g) any derivative transaction entered into in connection with protection against or benefit from fluctuation in any rate or price (and, when calculating the value of any derivative transaction, only the marked to market value (or, if any actual amount is due as a result of the termination or close-out of that derivative transaction, that amount) shall be taken into account);
- (h) any counter-indemnity obligation in respect of a guarantee, indemnity, bond, standby or documentary letter of credit or any other instrument (excluding any Trade Instrument) issued by a bank or financial institution (each, an "instrument") provided that the underlying obligation in respect of which the instrument was issued would, under one or more of the other paragraphs of this definition, be treated as being Financial Indebtedness; and
- (i) the amount of any liability in respect of any guarantee or indemnity for any of the items referred to in paragraphs (a) to (h) above.

"Fiscal Year" has the meaning given to that term in Clause 20.1 (*Financial definitions*) of this Agreement.

"Fixed HKD/USD Exchange Rate" means an exchange rate of HKD7.80:USD1

"**Funding Rate**" means any individual rate notified by a Disrupted Lender to the Agent pursuant to paragraph (d) of Clause 10.2 (*Market disruption*).

"**Funds Flow Statement**" means a funds flow statement (in respect of the flow of funds to occur on the date on which the Existing Facilities are repaid in full) in a form agreed between the Borrower and the Agent on or prior to the date of this Agreement.

"GAAP" means, in respect of the Borrower and other members of the Group, IFRS and, in respect of any person, generally accepted accounting principles applicable in the jurisdiction of that person from time to time.

"**Governing Documents**" means, collectively, as to any Person, the certificate of incorporation, the memorandum and articles of association or bylaws, any shareholders agreement, certificate of formation, limited liability company agreement, partnership agreement or other formation or constituent documents applicable to such Person.

"**Governmental Authority**" means, as to any Person, the government of the Macau SAR, any other national, state, provincial or local government (whether domestic or foreign), any political subdivision thereof or any other governmental, quasi-governmental, judicial, public or statutory instrumentality, authority, body, agency, bureau or entity, any entity exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to government, in each case having jurisdiction over such Person, or any arbitrator with authority to bind such Person at law.

"**Group**" means the Company and its Subsidiaries for the time being and "**Group Member**" shall mean any member of the Group.

"**Guarantee Obligation**" means any guarantee, indemnity, letter of credit or other legally binding assurance against loss granted by one Person in respect of any Financial Indebtedness or other liability or obligation of another Person, or any agreement to assume any Financial Indebtedness of any other Person or to supply funds or to invest in any manner whatsoever in such other Person by reason of Financial Indebtedness of such Person; **provided that** the term Guarantee Obligation shall not include endorsements of instruments for deposit or collection in the ordinary course of business. The amount of any Guarantee Obligation of any guaranteeing Person shall be deemed to be the lower of (1) an amount equal to the stated or determinable amount of the primary obligation in respect of which such Guarantee Obligation is made and (2) the maximum amount for which such guaranteeing Person may be liable pursuant to the terms of the instrument embodying such Guarantee Obligation (unless such primary obligation and the maximum amount for which such guaranteeing Person may be liable pursuant to such Guarantee Obligation shall be such guaranteeing Person's maximum reasonably anticipated liability in respect thereof as determined by the Borrower in good faith).

"HIBOR" means, in relation to any Loan:

(a) the applicable Screen Rate as of the Specified Time for the currency of that Loan and for a period equal in length to the Interest Period of that Loan;

- (b) (if no Screen Rate is available for HK dollars or for the Interest Period for that Loan) the relevant Reference Bank Rate; or
- (c) as otherwise determined pursuant to Clause 10 (*Changes to the Calculation of Interest*),

and if, in either case, that rate is less than zero, HIBOR shall be deemed to be zero.

"Holding Company" means, in relation to a person, any other person in respect of which it is a Subsidiary.

"IFRS" means International Financial Reporting Standards to the extent applicable to the relevant financial statements.

"Impaired Finance Party" means a Finance Party at any time when:

- (a) it has failed to make (or has notified a Party that it will not make) a payment required to be made by it under the Finance Documents by the due date for payment;
- (b) it otherwise rescinds or repudiates a Finance Document;
- (c) (if the Finance Party is also a Lender) it is a Defaulting Lender under paragraph (a) or (b) of the definition of "Defaulting Lender"; or
- (d) an Insolvency Event has occurred and is continuing with respect to the Agent;

unless, in the case of paragraph (a) above:

- (i) its failure to pay is caused by administrative or technical error or a Disruption Event and payment is made within five (5) Business Days of its due date; or
- (ii) the Finance Party is disputing in good faith whether it is contractually obliged to make the payment in question.

"**Increase Confirmation**" means a confirmation substantially in the form set out in Schedule 8 (*Form of Increase Confirmation*).

"Increase Lender" has the meaning given to that term in Clause 2.2 (Increase).

"Increased Costs" has the meaning given to it under Clause 13 (Increased Costs).

"Increased Costs Lender" means a Lender in respect of which:

- (a) an Obligor is required to pay Increased Costs under Clause 13 (*Increased Costs*), or a Tax Payment; or
- (b) on or after the date which is three Months before the earliest FATCA Application Date for any payment by a Party to a Lender (or to the Agent for the account of that Lender), that Lender is not, or has ceased to be, a FATCA Exempt Party and, as a consequence, a Party reasonably believes that it will be

required to make a FATCA Deduction from a payment to that Lender (or to the Agent for the account of that Lender) on or after that FATCA Application Date.

"Indirect Tax" means any goods and services tax, consumption tax, value added tax or any tax of a similar nature.

"**Information Memorandum**" means the lender presentation prepared by the Borrower concerning the Group which, at the Borrower's request and on its behalf, was prepared in relation to this transaction and distributed by the Arranger to selected financial institutions before the date of this Agreement.

"Insolvency Event" in relation to an entity means that the entity:

- (a) is dissolved (other than pursuant to a consolidation, amalgamation or merger);
- (b) becomes insolvent or is unable to pay its debts or fails or admits in writing its inability generally to pay its debts as they become due;
- (c) makes a general assignment, arrangement or composition with or for the benefit of its creditors;
- (d) institutes or has instituted against it, by a regulator, supervisor or any similar official with primary insolvency, rehabilitative or regulatory jurisdiction over it in the jurisdiction of its incorporation or organisation or the jurisdiction of its head or home office, a proceeding seeking a judgment of insolvency or bankruptcy or any other relief under any bankruptcy or insolvency law or other similar law affecting creditors' rights, or a petition is presented for its winding-up or liquidation by it or such regulator, supervisor or similar official;
- (e) has instituted against it a proceeding seeking a judgment of insolvency or bankruptcy or any other relief under any bankruptcy or insolvency law or other similar law affecting creditors' rights, or a petition is presented for its winding-up or liquidation, and, in the case of any such proceeding or petition instituted or presented against it, such proceeding or petition is instituted or presented by a person or entity not described in paragraph (d) above and:
 - (i) results in a judgment of insolvency or bankruptcy or the entry of an order for relief or the making of an order for its winding-up or liquidation; or
 - (ii) is not dismissed, discharged, stayed or restrained in each case within 30 days of the institution or presentation thereof;
- (f) has exercised in respect of it one or more of the stabilisation powers pursuant to Part 1 of the Banking Act 2009 and/or has instituted against it a bank insolvency proceeding pursuant to Part 2 of the Banking Act 2009 or a bank administration proceeding pursuant to Part 3 of the Banking Act 2009;

- (g) has a resolution passed for its winding-up, official management or liquidation (other than pursuant to a consolidation, amalgamation or merger);
- (h) seeks or becomes subject to the appointment of an administrator, provisional liquidator, conservator, receiver, trustee, custodian or other similar official for it or for all or substantially all its assets (other than, for so long as it is required by law or regulation not to be publicly disclosed, any such appointment which is to be made, or is made, by a person or entity described in paragraph (d) above);
- has a secured party take possession of all or substantially all its assets or has a distress, execution, attachment, sequestration or other legal process levied, enforced or sued on or against all or substantially all its assets and such secured party maintains possession, or any such process is not dismissed, discharged, stayed or restrained, in each case within 30 days thereafter;
- (j) causes or is subject to any event with respect to it which, under the applicable laws of any jurisdiction, has an analogous effect to any of the events specified in paragraphs (a) to (i) above; or
- (k) takes any action in furtherance of, or indicating its consent to, approval of, or acquiescence in, any of the foregoing acts.

"**Interest Period**" means, in relation to a Loan, each period determined in accordance with Clause 9 (*Interest Periods*) and, in relation to an Unpaid Sum, each period determined in accordance with Clause 8.3 (*Default interest*).

"**IP Agreement**" means the Amended and Restated Fourth Amendment to Intellectual Property Licence Agreement dated 19 September 2009.

"IP Fees" means the "Licensing Fee" as defined in the IP Agreement.

"Junior Party" has the meaning given to that term in the Subordination Agreement.

"**Land Concession Contract**" means the land concession contract agreed to by Wynn Resorts Macau with the Macau SAR on 4 June 2004 which forms an integral part of Dispatch number 81/2004 as amended by Dispatch number 58/2009 (including any extension or renewal thereof to the extent permitted pursuant to Clause 21.24 (*Amendments to documents*)).

"**Legal Requirements**" means all laws, statutes, orders, decrees, injunctions, licenses, permits, approvals, agreements and regulations of any Governmental Authority having jurisdiction over the matter in question.

"Legal Reservations" means:

(a) the principle that certain (including equitable) remedies may be granted or refused at the discretion of a court, the principle of reasonableness and fairness where implied by law and the limitation of enforcement by laws relating to



bankruptcy, insolvency, reorganisation, court schemes, administration, moratoria and other laws generally affecting the rights of creditors;

- (b) the time barring of claims under applicable statutes of limitation (or equivalent legislation), the possibility that an undertaking to assume liability for or indemnify a person against non-payment of stamp duty may be void and defences of acquiescence, set off or counterclaim;
- (c) similar principles, rights and defences in respect of the enforceability of a contract, agreement or undertaking under the laws of any relevant jurisdiction;
- (d) the principle that additional interest imposed pursuant to any relevant agreement may be held to be unenforceable on the grounds that it is a penalty and thus void;
- (e) the principle that a court may not give effect to an indemnity for legal costs incurred by an unsuccessful litigant; and
- (f) any other matters which are set out as qualifications or reservations as to matters of law of general application and which are set out in the Legal Opinions.

"**Legal Opinion**" means any legal opinion delivered to the Agent pursuant to Clause 4 (*Conditions of Utilisation*) or otherwise in connection with any Finance Document.

"Lender" means:

- (a) any Original Lender; and
- (b) any bank, financial institution, trust, fund or other entity which has become a Party as a "Lender" in accordance with Clause 2.2 (*Increase*) or Clause 23 (*Changes to the Lenders*),

which in each case has not ceased to be a Party as such in accordance with the terms of this Agreement.

"LIBOR" means, in relation to any Loan:

- (a) the applicable Screen Rate as of the Specified Time for the currency of that Loan and for a period equal in length to the Interest Period of that Loan;
- (b) (if no Screen Rate is available for US dollars or for the Interest Period for that Loan) the relevant Reference Bank Rate; or
- (c) as otherwise determined pursuant to Clause 10 (*Changes to the Calculation of Interest*),

and if, in either case, that rate is less than zero, LIBOR shall be deemed to be zero.

"Listing Rules" means the rules as in effect from time to time governing the listing of securities on The Stock Exchange of Hong Kong Limited ("HKSE").

"LMA" means the Loan Market Association.

"Loan" means a Facility A Loan or a Facility B Loan.

"**London Business Day**" means a day (other than a Saturday or Sunday) on which banks are open for general business in London.

"**Macau Gaming Laws**" means Law No. 16/2001 and Administrative Regulation No. 26/2001, as amended from time to time, and other laws promulgated by any Governmental Authority of the Macau SAR and applying to gaming operations in the Macau SAR.

"Macau SAR" means the Macau Special Administrative Region.

"Major Project Document" means any of:

- (a) the Concession Contract;
- (b) the Land Concession Contract;
- (c) the IP Agreement; and
- (d) the Cotai Land Concession Contract.

"**Majority Lenders**" means a Lender or Lenders whose US dollar equivalent amount of Commitments aggregate more than 50 per cent. of the US dollar equivalent amount of the Total Commitments (or, if the Total Commitments have been reduced to zero, aggregated more than 50 per cent. of the US dollar equivalent amount of the Total Commitments immediately prior to the reduction). For the purposes of the definition of "Majority Lenders", the "**equivalent**" of HKD in USD (and vice versa) shall be determined by the Agent by reference to the Fixed HKD/USD Exchange Rate.

"**Margin**" means, in respect of a Loan, 2.625% per annum, but if:

- (a) the first Quarterly Date has occurred; and
- (b) the Leverage Ratio as at the most recent Quarterly Date is within the range set out below and the Agent has received, in accordance with Clause 19.1 (*Financial statements*), the Borrower's financial statements for the period ending on such Quarterly Date together with the Compliance Certificate required under Clause 19.2 (*Compliance Certificate*) in respect of such financial statements,

then, the Margin will be the percentage per annum set out below in the column opposite that range:

Leverage Ratio	Margin
Less than 1.5	1.875%
Greater than or equal to 1.5 but less than 2.25	2.000%
Greater than or equal to 2.25 but less than 3.0	2.125%
Greater than or equal to 3.0 but less than 3.5	2.375%
Greater than or equal to 3.5 but less than 4.0	2.625%
4.0 or above	2.875%

provided that:

- (a) (in the case of any decrease in the Margin) no Event of Default has occurred and is continuing;
- (b) any increase or decrease in the Margin for a Loan shall take effect on (i) (save where paragraph (a) above applies) the date immediately following the date of delivery of the Compliance Certificate for that Relevant Period pursuant to Clause 19.2 (*Compliance Certificate*); or (ii) if a decrease has not taken effect because of an Event of Default continuing, then such change shall take effect on the first day on which that Event of Default ceases to be continuing; or (iii) if an increase is to take effect and the Borrower fails to deliver the Compliance Certificate on or before the date required pursuant to Clause 19.2 (*Compliance Certificate*), then such change shall take effect on the latest date the Borrower was permitted to deliver such Compliance Certificate; and
- (c) if, following receipt by the Agent of the Annual Financial Statements and related Compliance Certificate, those statements and Compliance Certificate do not confirm the basis for a reduced or increased Margin, then the provisions of paragraphs (b) and (c) of Clause 8.2 (*Payment of interest*) shall apply and (subject to such Clause) the Margin for that Loan shall be the percentage per annum determined using the table above and the revised Leverage Ratio calculated using the figures in that Compliance Certificate.

"Material Adverse Effect" means a material adverse effect on:

- (a) the business, operations, property or financial condition of the Group (taken as a whole);
- (b) the ability of the Company and the Borrower (taken as a whole) to perform their payment obligations under the Finance Documents;
 - 17

(c) subject to Legal Reservations, the validity or enforceability of any Finance Document or the rights and remedies of any Finance Party under any Finance Document.

"Money Laundering Laws" has the meaning given to that term in Clause 21.14 (Money Laundering Laws).

"**Month**" means a period starting on one day in a calendar month and ending on the numerically corresponding day in the next calendar month, except that:

- (a) (subject to paragraph (c) below) if the numerically corresponding day is not a Business Day, that period shall end on the next Business Day in that calendar month in which that period is to end if there is one, or if there is not, on the immediately preceding Business Day;
- (b) if there is no numerically corresponding day in the calendar month in which that period is to end, that period shall end on the last Business Day in that calendar month; and
- (c) if an Interest Period begins on the last Business Day of a calendar month, that Interest Period shall end on the last Business Day in the calendar month in which that Interest Period is to end.

The above rules will only apply to the last Month of any period.

"New Lender" has the meaning given to that term in Clause 23 (Changes to the Lenders).

"Non-Consenting Lender" has the meaning given to that term in Clause 34.6 (*Replacement of Lender*).

"**Obligor**" means the Borrower or the Company.

"OFAC" has the meaning given to that term in Clause 21.12 (OFAC).

"**Operatives**" means a shareholder, officer, employee, servant, controlling Person, executive, director, agent, authorised representative or Affiliate of any of the Obligors or of any Group Member.

"**Original Financial Statements**" means the audited financial statements of the Borrower for the financial year ended 31 December 2020 (being the audited balance sheets (each, on a consolidated basis) of the Borrower prepared in accordance with IFRS, as at the end of such financial year and the related audited statements of income and of cash flows (each, on a consolidated basis) for such financial year) prepared under IFRS.

"**Palo**" means Palo Real Estate Company Limited a company with limited liability incorporated in the Macau SAR with registration number 27319 SO.

"**Party**" means a party to this Agreement.

"Performance Bond" means:

- (a) the guarantee to be provided under article 61 of the Concession Contract; and/or
- (b) the guarantee or equivalent to be provided in connection with any renewal, extension, tender or other process conducted by the Macau SAR government in connection with the granting, extension or renewal of the Concession Contract.

"Performance Bond Facility" means:

- (a) a facility extended to Wynn Resorts Macau by the Performance Bond Provider in accordance with the terms of the Performance Bond Facility Agreement for the issuance of a Performance Bond; or
- (b) any other facility extended to a Borrower Group Member by a Performance Bond Provider for the issuance of a Performance Bond.

"**Performance Bond Facility Agreement**" means the agreement dated as of 14 September 2004 between the Performance Bond Provider and Wynn Resorts Macau.

"**Performance Bond Provider**" means Banco Nacional Ultramarino, S.A. or such other Person as may be acceptable to the Agent.

"**Permitted Businesses**" means:

- (a) the development, construction, ownership, operation, management, maintenance, refurbishment, enhancement and financing of hotel resorts and casinos in the Macau SAR as permitted under the Concession Contract;
- (b) the operation of casino games of chance or other forms of gaming in one or more locations in the Macau SAR in connection with Wynn Macau, the Cotai Property or any other integrated casino resort, in each case as permitted under the Concession Contract; and
- (c) the ownership, operation or oversight of food and beverage, spa, entertainment production, convention, retail, residential, foreign exchange, transportation and outsourcing of in-house facilities, businesses or other activities which are necessary for, incident to, arising out of, supportive of or connected to the development, construction, financing, ownership, operation, maintenance, refurbishment or enhancement of such hotel resorts and casinos,

including the holding of shares and other interests permitted hereunder in members of the Group (directly or indirectly) engaged in any of the foregoing.

"**Permitted Disposal**" means any sale, lease, transfer or other disposal:

(a) comprised in the entry into of any leases or licenses with respect to any space on or within the Site or the Cotai Site where the entry into of such lease or licence is part of Wynn Resorts Macau's or, as the case may be, Palo's

Permitted Businesses and is not otherwise prohibited by the terms of the Finance Documents;

- (b) arising as a result of any Permitted Security;
- (c) required as a result of any dedication of minor strips and/or gores of land to any Governmental Authority in Macau for public purposes;
- (d) in the case of the Company only, permitted pursuant to the 2019 Indenture (in the form existing as at the date of this Agreement);
- (e) which is a Permitted Selldown; or
- (f) not falling within any of the above paragraphs but made with the prior written consent of the Agent (acting on the instructions of the Majority Lenders).

"Permitted Financial Indebtedness" means:

- (a) until the first Utilisation Date, the Existing Facilities;
- (b) Financial Indebtedness under any Finance Document, any Performance Bond Facility, any Subordinated Loan or any Guarantee Obligations represented by the guarantee required to be issued pursuant to clause 7 of the Land Concession Contract or clause 7 of the Cotai Land Concession Contract;
- (c) Financial Indebtedness of any Borrower Group Member incurred for the purpose of repaying the balance of the principal amount owing to all Lenders *provided* the Agent is satisfied that upon the incurrence of such Financial Indebtedness or immediately thereafter that all amounts owing to the Finance Parties under the Finance Documents will be repaid and discharged in full;
- (d) Financial Indebtedness of the Borrower **provided that**:
 - (i) such Financial Indebtedness is incurred in the Borrower's ordinary course of trading;
 - (ii) no Event of Default has occurred and is continuing or would result from such incurrence of Financial Indebtedness;
 - (iii) the Leverage Ratio, as at the date of such incurrence of Financial Indebtedness, if determined on a *pro forma* basis after giving effect to the creation, assumption or sufferance to exist of such Financial Indebtedness (when taken together with all such other Financial Indebtedness of the Borrower Group permitted under the Finance Documents), would not exceed 4.50:1; and
 - (iv) the provider (or providers) of such Financial Indebtedness do not (in respect of such Financial Indebtedness) have the benefit of any Security (save for any Security referred to in paragraphs (h) and (n) of the definition of "Permitted Security" set out in this Clause 1.1) and

such Financial Indebtedness does not constitute Guarantee Obligations in respect of any WML Debt;

- (e) Financial Indebtedness owed by any Borrower Group Member to any other Borrower Group Member;
- (f) Financial Indebtedness under, or in respect of, any Permitted Swap Transaction;
- (g) Financial Indebtedness not permitted by any of the preceding paragraphs **provided that**:
 - (i) such Financial Indebtedness is incurred in the relevant Borrower Group Member's ordinary course of trading;
 - (ii) no Event of Default has occurred and is continuing or would result from such incurrence of Financial Indebtedness; and
 - (iii) the outstanding principal amount of which does not exceed USD25,000,000 (or its equivalent in other currencies) in aggregate for the Borrower Group at any time and does not constitute Guarantee Obligations in respect of any WML Debt; and
- (h) other Financial Indebtedness approved in writing by the Agent (acting on the instructions of the Majority Lenders).

"Permitted Loan" means:

- (a) extensions of trade credit in the ordinary course of business (including, without limitation, advances to patrons of Wynn Macau or the Cotai Property casino and gaming operations consistent with ordinary course gaming operations) provided that such extensions are in compliance with all Legal Requirements;
- (b) loans and advances to employees of Wynn Resorts Macau or Palo in the ordinary course of business (including, without limitation, for travel, entertainment and relocation expenses) in an aggregate amount not to exceed USD5,000,000 or its equivalent at any one time outstanding;
- (c) any advance, loan or extension of credit (by way of guarantee or otherwise) by a member of the Borrower Group to another member of the Borrower Group; and
- (d) any advance, loan or extension of credit (by way of guarantee or otherwise) not permitted by any of the preceding paragraphs provided that (i) no Event of Default shall have occurred and be continuing or would result therefrom, (ii) no Material Adverse Effect shall have occurred and be continuing or would result therefrom, (iii) such advance, loan or extension of credit is made in the relevant Borrower Group Member's ordinary course of trading, (iv) where such advance, loan or extension of credit is made after the delivery of

the Compliance Certificate in respect of the first Quarterly Date, had such advance, loan or extension of credit been made immediately prior to the most recent Quarterly Date, the Borrower would as at that Quarterly Date still have been in compliance with Clause 20.2 (*Financial condition*), and (v) such advance, loan or extension of credit (by way of guarantee or otherwise) does not constitute Guarantee Obligations in respect of any WML Debt, the outstanding amount of which does not exceed USD25,000,000 (or its equivalent in other currencies) in aggregate for the Borrower Group at any time.

"Permitted Payment" means:

- (a) any Restricted Payment by any Borrower Group Member (other than the Borrower) in favour of the holder(s) of shares or equity interests in such Borrower Group Member;
- (b) any Restricted Payment by the Borrower, **provided that**:
 - (i) the payment is made when no Event of Default is continuing or would occur immediately after the making of the payment;
 - (ii) the payment is made after the delivery of the Compliance Certificate in respect of the first Quarterly Date; and
 - (iii) had the payment been made immediately prior to the most recent Quarterly Date, the Borrower would, as that Quarterly Date, still have been in compliance with Clause 20.2 (*Financial condition*);
- (c) the payment of Approved IP Fees and Approved Corporate Administrative Fees **provided that**, in each case:
 - (i) no Event of Default has occurred and is continuing or would result from such payment; and
 - (ii) the aggregate amount paid pursuant to this paragraph (c) does not exceed in any Fiscal Year, any cap set out from time to time in any Corporate Administrative Fees Agreement or, as the case may be, the IP Agreement;
- (d) any repayment by any Subordinated Debtor (as defined in the Subordination Agreement) under any Subordinated Loan made to it solely for the purpose of the Company paying any amount with respect to WML Debt provided that (i) such repayments (when taken together) would not exceed (in aggregate) USD150,000,000 in any Fiscal Year, (ii) no Event of Default is continuing or would result from such repayment, (iii) such repayment is made after the delivery of the Compliance Certificate in respect of the first Quarterly Date, (iv) had the repayment been made immediately prior to the most recent Quarterly Date, the Borrower would, as at that Quarterly Date, still have been in compliance with Clause 20.2 (*Financial condition*) (for the avoidance of doubt, subject to paragraph (g) of Clause 20.4 (*Equity cure*)) and (v) following

such repayment, such repaid amount of such Subordinated Loan is not available for redrawing and the commitment of the relevant Junior Party in respect of such Subordinated Loan (equal to the amount so repaid) is cancelled on the date of such repayment; and

(e) the payment to any of its Affiliates of the direct cost incurred by that Affiliate (excluding any internal administration charges, internal handling fees, internal mark-ups or any other internal fees, costs, charges or impositions of a similar nature levied, imposed or charged by that Affiliate) for the acquisition of any assets, or the procurement or any services, to be used by Wynn Resorts Macau or Palo in its Permitted Business at or directly related to Wynn Macau and/or the Cotai Property.

"Permitted Security" means:

- (a) any Security granted in respect of the Existing Facilities, **provided that** such Existing Facilities have been repaid in full on the first Utilisation Date and the Borrower is cooperating with the Finance Parties in good faith to arrange for such Security to be discharged as soon as practicable;
- (b) liens for taxes not yet due or which are being contested in good faith by appropriate proceedings, **provided that** adequate reserves with respect thereto are maintained on the books of the relevant Borrower Group Member in conformity with applicable GAAP;
- (c) carriers', warehousemen's, mechanics', materialmen's, repairmen's or other like liens arising in the ordinary course of business for amounts which are not overdue for a period of more than 30 days or that are being contested in good faith by appropriate proceeding (such contested proceedings conclusively operating to stay the sale of any such Property on account of such lien);
- (d) pledges or deposits in connection with workers' compensation, unemployment insurance and other social security legislation **provided that** if such pledges are being contested, appropriate reserves (determined in accordance with the applicable GAAP are maintained on the books of the relevant Borrower Group Member);
- (e) deposits by or on behalf of any Borrower Group Member to secure the performance of bids, trade contracts (other than for borrowed money), leases, statutory obligations, appeal bonds and other obligations of a like nature incurred in the ordinary course of business;
- (f) easements, rights-of-way, restrictions, encroachments and other similar encumbrances and other minor defects and irregularities in title, in each case incurred in the ordinary course of business that, in the aggregate, do not materially detract from the value of the Property subject thereto or materially interfere with the ordinary conduct of the business of any Borrower Group Member;

- (g) licenses of patents, trademarks and other intellectual property rights granted by any Borrower Group Member in the ordinary course of business and not interfering in any material respect with the ordinary conduct of the business of any Borrower Group Member;
- (h) any attachment or judgment lien not constituting an Event of Default;
- (i) liens in favour of customs and revenue authorities arising as a matter of law to secure payment of customs duties in connection with the importation of goods;
- (j) any zoning or similar law or right reserved to or vested in any Governmental Authority to control or regulate the use of the Site and Site Easements or the Cotai Site and Cotai Site Easements;
- (k) any Security over or affecting any asset acquired by a Borrower Group Member after the date of this Agreement if:
 - (i) that Security was not created in contemplation of the acquisition of that asset by a Borrower Group Member;
 - (ii) the principal amount secured has not been increased in contemplation of or since the acquisition of that asset by that Borrower Group Member; and
 - (iii) such Security is removed or discharged within 6 months of the date of acquisition of such asset;
- any Security over or affecting any asset of any company which becomes a Borrower Group Member after the date of this Agreement, where that Security is created prior to the date on which that company becomes a Borrower Group Member and if:
 - (i) that Security was not created in contemplation of the acquisition of that asset by that company;
 - (ii) the principal amount secured has not been increased in contemplation of or since the acquisition of that asset by that company; and
 - (iii) such Security is removed or discharged within 6 months of that company becoming a Borrower Group Member;
- (m) any Security in favour of the Macau SAR government over assets of the Borrower Group securing the obligations of Wynn Resorts Macau under the Concession Contract and required pursuant to the Concession Contract;
- (n) any Security over cash deposited with, or held for the account of, any Borrower Group Member securing reimbursement obligations owing by that Borrower Group Member under performance bonds, guaranties, commercial or standby letters of credit, bankers' acceptances or similar instruments or the guarantee required to be issued pursuant to clause 7 of the Land Concession

Contract or clause 7 of the Cotai Land Concession Contract, granted in favour of the issuers of such performance bonds, guaranties, commercial letters of credit or bankers' acceptances, so long as (i) the performance bonds, guaranties, commercial or standby letters of credit, bankers' acceptances or similar instruments are issued for the benefit of a trade creditor of that Borrower Group Member or, in respect of the guarantee required to be issued pursuant to clause 7 of the Land Concession Contract or clause 7 of the Cotai Land Concession Contract, for the benefit of a Governmental Authority in the Macau SAR and (ii) the amount of cash secured by such Security does not exceed 110 % of the amount of the Financial Indebtedness secured thereby (ignoring any interest earned or paid on such cash);

- (o) any netting or set-off arrangements entered into by any Borrower Group Member under any Permitted Swap Transaction or in the normal course of its banking arrangements and the standard account operating procedures of the bank for the purpose of netting debit and credit balances;
- (p) any Security arising under title transfer or retention of title arrangements entered into by any Borrower Group Member in the normal course of its trading activities on the counterparty's standard or usual terms, provided that such arrangements shall be limited to Property of an aggregate value not exceeding USD5,000,000 or its equivalent;
- (q) any lien arising by operation of law and in the ordinary course of business;
- (r) any Quasi-Security arising as a result of a disposal which is a Permitted Disposal;
- (s) any other Security over assets the book value of which (when aggregated with the book value of any other assets subject to Security given by members of the Borrower Group other than any permitted under paragraphs (a) to (r) above) does not exceed USD25,000,000 (or its equivalent) and **provided that**:
 - (i) such Security is granted in the ordinary course of trading of the relevant Borrower Group Member; and
 - (ii) no Event of Default has occurred and is continuing or would result from such grant of Security; and
- (t) other Security approved in writing by the Agent (acting on the instructions of the Majority Lenders).

"**Permitted Selldown**" means, on or after the date of this Agreement, any sale, lease, transfer or other disposal (whether through a single transaction or a series of transactions and whether related or not) directly or indirectly of the outstanding Capital Stock of Wynn Resorts Macau **provided that**:

(a) the Borrower beneficially owns, directly or indirectly, at least 60 per cent. of the outstanding Capital Stock of Wynn Resorts Macau (for this purpose, any class of Capital Stock of Wynn Resorts Macau which has a de minimis

economic interest and which has been created and issued to the Executive Director solely for the purposes of complying with applicable Legal Requirements in the Macau SAR shall be deemed to not form part of the outstanding Capital Stock of Wynn Resorts Macau);

- (b) no Event of Default has occurred and is continuing or will occur as a result of such sale, lease, transfer or other disposal;
- (c) such sale, lease, transfer or other disposal does not, nor will it, result in a breach of or an event of default (howsoever described) under any Major Project Document (other than the IP Agreement);
- (d) such sale, lease, transfer or other disposal is effected and (once effected) will be, in compliance with all applicable Legal Requirements (including, without limitation, those of the Macau SAR);
- (e) such sale, lease, transfer or other disposal is effected after the delivery of the Compliance Certificate in respect of the first Quarterly Date; and
- (f) the Leverage Ratio and the Interest Coverage Ratio as at the most recent Quarterly Date, if determined on a *pro forma* basis after giving effect to such sale, lease, transfer or other disposal, would not (in respect of the Leverage Ratio) exceed 4.5:1.0 or (in respect of the Interest Coverage Ratio) be less than 2.5:1.0.

"**Permitted Swap Transaction**" means any derivative transaction entered into to protect against or benefit from any Group Member's exposure to fluctuations in any rate, price, index or credit rating (whether in relation to interest rates, commodity prices, currency exchange or otherwise) but excluding any transaction entered into for purely speculative purposes.

"**Person**" means any natural person, corporation, partnership, firm, association, Governmental Authority or any other entity whether acting in an individual, fiduciary or other capacity.

"**Property**" means any property or assets including without limitation any right or interest (whether legal or equitable) in or to property or assets of any kind whatsoever, whether real, personal or mixed and whether tangible or intangible, including, without limitation, Capital Stock.

"Property Mandatory Prepayment Event" means the occurrence of any of the following events or circumstances:

- (a) any temporary administrative intervention is made by the Macau SAR under article 44 of Law no. 16/2001, of 24 September and pursuant to clause 79 of the Concession Contract;
- (b) the Macau SAR takes any formal measure seeking the unilateral dissolution of the Concession Contract under article 45 of Law no. 16/2001, of 24 September and pursuant to clause 80 thereof or the Macau SAR gives notice pursuant to

article 80(3) of the Concession Contract and the Company fails to comply with the terms thereof within the grace period specified therein;

- (c) any event after which Wynn Resorts Macau does not have such licenses, concessions, subconcessions or other permits or authorisations as are necessary for the Company and its Subsidiaries to own or manage casino or gaming areas or operate casino games of fortune and chance in the Macau SAR in substantially the same manner and scope as the Company and its Subsidiaries are entitled to at the date of this Agreement, for a period of thirty (30) consecutive days or more, and such event has a material adverse effect (after taking into account all resources, insurance, indemnity and assurance to the Group) on the ability of the Company and the Borrower (taken as a whole) to perform their payment obligations under the Finance Documents;
- (d) the termination, rescission, revocation or modification of the Concession Contract resulting from or in connection with any renewal, extension tender or other process conducted by the Macau SAR government in connection with the granting, extension or renewal of the Concession Contract where such renewal, extension, tender or other process (i) does not result in the granting, extension or renewal of the Concession Contract or (ii) results in Wynn Resorts Macau not having such licenses, concessions, subconcessions or other permits or authorisations as are necessary for the Group to own or manage casino or gaming areas or operate casino games of fortune and chance in Macau SAR in substantially the same manner and scope as the Group is entitled to at the date of this Agreement for a period of thirty (30) consecutive days or more, and, in the case of a modification of the Concession Contract only, such event has a material adverse effect (after taking into account all resources, insurance, indemnity and assurance to the Group) on the ability of the Company and the Borrower (taken as a whole) to perform their payment obligations under the Finance Documents; and/or
- (e) the Land Concession Contract or the Cotai Land Concession Contract is terminated or rescinded or Macau SAR takes any formal measure seeking any termination of (i) the Land Concession Contract pursuant to clause 15 thereof or any rescission pursuant to Clause 16 thereof or (ii) the Cotai Land Concession Contract pursuant to clause 14 thereof or any rescission pursuant to clause 15 thereof.

"Quarterly Date" means:

- (a) 30 June 2022; and
- (b) the last day of each Fiscal Quarter ending after 30 June 2022.

"**Quasi-Security**" has the meaning given to that term in Clause 21.17 (*Negative pledge*).

"**Quotation Day**" means, in relation to any period for which an interest rate is to be determined, (if the currency US dollars) two London Business Days before the first day of that period or (if the currency is HK dollars), the first day of that period

(unless, in each case, market practice differs in the Relevant Market for that currency, in which case the Quotation Day for that currency will be determined by the Agent in accordance with market practice in the Relevant Market (and if quotations would normally be given on more than one day, the Quotation Day will be the last of those days)).

"**Reference Bank Quotation**" means any quotation supplied to the Agent by a Reference Bank.

"**Reference Bank Rate**" means the arithmetic mean of the rates (rounded upwards to four decimal places) as supplied to the Agent at its request by the Reference Banks:

(a) in relation to LIBOR as either:

(i) if:

- (A) the Reference Bank is a contributor to the applicable Screen Rate; and
- (B) it consists of a single figure,

the rate (applied to the relevant Reference Bank and the relevant currency and period) which contributors to the applicable Screen Rate are asked to submit to the relevant administrator; or

- (ii) in any other case, the rate at which the relevant Reference Bank could fund itself in the relevant currency for the relevant period with reference to the unsecured wholesale funding market; or
- (b) in relation to HIBOR as either:
 - (i) if:
 - (A) the Reference Bank is a contributor to the applicable Screen Rate; and
 - (B) it consists of a single figure,

the rate (applied to the relevant Reference Bank and the relevant currency and period) which contributors to the applicable Screen Rate are asked to submit to the relevant administrator; or

(ii) in any other case, the rate at which the relevant Reference Bank could fund itself in the relevant currency for the relevant period with reference to the unsecured wholesale funding market.

"Reference Banks" means, in relation to:

(a) LIBOR, the principal London office of Deutsche Bank AG; and

(b) HIBOR, the principal Hong Kong offices of Bank of China Limited, Hong Kong Branch, DBS Bank Ltd., Hong Kong Branch and Industrial and Commercial Bank of China (Asia) Limited,

or such other bank or banks designated from time to time by the Agent in consultation with the Borrower from time to time.

"**Related Fund**" in relation to a fund (the "**first fund**"), means a fund which is managed or advised by the same investment manager or investment adviser as the first fund or, if it is managed by a different investment manager or investment adviser, a fund whose investment manager or investment adviser is an Affiliate of the investment manager or investment adviser of the first fund.

"Relevant Market" means:

- (a) in the case of US dollars, the London interbank market; or
- (b) in the case of HK dollars, the Hong Kong interbank market.

"Relevant Ratio" means:

- (a) in relation to any Utilisation or cancellation of Facility A or Facility B, the ratio of the Available Facility of Facility B (and vice versa); or
- (b) in relation to any repayment or prepayment of Facility A or Facility B, the ratio of the aggregate amount of the outstanding Loan(s) under Facility A to the aggregate amount of the outstanding Loan(s) under Facility B (and vice versa),

in each case, determined by reference to the Fixed HKD/USD Exchange Rate.

"**Repeating Representations**" means each of the representations set out in Clauses 18.1 (*Status*) to 18.6 (*Governing law and enforcement*), 18.9 (*No default*), paragraph (b) of Clause 18.10 (*No misleading information*), 18.13 (*No proceedings*), 18.17 (*Solvency*) and 18.20 (*Fiscal Year*) to 18.24 (*Investment Company Act and U.S. margin regulations*).

"**Representative**" means any delegate, agent, manager, administrator, nominee, attorney, trustee or custodian.

"**Responsible Officer**" means, as to any Person in respect of any matter, the chief executive officer, president, managing director, chief financial officer, chief operating officer or treasurer of such Person duly authorised in respect of such matter, but in any event, with respect to financial matters, the chief financial officer or treasurer of such Person or the chief financial officer of a Borrower Group Member. Unless

otherwise qualified, all references to a "Responsible Officer" shall refer to a Responsible Officer of the Borrower.

"Resort Asset" means:

- (a) Wynn Macau (including, without limitation, the Site and the land and/or buildings constituting Wynn Macau);
- (b) the Cotai Property (including, without limitation, the Cotai Site and the land and/or buildings constituting the Cotai Site);
- (c) any asset which is necessary for the continued operation of the business conducted at Wynn Macau or the Cotai Property;
- (d) the Concession Contract;
- (e) the Land Concession Contract;
- (f) the Cotai Land Concession Contract; or
- (g) any share or equity interest in any person that directly or indirectly owns Wynn Macau or the Cotai Property or is a counterparty to the Concession Contract, the Land Concession Contract or the Cotai Land Concession Contract.

"Restricted Payment" has the meaning given in paragraph (a) of Clause 21.28 (Restricted Payments).

"Rollover Loan" means one or more Loans:

- (a) made or to be made on the same day that a maturing Loan is due to be repaid;
- (b) the aggregate amount of which is equal to or less than the amount of the maturing Loan;
- (c) in the same currency as the maturing Loan; and
- (d) made or to be made for the purpose of refinancing that maturing Loan.

"**Sanctions**" means any sanctions administered or enforced by the United States Government (including, without limitation, OFAC), the United Nations Security Council, the European Union, Her Majesty's Treasury, the Federal Government of Canada, the Government of the Hong Kong Special Administrative Region, the Government of France, the US Department of State or other relevant sanctions authority.

"**Screen Rate**" means, in relation to:

(a) LIBOR, the London interbank offered rate administered by ICE Benchmark Administration Limited (or any other person which takes over the administration of that rate) for US dollars for the relevant period displayed on

page LIBOR01 of the Thomson Reuters screen (or any replacement Thomson Reuters page which displays that rate), or on the appropriate page of such other information service which publishes that rate from time to time in place of Thomson Reuters; and

(b) HIBOR, the Hong Kong interbank offered rate administered by the Hong Kong Association of Banks (or any other person which takes over the administration of that rate) for the relevant period displayed on page HKABHIBOR of the Thomson Reuters screen (or any replacement Thomson Reuters page which displays that rate), or on the appropriate page of such other information service which publishes that rate from time to time in place of Thomson Reuters. If the agreed page is replaced or service ceases to be available, the Agent may specify another page or service displaying the appropriate rate after consultation with the Borrower and the Lenders.

"**Security**" means a mortgage, charge, pledge, lien or other security interest securing any obligation of any person or any other agreement or arrangement having a similar effect.

"Site" means the land described in the Land Concession Contract.

"**Site Easements**" the easements appurtenant, easements in gross, licence agreements and other rights running for the benefit of Wynn Resorts Macau and/or appurtenant to the Site.

"Specified Time" means a day or time determined in accordance with Schedule 7 (*Timetables*).

"**Subconcession**" means any subconcession for the operation of games of chance and other games in casinos in the Macau SAR granted by Wynn Resorts Macau under the Concession Contract with the approval of Macau SAR.

"**Subordinated Loan**" means any loan advanced to any Subordinated Debtor (as defined in the Subordination Agreement) by any person which is a Subordinated Creditor (under and as defined in the Subordination Agreement) that has been subordinated pursuant to the Subordination Agreement (together, the "**Subordinated Loans**").

"**Subordination Agreement**" means the subordination agreement entered into between, among others, the Borrower and the Company, in a form approved by the Majority Lenders.

"Subsidiary" means, in relation to any company or corporation, a company or corporation:

- (a) which is controlled, directly or indirectly, by the first mentioned company or corporation;
- (b) more than half the issued equity share capital of which is beneficially owned, directly or indirectly, by the first mentioned company or corporation; or

(c) which is a Subsidiary of another Subsidiary of the first mentioned company or corporation,

and, for this purpose, a company or corporation shall be treated as being controlled by another if that other company or corporation is able to direct its affairs and/or to control the composition of its board of directors or equivalent body.

"**Tax**" means any tax, levy, impost, duty or other charge or withholding of a similar nature (including any penalty or interest payable in connection with any failure to pay or any delay in paying any of the same).

"Termination Date" means the date falling 48 Months after the date of this Agreement.

"**Total Commitments**" means the Base Currency amount that is the aggregate of the Total Facility A Commitments and the USD equivalent of the Total Facility B Commitments (converted from HKD to USD at the Fixed HKD/USD Exchange Rate), being a Base Currency amount of USD1,500,000,000 at the date of this Agreement.

"**Total Facility A Commitments**" means the aggregate of the Facility A Commitments, being USD312,500,000 at the date of this Agreement.

"**Total Facility B Commitments**" means the aggregate of the Facility B Commitments, being HKD9,262,500,000 at the date of this Agreement.

"**Trade Instruments**" means any performance bonds or advance payment bonds or documentary letters of credit issued in respect of the obligations of any Borrower Group Member arising in the ordinary course of trading of that Borrower Group Member, but excluding any Performance Bonds.

"**Transfer Certificate**" means a certificate substantially in the form set out in Schedule 4 (*Form of Transfer Certificate*) or any other form agreed between the Agent and the Borrower.

"Transfer Date" means, in relation to an assignment or a transfer, the later of:

- (a) the proposed Transfer Date specified in the relevant Assignment Agreement or Transfer Certificate; and
- (b) the date on which the Agent executes the relevant Assignment Agreement or Transfer Certificate.

"Unpaid Sum" means any sum due and payable but unpaid by an Obligor under the Finance Documents.

"US" means the United States of America.

"US Tax Obligor" means:

(a) the Borrower if it is resident for tax purposes in the US; or

(b) an Obligor some or all of whose payments under the Finance Documents are from sources within the US for US federal income tax purposes.

"Utilisation" means a utilisation of a Facility.

"Utilisation Date" means the date of a Utilisation, being the date on which a Loan is to be made.

"Utilisation Request" means a notice substantially in the form set out in Schedule 3 (Utilisation Request).

"**Voting Stock**" means, with respect to any Person as of any date, the Capital Stock of such Person that is at the time entitled to vote in the election of the Board of Directors of such Person.

"WML Corp" means WML Corp. Ltd., an exempted company incorporated with limited liability in the Cayman Islands.

"**WML Debt**" means Financial Indebtedness of the Company (or a Subsidiary of the Company, other than a member of the Borrower Group).

"**WML Finance**" means WML Finance I Limited, an exempted company incorporated with limited liability in the Cayman Islands.

"Wynn HK" means Wynn Resorts (Macau), Limited, a company incorporated in the Hong Kong SAR.

"Wynn Holdings" means Wynn Resorts (Macau) Holdings, Ltd., a company incorporated in the Isle of Man.

"Wynn International" means Wynn Resorts International, Ltd., a company incorporated in the Isle of Man.

"**Wynn Macau**" means the luxury hotel resort, casino, retail and entertainment complex located in peninsular Macau, owned and operated by the Wynn Resorts Macau, and including "Encore at Wynn Macau".

"Wynn Resorts" means Wynn Resorts, Limited, a company incorporated in the State of Nevada.

"Wynn Resorts Macau" means Wynn Resorts (Macau) S.A, a company incorporated in Macau.

1.2 Construction

- (a) Unless a contrary indication appears, any reference in this Agreement to:
 - (i) the "**Agent**", the "**Arranger**", any "**Finance Party**", any "**Lender**", the "**Borrower**", the "**Company**" or any "**Party**" shall be construed so as to include its successors in title, permitted assigns and permitted

transferees to, or of, its rights and/or obligations under the Finance Documents;

- (ii) "assets" includes present and future properties, revenues and rights of every description;
- (iii) a "**Finance Document**" or any other agreement or instrument is a reference to that Finance Document or other agreement or instrument as amended, novated, supplemented, extended or restated;
- (iv) a "group of Lenders" includes all the Lenders;
- (v) "**indebtedness**" includes any obligation (whether incurred as principal or as surety) for the payment or repayment of money, whether present or future, actual or contingent;
- (vi) a "**person**" includes any individual, firm, company, corporation, government, state or agency of a state or any association, trust, joint venture, consortium, partnership or other entity (whether or not having separate legal personality);
- (vii) a "**regulation**" includes any regulation, rule, official directive, request or guideline (whether or not having the force of law but, if not having the force of law, one with which entities to which the same applies customarily comply) of any governmental, intergovernmental or supranational body, agency, department or of any regulatory, self-regulatory or other authority or organisation;
- (viii) a provision of law is a reference to that provision as amended or re-enacted from time to time; and
- (ix) a time of day is a reference to Hong Kong time.
- (b) The determination of the extent to which a rate is "**for a period equal in length**" to an Interest Period shall disregard any inconsistency arising from the last day of that Interest Period being determined pursuant to the terms of this Agreement.
- (c) Section, Clause and Schedule headings are for ease of reference only.
- (d) Unless a contrary indication appears, a term used in any other Finance Document or in any notice given under or in connection with any Finance Document has the same meaning in that Finance Document or notice as in this Agreement.
- (e) A Default or an Event of Default is "continuing" if it has not been remedied or waived.
- (f) If a Default (including an Event of Default) occurs for a failure to deliver a required certificate, notice or other document in connection with another

default (an "**Initial Default**") then at the time such Initial Default is remedied or waived, such Default (including an Event of Default) for a failure to report or deliver a required certificate, notice or other document in connection with the Initial Default will also be cured without any further action (save for prompt (and in any case within three Business Days') written notification by the Borrower to the Agent of such remedy or waiver of that Initial Default, together with reasonable details thereof), unless an Acceleration Event has occurred and is continuing. Any Default (including an Event of Default) for the failure to comply with the time periods prescribed in Clause 19 (*Information Undertakings*), or otherwise to deliver any notice, certificate or other document, as applicable, even though such delivery is not within the prescribed period specified in this Agreement or any other Finance Document, shall be deemed to be cured upon the delivery of any such report required by such covenant or notice, certificate or other document, as applicable (even though such delivery is not within the prescribed period specified in this Agreement or any other Finance Document) together with written notification to the Agent by the Borrower of such late delivery, unless an Acceleration Event has occurred and is continuing.

- (g) An Acceleration Event is "**continuing**" if the relevant Acceleration Event has occurred and the underlying notice of acceleration has not been withdrawn by the Agent (acting on the instructions of the Lenders).
- (h) Any references to any matter being "**permitted**" shall include references to such matters not being expressly prohibited by the Finance Documents or otherwise approved by the requisite Lenders.
- (i)
- (i) For the purposes of determining compliance with any basket amount, threshold and any other exceptions to any undertaking under Clause 21 (*General Undertakings*) and any Event of Default under Clause 22 (*Events of Default*), the equivalent to any amount in US Dollars shall be determined as at the time of the applicable incurrence, disposal, acquisition, investment, loan, guarantee or other relevant action.
- (ii) No breach of any undertaking under Clause 21 (*General Undertakings*) or Event of Default under Clause 22 (*Events of Default*) shall arise merely as a result of a subsequent change in the USD equivalent of any amount due to fluctuation in exchange rates.
- (j) Any reference to a specific provision of the Concession Contract, the Land Concession Contract or the Cotai Land Concession Contract shall, upon any Concession Extension, be interpreted as a reference to any provision in the amended Concession Contract, Land Concession Contract or Cotai Land Concession Contract, or in any documentation that replaces the original Concession Contract (as applicable), Land Concession Contract or Cotai Land Concession Contract (as applicable) that is equivalent in meaning or effect.

1.3 Currency symbols and definitions

- (a) "US\$", "USD" and "US dollars" denote the lawful currency of the United States of America.
- (b) "**HK\$**", "**HKD**" or "**HK dollars**" denote the lawful currency of Hong Kong SAR.

1.4 Third party rights

- (a) Unless expressly provided to the contrary in a Finance Document a person who is not a Party has no right under the Contracts (Rights of Third Parties) Act 1999 (the "**Third Parties Act**") to enforce or to enjoy the benefit of any term of this Agreement.
- (b) Subject to Clause 34.3 (*Other exceptions*) but otherwise notwithstanding any term of any Finance Document, the consent of any person who is not a Party is not required to rescind or vary this Agreement at any time.

1.5 Split Voting

A Lender may by notice to the Agent divide its participation(s) in any outstanding Loan(s) or Commitment(s) into separate amounts to reflect participations or similar arrangements and for the purposes solely of counting towards any decision or vote by that Lender require those separate amounts to be counted separately in that decision or vote for the purposes of the Finance Documents.

SECTION 2 THE FACILITIES

2. THE FACILITIES

2.1 The Facilities

Subject to the terms of this Agreement, the Lenders make available to the Borrower:

- (a) a USD revolving loan facility in an aggregate amount equal to the Total Facility A Commitments; and
- (b) a HKD revolving loan facility in an aggregate amount equal to the Total Facility B Commitments.

2.2 Increase

- (a) The Borrower may by giving prior notice to the Agent at any time, request that the Commitments relating to any Facility be increased (and the Commitments relating to that Facility shall be so increased) in an aggregate amount in the Base Currency (for the Facilities, when taken together and determined by converting any HKD amounts to USD at the Fixed HKD/USD Exchange Rate) of up to the amount of USD1,000,000,000 as follows:
 - (i) the increased Commitments will be assumed by one or more Eligible Institutions (each an "Increase Lender") each of which confirms in writing (whether in the relevant Increase Confirmation or otherwise) its willingness to assume and does assume all the obligations of a Lender corresponding to that part of the increased Commitments which it is to assume, as if it had been an Original Lender in respect of those Commitments;
 - (ii) each of the Obligors and any Increase Lender shall assume obligations towards one another and/or acquire rights against one another as the Obligors and the Increase Lender would have assumed and/or acquired had the Increase Lender been an Original Lender in respect of that part of the increased Commitments which it is to assume;
 - (iii) each Increase Lender shall become a Party as a "Lender" and any Increase Lender and each of the other Finance Parties shall assume obligations towards one another and acquire rights against one another as that Increase Lender and those Finance Parties would have assumed and/or acquired had the Increase Lender been an Original Lender in respect of that part of the increased Commitments which it is to assume;
 - (iv) the Commitments of the other Lenders shall continue in full force and effect; and

- (v) any increase in the Commitments relating to a Facility shall take effect on the date specified by the Borrower in the notice referred to above or any later date on which the Agent executes an otherwise duly completed Increase Confirmation delivered to it by the relevant Increase Lender.
- (b) The Agent shall, subject to paragraph (c) below, as soon as reasonably practicable after receipt by it of a duly completed Increase Confirmation appearing on its face to comply with the terms of this Agreement and delivered in accordance with the terms of this Agreement, execute that Increase Confirmation.
- (c) The Agent shall only be obliged to execute an Increase Confirmation delivered to it by an Increase Lender once:
 - (i) it is satisfied that the aggregate amount of the increased Commitments set out in the Increase Confirmation, when taken together with all other amounts of Total Commitments increased pursuant to this Clause 2.2, does not exceed USD1,000,000,000 (as determined by converting any HKD amounts to USD at the Fixed HKD/USD Exchange Rate);
 - (ii) it is satisfied it has complied with all necessary "know your customer" or other similar checks under all applicable laws and regulations in relation to the assumption of the increased Commitments by that Increase Lender; and
 - (iii) it has received a certificate from the Borrower dated no earlier than the date of the relevant Increase Confirmation and addressed to it (for and on behalf of the Finance Parties) and (if the relevant Increase Lender is not a Finance Party) the relevant Increase Lender, certifying that:
 - (A) no Default has occurred and is continuing nor would result from the increase in Commitment(s) contemplated by the relevant Increase Confirmation; and
 - (B) it is in compliance with Clause 20.2 (*Financial condition*) as determined on a *pro forma* basis at the date of the relevant Increase Confirmation after giving effect to the increase in the Commitment(s) contemplated by the relevant Increase Confirmation.
- (d) Each Increase Lender, by executing the Increase Confirmation, confirms (for the avoidance of doubt) that the Agent has authority to execute on its behalf any amendment or waiver that has been approved by or on behalf of the requisite Lender or Lenders in accordance with this Agreement on or prior to the date on which the increase becomes effective in accordance with this Agreement and that it is bound by that decision to the same extent as it would have been had it been an Original Lender.

- (e) The Borrower shall promptly on demand pay the Agent the amount of all costs and expenses (including legal fees) reasonably incurred by it in connection with any increase in Commitments under this Clause 2.2.
- (f) The Increase Lender shall, on the date upon which the increase takes effect, pay to the Agent (for its own account) a fee in an amount equal to the fee which would be payable under Clause 23.4 (*Assignment or transfer fee*) if the increase was a transfer pursuant to Clause 23.6 (*Procedure for transfer*) and if the Increase Lender was a New Lender.
- (g) The Borrower may pay to the Increase Lender a fee in the amount and at the times agreed between the Borrower and the Increase Lender in a letter between the Borrower and the Increase Lender setting out that fee. A reference in this Agreement to a Fee Letter shall include any letter referred to in this paragraph (g).
- (h) Neither the Agent nor any Lender shall have any obligation to find an Increase Lender and in no event shall any Lender whose Commitment is replaced by an Increase Lender be required to pay or surrender any of the fees received by such Lender pursuant to the Finance Documents.
- (i) Clause 23.5 (*Limitation of responsibility of Existing Lenders*) shall apply *mutatis mutandis* in this Clause 2.2 in relation to an Increase Lender as if references in that Clause to:
 - (i) an "Existing Lender" were references to all the Lenders immediately prior to the relevant increase;
 - (ii) the "New Lender" were references to that "Increase Lender"; and
 - (iii) a "**re-transfer**" and "**re-assignment**" were references to respectively a "**transfer**" and "**assignment**".

2.3 Finance Parties' rights and obligations

- (a) The obligations of each Finance Party under the Finance Documents are several. Failure by a Finance Party to perform its obligations under the Finance Documents does not affect the obligations of any other Party under the Finance Documents. No Finance Party is responsible for the obligations of any other Finance Party under the Finance Documents.
- (b) The rights of each Finance Party under or in connection with the Finance Documents are separate and independent rights and any debt arising under the Finance Documents to a Finance Party from an Obligor is a separate and independent debt in respect of which a Finance Party shall be entitled to enforce its rights in accordance with paragraph (c) below. The rights of each Finance Party include any debt owing to that Finance Party under the Finance Documents and, for the avoidance of doubt, any part of a Loan or any other amount owed by an Obligor which relates to a Finance Party's participation in a Facility or its role under a Finance Document (including any such amount

payable to the Agent on its behalf) is a debt owing to that Finance Party by that Obligor.

(c) A Finance Party may, except as specifically provided in the Finance Documents, separately enforce its rights under or in connection with the Finance Documents.

2.4 The Company's Agent

- (a) The Company, by its execution of this Agreement irrevocably appoints the Borrower (acting through one or more authorised signatories) to act on its behalf as its agent in relation to the Finance Documents and irrevocably authorises:
 - (i) the Borrower on its behalf to supply all information concerning the Company contemplated by this Agreement to the Finance Parties and to give all notices and instructions, to make such agreements and to effect the relevant amendments, supplements and variations capable of being given, made or effected by the Company notwithstanding that they may affect the Company, without further reference to or the consent of the Company; and
 - (ii) each Finance Party to give any notice, demand or other communication to the Company pursuant to the Finance Documents to the Borrower,

and in each case, the Company shall be bound as though the Company itself had given the notices and instructions or executed or made the agreements or effected the amendments, supplements or variations, or received the relevant notice, demand or other communication.

(b) Every act, omission, agreement, undertaking, settlement, waiver, amendment, supplement, variation, notice or other communication given or made by the Company's Agent or given to the Company's Agent under any Finance Document on behalf of the Company or in connection with any Finance Document (whether or not known to the Company) shall be binding for all purposes on the Company as if the Company had expressly made, given or concurred with it. In the event of any conflict between any notices or other communications of the Company's Agent and the Company, those of the Company's Agent shall prevail.

3. PURPOSE

3.1 Purpose

- (a) The Borrower shall apply all amounts borrowed by it under the Facilities towards:
 - (i) for financing costs, fees and expenses (including, without limitation, the commitment fee and the upfront fee, but excluding any interest

pursuant to Clause 8.2 (Payment of interest)) in respect of the Facilities;

- (ii) refinancing of the Existing Facilities; and
- (iii) for ongoing working capital needs and general corporate purposes of the Group,

as described, in respect of paragraphs (a)(i) and (ii) above, in the Funds Flow Statement.

(b) The Borrower shall not (and the Borrower shall ensure that no member of the Borrower Group will) apply any proceeds of any Loan towards the acquisition, maintenance or repair of equipment and utensils used in the operation of casino games of chance or other forms of gaming.

3.2 Monitoring

No Finance Party is bound to monitor or verify the application of any amount borrowed pursuant to this Agreement.

4. CONDITIONS OF UTILISATION

4.1 **Initial conditions precedent**

- (a) The Borrower may not deliver a Utilisation Request unless the Agent has received all of the documents and other evidence listed in Schedule 2 (*Conditions Precedent*) in form and substance satisfactory to the Agent. The Agent shall notify the Borrower and the Lenders promptly upon being so satisfied.
- (b) Other than to the extent that the Majority Lenders notify the Agent in writing to the contrary before the Agent gives the notification described in paragraph (a) above, the Lenders authorise (but do not require) the Agent to give that notification. The Agent shall not be liable for any damages, costs or losses whatsoever as a result of giving any such notification.

4.2 Further conditions precedent

The Lenders will only be obliged to comply with Clause 5.4 (*Lenders' participation*) if on the date of the Utilisation Request and on the proposed Utilisation Date:

- (a) in the case of a Rollover Loan, no Event of Default is continuing or would result from the proposed Loan and, in the case of any other Loan, no Default is continuing or would result from the proposed Loan; and
- (b) the Repeating Representations to be made by each Obligor are true in all material respects.

4.3 Maximum number of Loans

The Borrower may not deliver a Utilisation Request if as a result of the proposed Utilisation:

- (a) 10 or more Facility A Loans would be outstanding; or
- (b) 10 or more Facility B Loans would be outstanding.

SECTION 3 UTILISATION

5. UTILISATION

5.1 **Delivery of a Utilisation Request**

The Borrower may utilise a Facility by delivery to the Agent of a duly completed Utilisation Request not later than the Specified Time.

5.2 **Completion of a Utilisation Request**

- (a) Each Utilisation Request is irrevocable and will not be regarded as having been duly completed unless:
 - (i) Facility A and Facility B are utilised at the same time on a pro rata basis in accordance with the Relevant Ratio;
 - (ii) the proposed Utilisation Date is a Business Day within the Availability Period;
 - (iii) the currency and amount of the Utilisation comply with Clause 5.3 (Currency and amount); and
 - (iv) the proposed Interest Period complies with Clause 9 (Interest Periods).
- (b) Only one Loan under each Facility may be requested in each Utilisation Request.

5.3 **Currency and amount**

- (a) The currencies specified in a Utilisation Request must be USD and HKD.
- (b) The amount of the proposed Loans must be:
 - (i) in aggregate between the Facility A Loan and the Facility B Loan requested in the relevant Utilisation Request, a minimum of USD10,000,000 (as determined by converting any HKD amounts to USD at the Fixed HKD/USD Exchange Rate) and an integral multiple of USD2,000,000 (as determined by converting any HKD amounts to USD at the Fixed HKD/USD Exchange Rate) (save to the extent otherwise required to comply with the Relevant Ratio); and
 - (ii) in any event, such that its amount is less than or equal to the Available Facility in respect of that Facility.

5.4 Lenders' participation

(a) If the conditions set out in this Agreement have been met and subject to Clause 6.1 (*Repayment of Loans*) each Lender shall make its participation in each Loan available by the Utilisation Date through its Facility Office.

- (b) The amount of each Lender's participation in each Loan will be equal to the proportion borne by its Available Commitment to the Available Facility immediately prior to making the Loan.
- (c) The Agent shall notify each Lender of the amount and currency of each Loan, the amount of its participation in that Loan and, if different, the amount of that participation to be made available in accordance with Clause 28.1 (*Payments to the Agent*), in each case by the Specified Time.

5.5 Cancellation of Commitment

The Facility A Commitments and Facility B Commitments which, at that time, are unutilised shall be immediately cancelled at the end of the Availability Period.

5.6 **Right of cancellation in relation to a Defaulting Lender**

- (a) If any Lender becomes a Defaulting Lender, the Borrower may, at any time whilst the Lender continues to be a Defaulting Lender, give the Agent five Business Days' notice of cancellation of each Available Commitment of that Lender.
- (b) On the notice referred to in paragraph (a) above becoming effective, each Available Commitment of the Defaulting Lender shall be immediately reduced to zero.
- (c) The Agent shall as soon as practicable after receipt of a notice referred to in paragraph (a) above, notify all the Lenders.
- (d) For the avoidance of doubt, the Parties acknowledge that there will be a change in the then prevailing Relevant Ratio immediately after such cancellation and prepayment.

SECTION 4 REPAYMENT, PREPAYMENT AND CANCELLATION

6. REPAYMENT

6.1 Repayment of Loans

- (a) The Borrower shall repay each Loan on the last day of its Interest Period.
- (b) Without prejudice to the Borrower's obligation under paragraph (a) above, if:
 - (i) one or more Loans are to be made available to the Borrower:
 - (A) on the same day that a maturing Loan is due to be repaid by the Borrower;
 - (B) in the same currency as the maturing Loan; and
 - (C) in whole or in part for the purpose of refinancing the maturing Loan; and
 - (ii) the proportion borne by each Lender's participation in the maturing Loan to the amount of that maturing Loan is the same as the proportion borne by that Lender's participation in the new Loans to the aggregate amount of those new Loans,

the aggregate amount of the new Loans shall, unless the Borrower notifies the Agent to the contrary in the relevant Utilisation Request, be treated as if applied in or towards repayment of the maturing Loan so that:

- (D) if the amount of the maturing Loan exceeds the aggregate amount of the new Loans:
 - (1) the Borrower will only be required to make a payment under Clause 28.1 (*Payments to the Agent*) in an amount in the relevant currency equal to that excess; and
 - (2) each Lender's participation in the new Loans shall be treated as having been made available and applied by the Borrower in or towards repayment of that Lender's participation in the maturing Loan and that Lender will not be required to make a payment under Clause 28.1 (*Payments to the Agent*) in respect of its participation in the new Loans; and
- (E) if the amount of the maturing Loan is equal to or less than the aggregate amount of the new Loans:
 - (1) the Borrower will not be required to make a payment under Clause 28.1 (*Payments to the Agent*); and

(2) each Lender will be required to make a payment under Clause 28.1 (*Payments to the Agent*) in respect of its participation in the new Loans only to the extent that its participation in the new Loans exceeds that Lender's participation in the maturing Loan and the remainder of that Lender's participation in the new Loans shall be treated as having been made available and applied by the Borrower in or towards repayment of that Lender's participation in the maturing Loan.

7. PREPAYMENT AND CANCELLATION

7.1 Illegality

If, in any applicable jurisdiction, it becomes unlawful for any Lender to perform any of its obligations as contemplated by this Agreement or to fund or maintain its participation in any Loan:

- (a) that Lender shall promptly notify the Agent upon becoming aware of that event;
- (b) upon the Agent notifying the Borrower, the Available Commitment of that Lender will be immediately cancelled; and
- (c) to the extent that the Lender's participation has not been transferred pursuant to Clause 34.6 (*Replacement of Lenders*), the Borrower shall repay that Lender's participation in the Loans made to the Borrower on the last day of the Interest Period for each Loan occurring after the Agent has notified the Borrower or, if earlier, the date specified by the Lender in the notice delivered to the Agent (being no earlier than the last day of any applicable grace period permitted by law) and that Lender's corresponding Commitment shall be immediately cancelled in the amount of the participations repaid.

For the avoidance of doubt, the Parties acknowledge that there will be a change in the then prevailing Relevant Ratio immediately after such cancellation and prepayment.

7.2 Change of control

- (a) If a Change of Control occurs:
 - (i) the Borrower shall promptly notify the Agent upon becoming aware of that event;
 - (ii) a Lender shall not be obliged to fund a Utilisation (except for a Rollover Loan); and
 - (iii) if the Majority Lenders so require and notify the Agent within 45 days of the Borrower notifying the Agent of that event, the Agent shall, by not less than 30 days' notice to the Borrower, cancel the Available Commitment of each Lender and declare all Loans, together with

accrued interest, and all other amounts accrued or outstanding under the Finance Documents immediately due and payable, whereupon such Available Commitments will be immediately cancelled, the Facilities shall immediately cease to be available for further utilisation and all such Loans, accrued interest and other amounts shall become immediately due and payable.

- (b) For the purpose of paragraph (a) above "**Change of Control**" means:
 - (i) a sale, lease, transfer or other disposal by the Borrower of all or substantially all of the assets or business of the Borrower Group, in each case, whether through a single transaction or a series of transactions (and whether related or not, whether direct or indirect and whether or not voluntary);
 - (ii) Wynn Resorts ceasing to legally and beneficially own and control, directly or indirectly, more than fifty (50) per cent. of the entire outstanding Capital Stock of the Borrower measured by voting power; or
 - (iii) the Borrower ceasing to beneficially own and control, directly or indirectly, the entire outstanding Capital Stock of Wynn Resorts Macau measured by voting power, other than as a result of a Permitted Selldown.

In determining whether a Change of Control (as contemplated by paragraph (b)(iii) above) has occurred, any class of Capital Stock of Wynn Resorts Macau which has a de minimis economic interest and which has been created and issued to the Executive Director solely for the purposes of complying with applicable Legal Requirements in the Macau SAR shall be deemed to not form part of the outstanding Capital Stock of Wynn Resorts Macau.

7.3 Property Mandatory Prepayment Event

If a Property Mandatory Prepayment Event occurs:

- (a) the Borrower shall promptly notify the Agent upon becoming aware of that event;
- (b) a Lender shall not be obliged to fund a Utilisation (except for a Rollover Loan); and
- (c) if a Lender so requires and notifies the Agent within 90 days of the Borrower notifying the Agent of the event, the Agent shall, by not less than 30 days' notice to the Borrower at any time after the occurrence of such Property Mandatory Prepayment Event, cancel each Available Commitment of that Lender and declare the participations of that Lender in all Loans, together with accrued interest, and all other amounts accrued or outstanding under the Finance Documents immediately due and payable, whereupon such Available Commitments will be immediately cancelled, the Commitments of that Lender
 - 47

shall immediately cease to be available for further utilisation and all such Loans, accrued interest and other amounts shall become immediately due and payable.

For the avoidance of doubt, the Parties acknowledge that there will be a change in the then prevailing Relevant Ratio immediately after such cancellation and prepayment.

7.4 Voluntary cancellation

The Borrower may, if it gives the Agent not less than three Business Days' (or such shorter period as the Majority Lenders may agree) prior notice, cancel the whole or any part (being, in respect of Facility A, a minimum amount of USD2,000,000 and, in respect of Facility B, a minimum amount of HKD15,000,000 (save to the extent otherwise required to comply with the Relevant Ratio)) of an Available Facility **provided that** the ratio of the amount of the Available Facility under Facility A to be voluntarily cancelled to the amount of the Available Facility under Facility B to be voluntarily cancelled shall be equal to the Relevant Ratio (and vice versa) and both Facility A and Facility B must be voluntarily cancelled on a pro rata basis based on the Relevant Ratio at the same time. Any cancellation under this Clause 7.4 shall reduce the Commitments of the Lenders rateably under that Facility.

7.5 Voluntary prepayment of Loans

The Borrower may, if it gives the Agent not less than three Business Days' (or such shorter period as the Majority Lenders may agree) prior notice, prepay the whole or any part of a Loan (but if in part, being an amount that reduces, in respect of a Facility A Loan, the amount of the Loan by a minimum amount of USD2,000,000 and, in respect of a Facility B Loan, that reduces the amount of the Loan by a minimum amount of HKD15,000,000 (save to the extent otherwise required to comply with the Relevant Ratio)) **provided that** the Borrower shall prepay the Facility A Loan(s) and the Facility B Loan(s) at the same time on a pro rata basis in accordance with the Relevant Ratio.

7.6 **Right of repayment and cancellation in relation to a single Lender**

(a) If:

- (i) any sum payable to any Lender by an Obligor is required to be increased under Clause 12.2 (*Tax gross-up*);
- (ii) any Lender becomes a Disrupted Lender or a Non-Consenting Lender; or
- (iii) any Lender claims indemnification from the Borrower under Clause 12.3 (*Tax indemnity*) or Clause 13.1 (*Increased Costs*),

the Borrower may, whilst the circumstance giving rise to the requirement for that increase or indemnification continues, give the Agent notice of cancellation of the Commitment of that Lender and its intention to procure the repayment of that Lender's participation in the Loans.

- (b) On receipt of a notice of cancellation referred to in paragraph (a) above, the Available Commitment of that Lender shall be immediately reduced to zero.
- (c) On the last day of each Interest Period which ends after the Borrower has given notice of cancellation under paragraph (a) above (or, if earlier, the date specified by the Borrower in that notice), the Borrower shall repay that Lender's participation in the Loans and that Lender's corresponding Commitment shall be immediately cancelled in the amount of the participations repaid.
- (d) For the avoidance of doubt, the Parties acknowledge that there will be a change in the then prevailing Relevant Ratio immediately after such cancellation and prepayment.

7.7 Restrictions

- (a) Any notice of cancellation or prepayment given by any Party under this Clause 7 shall be irrevocable and, unless a contrary indication appears in this Agreement, shall specify the date or dates upon which the relevant cancellation or prepayment is to be made and the amount of that cancellation or prepayment.
- (b) Any prepayment under this Agreement shall be made together with accrued interest on the amount prepaid and, subject to any Break Costs, without premium or penalty.
- (c) Unless a contrary indication appears in this Agreement, any part of a Facility which is prepaid or repaid may be reborrowed in accordance with the terms of this Agreement.
- (d) The Borrower shall not repay or prepay all or any part of the Loans or cancel all or any part of the Commitments except at the times and in the manner expressly provided for in this Agreement.
- (e) Subject to Clause 2.2 (*Increase*), no amount of the Total Commitments cancelled under this Agreement may be subsequently reinstated.
- (f) If the Agent receives a notice under this Clause 7 it shall promptly forward a copy of that notice to either the Borrower or the affected Lender, as appropriate.
- (g) If all or part of any Lender's participation in a Loan under a Facility is repaid or prepaid and is not available for redrawing (other than by operation of Clause 4.2 (*Further conditions precedent*)), an amount of that Lender's Commitment (equal to the amount of the participation which is repaid or prepaid) in respect of that Facility will be deemed to be cancelled on the date of repayment or prepayment.

7.8 Application of prepayments

Any prepayment of a Loan pursuant to Clause 7.2 (*Change of control*), Clause 7.3 (*Property Mandatory Prepayment Event*) or Clause 7.5 (*Voluntary prepayment of Loans*) shall be applied *pro rata* to each Lender's participation in that Loan.

SECTION 5 COSTS OF UTILISATIONS

8. INTEREST

8.1 Calculation of interest

The rate of interest on each Loan for each Interest Period is the percentage rate per annum which is the aggregate of the applicable:

- (a) Margin; and
- (b) LIBOR or, in relation to any Loan in HK dollars, HIBOR.

8.2 **Payment of interest**

- (a) The Borrower shall pay accrued interest on that Loan on the last day of each Interest Period (and, if the Interest Period is longer than six Months, on the dates falling at six monthly intervals after the first day of the Interest Period).
- (b) If the Annual Financial Statements and related Compliance Certificate received by the Agent show that a higher Margin should have applied during a certain period (the "Applicable Period" for the purposes of this paragraph (b)), the next payment of interest falling due on each Loan shall be increased to the extent necessary (in the opinion of the Agent (acting reasonably)) to put the Lenders (but only in respect of payments to the Lenders participating in the relevant Facility or Facilities (as the case may be) both during such Applicable Period and at the time at which such increase is actually made) in the position they would have been in had the appropriate rate of the Margin applied during such Applicable Period.
- (c) If the Annual Financial Statements and related Compliance Certificate received by the Agent show that a lower Margin should have applied during a certain period (the "Applicable Period" for the purposes of this paragraph (c)), the next payment of interest falling due on any Loan (the "Next Interest Payment") shall be reduced to the extent necessary (in the opinion of the Agent (acting reasonably)) to put the Company (but only in respect of payments to the Lenders participating in the relevant Facility or Facilities (as the case may be) both during such Applicable Period and at the time at which such reduction is actually made) in the position it would have been in had the appropriate rate of the Margin applied during such Applicable Period, provided that the Next Interest Payment shall not be reduced by an amount greater than the portion of the Next Interest Payment which is attributable to accrued Margin (provided further that the next payment(s) of interest due on that Loan shall, subject (in each case) to the provisos in this paragraph (c), be reduced pro tanto until such time as the aggregate amount of such reductions equal the difference between the interest paid for such Applicable Period and the interest that would have been paid for such Applicable Period had the appropriate rate of the Margin applied during such Applicable Period and the interest paid for such Applicable Period and the interest paid for such Applicable Period and the interest paid for such Applicable Period.

8.3 **Default interest**

- (a) If an Obligor fails to pay any amount payable by it under a Finance Document on its due date, interest shall accrue on the overdue amount from the due date up to the date of actual payment (both before and after judgment) at a rate which, subject to paragraph (b) below, is two per cent. per annum higher than the rate which would have been payable if the overdue amount had, during the period of non-payment, constituted a Loan in the currency of the overdue amount for successive Interest Periods, each of a duration selected by the Agent (acting reasonably). Any interest accruing under this Clause 8.3 shall be immediately payable by the Obligor on demand by the Agent.
- (b) If any overdue amount consists of all or part of a Loan which became due on a day which was not the last day of an Interest Period relating to that Loan:
 - (i) the first Interest Period for that overdue amount shall have a duration equal to the unexpired portion of the current Interest Period relating to that Loan; and
 - (ii) the rate of interest applying to the overdue amount during that first Interest Period shall be two per cent. per annum higher than the rate which would have applied if the overdue amount had not become due.
- (c) Default interest (if unpaid) arising on an overdue amount will be compounded with the overdue amount at the end of each Interest Period applicable to that overdue amount but will remain immediately due and payable.

8.4 Notification of rates of interest

- (a) The Agent shall promptly notify the Lenders and the Borrower of the determination of a rate of interest under this Agreement.
- (b) The Agent shall promptly notify the Borrower of each Funding Rate relating to a Loan.

8.5 Modification and/or discontinuation of certain benchmarks

Without prejudice to any other provisions of this Agreement (including this Clause 8 and Clause 34.2 (*All Lender matters*)), each Obligor acknowledges and agrees for the benefit of the Finance Parties that:

- (a) IBOR benchmarks (i) may be subject to methodological or other changes which could affect their value, or (ii) may not comply with applicable laws and regulations (such as Regulation (EU) 2016/1011 of the European Parliament and of the Council, as amended (EU Benchmarks Regulation)) and/or (iii) may be permanently discontinued; and
- (b) the occurrence of any of the aforementioned events and/or a Screen Rate Replacement Event (as defined in paragraph (d) of Clause 34.4 (*Replacement of Screen Rate*) may have adverse consequences which may materially impact

the economics of the financing transaction contemplated under this Agreement.

9. INTEREST PERIODS

9.1 Selection of Interest Periods

- (a) The Borrower (or the Company on behalf of the Borrower) may select an Interest Period for a Loan in the Utilisation Request for that Loan.
- (b) Subject to this Clause 9, the Borrower (or the Company) may select:
 - (i) if the Loan is in US dollars, an Interest Period of one, three or six Months; or
 - (ii) if the Loan is in HK dollars, an Interest Period of one, three or six Months,

or, in each case, any other period agreed between the Borrower, the Agent and all the Lenders.

- (c) An Interest Period for a Loan shall not extend beyond the Termination Date.
- (d) Each Interest Period for a Loan shall start on the Utilisation Date.
- (e) A Loan has one Interest Period only.

9.2 Changes to Interest Periods

If the Agent makes any of the changes to an Interest Period referred to in this Clause 9.2, it shall promptly notify the Borrower and the Lenders.

9.3 Non-Business Days

If an Interest Period would otherwise end on a day which is not a Business Day, that Interest Period will instead end on the next Business Day in that calendar month (if there is one) or the preceding Business Day (if there is not).

10. CHANGES TO THE CALCULATION OF INTEREST

10.1 Absence of quotations

Subject to Clause 10.2 (*Market disruption*), if LIBOR or HIBOR is to be determined by reference to the Reference Banks but a Reference Bank does not supply a quotation on the Quotation Day, the applicable LIBOR or HIBOR shall be determined on the basis of the quotations of the remaining Reference Banks. Where LIBOR or HIBOR is to be determined by reference to the Reference Banks and none or only one Reference Bank supplies a quotation, then LIBOR or, as the case may be, HIBOR shall be treated as incapable of being determined.

10.2 Market disruption

- (a) If a Market Disruption Event occurs in relation to a Loan for any Interest Period, the Agent shall promptly notify the Borrower and the other Lenders of the fact and that this Clause 10.2 is in operation.
- (b) For the purpose of this Clause 10.2, "**Market Disruption Event**" means:
 - (i) in the case of a Facility A Loan:
 - (A) it is not possible to determine LIBOR in accordance with the provisions of this Agreement and Clause 10.1 (*Absence of quotations*); or
 - (B) before 12 p.m. on the next Business Day following the Quotation Day for the relevant Interest Period, the Agent has been notified by a Lender or Lenders (whose participations in such Loan exceed 50 per cent of the Loan) (each a "Disrupted Lender") that the cost to it/them of obtaining matching deposits in the London interbank market would be in excess of LIBOR;
 - (ii) in the case of a Facility B Loan:
 - (A) it is not possible to determine HIBOR in accordance with the provisions of this Agreement and Clause 10.1 (*Absence of quotations*); or
 - (B) before the close of business in Hong Kong on the Quotation Day for the relevant Interest Period, the Agent has been notified by a Lender or Lenders (whose participations in such Loan exceed 50 per cent of the Loan) (each a "Disrupted Lender") that the cost to it/them of obtaining matching deposits in the Hong Kong interbank market would be in excess of HIBOR.
- (c) Within five Business Days of the Agent notifying the Borrower in accordance with paragraph (a) above, the Borrower and the Agent shall enter into good faith negotiations for a period of up to thirty days with a view to agreeing an alternative basis for determining the rate of interest applicable to the relevant Loans. Any alternative basis agreed shall be binding on all parties hereto until (subject to the terms of such agreement) the Market Disruption Event referred to in paragraph (a) above is at an end and the Agent has notified the Borrower accordingly.
- (d) If no alternative basis is agreed pursuant to paragraph (c) above by the earlier of (i) the thirty-day period provided in paragraph (c) above and (ii) the relevant Utilisation Date (where the notification under paragraph (a) above applies to any Loan which has not been made) or the last day of the Interest Period (where the notification under paragraph (a) above applies to a Loan

which is outstanding), then each Lender participating in the relevant Loan shall, acting reasonably, certify an alternative basis for maintaining its participation in the relevant Loan which may include an alternative method of fixing the interest rate, alternative Interest Periods or alternative currencies but such alternative basis must reflect its cost of funding its participation in the relevant Loan from whatever sources it may in good faith select plus the applicable interest margin applicable to that Lender's participation in the relevant Loan. Each alternative basis so certified shall be binding on the Borrower and the certifying Lender and treated as part of this Agreement.

10.3 Break Costs

- (a) The Borrower shall, within five days of demand by a Finance Party, pay to that Finance Party its Break Costs attributable to all or any part of a Loan or Unpaid Sum being paid by the Borrower on a day other than the last day of an Interest Period for that Loan or Unpaid Sum.
- (b) Each Lender shall, as soon as reasonably practicable after a demand by the Agent or the Borrower, provide a certificate confirming the amount and providing reasonable supporting evidence of its Break Costs for any Interest Period in which they accrue.

11. **FEES**

11.1 Commitment fee

- (a) The Borrower shall pay to the Agent (for the account of each Lender) a fee computed at the rate of 40 per cent. of the applicable Margin (in respect of the period for which it is being calculated) on that Lender's Available Commitment for each Facility for each day during the Availability Period.
- (b) The accrued commitment fee is payable in arrears on the last day of each successive period of three Months which ends during the Availability Period, on the last day of the Availability Period and, if cancelled in full, on the cancelled amount of the relevant Lender's Commitment at the time the cancellation is effective.
- (c) No commitment fee is payable to the Agent (for the account of a Lender) on any Available Commitment of that Lender for any day on which that Lender is a Defaulting Lender.

11.2 Arrangement fee

The Borrower shall pay to the Arranger an arrangement fee in the amount and at the times agreed in a Fee Letter.

11.3 Agency fee

The Borrower shall pay to the Agent (for its own account) an agency fee in the amount and at the times agreed in a Fee Letter.

11.4 Fees Payable

No fees shall be payable to any Impaired Finance Party.

SECTION 6 ADDITIONAL PAYMENT OBLIGATIONS

12. TAX GROSS-UP AND INDEMNITIES

12.1 **Definitions**

In this Clause 12:

"Tax Credit" means a credit against, relief or remission for, or repayment of any Tax.

"**Tax Deduction**" means a deduction or withholding for or on account of Tax from a payment under a Finance Document, other than a FATCA Deduction.

"**Tax Payment**" means an increased payment made by an Obligor to a Finance Party under Clause 12.2 (*Tax gross-up*) or a payment under Clause 12.3 (*Tax indemnity*).

Unless a contrary indication appears, in this Clause 12 a reference to "**determines**" or "**determined**" means a determination made in the absolute discretion of the person making the determination.

12.2 Tax gross-up

- (a) All payments to be made by an Obligor to any Finance Party under the Finance Documents shall be made free and clear of and without any Tax Deduction unless such Obligor is required to make a Tax Deduction, in which case the sum payable by such Obligor (in respect of which such Tax Deduction is required to be made) shall be increased to the extent necessary to ensure that such Finance Party receives a sum net of any deduction or withholding equal to the sum which it would have received had no such Tax Deduction been made or required to be made.
- (b) The Borrower shall promptly upon becoming aware that an Obligor must make a Tax Deduction (or that there is any change in the rate or the basis of a Tax Deduction) notify the Agent accordingly. Similarly, a Lender shall notify the Agent on becoming so aware in respect of a payment payable to that Lender. If the Agent receives such notification from a Lender it shall notify the Borrower and that Obligor.
- (c) If an Obligor is required to make a Tax Deduction, that Obligor shall make that Tax Deduction and any payment required in connection with that Tax Deduction within the time allowed and in the minimum amount required by law.
- (d) Within 30 days of making either a Tax Deduction or any payment required in connection with that Tax Deduction, the Obligor making that Tax Deduction shall deliver to the Agent for the Finance Party entitled to the payment evidence reasonably satisfactory to that Finance Party that the Tax Deduction has been made or (as applicable) any appropriate payment paid to the relevant taxing authority.

12.3 Tax indemnity

- (a) Without prejudice to Clause 12.2 (*Tax gross-up*), if any Finance Party is required to make any payment of or on account of Tax on or in relation to any sum received or receivable under the Finance Documents (including any sum deemed for the purposes of Tax to be received or receivable by such Finance Party whether or not actually received or receivable) or if any liability in respect of any such payment is asserted, imposed, levied or assessed against any Finance Party, the Borrower shall, within 15 days of demand of the Agent, promptly indemnify the Finance Party which suffers a loss or liability as a result against such payment or liability, together with any interest, penalties, costs and expenses payable or incurred in connection therewith, **provided that** this Clause 12.3 shall not apply to:
 - (i) any Tax imposed on and calculated by reference to the net income actually received or receivable by such Finance Party (but, for the avoidance of doubt, not including any sum deemed for the purposes of Tax to be received or receivable by such Finance Party but not actually receivable) by the jurisdiction in which such Finance Party is incorporated;
 - (ii) any Tax imposed on and calculated by reference to the net income of the Facility Office of such Finance Party actually received or receivable by such Finance Party (but, for the avoidance of doubt, not including any sum deemed for the purposes of Tax to be received or receivable by such Finance Party but not actually receivable) by the jurisdiction in which its Facility Office is located; or
 - (iii) a FATCA Deduction required to be made by a Party.
- (b) A Finance Party intending to make a claim under paragraph (a) above shall notify the Agent of the event giving rise to the claim, whereupon the Agent shall notify the Borrower thereof.
- (c) A Finance Party shall, on receiving a payment from an Obligor under this Clause 12.3, notify the Agent.

12.4 Tax credit

If an Obligor makes a Tax Payment and the relevant Finance Party determines that:

- (a) a Tax Credit is attributable to an increased payment of which that Tax Payment forms part, to that Tax Payment or to a Tax Deduction in consequence of which that Tax Payment was required; and
- (b) that Finance Party has obtained and utilised that Tax Credit,

the Finance Party shall pay an amount to the Obligor which that Finance Party determines will leave it (after that payment) in the same after-Tax position as it would have been in had the Tax Payment not been required to be made by the Obligor.

12.5 Stamp taxes

The Borrower shall:

- (a) pay all stamp duty, registration and other similar Taxes payable in respect of any Finance Document; and
- (b) within 15 days of demand, indemnify each Finance Party against any cost, loss or liability that Finance Party incurs in relation to any stamp duty, registration or other similar Tax paid or payable in respect of any Finance Document.

12.6 Indirect tax

- (a) All amounts set out or expressed in a Finance Document to be payable by any Party to a Finance Party shall be deemed to be exclusive of any Indirect Tax. If any Indirect Tax is chargeable on any supply made by any Finance Party to any Party in connection with a Finance Document, that Party shall pay to the Finance Party (in addition to and at the same time as paying the consideration) an amount equal to the amount of the Indirect Tax.
- (b) Where a Finance Document requires any Party to reimburse or indemnify a Finance Party for any costs or expenses, that Party shall also at the same time pay and indemnify the Finance Party against all Indirect Tax incurred by that Finance Party in respect of the costs or expenses to the extent that the Finance Party reasonably determines that it is not entitled to credit or repayment in respect of the Indirect Tax.

12.7 FATCA information

- (a) Subject to paragraph (c) below, each Party shall, within 10 Business Days of a reasonable request by another Party:
 - (i) confirm to that other Party whether it is:
 - (A) a FATCA Exempt Party; or
 - (B) not a FATCA Exempt Party;
 - supply to that other Party such forms, documentation and other information relating to its status under FATCA as that other Party reasonably requests for the purposes of that other Party's compliance with FATCA; and
 - (iii) supply to that other Party such forms, documentation and other information relating to its status as that other Party reasonably requests for the purposes of that other Party's compliance with any other law, regulation, or exchange of information regime.
- (b) If a Party confirms to another Party pursuant to paragraph (a)(i) above that it is a FATCA Exempt Party and it subsequently becomes aware that it is not or

has ceased to be a FATCA Exempt Party, that Party shall notify that other Party reasonably promptly.

- (c) Paragraph (a) above shall not oblige any Finance Party to do anything, and paragraph (a)(iii) above shall not oblige any other Party to do anything, which would or might in its reasonable opinion constitute a breach of:
 - (i) any law or regulation;
 - (ii) any fiduciary duty; or
 - (iii) any duty of confidentiality.
- (d) If a Party fails to confirm whether or not it is a FATCA Exempt Party or to supply forms, documentation or other information requested in accordance with paragraph (a)(i) or (a)(ii) above (including, for the avoidance of doubt, where paragraph (c) above applies), then such Party shall be treated for the purposes of the Finance Documents (and payments under them) as if it is not a FATCA Exempt Party until such time as the Party in question provides the requested confirmation, forms, documentation or other information.
- (e) If the Borrower is a US Tax Obligor or the Agent reasonably believes that its obligations under FATCA or any other applicable law or regulation require it, each Lender shall, within 10 Business Days of:
 - (i) where the Borrower is a US Tax Obligor and the relevant Lender is an Original Lender, the date of this Agreement;
 - (ii) where the Borrower is a US Tax Obligor on a date on which any other Lender becomes a Party as a Lender, that date; or
 - (iii) where the Borrower is not a US Tax Obligor, the date of a request from the Agent,

supply to the Agent:

- (A) a withholding certificate on Form W-8, Form W-9 or any other relevant form; or
- (B) any withholding statement or other document, authorisation or waiver as the Agent may require to certify or establish the status of such Lender under FATCA or that other law or regulation.
- (f) The Agent shall provide any withholding certificate, withholding statement, document, authorisation or waiver it receives from a Lender pursuant to paragraph (e) above to the Borrower.
- (g) If any withholding certificate, withholding statement, document, authorisation or waiver provided to the Agent by a Lender pursuant to paragraph (e) above is or becomes materially inaccurate or incomplete, that Lender shall promptly

update it and provide such updated withholding certificate, withholding statement, document, authorisation or waiver to the Agent unless it is unlawful for the Lender to do so (in which case the Lender shall promptly notify the Agent). The Agent shall provide any such updated withholding certificate, withholding statement, document, authorisation or waiver to the Borrower.

(h) The Agent may rely on any withholding certificate, withholding statement, document, authorisation or waiver it receives from a Lender pursuant to paragraph (e) or (g) above without further verification. The Agent shall not be liable for any action taken by it under or in connection with paragraph (e), (f) or (g) above.

12.8 FATCA Deduction

- (i) Each Party may make any FATCA Deduction it is required to make by FATCA, and any payment required in connection with that FATCA Deduction, and no Party shall be required to increase any payment in respect of which it makes such a FATCA Deduction or otherwise compensate the recipient of the payment for that FATCA Deduction.
- (j) Each Party shall promptly, upon becoming aware that it must make a FATCA Deduction (or that there is any change in the rate or the basis of such FATCA Deduction), notify the Party to whom it is making the payment and, in addition, shall notify the Borrower and the Agent and the Agent shall notify the other Finance Parties.

13. INCREASED COSTS

13.1 Increased Costs

- (a) Subject to Clause 13.3 (*Exceptions*) the Borrower shall, within fifteen Business Days of a demand by the Agent, pay for the account of a Finance Party the amount of any Increased Costs incurred by that Finance Party or any of its Affiliates as a result of:
 - (i) the introduction of or change in (or in the interpretation, administration or application of) any law or regulation after the date of this Agreement;
 - (ii) compliance with any request or requirement relating to the maintenance of capital or any other request from or requirement of any central bank or other fiscal, monetary, regulatory or other authority that is made after the date of this Agreement;
 - (iii) the implementation or application of, or compliance with, Basel III or any law or regulation that implements or applies Basel III; or
 - (iv) the implementation or application of, or compliance with, the Dodd-Frank Wall Street Reform and Consumer Protection Act of the United States of America and all requests, rules, guidelines or directives

thereunder or issued in connection therewith (in each case, regardless of the date of implementation, application, enactment, adoption or issuance).

- (b) In this Agreement:
 - (i) "Increased Costs" means:
 - (A) a reduction in the rate of return from a Facility or on a Finance Party's (or its Affiliate's) overall capital (including as a result of any reduction in the rate of return on capital brought about by more capital being required to be allocated by such Lender or Affiliate);
 - (B) an additional or increased cost; or
 - (C) a reduction of any amount due and payable under any Finance Document,

which is incurred or suffered by a Finance Party or any of its Affiliates to the extent that it is attributable to that Finance Party having entered into or maintaining its Commitment or funding or performing its obligations under any Finance Document.

- (ii) "**Basel III**" means:
 - (i) the agreements on capital requirements, a leverage ratio and liquidity standards contained in "Basel III: A global regulatory framework for more resilient banks and banking systems", "Basel III: International framework for liquidity risk measurement, standards and monitoring" and "Guidance for national authorities operating the countercyclical capital buffer" published by the Basel Committee on Banking Supervision in December 2010, each as amended, supplemented or restated; and
 - (ii) any further guidance or standards published by the Basel Committee on Banking Supervision relating to "Basel III".

13.2 Increased Cost claims

- (a) A Finance Party intending to make a claim pursuant to Clause 13.1 (*Increased Costs*) shall notify the Agent of the event giving rise to the claim, following which the Agent shall promptly notify the Borrower.
- (b) Each Finance Party shall, as soon as practicable after a demand by the Agent, provide a certificate confirming the amount of its Increased Costs and the basis thereof.

13.3 Exceptions

- (a) Clause 13.1 (*Increased Costs*) does not apply to the extent any Increased Cost is:
 - (i) attributable to a Tax Deduction required by law to be made by an Obligor;
 - (ii) attributable to a FATCA Deduction required to be made by a Party;
 - (iii) compensated for by Clause 12.3 (*Tax indemnity*) (or would have been compensated for under Clause 12.3 (*Tax indemnity*) but was not so compensated solely because any of the exclusions in paragraph (a) of Clause 12.3 (*Tax indemnity*) applied);
 - (iv) attributable to the wilful breach by the relevant Finance Party or its Affiliates of any law or regulation; or
 - (v) is attributable to the implementation or application of or compliance with Basel II or any other law or regulation existing on the date of this Agreement which implements Basel II (whether such implementation, application or compliance is by a government regulator, Finance Party or any of its Affiliates or otherwise) but excluding any Increased Costs attributable to the implementation or application of Basel III or any law or regulation which implements Basel III.
- (b) In this Clause 13.3, a reference to a "**Tax Deduction**" has the same meaning given to that term in Clause 12.1 (*Definitions*).

14. **OTHER INDEMNITIES**

14.1 **Currency indemnity**

- (a) If any sum due from an Obligor under the Finance Documents (a "Sum"), or any order, judgment or award given or made in relation to a Sum, has to be converted from the currency (the "First Currency") in which that Sum is payable into another currency (the "Second Currency") for the purpose of:
 - (i) making or filing a claim or proof against that Obligor;
 - (ii) obtaining or enforcing an order, judgment or award in relation to any litigation or arbitration proceedings,

that Obligor shall as an independent obligation, within 15 days of demand, indemnify each Finance Party to whom that Sum is due against any cost, loss or liability arising out of or as a result of the conversion including any discrepancy between (A) the rate of exchange used to convert that Sum from the First Currency into the Second Currency and (B) the rate or rates of exchange available to that person at the time of its receipt of that Sum.

(b) Each Obligor waives any right it may have in any jurisdiction to pay any amount under the Finance Documents in a currency or currency unit other than that in which it is expressed to be payable.

14.2 Other indemnities

The Borrower shall (or shall procure that an Obligor will), within 15 days of demand, indemnify each Finance Party against any cost, loss or liability incurred by that Finance Party as a result of:

- (a) the occurrence of any Event of Default;
- (b) a failure by an Obligor to pay any amount due under a Finance Document on its due date, including without limitation, any cost, loss or liability arising as a result of Clause 27 (*Sharing Among the Finance Parties*);
- (c) funding, or making arrangements to fund, its participation in a Loan requested by the Borrower in a Utilisation Request but not made by reason of the operation of any one or more of the provisions of this Agreement (other than by reason of default or negligence by that Finance Party alone);
- (d) a Loan (or part of a Loan) not being prepaid in accordance with a notice of prepayment given by the Borrower; or
- (e) any claim concerning Wynn Macau or the Cotai Property (including, in each case, its participation therein) to the extent that loss or liability is suffered or incurred by that Finance Party (other than by reason of default or negligence by a Finance Party),

provided that, unless any such cost, loss or liability is incurred while an Acceleration Event is continuing, any such cost, loss or liability shall be reasonable.

14.3 **Indemnity to the Agent**

The Borrower shall promptly indemnify the Agent against any cost, loss or liability incurred by the Agent (acting reasonably) as a result of:

- (a) investigating any event which it reasonably believes is a Default;
- (b) acting or relying on any notice, request or instruction which it reasonably believes to be genuine, correct and appropriately authorised; or
- (c) instructing lawyers, accountants, tax advisers, surveyors or other professional advisers or experts as permitted under this Agreement,

provided that, unless any such cost, loss or liability is incurred while an Acceleration Event is continuing, any such cost, loss or liability shall be reasonable.

15. MITIGATION BY THE LENDERS

15.1 Mitigation

- (a) Each Finance Party shall, in consultation with the Borrower, take all reasonable steps to mitigate any circumstances which arise and which would result in any amount becoming payable under or pursuant to, or cancelled pursuant to, any of Clause 7.1 (*Illegality*), Clause 10 (*Changes to the Calculation of Interest*), Clause 12 (*Tax Gross-up and Indemnities*) or Clause 13 (*Increased Costs*) including (but not limited to) transferring its rights and obligations under the Finance Documents to another Affiliate or Facility Office.
- (b) Paragraph (a) above does not in any way limit the obligations of any Obligor under the Finance Documents.

15.2 Limitation of liability

- (a) The Borrower shall promptly indemnify each Finance Party for all costs and expenses reasonably incurred by that Finance Party as a result of steps taken by it under Clause 15.1 (*Mitigation*).
- (b) A Finance Party is not obliged to take any steps under Clause 15.1 (*Mitigation*) if, in the opinion of that Finance Party (acting reasonably), to do so might be prejudicial to it.

16. COSTS AND EXPENSES

16.1 **Transaction expenses**

The Borrower shall, within fifteen days of receipt of a written demand, pay the Agent and the Arranger the amount of all costs and expenses (including legal fees) reasonably incurred by any of them in connection with the negotiation, preparation, printing, execution and syndication of:

- (a) this Agreement and any other documents referred to in this Agreement; and
- (b) any other Finance Documents executed after the date of this Agreement.

16.2 Amendment costs

If:

- (a) an Obligor requests an amendment, waiver or consent; or
- (b) an amendment is required pursuant to Clause 28.9 (*Change of currency*) and/or Clause 34.4 (*Replacement of Screen Rate*),

the Borrower shall, within 15 days of demand, reimburse the Agent for the amount of all costs and expenses (including legal fees) reasonably incurred by the Agent in responding to, evaluating, negotiating or complying with that request or requirement.

16.3 Enforcement costs

The Borrower shall, within fifteen days of written demand, pay to each Finance Party the amount of all costs and expenses (including legal fees) incurred by that Finance Party in connection with the enforcement of, or the preservation of any rights under, any Finance Document.

SECTION 7 GUARANTEE

17. GUARANTEE AND INDEMNITY

17.1 Guarantee and indemnity

The Company irrevocably and unconditionally:

- (a) guarantees to each Finance Party punctual performance by the Borrower of all the Borrower's obligations under the Finance Documents;
- (b) undertakes with each Finance Party that whenever the Borrower does not pay any amount when due under or in connection with any Finance Document, the Company shall immediately on demand pay that amount as if it was the principal obligor; and
- (c) agrees with each Finance Party that if any obligation guaranteed by it is or becomes unenforceable, invalid or illegal, it will, as an independent and primary obligation, indemnify that Finance Party immediately on demand against any cost, loss or liability it incurs as a result of the Borrower not paying any amount which would, but for such unenforceability, invalidity or illegality, have been payable by it under any Finance Document on the date when it would have been due. The amount payable by the Company under this indemnity will not exceed the amount it would have had to pay under this Clause 17 if the amount claimed had been recoverable on the basis of a guarantee.

17.2 Continuing guarantee

This guarantee is a continuing guarantee and will extend to the ultimate balance of sums payable by any Obligor under the Finance Documents, regardless of any intermediate payment or discharge in whole or in part.

17.3 Reinstatement

If any discharge, release or arrangement (whether in respect of the obligations of any Obligor or any security for those obligations or otherwise) is made by a Finance Party in whole or in part on the basis of any payment, security or other disposition which is avoided or must be restored in insolvency, liquidation, administration or otherwise, without limitation, then the liability of the Company under this Clause 17 will continue or be reinstated as if the discharge, release or arrangement had not occurred.

17.4 Waiver of defences

The obligations of the Company under this Clause 17 will not be affected by an act, omission, matter or thing which, but for this Clause 17, would reduce, release or prejudice any of its obligations under this Clause 17 (without limitation and whether or not known to it or any Finance Party) including:

- (a) any time, waiver or consent granted to, or composition with, any Obligor or other person;
- (b) the release of any other Obligor or any other person under the terms of any composition or arrangement with any creditor of any Group Member;
- (c) the taking, variation, compromise, exchange, renewal or release of, or refusal or neglect to perfect, take up or enforce, any rights against, or security over assets of, any Obligor or other person or any non-presentation or nonobservance of any formality or other requirement in respect of any instrument or any failure to realise the full value of any security;
- (d) any incapacity or lack of power, authority or legal personality of or dissolution or change in the members or status of an Obligor or any other person;
- (e) any amendment, novation, supplement, extension, restatement (however fundamental and whether or not more onerous) or replacement of any Finance Document or any other document or security including without limitation any change in the purpose of, any extension of or any increase in any facility or the addition of any new facility under any Finance Document or other document or security;
- (f) any unenforceability, illegality or invalidity of any obligation of any person under any Finance Document or any other document or security; or
- (g) any insolvency or similar proceedings.

17.5 Immediate recourse

The Company waives any right it may have of first requiring any Finance Party (or any trustee or agent on its behalf) to proceed against or enforce any other rights or security or claim payment from any person before claiming from the Company under this Clause 17. This waiver applies irrespective of any law or any provision of a Finance Document to the contrary.

17.6 **Appropriations**

Until all amounts which may be or become payable by the Obligors under or in connection with the Finance Documents have been irrevocably paid in full, each Finance Party (or any trustee or agent on its behalf) may:

- (a) refrain from applying or enforcing any other moneys, security or rights held or received by that Finance Party (or any trustee or agent on its behalf) in respect of those amounts, or apply and enforce the same in such manner and order as it sees fit (whether against those amounts or otherwise) and the Company shall not be entitled to the benefit of the same; and
- (b) hold in a non interest-bearing or an interest-bearing suspense account any moneys received from the Company or on account of the Company's liability under this Clause 17.



17.7 Deferral of Company's rights

Until all amounts which may be or become payable by the Obligors under or in connection with the Finance Documents have been irrevocably paid in full and unless the Agent otherwise directs, the Company will not exercise any rights which it may have by reason of performance by it of its obligations under the Finance Documents or by reason of any amount being payable, or liability arising, under this Clause 17:

- (a) to be indemnified by an Obligor;
- (b) to claim any contribution from the Company of any Obligor's obligations under the Finance Documents;
- (c) to take the benefit (in whole or in part and whether by way of subrogation or otherwise) of any rights of the Finance Parties under the Finance Documents or of any other guarantee or security taken pursuant to, or in connection with, the Finance Documents by any Finance Party;
- (d) to bring legal or other proceedings for an order requiring any Obligor to make any payment, or perform any obligation, in respect of which the Company has given a guarantee, undertaking or indemnity under Clause 17.1 (*Guarantee and indemnity*);
- (e) to exercise any right of set-off against any Obligor; and/or
- (f) to claim or prove as a creditor of any Obligor in competition with any Finance Party.

If the Company receives any benefit, payment or distribution in relation to such rights it shall hold that benefit, payment or distribution to the extent necessary to enable all amounts which may be or become payable to the Finance Parties by the Obligors under or in connection with the Finance Documents to be repaid in full on trust for the Finance Parties and shall promptly pay or transfer the same to the Agent or as the Agent may direct for application in accordance with Clause 28 (*Payment Mechanics*).

17.8 Additional security

This guarantee is in addition to and is not in any way prejudiced by any other guarantee or security now or subsequently held by any Finance Party.

SECTION 8

REPRESENTATIONS, UNDERTAKINGS AND EVENTS OF DEFAULT

18. **REPRESENTATIONS**

Each Obligor makes the representations and warranties set out in this Clause 18 to each Finance Party on the date of this Agreement other than the representations and warranties in Clause 18.11 (*Original financial statements*) which shall be made upon the delivery of the Original Financial Statements to the Agent.

18.1 Status

- (a) It is an exempted company, duly incorporated, validly existing and in good standing under the law of its jurisdiction of incorporation.
- (b) It and each of its Subsidiaries has the power to own its assets and carry on its business as it is being conducted.

18.2 **Binding obligations**

Subject to Legal Reservations, the obligations expressed to be assumed by it in each Finance Document to which it is a party are, legal, valid, binding and enforceable obligations.

18.3 Non-conflict with other obligations

The entry into and performance by it of, and the transactions contemplated by, the Finance Documents (to which it is a party) do not and will not conflict with:

- (a) subject to Legal Reservations, any material Legal Requirement applicable to it;
- (b) its or any of its Subsidiaries' constitutional documents; or
- (c) any agreement or instrument binding upon it or any of its Subsidiaries or any of its or any of its Subsidiaries' assets to an extent which could reasonably be expected to have a Material Adverse Effect.

18.4 **Power and authority**

It has the power to enter into, perform and deliver, and has taken all necessary action to authorise its entry into, performance and delivery of, the Finance Documents to which it is a party and the transactions contemplated by those Finance Documents.

18.5 Validity and admissibility in evidence

Subject to Legal Reservations, all Authorisations required:

(a) to enable it lawfully to enter into, exercise its rights and comply with its obligations in the Finance Documents to which it is a party; and

(b) to make the Finance Documents to which it is a party admissible in evidence in its jurisdiction of incorporation (other than court filings in the normal course of proceedings),

have been (or will by the required date be) obtained or effected and are (or will by the required date be) in full force and effect.

18.6 Governing law and enforcement

- (a) Subject to Legal Reservations, the choice of English law as the governing law of the Finance Documents to which it is a party will be recognised and enforced in its jurisdiction of incorporation.
- (b) Subject to Legal Reservations, any judgment obtained in England in relation to a Finance Document to which it is a party will be recognised and enforced in its jurisdiction of incorporation.

18.7 **Deduction of Tax**

It is not required to make any Tax Deduction (as defined in Clause 12.1 (*Definitions*)) from any payment it may make under any Finance Document to a Lender.

18.8 No filing or stamp fees or taxes

Under the law of its jurisdiction of incorporation it is not necessary that the Finance Documents be filed, recorded or enrolled with any court or other authority in that jurisdiction or that any stamp, registration or similar tax or fees be paid on or in relation to the Finance Documents or the transactions contemplated by the Finance Documents, **provided that**, Cayman Islands stamp duty shall be payable on any Finance Document that is executed in, brought into or produced before a court in, the Cayman Islands.

18.9 No default

- (a) No Event of Default is continuing or might reasonably be expected to result from the making of any Utilisation.
- (b) To the best of its knowledge and belief, no other event or circumstance is outstanding which constitutes a default under any other agreement or instrument which is binding on it or any of its Subsidiaries or to which its (or any of its Subsidiaries') assets are subject which might have a Material Adverse Effect.

18.10 No misleading information

- (a) Except as disclosed to the Agent or the Arrangers in writing prior to the date of this Agreement, to the best of its knowledge and belief:
 - (i) any factual information (other than information of a general economic nature) provided by any Group Member for the purposes of the Information Memorandum was true and accurate in all material

respects as at the date it was provided or as at the date (if any) at which it is stated;

- (ii) the financial projections contained in the Information Memorandum have been prepared on the basis of recent historical information and on the basis of assumptions believed by it to be reasonable at the time of being prepared (it being understood that such projections may be subject to significant uncertainties and contingencies which are beyond its or the Group's control and accordingly no assurance can be given that such projections will be realised or accurate); and
- (iii) nothing has occurred or been omitted from the Information Memorandum and no information has been given or withheld that results in the information contained in the Information Memorandum being untrue or misleading in any material respect.
- (b) As of the date of this Agreement and the first Utilisation Date (in respect of each Beneficial Ownership Certification delivered on or prior to the first Utilisation Date) or (in respect of each Beneficial Ownership Certification delivered after the first Utilisation Date) as at the date of that Beneficial Ownership Certification, the information included in each Beneficial Ownership Certification (as defined in Clause 42 (*Beneficial Ownership*)) is true and correct in all respects.

18.11 Original financial statements

- (a) The Original Financial Statements were prepared in accordance with IFRS consistently applied.
- (b) The Original Financial Statements fairly present its consolidated financial condition as at 31 December 2020 and its consolidated results of operations during the relevant financial year ending 31 December 2020.
- (c) There has been no material adverse change in the Borrower Group's business or financial condition since 31 December 2020.

18.12 Pari passu ranking

Its payment obligations under the Finance Documents rank at least *pari passu* with the claims of all its other unsecured and unsubordinated creditors, except for obligations mandatorily preferred by law applying to companies generally.

18.13 No proceedings

(a) No litigation, arbitration or administrative proceedings of or before any court, arbitral body or agency which, if adversely determined, might reasonably be expected to have a Material Adverse Effect has or have been started or threatened in writing against it or any of its Subsidiaries; and

(b) No judgment or order of a court, arbitral body or agency which might reasonably be expected to have a Material Adverse Effect has been made against it or any of its Subsidiaries.

18.14 Security, Financial Indebtedness and Immunity

- (a) No Security or Quasi-Security exists over all or any of the present or future assets of any Borrower Group Member, other than Permitted Security.
- (b) No Borrower Group Member has any Financial Indebtedness outstanding, other than Permitted Financial Indebtedness.
- (c) The entry into by it of each Finance Document constitutes, and the exercise by it of its rights and performance of its obligations under each Finance Document will constitute, private and commercial acts performed for private and commercial purposes.
- (d) It will not be entitled to claim immunity from suit, execution, attachment or other legal process in any proceedings taken in its jurisdiction of incorporation in relation to any Finance Document.

18.15 Environmental Laws

To the best of its knowledge and belief:

- (a) each Borrower Group Member is in compliance with all applicable material Environmental Laws in all material respects and, so far as it is aware, there are no circumstances that could at any time be reasonably expected to prevent or interfere with such compliance; and
- (b) no Environmental Claim has been made against any Borrower Group Member which has not been fully discharged, released, satisfied or withdrawn, except for any Environmental Claim that would not have, or would not reasonably be expected to have, a Material Adverse Effect.

18.16 Intellectual Property

Each Borrower Group Member owns, or is licensed to use, all intellectual property necessary for the conduct of its business as currently conducted. The use by the Borrower Group of the intellectual property related to or otherwise associated with the Group's use of the "Wynn" and " $\dot{\chi}$ $\dot{\chi}$ J" names does not infringe on the rights of any Person. The use by the Borrower Group of intellectual property other than intellectual property related to or otherwise associated with the Borrower Group's use of the "Wynn" and " $\dot{\chi}$ $\dot{\chi}$ J" names, does not infringe on the rights of any Person, except where such infringement, individually or collectively, would not reasonably be expected to have a Material Adverse Effect.

18.17 Solvency

(a) No:

- (i) corporate action, legal proceeding or other formal procedure or formal step described in paragraph (a) of Clause 22.7 (Insolvency proceedings); or
- (ii) creditors' process described in Clause 22.8 (Creditors' process),

has been taken or, to the best of its knowledge and belief of the Borrower, threatened in writing against any Group Member and none of the circumstances described in Clause 22.6 (Insolvency) applies to any Group Member.

- (b) Each of the Obligors is, in accordance with IFRS consistently applied, a "going concern".
- (c) The auditors of the Obligors (or any of them) have not qualified the most recent audited annual financial statements of any Obligor with a "going concern" or like qualification or exception arising out of the scope of their audit or any other material qualification.

18.18 Subsidiaries

- (a) The Executive Director legally and beneficially owns 20,010 Class A Shares (as defined in the Governing Documents of Wynn Resorts Macau), representing 10% of the total issued share capital and 10% of the Voting Stock of Wynn Resorts Macau.
- (b) Wynn HK legally and beneficially owns 102,051 Class B Shares (as defined in the Governing Documents of Wynn Resorts Macau), representing 51% of the total issued share capital and 51% of the Voting Stock of Wynn Resorts Macau.
- (c) Wynn International legally and beneficially owns 78,039 Class C Shares (as defined in the Governing Documents of Wynn Resorts Macau), representing 39% of the total issued share capital and 39% of the Voting Stock of Wynn Resorts Macau.
- (d) Wynn Holdings legally and beneficially owns 99% and beneficially owns 1%, and Wynn International, as nominee of Wynn Holdings, legally owns 1%, of the total issued share capital of Wynn HK.
- (e) Wynn International legally and beneficially owns 100% of Wynn Holdings.
- (f) The Borrower legally and beneficially owns 100% of the total issued share capital of Wynn International.
- (g) WML Corp legally and beneficially owns 100% of the total issued share capital of the Borrower.
- (h) The Company legally and beneficially owns 100% of the total issued share capital of WML Corp.

- (i) Wynn Resorts Macau legally and beneficially owns 99.8%, and Wynn International and Wynn HK each legally and beneficially own 0.1%, of the total issued share capital of Palo.
- (j) Save as permitted hereunder, other than Palo, Wynn Resorts Macau has no Subsidiaries and does not legally or beneficially own any Capital Stock in any Person.

18.19 Insurance

- (a) Wynn Resorts Macau and Palo are each insured by insurers of recognised financial responsibility against such losses and risks and in such amounts as are prudent and customary in the businesses in which they are each engaged.
- (b) There is no outstanding insured loss or liability incurred by either Wynn Resorts Macau or Palo which is US\$10,000,000 (or its equivalent) or more which is not expected to be covered to the full extent of that loss or liability.
- (c) There has been no non-disclosure, misrepresentation or breach of any term of any material insurance which would entitle any insurer of that insurance to repudiate, rescind or cancel it or to treat it as avoided in whole or in part or otherwise decline any valid claim under it by or on behalf of any Group Member.
- (d) No Obligor is aware of any insurer of any material insurance being in run-off or having entered into any insolvency proceedings which are still pending or current.

18.20 Fiscal Year

The fiscal year of the Borrower ends on 31 December of each calendar year.

18.21 Compliance with laws and Authorisations required for business

- (a) Each Borrower Group Member is in compliance with all Legal Requirements (including Sanctions and Authorisations) in all material respects and no notices of any material violation of any Authorisation made or issued by or with a Governmental Authority relating to Wynn Macau or the Cotai Property have been issued, entered or received by any such Borrower Group Member (and which violation is continuing).
- (b) All Authorisations required under any law or regulation to enable each Borrower Group Member to carry on its business (in the ordinary course of its business) have been obtained or effected and are in full force and effect where failure to do so has or would reasonably be expected to have a Material Adverse Effect.

18.22 **OFAC**

None of its directors, officers, agents, employees or affiliates (as defined in Rule 405 under the U.S. Securities Act of 1933, as amended) (a "**relevant OFAC person**") are the subject of any U.S. sanctions administered by OFAC or any other Sanctions.

18.23 Money Laundering Laws, Anti-Terrorism Laws and Anti-Bribery Laws

- (a) No action, suit or proceeding by or before any court or Governmental Authority, authority or body or any arbitrator involving any Group Member with respect to Money Laundering Laws or Anti-Terrorism Laws is pending and, to the best of its knowledge and belief, no such actions, suits or proceedings are threatened in writing or contemplated; and
- (b) To the best of its knowledge and belief, there is no such action, suit or proceeding pending which may reasonably be expected to have a Material Adverse Effect.
- (c) Its' and its Subsidiaries' operations are being and have been conducted at all times in material compliance with all Money Laundering Laws.
- (d) Neither it nor any of its Subsidiaries:
 - (i) has; and
 - (ii) subject to the proviso at the end of this paragraph (d), none of its (and each other Group Member's) directors, officers, agents, employees or other Persons associated with or acting on behalf of any Group Member (a "relevant FCPA person") have,

used any corporate funds for any unlawful contribution, gift, entertainment or other unlawful expense relating to political activity; made any direct or indirect unlawful payment to any foreign or domestic government official or employee from corporate funds; violated any provision of the U.S. Foreign Corrupt Practices Act of 1977 (the "**FCPA**") or the Bribery Act 2010 or made any bribe, rebate, payoff, influence payment, kickback or other unlawful payment prohibited under any applicable law or regulation equivalent to the FCPA or the Bribery Act 2010, **provided that** (in the case of paragraph (d)(ii) above) a Group Member will not be in breach of this paragraph if the relevant Group Member did not have knowledge or reason to believe that the relevant FCPA person engaged in any of the activities set out in this paragraph (d).

(e) It and each of its Subsidiaries has in place customary procedures designed to monitor the engagement by its (and each of its Subsidiaries') directors, officers and employees in any of the activities set out in paragraph (d) above.

18.24 Investment Company Act and U.S. margin regulations

(a) No Group Member is an "investment company" or a company "controlled" by an "investment company," as defined in, or subject to regulation under, the

Investment Company Act of 1940 of the United States of America, as amended.

- (b) No Group Member is engaged principally, or as one of its important activities, in the business of purchasing or carrying Margin Stock, or extending credit for the purpose of purchasing or carrying Margin Stock.
- (c) No part of the proceeds of any Facility will be used, whether directly or indirectly, and whether immediately, incidentally or ultimately, for purchasing or carrying Margin Stock or for any other purpose that entails a violation of, or that is inconsistent with, the provisions of the regulations of the Board of Governors of the Federal Reserve System of the United States (or any successor), including Regulations T, U and X.
- (d) In this Clause 18.24, "**Margin Stock**" means margin stock or "margin security" within the meaning of Regulations T, U and X of the Board of Governors of the Federal Reserve System of the United States (or any successor).

18.25 Repetition

The Repeating Representations are deemed to be made by each Obligor by reference to the facts and circumstances then existing on the date of each Utilisation Request and the first day of each Interest Period.

19. INFORMATION UNDERTAKINGS

The undertakings in this Clause 19 remain in force from the date of this Agreement for so long as any amount is outstanding under the Finance Documents or any Commitment is in force.

19.1 Financial statements

The Borrower shall supply to the Agent in sufficient copies for all the Lenders:

- (a) as soon as the same become available, but in any event within 90 days after the end of each Fiscal Year, its audited consolidated financial statements for that Fiscal Year (being the audited balance sheets (on a consolidated basis) of the Borrower prepared in accordance with IFRS, as at the end of such Fiscal Year and the related audited statements of income and of cash flows (each, on a consolidated basis) for such Fiscal Year); and
- (b) as soon as the same become available, but in any event within 45 days after the end of each Fiscal Quarter ending on or after 30 June 2022), its unaudited condensed consolidated financial statements for that Fiscal Quarter (being the unaudited (on a consolidated basis) balance sheets of the Borrower prepared in accordance with IFRS, as at the end of such quarter and the related unaudited

statements of income and of cash flows (each, on a consolidated basis) for such quarter).

19.2 **Compliance Certificate**

- (a) The Borrower shall supply to the Agent, with each set of financial statements delivered pursuant to paragraph (a) or
 (b) of Clause 19.1 (*Financial statements*) commencing with the Fiscal Quarter ending 30 June 2022, a Compliance Certificate setting out (in reasonable detail) computations as to compliance with Clause 20 (*Financial Covenants*) as at the date as at which those financial statements were drawn up.
- (b) Each Compliance Certificate shall be signed by a Responsible Officer.

19.3 Requirements as to financial statements

- (a) Each set of financial statements delivered by the Borrower pursuant to paragraph (a) of Clause 19.1 (*Financial statements*) shall be certified by a Responsible Officer as fairly presenting its financial condition as at the date as at which those financial statements were drawn up.
- (b) Each set of financial statements delivered by the Borrower pursuant to paragraph (b) of Clause 19.1 (*Financial statements*) shall be certified by a Responsible Officer as fair in all material respects (subject, where relevant, to normal year-end audit adjustments).
- (c) The Borrower shall procure that each set of financial statements of the Borrower delivered pursuant to Clause 19.1 (*Financial statements*) is prepared using GAAP, accounting practices and financial reference periods consistent with those applied in the preparation of the Original Financial Statements for the Borrower unless, in relation to any set of financial statements, it notifies the Agent that there has been a change in GAAP, the accounting practices or reference periods and its auditors deliver to the Agent:
 - (i) a description of any change necessary for those financial statements to reflect the GAAP, accounting practices and reference periods upon which the Borrower's Original Financial Statements were prepared; and
 - (ii) sufficient information, in form and substance as may be reasonably required by the Agent, to enable the Lenders to determine whether Clause 20 (*Financial Covenants*) has been complied with and make an accurate comparison between the financial position indicated in those financial statements and the Borrower's Original Financial Statements.

Any reference in this Agreement to those financial statements shall be construed as a reference to those financial statements as adjusted to reflect the basis upon which the Original Financial Statements were prepared.

19.4 Information: miscellaneous

The Borrower shall supply to the Agent (in sufficient copies for all the Lenders, if the Agent so requests):

- (a) copies of all documents required by law to be dispatched by the Company or the Borrower to its shareholders (or any class of them) or its creditors generally at the same time as they are dispatched, other than to the extent that such documents have been made available on to the public on the website of the HKSE;
- (b) promptly upon becoming aware of them, the details of any litigation, arbitration or administrative proceedings which are current, threatened in writing or pending against any Borrower Group Member, and which, if adversely determined, has or is reasonably likely to have a Material Adverse Effect;
- (c) promptly upon becoming aware of them, the details of any judgment or order of a court, arbitral body or agency which is made against any Borrower Group Member, and which has or is reasonably likely to have a Material Adverse Effect;
- (d) promptly upon receipt or dispatch (as applicable), any notice of termination (other than in respect of expiration in accordance with the terms thereof) or any notice of default sent or received by an Obligor or any other Group Member, in each case under any Major Project Document (other than the Concession Contract, the Land Concession Contract or the Cotai Land Concession Contract); and
- (e) promptly on request, such further information regarding the financial condition, business and operations of any Borrower Group Member as any Finance Party (through the Agent) may reasonably request.

19.5 Notification of default

- (a) Each Obligor shall notify the Agent of any Default (and the steps, if any, being taken to remedy it) promptly upon becoming aware of its occurrence (unless that Obligor is aware that a notification has already been provided by another Obligor).
- (b) Promptly upon a request by the Agent, the Borrower shall supply to the Agent a certificate signed by one of its directors or senior officers on its behalf certifying that no Default is continuing (or if a Default is continuing, specifying the Default and the steps, if any, being taken to remedy it).

19.6 **Direct electronic delivery by an Obligor**

An Obligor may satisfy its obligation under this Agreement to deliver any information in relation to a Lender by delivering that information directly to that Lender in

accordance with Clause 30.5 (*Electronic communication*) to the extent that Lender and the Agent agree to this method of delivery.

19.7 "Know your customer" checks

(a) If:

- (i) the introduction of or any change in (or in the interpretation, administration or application of) any law or regulation made after the date of this Agreement;
- (ii) any change in the status of an Obligor (or of a Holding Company of an Obligor) after the date of this Agreement; or
- (iii) a proposed assignment or transfer by a Lender of any of its rights and obligations under this Agreement to a party that is not a Lender prior to such assignment or transfer,

obliges the Agent or any Lender (or, in the case of paragraph (iii) above, any prospective new Lender) to comply with "know your customer" or similar identification procedures in circumstances where the necessary information is not already available to it, each Obligor shall promptly upon the request of the Agent or any Lender supply, or procure the supply of, such documentation and other evidence as is reasonably requested by the Agent (for itself or on behalf of any Lender) or any Lender (for itself or, in the case of the event described in paragraph (iii) above, on behalf of any prospective new Lender) in order for the Agent, such Lender or, in the case of the event described in paragraph (iii) above, any prospective new Lender to carry out and be satisfied it has complied with all necessary "know your customer" or other similar checks under all applicable laws and regulations pursuant to the transactions contemplated in the Finance Documents.

(b) Each Lender shall promptly upon the request of the Agent supply, or procure the supply of, such documentation and other evidence as is reasonably requested by the Agent (for itself) in order for the Agent to carry out and be satisfied it has complied with all necessary "know your customer" or other similar checks under all applicable laws and regulations pursuant to the transactions contemplated in the Finance Documents.

19.8 Disclosure of Customer Data

Notwithstanding anything else in this Agreement, no Obligor shall be required to disclose any data in relation to individual customers of the Group to any Finance Parties to the extent that any Group Member is prohibited from making such a disclosure without the consent of the Macau SAR government by the laws or regulations of the Macau SAR in respect of personal data and such consent has not been obtained.

19.9 SFO "Safe Harbour"

Notwithstanding any other provision of this Agreement or any other Finance Document, no Obligor shall be required to provide any document or information to the Finance Parties that may otherwise be required to be provided under any Finance Document if such information is inside information (as defined in the Securities and Futures Ordinance Cap. 571 of the Laws of Hong Kong (the "**SFO**")) but only if and for as long as the Company is not required to disclose such inside information to the market as a result of the Safe Harbour in section 307D of the SFO.

20. FINANCIAL COVENANTS

20.1 Financial definitions

In this Agreement:

"Acceptable Bank" means:

- (a) a bank or financial institution which has a rating for its long-term unsecured and non credit-enhanced debt obligations of A– or higher by Standard & Poor's Rating Services or Fitch Ratings Ltd or A3 or higher by Moody's Investors Service Limited or a comparable rating from an internationally recognised credit rating agency; or
- (b) any other bank or financial institution approved by the Agent.

"**Cash**" means, at any time, cash in hand or at bank and (in the latter case) credited to an account in the name of a member of the Borrower Group with an Acceptable Bank and to which a member of the Borrower Group is alone (or together with other members of the Borrower Group) beneficially entitled and for so long as:

- (a) that cash is repayable within 30 days after the relevant date of calculation;
- (b) repayment of that cash is not contingent on the prior discharge of any other indebtedness of any member of the Borrower Group or of any other person whatsoever or on the satisfaction of any other condition;
- (c) there is no Security over that cash except for any Permitted Security constituted by a netting or set-off arrangement entered into by members of the Borrower Group in the ordinary course of their banking arrangements; and
- (d) the cash is freely and (except as mentioned in paragraph (a) above) immediately available to be applied in repayment or prepayment of the Facilities.

"Cash Equivalent Investments" means at any time:

(a) certificates of deposit maturing within one year after the relevant date of calculation and issued by an Acceptable Bank;

- (b) any investment in marketable debt obligations issued or guaranteed by the government of the United States of America or the United Kingdom or by an instrumentality or agency of any of them having an equivalent credit rating, maturing within one year after the relevant date of calculation and not convertible or exchangeable to any other security;
- (c) commercial paper not convertible or exchangeable to any other security:
 - (i) for which a recognised trading market exists;
 - (ii) issued by an issuer incorporated in the United States of America or the United Kingdom;
 - (iii) which matures within one year after the relevant date of calculation; and
 - (iv) which has a credit rating of either A-1 or higher by Standard & Poor's Rating Services or F1 or higher by Fitch Ratings Ltd or P-1 or higher by Moody's Investors Service Limited, or, if no rating is available in respect of the commercial paper, the issuer of which has, in respect of its long-term unsecured and noncredit enhanced debt obligations, an equivalent rating;
- (d) any investment in money market funds which:
 - (i) have a credit rating of either A-1 or higher by Standard & Poor's Rating Services or F1 or higher by Fitch Ratings Ltd or P-1 or higher by Moody's Investors Service Limited; and
 - (ii) invest substantially all their assets in securities of the types described in paragraphs (a) to (c) above,

to the extent that investment can be turned into cash on not more than 30 days' notice; or

(e) any other debt security approved by the Majority Lenders,

in each case, to which any member of the Borrower Group is alone (or together with other members of the Borrower Group) beneficially entitled at that time and which is not issued or guaranteed by any member of the Borrower Group or subject to any Security.

"**EBITDA**" means, in relation to any period, the Net Income of the Borrower Group for such period *plus*, without duplication and to the extent reflected as a charge in the Borrower's statement of such Net Income for such period, the sum of:

(a) income Tax expense (whether or not paid during such period) other than Tax on gross gaming revenue;

- (b) amortization or write-off of debt discount and debt issuance costs and interest, commissions, discounts and other fees and charges associated with Financial Indebtedness (including the Loans);
- (c) depreciation and amortization expense;
- (d) amortization of intangibles (including goodwill);
- (e) an amount equal to the aggregate net non-cash loss on the disposal of Property during such period (other than sales of inventory in the ordinary course of business);
- (f) any extraordinary expenses or losses;
- (g) any foreign currency translation losses;
- (h) any corporate expenses;
- (i) any expense that is non-recurring in nature;
- (j) stock-based compensation;
- (k) pre-opening and development expenses;
- (l) Approved IP Fees; and
- (m) any EBITDA Cure (if any) made pursuant to (and in accordance with) Clause 20.4 (*Equity cure*),

and minus, without duplication and to the extent included in the statement of such Net Income for such period, the sum of:

- (n) interest income;
- (o) an amount equal to the aggregate net non-cash gain on the disposal of Property during such period (other than sales of inventory in the ordinary course of business);
- (p) any extraordinary income or gains (including, without limitation, any upfront premium or similar income or gains derived from, or in connection with the grant of, any Subconcession); and
- (q) any foreign currency translation gains,

all (including Net Income) as determined on a consolidated basis which includes members of the Borrower Group only (before taking into account any minority interests) and otherwise in accordance with applicable GAAP **provided that**, if one or more Permitted Selldowns have occurred, such amount shall then be reduced *pro rata* to reflect the Borrower's percentage interest in the outstanding Capital Stock of Wynn Resorts Macau on the last day of such period.

"**Financial Covenant**" means any of the financial covenants contained in Clause 20.2 (*Financial condition*) (together, the "**Financial Covenants**").

"Financing Costs" means:

- (a) interest, fees, commissions, costs and expenses payable by the Borrower under the Finance Documents;
- (b) interest, fees, commissions, costs and expenses payable by Wynn Resorts Macau under the Performance Bond Facility;
- (c) amounts payable by an Obligor under Clause 10 (*Changes to the Calculation of Interest*), Clause 12 (*Tax Gross-Up and Indemnities*), Clause 13 (*Increased Costs*) and Clause 14 (*Other Indemnities*);
- (d) any other amounts of interest, fees, commissions, discounts, prepayment penalties or premiums and other finance payments payable in respect of any Financial Indebtedness permitted to be incurred under paragraphs (b) (excluding any Subordinated Loans (save as set out in paragraph (e) below)), (d), (g) and/or (h) of the definition of "Permitted Financial Indebtedness" in Clause 1.1 (*Definitions*);
- (e) any amounts of interest, fees, commissions, discounts, payment penalties, prepayment penalties, premiums or other payments of any nature whatsoever for the purpose of making any payment pursuant to paragraph (d) of the definition of "Permitted Payment" in Clause 1.1 (*Definitions*);
- (f) net amounts payable by any Borrower Group Member under any Permitted Swap Transaction; and
- (g) any value added or other taxes payable by the Borrower or any other Borrower Group Member in respect of paragraphs (a) through (e) above and, save to the extent already included in paragraph (c) above, any withholding tax on a party under a Finance Document or any other agreement relating to the provision of Financial Indebtedness referred to above in respect of which the Borrower or any other Borrower Group Member has an obligation to gross up.

"**Fiscal Quarter**" means any one of the four consecutive three calendar month periods comprised in a Fiscal Year, **provided that** the first Fiscal Quarter shall end on 30 June 2022.

"Fiscal Year" means the fiscal year of the Borrower ending on 31 December of each calendar year.

"Interest Coverage Ratio" means, in relation to any period, the ratio of EBITDA to Financing Costs for such period.

"**Leverage Ratio**" means, in relation to any period, the ratio of Total Net Debt on the last day of such period to EBITDA for such period.

"**Net Income**" means, in relation to any period, the net income (or loss) of the Borrower Group for such period, determined on a consolidated basis which includes members of the Borrower Group only and otherwise in accordance with GAAP before any reduction in respect of preferred equity dividends.

"**Relevant Period**" means each period of four consecutive Fiscal Quarters ending on a Quarterly Date.

"**Total Net Debt**" means, in relation to the Borrower Group at any time, the aggregate principal amount of all Financial Indebtedness of each Borrower Group Member at such time (without double counting) but:

- (a) excluding Financial Indebtedness referred to in paragraph (g) of the definition thereof;
- (b) excluding Financial Indebtedness arising in respect of any Performance Bond Facility (save in relation to any drawing under the Performance Bond);
- (c) excluding Financial Indebtedness arising in respect of any Subordinated Loans;
- (d) excluding such Financial Indebtedness to the extent it is owed to another Borrower Group Member; and
- (e) deducting the aggregate amount of Cash and Cash Equivalent Investments held by any Borrower Group Member at that time.

20.2 Financial condition

The Borrower shall ensure that:

- (a) *Leverage Ratio* The Leverage Ratio as at the last day of any Relevant Period is not greater than 4.5:1.0.
- (b) *Interest Coverage Ratio* The Interest Coverage Ratio for each Relevant Period is not less than 2.5:1.0.

20.3 Financial testing

- (a) Subject to paragraph (b) below, the Financial Covenants shall be calculated in accordance with GAAP and tested by reference to each of the financial statements delivered pursuant to paragraphs (a) and (b) of Clause 19.1 (*Financial statements*) and/or each Compliance Certificate delivered pursuant to Clause 19.2 (*Compliance Certificate*).
- (b) For the purpose of calculating the Financial Covenants in Clause 20.2 (*Financial condition*) for the Quarterly Dates falling 30 June 2022 and 30 September 2022, EBITDA shall be calculated, at the option of the Borrower, by reference to either:

- (i) the amount of EBITDA for the four full consecutive Fiscal Quarters ending on such Quarterly Date; or
- (ii) the amount of EBITDA as disclosed in the financial statements and/or Compliance Certificates for the Fiscal Quarter ending on such Quarterly Date, annualised on a straight line basis,

provided that, paragraph (ii) above shall not be applicable for the purposes of determining the "Margin".

(c) For the purpose of calculating the Financial Covenant in paragraph (b) of Clause 20.2 (*Financial condition*) for the Quarterly Dates falling 30 June 2022 and 30 September 2022, Financing Costs shall be determined based on the amount of Financing Costs from the first Utilisation Date until the relevant Quarterly Date, annualised on a straight line basis.

20.4 Equity cure

- (a) If the requirements of any of the financial covenants in Clause 20.2 (*Financial condition*) are not met (or would but for this Clause 20.4 not be met) in respect of any Relevant Period (the "Breach Period") but cash proceeds of any Equity Injection are received by the Borrower (the "Cure Amount") after the end of the Breach Period and no later than the date falling 10 Business Days after the date on which the relevant Compliance Certificate in respect of such Breach Period is delivered (or if earlier, after the date on which the relevant Compliance Certificate in respect of such Breach Period is required to be delivered) pursuant to Clause 19.2 (*Compliance Certificate*), then the financial covenants under Clause 20.2 (*Financial condition*) will be tested or, as applicable, retested by giving effect to the following adjustments:
 - (i) (for the purpose of calculating the Leverage Ratio) the Cure Amount shall (at the Borrower's sole direction) be added to EBITDA for the last Fiscal Quarter of the Breach Period ("EBITDA Cure"); and
 - (ii) (for the purpose of calculating the Interest Coverage Ratio) the Cure Amount shall (at the Borrower's sole discretion) be applied as an EBITDA Cure in respect of such Breach Period,

in each case, solely for the purposes of ascertaining compliance with the financial covenants in Clause 20.2 (*Financial condition*) and not for any other purpose. If, after giving effect to the adjustments referred to above, the requirements of the financial covenants in Clause 20.2 (*Financial condition*) in respect of that Breach Period are met, then such requirements shall be deemed to have been satisfied as at the original date of determination (and as at the Quarterly Date on which such Breach Period ends) as though there had been no failure to comply and any Default or Event of Default occasioned thereby shall be deemed to have been remedied for all purposes under the Finance Documents.

- (b) Where a Cure Amount is received by the Borrower in respect of any Breach Period, then in respect of each Relevant Period ending on each Quarterly Date that falls less than 12 Months after the last day of such Breach Period (such Relevant Period being a "Subsequent Relevant Period" relating to such Breach Period), if the Borrower has elected EBITDA Cure in respect of such Breach Period, then for the purposes of calculating the Leverage Ratio and the Interest Coverage Ratio for such Subsequent Relevant Period, the EBITDA for such Subsequent Relevant Period shall be deemed to be increased by the amount of such Cure Amount.
- (c) The Borrower shall not be entitled to exercise any rights it may have to prevent or cure breaches of Financial Covenants on more than two occasions in any four consecutive Quarterly Dates (without prejudice to a relevant Cure Amount continuing to be included in any Subsequent Relevant Period).
- (d) If the Borrower exercises its rights to prevent or cure breaches of Financial Covenants on more than two occasions in respect of any four consecutive Quarterly Dates (without prejudice to a relevant Cure Amount continuing to be included in any Subsequent Relevant Period), it shall not be entitled to exercise any rights it may have to prevent or cure breaches of Financial Covenants in respect of any of the immediately following four Quarterly Dates.
- (e) Any Cure Amount received by the Borrower in accordance with this Clause 20.4 may exceed the amount required to rectify any breach or non-compliance with the financial undertakings in Clause 20.2 (*Financial condition*) in respect of any Relevant Period.
- (f) No Cure Amount shall be required to be applied in prepayment of the Loans.
- (g) No Cure Amount may be used, at any time, to make (or for the purposes of making) a Permitted Payment. Without limiting the foregoing, no Cure Amount shall be included or taken into account for any purpose (other than as set out in paragraphs (a) and (b) above) and (without limitation) the effect of each Cure Amount shall be ignored or excluded (to the extent it would otherwise be taken into account or included under the terms of any Finance Document) in any calculation (other than those specified in paragraphs (a) and (b) above) for the purposes of (or definitions set out in) the Finance Documents.
- (h) Save as contemplated by paragraph (d) above, there is no limit on the number of EBITDA Cures that may be made during the life of the Facilities.

21. GENERAL UNDERTAKINGS

The undertakings in this Clause 21 remain in force from the date of this Agreement for so long as any amount is outstanding under the Finance Documents or any Commitment is in force.

21.1 Authorisations and Permits

Each Obligor shall promptly:

- (a) obtain, comply with and do all that is necessary to maintain in full force and effect; and
- (b) upon request by the Agent, supply copies to the Agent of,

any Authorisation required under any law or regulation of its jurisdiction of incorporation to: (i) enable it to perform its obligations under the Finance Documents to which it is a party and (ii) subject to Legal Reservations, ensure the legality, validity, enforceability or admissibility in evidence in its jurisdiction of incorporation of any Finance Document to which it is a party.

21.2 **Compliance with laws**

- (a) Each Obligor shall comply in all respects with all laws to which it may be subject, if failure so to comply has or would materially impair its ability to perform its obligations under the Finance Documents.
- (b) Without prejudice to Clauses 21.12 (*OFAC*), 21.13 (*FCPA*), 21.14 (*Money Laundering Laws*) and 21.15 (*Anti-Terrorism Laws*), the Borrower shall ensure that it (and each other Group Member) is in compliance in all material respects with all applicable laws in respect of Sanctions, all applicable Anti-Bribery Laws, all applicable Money Laundering Laws and all applicable Anti-Terrorism Laws.

21.3 Payment of Obligations

To the extent not otherwise subject to valid subordination, standstill, intercreditor or similar arrangements, each Obligor shall pay, discharge or otherwise satisfy (and ensure each other Group Member shall pay, discharge or otherwise satisfy) at or before maturity or before they become delinquent, as the case may be, all its material obligations of whatever nature, except where the amount or validity thereof is currently being contested in good faith by appropriate (under the circumstances) proceedings and reserves in conformity with applicable GAAP with respect thereto have been provided on the books of such Group Member.

21.4 Conduct of Business and Maintenance of Existence, etc.

Each Obligor shall:

- (a) preserve, renew and keep in full force and effect (and ensure each other Group Member shall preserve, renew and keep in full force and effect) its corporate or limited liability company existence; and
- (b) take all reasonable action to maintain all rights, privileges, franchises, Authorisations and licenses necessary (and ensure each other Group Member shall take all reasonable action to maintain all rights, privileges, franchises,

Authorisations and licenses necessary) in the normal conduct of its business, except to the extent that failure to do so could not (individually or in the aggregate) reasonably be expected to have a Material Adverse Effect.

21.5 Insurance

Each Obligor shall (and shall ensure that each Group Member will):

- (a) maintain insurances on and in relation to its business and assets against those risks and to the extent as is usual for companies carrying on the same or substantially similar business (as reasonably determined by that Obligor or, as the case may be, Group Member); and
- (b) ensure that all such insurances and reinsurances must be with reputable independent insurance companies or underwriters.

21.6 Inspection of Property; Books and Records; Discussions

Each Obligor shall:

- (a) keep (and ensure each other Group Member shall keep) proper books of records and account in which full, true and correct entries in conformity with applicable GAAP and all Legal Requirements; and
- (b) subject to any Macau Gaming Laws, data protection laws or other applicable Legal Requirements restricting such actions and, where no Event of Default has occurred which is continuing, prior reasonable request and notice, procure (and ensure each other Group Member procures) that the Agent or its nominees (for purposes of this paragraph (b), each a "**Relevant Person**") be allowed reasonable rights of inspection and access during normal business hours to the Site, the Cotai Site, the Cotai Property, the Auditors and Company and Group Member officers, the Borrower's and each other Group Member's accounting books and records and any other documents relating to Wynn Resorts Macau or Palo (or their respective businesses or assets) as they may reasonably require, and so as not unreasonably to interfere with the operations of the Group and to take copies of any documents inspected. Any information and documents made available for inspection by a Relevant Person pursuant to this paragraph (b) shall be made available subject to customary confidentiality undertakings being executed by such Relevant Person.

21.7 Environmental Laws

Each Obligor shall:

(a) comply (and ensure each other Group Member complies) with all applicable Environmental Laws and Environmental Licences (save for any failure to comply that could not reasonably be expected to have a Material Adverse Effect);

- (b) provide copies of any notices from any Governmental Authority of non-compliance with any material Environmental Law or Environmental Licence and any notices of any material Environmental Claims to the Agent; and
- (c) deliver to the Agent with reasonable promptness, such other documents and information as from time to time may be reasonably requested by the Agent in relation to any matters disclosed pursuant to this Clause 21.7.

21.8 Compliance with Major Project Documents and Concession Contracts

Each Obligor shall:

- (a) comply (and ensure that each other Group Member complies), duly and promptly, in all material respects with its material obligations and enforce all of its material rights under all Major Project Documents;
- (b) from time to time obtain, maintain, retain, observe, keep in full force and effect and comply (and ensure that each other Group Member) in all material respects with the terms, conditions and provisions of all Authorisations made or issued by or with a Governmental Authority as shall now or hereafter be necessary under applicable laws; and
- (c) comply (and ensure each other Group Member complies) with all Legal Requirements related to the Cotai Land Concession Contract, the Land Concession Contract and the Concession Contract.

21.9 Pari Passu Ranking

Each Obligor shall procure that its obligations under the Finance Documents do and shall rank at least *pari passu* with all its other present and future unsecured and unsubordinated obligations, except for obligations mandatorily preferred by law applying to companies generally.

21.10 Further Assurances

The Borrower shall from time to time execute and deliver, or cause to be executed and delivered, such additional instruments, certificates or documents, and take all such actions, as the Agent may reasonably request, for the purposes of implementing or effectuating the provisions of the Finance Documents. Upon the exercise by the Agent, or any Finance Party of any power, right, privilege or remedy pursuant to any of the Finance Documents which requires any consent, approval, notification, registration or authorisation of any Governmental Authority, the Borrower shall execute and deliver, or will cause the execution and delivery of, all applications, certifications, instruments and other documents and papers that the Agent or such other Finance Party may reasonably be required to obtain from the Company, the Borrower or any other Borrower Group Member for such governmental consent, approval, notification, registration or authorisation.

21.11 Termination of Concession Contract and other Concession Contract matters

- (a) The Borrower shall notify the Agent promptly upon any Group Member receiving:
 - notice of any formal consultations with the Macau SAR in relation to any termination and cessation of the Concession Contract (but for the avoidance of doubt, not including consultations relating to the potential extension of the Concession Contract);
 - (ii) notice of any formal consultations with the Macau SAR in relation to any termination or rescission of the Land Concession Contract or in relation to any termination or rescission of the Cotai Land Concession Contract (but for the avoidance of doubt, not including consultations relating to the potential extension of the Concession Contract);
 - (iii) notice of any negotiations with the Macau SAR pursuant to Article 83 of the Concession Contract;
 - (iv) any notice from the Macau SAR pursuant to clause 3 of article 80 of the Concession Contract; or
 - (v) any notice from the Macau SAR pursuant to clause 4 of article 80 of the Concession Contract,

and keep the Agent fully appraised thereof.

- (b) The Borrower shall promptly upon receipt of a request from the Agent (acting on the instructions of the Majority Lenders), supply to the Agent (in sufficient copies for all the Lenders, if the Agent so requests), a copy of each written notice which is delivered under or in connection with the Concession Contract, the Land Concession Contract or the Cotai Land Concession Contract (and which is relevant and material to the interest of the Finance Parties (taken as a whole)). For these purposes (and without prejudice to, or limitation of, the foregoing), it is acknowledged that any notices in connection with or pursuant to Article 82 of the Concession Contract are relevant and material to the interest of the Finance Parties (taken as a whole).
- (c) Nothing in this Clause 21.11 shall require the Borrower to take any action that will render any party to be in breach of the Listing Rules or any other applicable laws and regulations.

21.12 **OFAC**

- (a) The Company shall ensure that it (and ensure that each other Group Member):
 - (i) is not the subject of any U.S. sanctions administered by the Office of Foreign Assets Control of the U.S. Department of the Treasury ("**OFAC**") or any other Sanctions; and

 (ii) is not located, organised or residing in any Designated Jurisdiction, or participating in or facilitating a transaction or business in a Designated Jurisdiction or, subject to the proviso below, involving any Person who is the subject of Sanctions,

provided that the Company will not (and no other Group Member will) be in breach of the relevant part of paragraph (ii) above, if the relevant Group Member does not have knowledge or reason to believe that the relevant Person is the subject of Sanctions.

- (b) The Company shall ensure that it shall not (and that no other Group Member shall) directly or indirectly use the proceeds of the Facilities, or lend, contribute or otherwise make available such proceeds to any other person, for the purpose of financing the activities of any person that, at the time of such financing, is the subject of any U.S. sanctions administered by OFAC or any other Sanctions, or in any Designated Jurisdiction.
- (c) Without prejudice to paragraph (b) above, the Company shall ensure that none of its funds or assets (and none of the funds or assets of any other Group Member) which are used to pay any amount due pursuant to this Agreement or any other Finance Document shall constitute funds knowingly obtained from transactions with or relating to Designated Persons or Designated Jurisdiction.
- (d) The Company shall ensure that it (and each other Group Member):
 - (i) has in place customary procedures designed to identify if any of its (or any other Group Member) directors, officers and agents, at the time of its engagement with or appointment of such relevant OFAC persons, is the subject of any U.S. sanctions administered by OFAC or any other Sanctions; and
 - (ii) will terminate its engagement with or appointment of any of its relevant OFAC person promptly upon it or any other Group Member having knowledge of such relevant OFAC person being the subject of any U.S. sanctions administered by OFAC or any other Sanctions.

21.13 FCPA

- (a) The Company shall ensure that it (and each other Group Member):
 - (i) does not; and
 - (ii) subject to the proviso at the end of this paragraph (a), none of its (and each other Group Member's) directors, officers, agents, employees or other Persons associated with or acting on behalf of any Group Member (a "relevant FCPA person") shall,

use any corporate funds for any unlawful contribution, gift, entertainment or other unlawful expense relating to political activity; make any direct or indirect unlawful payment to any foreign or domestic government official or

employee from corporate funds; violate any provision of the U.S. Foreign Corrupt Practices Act of 1977 (the "**FCPA**") or the Bribery Act 2010 (the "**Bribery Act**") or make any bribe, rebate, payoff, influence payment, kickback or other unlawful payment prohibited under any applicable law or regulation equivalent to the FCPA or the Bribery Act (such law or regulation, together with the FCPA and the Bribery Act, being "**Anti-Bribery Laws**"), **provided that** (in the case of paragraph (a)(ii) above) a Group Member will not be in breach of this paragraph (a) if the relevant Group Member does not have knowledge or reason to believe that the relevant FCPA person is engaged in any of the activities set out in this paragraph (a).

- (b) Without prejudice to paragraph (a) above or (c) below, the Borrower will not directly or indirectly use, lend or contribute the proceeds raised or otherwise utilised under this Agreement for any purpose that would breach the Anti-Bribery Laws (or any of them).
- (c) The Company shall ensure that it (and each other Group Member):
 - (i) has in place customary procedures designed to monitor the engagement by its (and each other Group Member's) directors, officers and employees in any of the activities set out in paragraph (a) above; and
 - (ii) will terminate or, if prohibited under applicable law from so terminating, take appropriate remedial or disciplinary action (including, without limitation, promptly relieving such relevant FCPA person of his or her duties and other responsibilities) and commence the termination of (and diligently pursue such action and termination) its (and each other Group Member's) engagement with or appointment of any of its (and each other Group Member's) relevant FCPA person promptly upon it or any other Group Member having knowledge that that relevant FCPA person has engaged in any of the activities set out in paragraph (a) above.

21.14 Money Laundering Laws

- (a) The Company shall ensure that it (and each other Group Member) is in material compliance with all applicable antimoney laundering laws (including but not limited to applicable financial record keeping and reporting requirements and money laundering statutes in Macau, and, to the best of its (and each other Group Member's) knowledge and belief, all jurisdictions in which it, or each Group Member, conducts business or which are otherwise applicable to it or each other Group Member) and all applicable rules, regulations and guidelines issued, administered or enforced by any Governmental Authority pursuant to such laws (collectively, "**Money Laundering Laws**").
- (b) The Company shall ensure that it (and each other Group Member) is in material compliance with the U.S. International Money Laundering Abatement and the U.S. Terrorism Financing Act of 2001.

- (c) Without prejudice to paragraphs (a) and (b) above, the Borrower shall not permit or authorise any person directly or indirectly to:
 - (i) use the proceeds under this Agreement to violate any Money Laundering Laws; or
 - (ii) lend, invest, contribute or otherwise make available the proceeds raised or otherwise utilised under this Agreement to or for the benefit of any Subsidiaries, any joint venture partner or any other person in a manner that would result in a violation of any Money Laundering Laws.

21.15 Anti-Terrorism Laws

- (a) Subject to the proviso at the end of this paragraph (a), the Company shall ensure that neither it nor any other Group Member, nor any of its or their respective brokers or other agents acting or benefiting in any capacity in connection with any Loan:
 - (i) is in violation of any Anti-Terrorism Law;
 - (ii) is a Designated Person; or
 - (iii) is dealing in any property or interest in property blocked pursuant to any Anti-Terrorism Law,

provided that such Group Member will not be in breach of this paragraph (a) in respect of the circumstances or activities of any brokers or agents of any Group Member which are restricted under paragraphs (a)(i) to (iii) above (inclusive) if that Group Member does not have knowledge of the relevant circumstances or activities.

- (b) The Company shall ensure that it (and each other Group Member):
 - (i) has in place customary procedures designed to identify, at the time of engagement with or appointment of its (and each other Group Member's) broker or agent acting or benefiting in any capacity in connection with any Loans, if any of the circumstances or activities that are restricted under paragraphs (a)(i) to (iii) above (inclusive) applies to or is undertaken by such broker or agent; and
 - (ii) terminates its (and each other Group Member's) engagement with or appointment of any such broker or agent promptly upon it, any other Group Member having knowledge that any of the circumstances or activities that are restricted under paragraphs (a)(i) to (iii) above (inclusive) applies to or is undertaken by that broker or agent.

21.16 Listed Status

The Company shall maintain its listing status on The Stock Exchange of Hong Kong Limited at all times.



21.17 Negative pledge

In this Clause 21.17, "Quasi-Security" means an arrangement or transaction described in paragraph (b) below.

- (a) The Borrower shall not (and the Borrower shall ensure that no other Borrower Group Member will) create or permit to subsist any Security over any of its assets.
- (b) The Borrower shall not (and the Borrower shall ensure that no other Borrower Group Member will):
 - (i) sell, transfer or otherwise dispose of any of its assets on terms whereby they are or may be leased to or reacquired by an Obligor or any other Group Member;
 - (ii) sell, transfer or otherwise dispose of any of its receivables on recourse terms;
 - (iii) enter into any arrangement under which money or the benefit of a bank or other account may be applied, set-off or made subject to a combination of accounts; or
 - (iv) enter into any other preferential arrangement having a similar effect,

in circumstances where the arrangement or transaction is entered into primarily as a method of raising Financial Indebtedness or of financing the acquisition of an asset.

(c) Paragraphs (a) and (b) above do not apply to any Permitted Security.

21.18 Disposals

- (a) No Obligor shall (and the Borrower shall ensure that no other Borrower Group Member will), enter into a single transaction or a series of transactions (whether related or not) and whether voluntary or involuntary to sell, lease, transfer or otherwise dispose of all or any part of any Resort Asset.
- (b) Paragraph (a) above does not apply to any Permitted Disposal.

21.19 Limitation on Transactions with Affiliates

The Borrower shall not (and the Borrower shall ensure that no other Borrower Group Member will) enter into any single transaction, or series of transactions, whether related or not (including, without limitation, any purchase, sale, lease or exchange of property, the rendering of any service or the payment of any management, advisory or similar fees, the entry into of any contract or the amendment, novation, supplementation, extension or restatement of any existing contract) with any Affiliate of the Company unless:

(a) the relevant transaction is:

- (i) a genuine commercial transaction on terms that are not less favourable to the Borrower or applicable Borrower Group Member than those that might be obtained at the time in a comparable arm's length transaction with Persons who are not Affiliates of the Company; and
- (ii) is not prohibited by the Listing Rules, or
- (b) the relevant transaction is between the Company or the Borrower and any of their respective Subsidiaries.

21.20 Merger

- (a) No Obligor shall (and the Borrower shall ensure that no other Borrower Group Member will) enter into any amalgamation, demerger, merger or corporate reconstruction.
- (b) Paragraph (a) above does not apply to any sale, lease, transfer or other disposal permitted pursuant to Clause 21.18 (*Disposals*).

21.21 Change of business

- (a) The Borrower shall procure that no substantial change is made to the general nature of the business of the Company or the Group from that carried on at the date of this Agreement.
- (b) The Borrower must ensure that the Group continues to own and operate Wynn Macau and the Cotai Property as integrated casino resorts.

21.22 Permitted Swap Transactions

The Borrower shall not (and shall ensure that no Borrower Group Member shall) enter into any derivatives transaction other than a Permitted Swap Transaction.

21.23 IP Rights

- (a) The Company shall (and shall ensure that each Group Member shall) take such steps as are necessary and commercially reasonable (including, where appropriate, the institution of legal proceedings) to prevent third parties infringing the Group's intellectual property rights.
- (b) The Company must ensure that it and, where relevant, its Subsidiaries have the legal right to use the trade names "Wynn" and "沈利" in connection with the pursuit of the Group's resort casino business and other related lodging and ancillary businesses in Macau.

21.24 Amendments to documents

Each Obligor shall not (and shall ensure that no Borrower Group Member shall):

(a) amend its memorandum or articles of association or other constitutional documents; or

(b) amend, waive or replace any term of any Major Project Document (including as a result of any Concession Extension),

in a manner which will, or is reasonably likely to, have a Material Adverse Effect.

21.25 Taxes

- (a) The Borrower shall (and shall procure that each Group Member shall) pay all Taxes due and payable (or, where payments of Tax must be made by reference to estimated amounts, such estimated Tax (calculated in good faith) as due and payable for the relevant period) by it prior to the accrual of any fine or penalty for late payment, unless (and only to the extent that):
 - (i) payment of those Taxes is being contested in good faith;
 - (ii) adequate reserves are being maintained for those Taxes and the costs required to contest them to the extent required by applicable GAAP; and
 - (iii) failure to pay those Taxes does not constitute a Material Adverse Effect.
- (b) No Obligor may change its residence for Tax purposes.

21.26 Financial Indebtedness and investments

- (a) The Borrower shall not, and shall ensure that no Borrower Group Member shall, incur or allow to remain outstanding any Financial Indebtedness, other than Permitted Financial Indebtedness.
- (b) The Borrower shall not, and shall ensure that no Borrower Group Member shall, make any advance, loan or extension of credit (including by way of guarantee or otherwise) to any other Person, other than Permitted Loans.

21.27 Permitted Businesses

- (a) The Borrower shall not, and shall ensure that no Borrower Group Member shall, enter into any business activities, whether directly or indirectly, other than Permitted Businesses.
- (b) The Borrower shall not, and shall ensure that no Borrower Group Member shall, establish any representative office other than (in the case of Wynn Resorts Macau) marketing offices or any other place of business in a jurisdiction outside the Macau SAR or its place of incorporation unless such office or place of business has been established and is operated in accordance with all applicable laws and regulations (where failure to do so would have an adverse effect on the interests of the Finance Parties under the Finance Documents) and the terms of the Concession Contract, and, in each case, is notified to the Agent within 30 days of such establishment.

21.28 Restricted Payments

- (a) Except as permitted under paragraph (b) below, the Borrower shall not (and will ensure that no other Borrower Group Member will):
 - (i) declare, make or pay any dividend, charge, fee or other distribution (or interest on any unpaid dividend, charge, fee or other distribution) (whether in cash or in kind) on or in respect of its share capital (or any class of its share capital);
 - (ii) repay or distribute any dividend or share premium reserve;
 - (iii) pay or allow any Borrower Group Member to pay any management, advisory or other fee to or to the order of any of the shareholders of the Borrower;
 - (iv) redeem, repurchase, defease, retire or repay any of its share capital;
 - (v) make any repayment of, or pay any interest on or other amount in respect of any Performance Bond Facility or any Subordinated Loans or any WML Debt;
 - (vi) make any payment of IP Fees; or
 - (vii) make any payment of Corporate Administrative Fees,

(collectively, the "Restricted Payments").

(b) Paragraph (a) above does not apply to a Permitted Payment.

21.29 **Provision of Original Financial Statements**

The Borrower shall procure that copies of the Original Financial Statements are provided to the Agent (in form and substance reasonably satisfactory to the Agent (acting on the instructions of the Majority Lenders)) by no later than the date falling 60 days from the first Utilisation Date under this Agreement.

22. EVENTS OF DEFAULT

Each of the events or circumstances set out in Clause 22 is an Event of Default (save for Clause 22.14 (Acceleration)).

22.1 Non-payment

An Obligor does not pay on the due date any amount payable pursuant to a Finance Document at the place and in the currency in which it is expressed to be payable unless:

- (a) its failure to pay is caused by:
 - (i) administrative or technical error; or

(ii) a Disruption Event; and

payment is made within three Business Days of its due date;

- (b) in the case of a payment of interest on any Loan, payment is made within five Business Days after such interest becomes due in accordance with the terms of this Agreement; or
- (c) in the case of a payment of any amount (other than an amount of principal or interest on any Loan) under the Finance Documents, payment is made within 10 days after such amount becomes due in accordance with the terms of the Finance Documents (or any of them).

22.2 Financial covenants

Subject to Clause 20.4 (Equity cure), any requirement of Clause 20.2 (Financial condition) is not satisfied.

22.3 **Other obligations**

- (a) An Obligor does not comply with any provision of the Finance Documents (other than those referred to in Clause 22.1 (*Non-payment*) or Clause 22.2 (*Financial covenants*)).
- (b) No Event of Default under paragraph (a) above will occur if the failure to comply is capable of remedy and is remedied within 30 days of the earlier of (A) the Agent giving notice to an Obligor and (B) an Obligor becoming aware of the failure to comply.

22.4 Misrepresentation

- (a) Any representation or written statement made or deemed to be made by an Obligor in the Finance Documents or any other document delivered by or on behalf of any Obligor under or in connection with any Finance Document is or proves to have been incorrect or misleading in any material respect when made or deemed to be made.
- (b) No Event of Default under paragraph (a) above will occur if the misrepresentation is capable of remedy and remedied within 30 days of the earlier to occur of (A) the Agent giving notice to an Obligor and (B) an Obligor becoming aware of such misrepresentation.

22.5 Cross default

- (a) Any Financial Indebtedness of any Obligor or any Borrower Group Member is not paid when due nor within any originally applicable grace period.
- (b) Any Financial Indebtedness of any Obligor or any Borrower Group Member is declared to be or otherwise becomes due and payable prior to its specified maturity as a result of an event of default (however described).

- (c) Any commitment for any Financial Indebtedness of any Obligor or any Borrower Group Member is cancelled or suspended by a creditor of any Obligor or any Borrower Group Member as a result of an event of default (however described).
- (d) Any creditor of any Obligor or any Borrower Group Member becomes entitled to declare any Financial Indebtedness of any Obligor or any Borrower Group Member due and payable prior to its specified maturity as a result of an event of default (however described).
- (e) An event of default (however described) occurs in respect of any WML Debt.
- (f) No Event of Default will occur under this Clause 22.5 in respect of any Financial Indebtedness:
 - (i) if that Financial Indebtedness constitutes Subordinated Loans;
 - (ii) if that Financial Indebtedness is in respect of any derivative transaction which is terminated as a result of an event of default (however described) with respect to any counterparty or credit support provider for or specified entity of any counterparty rather than with respect to an Obligor, a Borrower Group Member or a Junior Party; or
 - (iii) if the aggregate amount of Financial Indebtedness or commitment for Financial Indebtedness falling within paragraphs (a) to (e) above (excluding any falling within any of paragraphs (f)(i) to (f)(ii) above) is less than US\$25,000,000 (or its equivalent in any other currency or currencies).

22.6 Insolvency

- (a) A Group Member:
 - (i) is unable or admits inability to pay its debts as they fall due;
 - (ii) suspends making payments on any of its debts; or
 - (iii) by reason of actual or anticipated financial difficulties, commences negotiations with one or more of its creditors (excluding any Finance Party in its capacity as such) with a view to rescheduling any of its indebtedness.
- (b) The value of the assets of the Borrower Group (on a consolidated basis) is less than its liabilities (excluding the Subordinated Loans, but including all other contingent and prospective liabilities) on a consolidated basis.
- (c) A moratorium is declared in respect of any indebtedness of any Group Member.

22.7 Insolvency proceedings

Any corporate action, legal proceedings or other procedure or step is taken in relation to:

- (a) the suspension of payments, a moratorium of any indebtedness, winding-up, dissolution, administration or reorganisation (by way of voluntary arrangement, scheme of arrangement or otherwise) of any Group Member other than a solvent liquidation or reorganisation of any Group Member which is not an Obligor;
- (b) a composition, compromise, assignment or arrangement with any creditor of any Group Member for reasons of financial difficulty;
- (c) the appointment of a liquidator (other than in respect of a solvent liquidation of a Group Member which is not an Obligor), receiver, administrative receiver, administrator, compulsory manager or other similar officer in respect of any Group Member or any of its assets having an aggregate value of at least US\$50,000,000 (or its equivalent); or
- (d) enforcement of any Security over any assets of any Group Member having an aggregate value of at least US\$50,000,000 (or its equivalent),

or any analogous procedure or step is taken in any jurisdiction.

This Clause 22.7 shall not apply to any winding-up petition which is frivolous or vexatious and is discharged, stayed or dismissed within 60 days of commencement.

22.8 Creditors' process

Any expropriation, attachment, sequestration, distress or execution affects any asset or assets of a Group Member having an aggregate value of at least US\$50,000,000 (or its equivalent) and is not discharged within 30 days.

22.9 Unlawfulness

It is or becomes unlawful for an Obligor to perform any of its obligations under the Finance Documents.

22.10 Repudiation

An Obligor repudiates a Finance Document or evidences an intention in writing to repudiate a Finance Document.

22.11 **Expropriation**

The authority or ability of the Borrower or any Borrower Group Member to conduct its business or operations as currently conducted and as proposed to be conducted, or a material part thereof, is wholly or substantially curtailed by any seizure, expropriation, nationalisation, intervention, restriction or other action by or on behalf of any Governmental Authority.

22.12 Judgments

One or more judgments or decrees are entered against any Obligor or any other Group Member involving (for the Group taken as a whole) a liability (that is not paid or covered by insurance) of US\$50,000,000 or its equivalent or more, and all such judgments or decrees, in either case, shall not have been vacated, discharged, stayed or bonded pending appeal within 60 days from the entry thereof.

22.13 Material adverse change

Any other event or circumstance (other than, at any time prior to the time limit for exercising any cure pursuant to Clause 20.4 (*Equity cure*), any circumstances where it is reasonably likely that any of the financial covenants set out in Clause 20 (*Financial Covenants*) may not be complied with) occurs which has a Material Adverse Effect.

22.14 Acceleration

On and at any time after the occurrence of an Event of Default which is continuing the Agent shall if so directed by the Majority Lenders, by notice to the Borrower:

- (a) cancel the Available Commitment of each Lender whereupon each such Available Commitment shall immediately be cancelled and each Facility shall immediately cease to be available for further utilisation;
- (b) declare that all or part of the Loans, together with accrued interest, and all other amounts accrued or outstanding under the Finance Documents be immediately due and payable, whereupon they shall become immediately due and payable; and/or
- (c) declare that all or part of the Loans be payable on demand, whereupon they shall immediately become payable on demand by the Agent on the instructions of the Majority Lenders.

SECTION 9 CHANGES TO PARTIES

23. CHANGES TO THE LENDERS

23.1 Assignments and transfers by the Lenders

Subject to this Clause 23, a Lender (the "Existing Lender") may:

- (a) assign any of its rights; or
- (b) transfer by novation any of its rights and obligations,

to another bank or financial institution (including, without limitation, any insurer or reinsurer) or to a trust, fund or other entity which is regularly engaged in or established for the purpose of making, purchasing or investing in loans, securities or other financial assets (the "**New Lender**").

23.2 Borrower consent

- (a) The consent of the Borrower is required for an assignment or transfer by an Existing Lender, unless the assignment or transfer is:
 - (i) to another Lender (other than a Defaulting Lender); or
 - (ii) an Affiliate of any Lender (other than a Defaulting Lender); or
 - (iii) made at a time when an Event of Default is continuing.
- (b) The consent of the Borrower to an assignment or transfer must not be unreasonably withheld or delayed. The Borrower will be deemed to have given its consent ten Business Days after the Existing Lender has requested it unless consent is expressly refused by the Borrower within that time.

23.3 Other conditions of assignment or transfer

- (a) An assignment will only be effective on:
 - (i) receipt by the Agent (whether in the Assignment Agreement or otherwise) of written confirmation from the New Lender (in form and substance satisfactory to the Agent) that the New Lender will assume the same obligations to the other Finance Parties as it would have been under if it had been an Original Lender; and
 - (ii) performance by the Agent of all necessary "know your customer" or other similar checks under all applicable laws and regulations in relation to such assignment to a New Lender, the completion of which the Agent shall promptly notify to the Existing Lender and the New Lender.

- (b) A transfer will only be effective if the procedure set out in Clause 23.6 (*Procedure for transfer*) is complied with.
- (c) If:
 - (i) a Lender assigns or transfers any of its rights or obligations under the Finance Documents or changes its Facility Office; and
 - (ii) as a result of circumstances existing at the date the assignment, transfer or change occurs, an Obligor would be obliged to make a payment to the New Lender or Lender acting through its new Facility Office under Clause 12 (*Tax Gross-up and Indemnities*) or Clause 13 (*Increased Costs*),

then the New Lender or Lender acting through its new Facility Office is only entitled to receive payment under those Clauses to the same extent as the Existing Lender or Lender acting through its previous Facility Office would have been if the assignment, transfer or change had not occurred. This paragraph (c) shall not apply in respect of an assignment or transfer made in the ordinary course of the primary syndication of any of the Facilities.

(d) Each New Lender, by executing the relevant Transfer Certificate or Assignment Agreement, confirms, for the avoidance of doubt, that the Agent has authority to execute on its behalf any amendment or waiver that has been approved by or on behalf of the requisite Lender or Lenders in accordance with this Agreement on or prior to the date on which the transfer or assignment becomes effective in accordance with this Agreement and that it is bound by that decision to the same extent as the Existing Lender would have been had it remained a Lender.

23.4 Assignment or transfer fee

The New Lender shall, on the date upon which an assignment or transfer takes effect, pay to the Agent (for its own account) a fee of US\$5,000.

23.5 Limitation of responsibility of Existing Lenders

- (a) Unless expressly agreed to the contrary, an Existing Lender makes no representation or warranty and assumes no responsibility to a New Lender for:
 - (i) the legality, validity, effectiveness, adequacy or enforceability of the Finance Documents or any other documents;
 - (ii) the financial condition of any Obligor;
 - (iii) the performance and observance by any Obligor of its obligations under the Finance Documents or any other documents; or
 - (iv) the accuracy of any statements (whether written or oral) made in or in connection with any Finance Document or any other document,

and any representations or warranties implied by law are excluded.

- (b) Each New Lender confirms to the Existing Lender and the other Finance Parties that it:
 - (i) has made (and shall continue to make) its own independent investigation and assessment of the financial condition and affairs of each Obligor and its related entities in connection with its participation in this Agreement and has not relied exclusively on any information provided to it by the Existing Lender in connection with any Finance Document; and
 - (ii) will continue to make its own independent appraisal of the creditworthiness of each Obligor and its related entities whilst any amount is or may be outstanding under the Finance Documents or any Commitment is in force.
- (c) Nothing in any Finance Document obliges an Existing Lender to:
 - (i) accept a re-transfer or re-assignment from a New Lender of any of the rights and obligations assigned or transferred under this Clause 23; or
 - (ii) support any losses directly or indirectly incurred by the New Lender by reason of the non-performance by any Obligor of its obligations under the Finance Documents or otherwise.

23.6 **Procedure for transfer**

- (a) Subject to the conditions set out in Clause 23.2 (*Borrower consent*) and Clause 23.3 (*Other conditions of assignment or transfer*) a transfer is effected in accordance with paragraph (c) below when the Agent executes an otherwise duly completed Transfer Certificate delivered to it by the Existing Lender and the New Lender. The Agent shall, subject to paragraph (b) below, as soon as reasonably practicable after receipt by it of a duly completed Transfer Certificate appearing on its face to comply with the terms of this Agreement and delivered in accordance with the terms of this Agreement, execute that Transfer Certificate.
- (b) The Agent shall only be obliged to execute a Transfer Certificate delivered to it by the Existing Lender and the New Lender once it is satisfied it has complied with all necessary "know your customer" or other similar checks under all applicable laws and regulations in relation to the transfer to such New Lender.
- (c) On the Transfer Date:
 - (i) to the extent that in the Transfer Certificate the Existing Lender seeks to transfer by novation its rights and obligations under the Finance Documents each of the Obligors and the Existing Lender shall be released from further obligations towards one another under the

Finance Documents and their respective rights against one another under the Finance Documents shall be cancelled (being the "**Discharged Rights and Obligations**");

- (ii) each of the Obligors and the New Lender shall assume obligations towards one another and/or acquire rights against one another which differ from the Discharged Rights and Obligations only insofar as that Obligor and the New Lender have assumed and/or acquired the same in place of that Obligor and the Existing Lender;
- (iii) the Agent, the Arranger, the New Lender and other Lenders shall acquire the same rights and assume the same obligations between themselves as they would have acquired and assumed had the New Lender been an Original Lender with the rights and/or obligations acquired or assumed by it as a result of the transfer and to that extent the Agent, the Arranger and the Existing Lender shall each be released from further obligations to each other under the Finance Documents; and
- (iv) the New Lender shall become a Party as a "Lender".

23.7 **Procedure for assignment**

- (a) Subject to the conditions set out in Clause 23.2 (*Borrower consent*) and Clause 23.3 (*Other conditions of assignment or transfer*) an assignment may be effected in accordance with paragraph (c) below when the Agent executes an otherwise duly completed Assignment Agreement delivered to it by the Existing Lender and the New Lender. The Agent shall, subject to paragraph (b) below, as soon as reasonably practicable after receipt by it of a duly completed Assignment Agreement appearing on its face to comply with the terms of this Agreement and delivered in accordance with the terms of this Agreement, execute that Assignment Agreement.
- (b) The Agent shall only be obliged to execute an Assignment Agreement delivered to it by the Existing Lender and the New Lender once it is satisfied it has complied with all necessary "know your customer" or other similar checks under all applicable laws and regulations in relation to the assignment to such New Lender.
- (c) On the Transfer Date:
 - (i) the Existing Lender will assign absolutely to the New Lender the rights under the Finance Documents expressed to be the subject of the assignment in the Assignment Agreement;
 - (ii) the Existing Lender will be released by each Obligor and the other Finance Parties from the obligations owed by it (the "**Relevant Obligations**") and expressed to be the subject of the release in the Assignment Agreement; and

- (iii) the New Lender shall become a Party as a "Lender" and will be bound by obligations equivalent to the Relevant Obligations.
- (d) Lenders may utilise procedures other than those set out in this Clause 23.7 to assign their rights under the Finance Documents (but not, without the consent of the relevant Obligor or unless in accordance with Clause 23.6 (*Procedure for transfer*), to obtain a release by that Obligor from the obligations owed to that Obligor by the Lenders nor the assumption of equivalent obligations by a New Lender) **provided that** they comply with the conditions set out in Clause 23.2 (*Borrower consent*) and Clause 23.3 (*Other conditions of assignment or transfer*).

23.8 Copy of Transfer Certificate, Assignment Agreement or Increase Confirmation to Borrower

The Agent shall, as soon as reasonably practicable after it has executed a Transfer Certificate, an Assignment Agreement or an Increase Confirmation, send to the Borrower a copy of that Transfer Certificate, Assignment Agreement or Increase Confirmation.

23.9 Security over Lenders' rights

In addition to the other rights provided to Lenders under this Clause 23, each Lender may without consulting with or obtaining consent from any Obligor, at any time charge, assign or otherwise create Security in or over (whether by way of collateral or otherwise) all or any of its rights under any Finance Document to secure obligations of that Lender including, without limitation:

- (a) any charge, assignment or other Security to secure obligations to a federal reserve or central bank; and
- (b) any charge, assignment or other Security granted to any holders (or trustee or representatives of holders) of obligations owed, or securities issued, by that Lender as security for those obligations or securities,

except that no such charge, assignment or Security shall:

- (i) release a Lender from any of its obligations under the Finance Documents or substitute the beneficiary of the relevant charge, assignment or Security for the Lender as a party to any of the Finance Documents; or
- (ii) require any payments to be made by an Obligor other than or in excess of, or grant to any person any more extensive rights than, those required to be made or granted to the relevant Lender under the Finance Documents.

24. CHANGES TO THE OBLIGORS

No Obligor may assign any of its rights or transfer any of its rights or obligations under the Finance Documents.

SECTION 10 THE FINANCE PARTIES

25. ROLE OF THE AGENT AND THE ARRANGER

25.1 Appointment of the Agent

- (a) Each of the Arranger and the Lenders appoints the Agent to act as its agent under and in connection with the Finance Documents.
- (b) Each of the Arranger and the Lenders authorises the Agent to perform the duties, obligations and responsibilities and to exercise the rights, powers, authorities and discretions specifically given to the Agent under or in connection with the Finance Documents together with any other incidental rights, powers, authorities and discretions.

25.2 Instructions

- (a) The Agent shall:
 - (i) unless a contrary indication appears in a Finance Document, exercise or refrain from exercising any right, power, authority or discretion vested in it as Agent in accordance with any instructions given to it by:
 - (A) all Lenders if the relevant Finance Document stipulates the matter is an all Lender decision; and
 - (B) in all other cases, the Majority Lenders; and
 - (ii) not be liable for any act (or omission) if it acts (or refrains from acting) in accordance with paragraph (i) above.
- (b) The Agent shall be entitled to request instructions, or clarification of any instruction, from the Majority Lenders (or, if the relevant Finance Document stipulates the matter is a decision for any other Lender or group of Lenders, from that Lender or group of Lenders) as to whether, and in what manner, it should exercise or refrain from exercising any right, power, authority or discretion. The Agent may refrain from acting unless and until it receives any such instructions or clarification that it has requested.
- (c) Save in the case of decisions stipulated to be a matter for any other Lender or group of Lenders under the relevant Finance Document and unless a contrary indication appears in a Finance Document, any instructions given to the Agent by the Majority Lenders shall override any conflicting instructions given by any other Parties and will be binding on all Finance Parties.
- (d) The Agent may refrain from acting in accordance with any instructions of any Lender or group of Lenders until it has received any indemnification and/or security that it may in its discretion require (which may be greater in extent than that contained in the Finance Documents and which may include payment

in advance) for any cost, loss or liability which it may incur in complying with those instructions.

- (e) In the absence of instructions, the Agent may act (or refrain from acting) as it considers to be in the best interest of the Lenders.
- (f) The Agent is not authorised to act on behalf of a Lender (without first obtaining that Lender's consent) in any legal or arbitration proceedings relating to any Finance Document.

25.3 **Duties of the Agent**

- (a) The Agent's duties under the Finance Documents are solely mechanical and administrative in nature.
- (b) Subject to paragraph (c) below, the Agent shall promptly forward to a Party the original or a copy of any document which is delivered to the Agent for that Party by any other Party.
- (c) Without prejudice to Clause 23.8 (*Copy of Transfer Certificate, Assignment Agreement or Increase Confirmation to Borrower*), paragraph (b) above shall not apply to any Transfer Certificate, any Assignment Agreement or any Increase Confirmation.
- (d) Except where a Finance Document specifically provides otherwise, the Agent is not obliged to review or check the adequacy, accuracy or completeness of any document it forwards to another Party.
- (e) If the Agent receives notice from a Party referring to this Agreement, describing a Default and stating that the circumstance described is a Default, it shall promptly notify the other Finance Parties.
- (f) If the Agent is aware of the non-payment of any principal, interest, commitment fee or other fee payable to a Finance Party (other than the Agent or the Arranger) under this Agreement it shall promptly notify the other Finance Parties.
- (g) The Agent shall have only those duties, obligations and responsibilities expressly specified in the Finance Documents to which it is expressed to be a party (and no others shall be implied).

25.4 Role of the Arranger

Except as specifically provided in the Finance Documents, the Arranger has no obligations of any kind to any other Party under or in connection with any Finance Document.

25.5 **No fiduciary duties**

(a) Nothing in any Finance Document constitutes the Agent or the Arranger as a trustee or fiduciary of any other person.

(b) Neither the Agent nor the Arranger shall be bound to account to any Lender for any sum or the profit element of any sum received by it for its own account.

25.6 Business with the Group

The Agent and the Arranger may accept deposits from, lend money to and generally engage in any kind of banking or other business with any Group Member.

25.7 Rights and discretions

- (a) The Agent may:
 - (i) rely on any representation, communication, notice or document believed by it to be genuine, correct and appropriately authorised;
 - (ii) assume that:
 - (A) any instructions received by it from the Majority Lenders, any Lenders or any group of Lenders are duly given in accordance with the terms of the Finance Documents; and
 - (B) unless it has received notice of revocation, that those instructions have not been revoked; and
 - (iii) rely on a certificate from any person:
 - (A) as to any matter of fact or circumstance which might reasonably be expected to be within the knowledge of that person; or
 - (B) to the effect that such person approves of any particular dealing, transaction, step, action or thing,

as sufficient evidence that that is the case and, in the case of paragraph (A) above, may assume the truth and accuracy of that certificate.

- (b) The Agent may assume (unless it has received notice to the contrary in its capacity as agent for the Lenders) that:
 - (i) no Default has occurred (unless it has actual knowledge of a Default arising under Clause 22.1 (*Non-payment*));
 - (ii) any right, power, authority or discretion vested in any Party or any group of Lenders has not been exercised; and
 - (iii) any notice or request made by the Borrower (other than a Utilisation Request) is made on behalf of and with the consent and knowledge of all the Obligors.

- (c) The Agent may engage and pay for the advice or services of any lawyers, accountants, tax advisers, surveyors or other professional advisers or experts.
- (d) Without prejudice to the generality of paragraph (c) above or paragraph (e) below, the Agent may at any time engage and pay for the services of any lawyers to act as independent counsel to the Agent (and so separate from any lawyers instructed by the Lenders) if the Agent in its reasonable opinion deems this to be necessary.
- (e) The Agent may rely on the advice or services of any lawyers, accountants, tax advisers, surveyors or other professional advisers or experts (whether obtained by the Agent or by any other Party) and shall not be liable for any damages, costs or losses to any person, any diminution in value or any liability whatsoever arising as a result of its so relying.
- (f) The Agent may act in relation to the Finance Documents through its officers, employees and agents.
- (g) Unless a Finance Document expressly provides otherwise the Agent may disclose to any other Party any information it reasonably believes it has received as agent under this Agreement.
- (h) Without prejudice to the generality of paragraph (g) above, the Agent:
 - (i) may disclose; and
 - (ii) on the written request of the Borrower or the Majority Lenders shall, as soon as reasonably practicable, disclose,

the identity of a Defaulting Lender to the Borrower and to the other Finance Parties.

- (i) Notwithstanding any other provision of any Finance Document to the contrary, neither the Agent nor the Arranger is obliged to do or omit to do anything if it would, or might in its reasonable opinion, constitute a breach of any law or regulation or a breach of a fiduciary duty or duty of confidentiality.
- (j) Notwithstanding any provision of any Finance Document to the contrary, the Agent is not obliged to expend or risk its own funds or otherwise incur any financial liability in the performance of its duties, obligations or responsibilities or the exercise of any right, power, authority or discretion if it has grounds for believing the repayment of such funds or adequate indemnity against, or security for, such risk or liability is not reasonably assured to it.

25.8 Responsibility for documentation

Neither the Agent nor the Arranger is responsible or liable for:

(a) the adequacy, accuracy or completeness of any information (whether oral or written) supplied by the Agent, the Arranger, an Obligor or any other person

in or in connection with any Finance Document or the Information Memorandum or the transactions contemplated in the Finance Documents or any other agreement, arrangement or document entered into, made or executed in anticipation of, under or in connection with any Finance Document;

- (b) the legality, validity, effectiveness, adequacy or enforceability of any Finance Document or any other agreement, arrangement or document entered into, made or executed in anticipation of, under or in connection with any Finance Document; or
- (c) any determination as to whether any information provided or to be provided to any Finance Party is non-public information the use of which may be regulated or prohibited by applicable law or regulation relating to insider dealing or otherwise.

25.9 No duty to monitor

The Agent shall not be bound to enquire:

- (a) whether or not any Default has occurred;
- (b) as to the performance, default or any breach by any Party of its obligations under any Finance Document; or
- (c) whether any other event specified in any Finance Document has occurred.

25.10 Exclusion of liability

- (a) Without limiting paragraph (b) below (and without prejudice to any other provision of any Finance Document excluding or limiting the liability of the Agent), the Agent will not be liable for:
 - (i) any damages, costs or losses to any person, any diminution in value, or any liability whatsoever arising as a result of taking or not taking any action under or in connection with any Finance Document, unless directly caused by its gross negligence or wilful misconduct;
 - (ii) exercising, or not exercising, any right, power, authority or discretion given to it by, or in connection with, any Finance Document or any other agreement, arrangement or document entered into, made or executed in anticipation of, under or in connection with, any Finance Document, other than by reason of its gross negligence or wilful misconduct; or
 - (iii) without prejudice to the generality of paragraphs (i) and (ii) above, any damages, costs or losses to any person, any diminution in value or any liability whatsoever (including, without limitation, for negligence or any other category of liability whatsoever but not including any claim based on the fraud of the Agent) arising as a result of:

- (A) any act, event or circumstance not reasonably within its control; or
- (B) the general risks of investment in, or the holding of assets in, any jurisdiction,

including (in each case and without limitation) such damages, costs, losses, diminution in value or liability arising as a result of: nationalisation, expropriation or other governmental actions; any regulation, currency restriction, devaluation or fluctuation; market conditions affecting the execution or settlement of transactions or the value of assets (including any Disruption Event); breakdown, failure or malfunction of any third party transport, telecommunications, computer services or systems; natural disasters or acts of God; war, terrorism, insurrection or revolution; or strikes or industrial action.

- (b) No Party (other than the Agent) may take any proceedings against any officer, employee or agent of the Agent in respect of any claim it might have against the Agent or in respect of any act or omission of any kind by that officer, employee or agent in relation to any Finance Document and any officer, employee or agent of the Agent may rely on this paragraph (b) subject to Clause 1.4 (*Third party rights*) and the provisions of the Third Parties Act.
- (c) The Agent will not be liable for any delay (or any related consequences) in crediting an account with an amount required under the Finance Documents to be paid by the Agent if the Agent has taken all necessary steps as soon as reasonably practicable to comply with the regulations or operating procedures of any recognised clearing or settlement system used by the Agent for that purpose.
- (d) Nothing in this Agreement shall oblige the Agent or the Arranger to carry out:
 - (i) any "know your customer" or other checks in relation to any person; or
 - (ii) any check on the extent to which any transaction contemplated by this Agreement might be unlawful for any Lender or for any Affiliate of any Lender,

on behalf of any Lender and each Lender confirms to the Agent and the Arranger that it is solely responsible for any such checks it is required to carry out and that it may not rely on any statement in relation to such checks made by the Agent or the Arranger.

(e) Without prejudice to any provision of any Finance Document excluding or limiting the Agent's liability, any liability of the Agent arising under or in connection with any Finance Document shall be limited to the amount of actual loss which has been suffered (as determined by reference to the date of default of the Agent or, if later, the date on which the loss arises as a result of such default) but without reference to any special conditions or circumstances known to the Agent at any time which increase the amount of that loss. In no

event shall the Agent be liable for any loss of profits, goodwill, reputation, business opportunity or anticipated saving, or for special, punitive, indirect or consequential damages, whether or not the Agent has been advised of the possibility of such loss or damages.

25.11 Lenders' indemnity to the Agent

Each Lender shall (in proportion to its share of the Total Commitments or, if the Total Commitments are then zero, to its share of the Total Commitments immediately prior to their reduction to zero) indemnify the Agent, within three Business Days of demand, against any cost, loss or liability (including, without limitation, for negligence or any other category of liability whatsoever) incurred by the Agent (otherwise than by reason of the Agent's gross negligence or wilful misconduct) (or, in the case of any cost, loss or liability pursuant to Clause 28.10 (*Disruption to payment systems etc.*), notwithstanding the Agent's negligence, gross negligence or any other category of liability whatsoever but not including any claim based on the fraud of the Agent) in acting as Agent under the Finance Documents (unless the Agent has been reimbursed by an Obligor pursuant to a Finance Document).

25.12 Resignation of the Agent

- (a) The Agent may resign and appoint one of its Affiliates acting through an office in Macau or Hong Kong as successor by giving notice to the Lenders and the Borrower.
- (b) Alternatively the Agent may resign by giving 30 days' notice to the Lenders and the Borrower, in which case the Majority Lenders (after consultation with the Borrower) may appoint a successor Agent.
- (c) If the Majority Lenders have not appointed a successor Agent in accordance with paragraph (b) above within 20 days after notice of resignation was given, the retiring Agent (after consultation with the Borrower) may appoint a successor Agent (acting through an office in Macau or Hong Kong).
- (d) The retiring Agent shall, at its own cost, make available to the successor Agent such documents and records and provide such assistance as the successor Agent may reasonably request for the purposes of performing its functions as Agent under the Finance Documents.
- (e) The Agent's resignation notice shall only take effect upon the appointment of a successor.
- (f) Upon the appointment of a successor, the retiring Agent shall be discharged from any further obligation in respect of the Finance Documents (other than its obligations under paragraph (d) above) but shall remain entitled to the benefit of Clause 14.3 (*Indemnity to the Agent*) and this Clause 25 (and any agency fees for the account of the retiring Agent shall cease to accrue from (and shall be payable on) that date). Any successor and each of the other Parties shall have the same rights and obligations amongst themselves as they would have had if such successor had been an original Party.

- (g) After consultation with the Borrower, the Majority Lenders may, by notice to the Agent, require it to resign in accordance with paragraph (b) above. In this event, the Agent shall resign in accordance with paragraph (b) above.
- (h) The Agent shall resign in accordance with paragraph (b) above (and, to the extent applicable, shall use reasonable endeavours to appoint a successor Agent pursuant to paragraph (c) above) if on or after the date which is three months before the earliest FATCA Application Date relating to any payment to the Agent under the Finance Documents, either:
 - (i) the Agent fails to respond to a request under Clause 12.7 (*FATCA information*) and the Borrower or a Lender reasonably believes that the Agent will not be (or will have ceased to be) a FATCA Exempt Party on or after that FATCA Application Date;
 - (ii) the information supplied by the Agent pursuant to Clause 12.7 (*FATCA information*) indicates that the Agent will not be (or will have ceased to be) a FATCA Exempt Party on or after that FATCA Application Date; or
 - (iii) the Agent notifies the Borrower and the Lenders that the Agent will not be (or will have ceased to be) a FATCA Exempt Party on or after that FATCA Application Date;

and (in each case) the Borrower or a Lender reasonably believes that a Party will be required to make a FATCA Deduction that would not be required if the Agent were a FATCA Exempt Party, and the Borrower or that Lender, by notice to the Agent, requires it to resign.

25.13 Confidentiality

- (a) In acting as agent for the Finance Parties, the Agent shall be regarded as acting through its agency division which shall be treated as a separate entity from any other of its divisions or departments.
- (b) If information is received by another division or department of the Agent, it may be treated as confidential to that division or department and the Agent shall not be deemed to have notice of it.

25.14 Relationship with the Lenders

- (a) The Agent may treat the person shown in its records as Lender at the opening of business (in the place of the Agent's principal office as notified to the Finance Parties from time to time) as the Lender acting through its Facility Office:
 - (i) entitled to or liable for any payment due under any Finance Document on that day; and

(ii) entitled to receive and act upon any notice, request, document or communication or make any decision or determination under any Finance Document made or delivered on that day,

unless it has received not less than five Business Days' prior notice from that Lender to the contrary in accordance with the terms of this Agreement.

(b) Any Lender may by notice to the Agent appoint a person to receive on its behalf all notices, communications, information and documents to be made or despatched to that Lender under the Finance Documents. Such notice shall contain the address, fax number and (where communication by electronic mail or other electronic means is permitted under Clause 30.5 (*Electronic communication*)) electronic mail address and/or any other information required to enable the transmission of information by that means (and, in each case, the department or officer, if any, for whose attention communication is to be made) and be treated as a notification of a substitute address, fax number, electronic mail address (or such other information), department and officer by that Lender for the purposes of Clause 30.2 (*Addresses*) and paragraph (a)(ii) of Clause 30.5 (*Electronic communication*) and the Agent shall be entitled to treat such person as the person entitled to receive all such notices, communications, information and documents as though that person were that Lender.

25.15 Credit appraisal by the Lenders

Without affecting the responsibility of any Obligor for information supplied by it or on its behalf in connection with any Finance Document, each Lender confirms to the Agent and the Arranger that it has been, and will continue to be, solely responsible for making its own independent appraisal and investigation of all risks arising under or in connection with any Finance Document including but not limited to:

- (a) the financial condition, status and nature of each Group Member;
- (b) the legality, validity, effectiveness, adequacy or enforceability of any Finance Document and any other agreement, arrangement or document entered into, made or executed in anticipation of, under or in connection with any Finance Document;
- (c) whether that Lender has recourse, and the nature and extent of that recourse, against any Party or any of its respective assets under or in connection with any Finance Document, the transactions contemplated by the Finance Documents or any other agreement, arrangement or document entered into, made or executed in anticipation of, under or in connection with any Finance Document; and
- (d) the adequacy, accuracy or completeness of the Information Memorandum and any other information provided by the Agent, any Party or by any other person under or in connection with any Finance Document, the transactions contemplated by any Finance Document or any other agreement, arrangement

or document entered into, made or executed in anticipation of, under or in connection with any Finance Document.

25.16 **Deduction from amounts payable by the Agent**

If any Party owes an amount to the Agent under the Finance Documents the Agent may, after giving notice to that Party, deduct an amount not exceeding that amount from any payment to that Party which the Agent would otherwise be obliged to make under the Finance Documents and apply the amount deducted in or towards satisfaction of the amount owed. For the purposes of the Finance Documents that Party shall be regarded as having received any amount so deducted.

25.17 Role of Reference Banks

- (a) No Reference Bank is under any obligation to provide a quotation or any other information to the Agent.
- (b) No Reference Bank will be liable for any action taken by it under or in connection with any Finance Document, or for any Reference Bank Quotation, unless directly caused by its gross negligence or wilful misconduct.
- (c) No Party (other than the relevant Reference Bank) may take any proceedings against any officer, employee or agent of any Reference Bank in respect of any claim it might have against that Reference Bank or in respect of any act or omission of any kind by that officer, employee or agent in relation to any Finance Document, or to any Reference Bank Quotation, and any officer, employee or agent of each Reference Bank may rely on this Clause 25.17 subject to Clause 1.4 (*Third party rights*) and the provisions of the Third Parties Act.

25.18 Third party Reference Banks

A Reference Bank which is not a Party may rely on Clause 25.17 (*Role of Reference Banks*), Clause 34.3 (*Other exceptions*) and Clause 36 (*Confidentiality of Funding Rates and Reference Bank Quotations*) subject to Clause 1.4 (*Third party rights*) and the provisions of the Third Parties Act.

26. CONDUCT OF BUSINESS BY THE FINANCE PARTIES

No provision of this Agreement will:

- (a) interfere with the right of any Finance Party to arrange its affairs (tax or otherwise) in whatever manner it thinks fit;
- (b) oblige any Finance Party to investigate or claim any credit, relief, remission or repayment available to it or the extent, order and manner of any claim; or
- (c) oblige any Finance Party to disclose any information relating to its affairs (tax or otherwise) or any computations in respect of Tax.

27. SHARING AMONG THE FINANCE PARTIES

27.1 **Payments to Finance Parties**

If a Finance Party (a "**Recovering Finance Party**") receives or recovers any amount from an Obligor other than in accordance with Clause 28 (*Payment Mechanics*) (a "**Recovered Amount**") and applies that amount to a payment due under the Finance Documents then:

- (a) the Recovering Finance Party shall, within three Business Days, notify details of the receipt or recovery to the Agent;
- (b) the Agent shall determine whether the receipt or recovery is in excess of the amount the Recovering Finance Party would have been paid had the receipt or recovery been received or made by the Agent and distributed in accordance with Clause 28 (*Payment Mechanics*), without taking account of any Tax which would be imposed on the Agent in relation to the receipt, recovery or distribution; and
- (c) the Recovering Finance Party shall, within three Business Days of demand by the Agent, pay to the Agent an amount (the "**Sharing Payment**") equal to such receipt or recovery less any amount which the Agent determines may be retained by the Recovering Finance Party as its share of any payment to be made, in accordance with Clause 28.5 (*Partial payments*).

27.2 Redistribution of payments

The Agent shall treat the Sharing Payment as if it had been paid by the relevant Obligor and distribute it between the Finance Parties (other than the Recovering Finance Party) (the "**Sharing Finance Parties**") in accordance with Clause 28.5 (*Partial payments*) towards the obligations of that Obligor to the Sharing Finance Parties.

27.3 **Recovering Finance Party's rights**

On a distribution by the Agent under Clause 27.2 (*Redistribution of payments*) of a payment received by a Recovering Finance Party from an Obligor, as between the relevant Obligor and the Recovering Finance Party, an amount of the Recovered Amount equal to the Sharing Payment will be treated as not having been paid by that Obligor.

27.4 **Reversal of redistribution**

If any part of the Sharing Payment received or recovered by a Recovering Finance Party becomes repayable and is repaid by that Recovering Finance Party, then:

(a) each Sharing Finance Party shall, upon request of the Agent, pay to the Agent for the account of that Recovering Finance Party an amount equal to the appropriate part of its share of the Sharing Payment (together with an amount as is necessary to reimburse that Recovering Finance Party for its proportion

of any interest on the Sharing Payment which that Recovering Finance Party is required to pay) (the "**Redistributed Amount**"); and

(b) as between the relevant Obligor and each relevant Sharing Finance Party, an amount equal to the relevant Redistributed Amount will be treated as not having been paid by that Obligor.

27.5 Exceptions

- (a) This Clause 27 shall not apply to the extent that the Recovering Finance Party would not, after making any payment pursuant to this Clause 27, have a valid and enforceable claim against the relevant Obligor.
- (b) A Recovering Finance Party is not obliged to share with any other Finance Party any amount which the Recovering Finance Party has received or recovered as a result of taking legal or arbitration proceedings, if:
 - (i) it notified that other Finance Party of the legal or arbitration proceedings; and
 - (ii) that other Finance Party had an opportunity to participate in those legal or arbitration proceedings but did not do so as soon as reasonably practicable having received notice and did not take separate legal or arbitration proceedings.

SECTION 11 ADMINISTRATION

28. PAYMENT MECHANICS

28.1 **Payments to the Agent**

- (a) On each date on which an Obligor or a Lender is required to make a payment under a Finance Document, that Obligor or Lender shall make the same available to the Agent (unless a contrary indication appears in a Finance Document) for value on the due date at the time and in such funds specified by the Agent as being customary at the time for settlement of transactions in the relevant currency in the place of payment.
- (b) Payment shall be made to such account in the principal financial centre of the country of that currency and with such bank as the Agent, in each case, specifies.

28.2 Distributions by the Agent

Each payment received by the Agent under the Finance Documents for another Party shall, subject to Clause 28.3 (*Distributions to an Obligor*) and Clause 28.4 (*Clawback and pre-funding*) be made available by the Agent as soon as practicable after receipt to the Party entitled to receive payment in accordance with this Agreement (in the case of a Lender, for the account of its Facility Office), to such account as that Party may notify to the Agent by not less than five Business Days' notice with a bank specified by that Party in the principal financial centre of the country of that currency.

28.3 Distributions to an Obligor

The Agent may (with the consent of the Obligor or in accordance with Clause 29 (*Set-off*)) apply any amount received by it for that Obligor in or towards payment (on the date and in the currency and funds of receipt) of any amount due from that Obligor under the Finance Documents or in or towards purchase of any amount of any currency to be so applied.

28.4 Clawback and pre-funding

- (a) Where a sum is to be paid to the Agent under the Finance Documents for another Party, the Agent is not obliged to pay that sum to that other Party (or to enter into or perform any related exchange contract) until it has been able to establish to its satisfaction that it has actually received that sum.
- (b) Unless paragraph (c) below applies, if the Agent pays an amount to another Party and it proves to be the case that the Agent had not actually received that amount, then the Party to whom that amount (or the proceeds of any related exchange contract) was paid by the Agent shall on demand refund the same to the Agent together with interest on that amount from the date of payment to the date of receipt by the Agent, calculated by the Agent to reflect its cost of funds.

- (c) If the Agent has notified the Lenders that it is willing to make available amounts for the account of the Borrower before receiving funds from the Lenders then if and to the extent that the Agent does so but it proves to be the case that it does not then receive funds from a Lender in respect of a sum which it paid to the Borrower:
 - (i) the Agent shall notify the Borrower of that Lender's identity and the Borrower shall on demand refund it to the Agent; and
 - (ii) the Lender by whom those funds should have been made available or, if that Lender fails to do so, the Borrower, shall on demand pay to the Agent the amount (as certified by the Agent) which will indemnify the Agent against any funding cost incurred by it as a result of paying out that sum before receiving those funds from that Lender.

28.5 **Partial payments**

- (a) If the Agent receives a payment that is insufficient to discharge all the amounts then due and payable by an Obligor under the Finance Documents, the Agent shall apply that payment towards the obligations of that Obligor under the Finance Documents in the following order:
 - (i) **first**, in or towards payment *pro rata* of any unpaid amount owing to the Agent under the Finance Documents;
 - (ii) **secondly**, in or towards payment *pro rata* of any accrued interest, fee or commission due but unpaid under this Agreement;
 - (iii) thirdly, in or towards payment *pro rata* of any principal due but unpaid under this Agreement; and
 - (iv) **fourthly**, in or towards payment *pro rata* of any other sum due but unpaid under the Finance Documents.
- (b) The Agent shall, if so directed by the Majority Lenders, vary the order set out in paragraphs (a)(ii) to (a)(iv) above.
- (c) Paragraphs (a) and (b) above will override any appropriation made by an Obligor.

28.6 No set-off by Obligors

All payments to be made by an Obligor under the Finance Documents shall be calculated and be made without (and free and clear of any deduction for) set-off or counterclaim.

28.7 Business Days

(a) Any payment under any Finance Document which is due to be made on a day that is not a Business Day shall be made on the next Business Day in the same calendar month (if there is one) or the preceding Business Day (if there is not).

- (b) During any extension of the due date for payment of any principal or Unpaid Sum under this Agreement interest is payable on the principal or Unpaid Sum at the rate payable on the original due date.
- (c) If the Termination Date falls on a day that is not a Business Day, the Termination Date shall fall on the immediately preceding Business Day instead.

28.8 Currency of account

- (a) Subject to paragraphs (b) to (e) below, the Base Currency is the currency of account and payment for any sum due from an Obligor under any Finance Document.
- (b) A repayment of a Loan or Unpaid Sum or a part of a Loan or Unpaid Sum shall be made in the currency in which that Loan or Unpaid Sum is denominated, pursuant to this Agreement, on its due date.
- (c) Each payment of interest shall be made in the currency in which the sum in respect of which the interest is payable was denominated, pursuant to this Agreement, when that interest accrued.
- (d) Each payment in respect of costs, expenses or Taxes shall be made in the currency in which the costs, expenses or Taxes are incurred.
- (e) Any amount expressed to be payable in a currency other than the Base Currency shall be paid in that other currency.

28.9 **Change of currency**

- (a) Unless otherwise prohibited by law, if more than one currency or currency unit are at the same time recognised by the central bank of any country as the lawful currency of that country, then:
 - (i) any reference in the Finance Documents to, and any obligations arising under the Finance Documents in, the currency of that country shall be translated into, or paid in, the currency or currency unit of that country designated by the Agent (after consultation with the Borrower); and
 - (ii) any translation from one currency or currency unit to another shall be at the official rate of exchange recognised by the central bank for the conversion of that currency or currency unit into the other, rounded up or down by the Agent (acting reasonably).
- (b) If a change in any currency of a country occurs, this Agreement will, to the extent the Agent (acting reasonably and after consultation with the Borrower) specifies to be necessary, be amended to comply with any generally accepted conventions and market practice in the Relevant Market and otherwise to reflect the change in currency.

28.10 **Disruption to payment systems etc.**

If either the Agent determines (in its discretion) that a Disruption Event has occurred or the Agent is notified by the Borrower that a Disruption Event has occurred:

- (a) the Agent may, and shall if requested to do so by the Borrower, consult with the Borrower with a view to agreeing with the Borrower such changes to the operation or administration of the Facilities as the Agent may deem necessary in the circumstances;
- (b) the Agent shall not be obliged to consult with the Borrower in relation to any changes mentioned in paragraph (a) above if, in its opinion, it is not practicable to do so in the circumstances and, in any event, shall have no obligation to agree to such changes;
- (c) the Agent may consult with the Finance Parties in relation to any changes mentioned in paragraph (a) above but shall not be obliged to do so if, in its opinion, it is not practicable to do so in the circumstances;
- (d) any such changes agreed upon by the Agent and the Borrower shall (whether or not it is finally determined that a Disruption Event has occurred) be binding upon the Parties as an amendment to (or, as the case may be, waiver of) the terms of the Finance Documents notwithstanding the provisions of Clause 34 (*Amendments and Waivers*);
- (e) the Agent shall not be liable for any damages, costs or losses to any person, any diminution in value or any liability whatsoever (including, without limitation for negligence, gross negligence or any other category of liability whatsoever but not including any claim based on the fraud of the Agent) arising as a result of its taking, or failing to take, any actions pursuant to or in connection with this Clause 28.10; and
- (f) the Agent shall notify the Finance Parties of all changes agreed pursuant to paragraph (d) above.

29. SET-OFF

While an Event of Default is continuing, a Finance Party may set off any matured obligation due from an Obligor under the Finance Documents (to the extent beneficially owned by that Finance Party) against any matured obligation owed by that Finance Party to that Obligor, regardless of the place of payment, booking branch or currency of either obligation. If the obligations are in different currencies, the Finance Party may convert either obligation at a market rate of exchange in its usual course of business for the purpose of the set-off.

30. NOTICES

30.1 **Communications in writing**

Any communication to be made under or in connection with the Finance Documents shall be made in writing and, unless otherwise stated, may be made by fax or letter.

30.2 Addresses

The address and fax number (and the department or officer, if any, for whose attention the communication is to be made) of each Party for any communication or document to be made or delivered under or in connection with the Finance Documents is:

- (a) in the case of an Obligor, that identified with its name below;
- (b) in the case of each Lender, that notified in writing to the Agent on or prior to the date on which it becomes a Party; and
- (c) in the case of the Agent, that identified with its name below,

or any substitute address or fax number or department or officer as the Party may notify to the Agent (or the Agent may notify to the other Parties, if a change is made by the Agent) by not less than five Business Days' notice.

30.3 Delivery

- (a) Any communication or document made or delivered by one person to another under or in connection with the Finance Documents will only be effective:
 - (i) if by way of fax, when received in legible form; or
 - (ii) if by way of letter, when it has been left at the relevant address or five Business Days after being deposited in the post postage prepaid in an envelope addressed to it at that address;

and, if a particular department or officer is specified as part of its address details provided under Clause 30.2 (*Addresses*), if addressed to that department or officer.

- (b) Any communication or document to be made or delivered to the Agent will be effective only when actually received by the Agent and then only if it is expressly marked for the attention of the department or officer identified with the Agent's signature below (or any substitute department or officer as the Agent shall specify for this purpose).
- (c) All notices from or to an Obligor shall be sent through the Agent.
- (d) Any communication or document made or delivered to the Borrower in accordance with this Clause 30.3 will be deemed to have been made or delivered to each of the Obligors.

(e) Any communication or document which becomes effective, in accordance with paragraphs (a) to (d) above, after 5:00 p.m. in the place of receipt shall be deemed only to become effective on the following day.

30.4 Notification of address and fax number

Promptly upon changing its address or fax number, the Agent shall notify the other Parties.

30.5 Electronic communication

- (a) Any communication or document to be made or delivered by one Party to another under or in connection with the Finance Documents may be made or delivered by electronic mail or other electronic means (including, without limitation, by way of posting to a secure website) if those two Parties:
 - (i) notify each other in writing of their electronic mail address and/or any other information required to enable the transmission of information by that means; and
 - (ii) notify each other of any change to their address or any other such information supplied by them by not less than five Business Days' notice.
- (b) Any such electronic communication or delivery as specified in paragraph (a) above to be made between an Obligor and a Finance Party may only be made in that way to the extent that those two Parties agree that, unless and until notified to the contrary, this is to be an accepted form of communication or delivery.
- (c) Any such electronic communication or document as specified in paragraph (a) above made or delivered by one Party to another will be effective only when actually received (or made available) in readable form and in the case of any electronic communication or document made or delivered by a Party to the Agent only if it is addressed in such a manner as the Agent shall specify for this purpose.
- (d) Any electronic communication or document which becomes effective, in accordance with paragraph (c) above, after 5:00 p.m. in the place in which the Party to whom the relevant communication or document is sent or made available has its address for the purpose of this Agreement shall be deemed only to become effective on the following day.
- (e) Any reference in a Finance Document to a communication being sent or received or a document being delivered shall be construed to include that communication or document being made available in accordance with this Clause 30.5.

30.6 English language

- (a) Any notice given under or in connection with any Finance Document must be in English.
- (b) All other documents provided under or in connection with any Finance Document must be:
 - (i) in English; or
 - (ii) if not in English, and if so required by the Agent, accompanied by a certified English translation and, in this case, the English translation will prevail unless the document is a constitutional, statutory or other official document.

31. CALCULATIONS AND CERTIFICATES

31.1 Accounts

In any litigation or arbitration proceedings arising out of or in connection with a Finance Document, the entries made in the accounts maintained by a Finance Party are *prima facie* evidence of the matters to which they relate.

31.2 Certificates and determinations

Any certification or determination by a Finance Party of a rate or amount under any Finance Document is, in the absence of manifest error, conclusive evidence of the matters to which it relates.

31.3 Day count convention

Any interest, commission or fee accruing under a Finance Document will accrue from day to day and is calculated on the basis of the actual number of days elapsed and a year of 360 days (where due in US dollars) and 365 days (where due in HK dollars) or, in any case where the practice in the Relevant Market differs, in accordance with that market practice.

32. PARTIAL INVALIDITY

If, at any time, any provision of a Finance Document is or becomes illegal, invalid or unenforceable in any respect under any law of any jurisdiction, neither the legality, validity or enforceability of the remaining provisions nor the legality, validity or enforceability of such provision under the law of any other jurisdiction will in any way be affected or impaired.

33. REMEDIES AND WAIVERS

No failure to exercise, nor any delay in exercising, on the part of any Finance Party, any right or remedy under a Finance Document shall operate as a waiver of any such right or remedy or constitute an election to affirm any of the Finance Documents. No election to affirm any Finance Document on the part of any Finance Party shall be

effective unless it is in writing. No single or partial exercise of any right or remedy shall prevent any further or other exercise or the exercise of any other right or remedy. The rights and remedies provided in each Finance Document are cumulative and not exclusive of any rights or remedies provided by law.

34. AMENDMENTS AND WAIVERS

34.1 **Required consents**

- (a) Subject to Clause 34.2 (*All Lender matters*) and Clause 34.3 (*Other exceptions*) any term of the Finance Documents may be amended or waived only with the consent of the Majority Lenders and the Obligors and any such amendment or waiver will be binding on all Parties.
- (b) The Agent may effect, on behalf of any Finance Party, any amendment or waiver permitted by this Clause 34.

34.2 All Lender matters

Subject to Clause 34.4 (*Replacement of Screen Rate*) and Clause 34.5 (*Concession Reorganisation*), an amendment or waiver of any term of any Finance Document that has the effect of changing or which relates to:

- (a) the definition of "Anti-Terrorism Laws", "Majority Lenders", "Relevant Ratio" or "Sanctions" in Clause 1.1 (*Definitions*);
- (b) an extension to the date of payment of any amount under the Finance Documents;
- (c) a reduction in the Margin or a reduction in the amount of any payment of principal, interest, fees or commission payable;
- (d) a change in currency of payment of any amount under the Finance Documents;
- (e) an increase in any Commitment, an extension of the Availability Period or any requirement that a cancellation of Commitments reduces the Commitments of the Lenders rateably under the relevant Facility;
- (f) any provision which expressly requires the consent of all the Lenders;
- (g) Clause 2.3 (Finance Parties' rights and obligations), Clause 5.1 (Delivery of a Utilisation Request), Clause 7.1 (Illegality), Clause 7.3 (Property Mandatory Prepayment Event), Clause 7.8 (Application of prepayments), Clause 21.12 (OFAC), Clause 21.13 (FCPA), Clause 21.14 (Money Laundering Laws), Clause 21.15 (Anti-Terrorism Laws), Clause 23 (Changes to the Lenders), Clause 24 (Changes to the Obligors), Clause 27 (Sharing Among the Finance Parties), this Clause 34, Clause 43 (Governing Law) or Clause 44.1 (Jurisdiction);

- (h) the nature or scope of the guarantee and indemnity granted under Clause 17 (*Guarantee and Indemnity*), shall not be made without the prior consent of all the Lenders;
- (i) the definition of "Money Laundering Laws" in paragraph (a) of Clause 21.14 (Money Laundering Laws); or
- (j) the definition of "Anti-Bribery Laws" in paragraph (a) of Clause 21.13 (*FCPA*).

34.3 **Other exceptions**

An amendment or waiver which relates to the rights or obligations of the Agent or the Arranger or a Reference Bank (each in their capacity as such) may not be effected without the consent of the Agent, the Arranger or that Reference Bank, as the case may be.

34.4 Replacement of Screen Rate

- (a) Subject to Clause 34.3 (*Other exceptions*), if a Screen Rate Replacement Event has occurred in relation to any Screen Rate for a currency which can be selected for a Loan, any amendment or waiver which relates to:
 - (i) providing for the use of a Replacement Benchmark in relation to that currency in place of that Screen Rate; and

(ii)

- (A) aligning any provision of any Finance Document to the use of that Replacement Benchmark;
- (B) enabling that Replacement Benchmark to be used for the calculation of interest under this Agreement (including, without limitation, any consequential changes required to enable that Replacement Benchmark to be used for the purposes of this Agreement);
- (C) implementing market conventions applicable to that Replacement Benchmark;
- (D) providing for appropriate fallback (and market disruption) provisions for that Replacement Benchmark; or
- (E) adjusting the pricing to reduce or eliminate, to the extent reasonably practicable, any transfer of economic value from one Party to another as a result of the application of that Replacement Benchmark (and if any adjustment or method for calculating any adjustment has been formally designated, nominated or recommended by the Relevant Nominating Body,

the adjustment shall be determined on the basis of that designation, nomination or recommendation),

may be made with the consent of the Agent (acting on the instructions of the Majority Lenders) and the Obligors.

- (b) If, as at 31 July 2022, this Agreement provides that the rate of interest for a Loan in US dollars is to be determined by reference to the Screen Rate for LIBOR:
 - (i) a Screen Rate Replacement Event shall be deemed to have occurred on that date in relation to the Screen Rate for US dollars; and
 - (ii) the Agent (acting on the instructions of the Majority Lenders) and the Obligors shall enter into negotiations in good faith with a view to agreeing the use of a Replacement Benchmark in relation to US dollars in place of that Screen Rate from and including a date no later than 30 November 2022, which is consistent with market standards, prevailing market conventions and any regulatory requirements from the relevant regulators at that time, and which may include a term or daily secured overnight funding rate ("SOFR") plus any related benchmark replacement adjustment or credit adjustment spread over SOFR.
- (c) If any Lender fails to respond to a request for an amendment or waiver described in. or for any other vote of Lenders in relation to, paragraph (a) or (b) above within ten Business Days (or such longer time period in relation to any request which the Company and the Agent may agree) of that request being made:
 - (i) its Commitment shall not be included for the purpose of calculating the Total Commitments when ascertaining whether any relevant percentage of Total Commitments has been obtained to approve that request; and
 - (ii) its status as a Lender shall be disregarded for the purpose of ascertaining whether the agreement of any specified group of Lenders has been obtained to approve that request.
- (d) In this Clause 34.4:

"**Relevant Nominating Body**" means any applicable central bank, regulator or other supervisory authority or a group of them, or any working group or committee sponsored or chaired by, or constituted at the request of, any of them or the Financial Stability Board.

"Replacement Benchmark" means a benchmark rate which is:

(a) formally designated, nominated or recommended as the replacement for a Screen Rate by:

- (i) the administrator of that Screen Rate (**provided that** the market or economic reality that such benchmark rate measures is the same as that measured by that Screen Rate); or
- (ii) any Relevant Nominating Body,

and if replacements have, at the relevant time, been formally designated, nominated or recommended under both paragraphs, the "Replacement Benchmark" will be the replacement under paragraph (ii) above;

- (b) in the opinion of the Majority Lenders and the Obligors, generally accepted in the international or any relevant domestic syndicated loan markets as the appropriate successor to a Screen Rate; or
- (c) in the opinion of the Majority Lenders and the Obligors, an appropriate successor to a Screen Rate.

"Screen Rate Replacement Event" means, in relation to a Screen Rate:

- (a) the methodology, formula or other means of determining that Screen Rate has, in the opinion of the Majority Lenders and the Obligors, materially changed;
- (b)
- (i)
- (A) the administrator of that Screen Rate or its supervisor publicly announces that such administrator is insolvent; or
- (B) information is published in any order, decree, notice, petition or filing, however described, of or filed with a court, tribunal, exchange, regulatory authority or similar administrative, regulatory or judicial body which reasonably confirms that the administrator of that Screen Rate is insolvent,

provided that, in each case, at that time, there is no successor administrator to continue to provide that Screen Rate;

- (ii) the administrator of that Screen Rate publicly announces that it has ceased or will cease to provide that Screen Rate permanently or indefinitely and, at that time, there is no successor administrator to continue to provide that Screen Rate;
- (iii) the supervisor of the administrator of that Screen Rate publicly announces that such Screen Rate has been or will be permanently or indefinitely discontinued;

- (iv) the administrator of that Screen Rate or its supervisor announces that that Screen Rate may no longer be used; or
- (v) in the case of a Screen Rate for LIBOR, the supervisor of the administrator of that Screen Rate makes a public announcement or publishes information:
 - (A) stating that that Screen Rate is no longer or, as of a specified future date will no longer be, representative of the underlying market or economic reality that it is intended to measure and that representativeness will not be restored (as determined by such supervisor); and
 - (B) with awareness that any such announcement or publication will engage certain triggers for fallback provisions in contracts which may be activated by any such pre-cessation announcement or publication;
- (c) the administrator of that Screen Rate determines that that Screen Rate should be calculated in accordance with its reduced submissions or other contingency or fallback policies or arrangements and either:
 - (i) the circumstance(s) or event(s) leading to such determination are not (in the opinion of the Majority Lenders and the Obligors) temporary; or
 - (ii) that Screen Rate is calculated in accordance with any such policy or arrangement for a period no less than 30 days; or
- (d) in the opinion of the Majority Lenders and the Obligors, that Screen Rate is otherwise no longer appropriate for the purposes of calculating interest under this Agreement.

34.5 Concession Reorganisation

- (a) Subject to Clause 34.3 (*Other exceptions*), if a Concession Reorganisation Event has occurred, any amendment or waiver which relates to:
 - the definitions in Clause 1.1 (*Definitions*) of any of "Wynn Macau", "Wynn Resorts Macau", the "Site", "Palo", the "Cotai Property", the "Cotai Site", "Site Easements", "Cotai Site Easements", the "Concession Contract", the "Land Concession Contract" and/or the "Cotai Land Concession Contract";
 - (ii) Clause 21.8 (*Compliance with Major Project Documents and Concession Contracts*), Clause 21.19 (*Limitation on Transactions with Affiliates*) and/or 21.24 (*Amendments to documents*); and/or
 - (iii) any other provision of any Finance Document,

in each case, in order to provide for a Concession Reorganisation, may be made with the consent of the Agent (acting on the instructions of the Majority Lenders) and the Borrower.

(b) In this Clause 34.5:

"**Concession Reorganisation Event**" means any Governmental Authority in the Macau SAR notifies Wynn Resorts Macau in writing prior to the extension or renewal of the Concession Contract that a Concession Reorganisation is required as a condition to the extension or renewal of the Concession Contract; and

"**Concession Reorganisation**" means any transaction or series of transactions in which Wynn Resorts Macau disposes to another directly or indirectly wholly owned Subsidiary of the Borrower all of Wynn Resorts Macau's assets and undertaking (including its Resort Assets) in accordance with all applicable Legal Requirements.

34.6 Replacement of Lenders

(a) If:

- (i) any Lender becomes a Non-Consenting Lender (as defined in paragraph (d) below);
- (ii) any Lender becomes a Disrupted Lender; or
- (iii) an Obligor becomes obliged to repay any amount in accordance with Clause 7.1 (*Illegality*) or to pay additional amounts pursuant to Clause 12.2 (*Tax gross-up*), Clause 12.3 (*Tax indemnity*) or Clause 13 (*Increased Costs*) to any Lender,

then the Borrower may, on five Business Days' prior written notice to the Agent and such Lender, replace such Lender by requiring such Lender to (and, to the extent permitted by law, such Lender shall) transfer pursuant to Clause 23.6 (*Procedure for transfer*) all (and not part only) of its rights and obligations under this Agreement to an Eligible Institution (a "**Replacement Lender**") which is acceptable to the Borrower and which confirms its willingness to assume and does assume all the obligations of the transferring Lender in accordance with Clause 23 (*Changes to the Lenders*) for a purchase price in cash payable at the time of transfer in an amount equal to the outstanding principal amount of such Lender's participation in the outstanding Loans and all accrued interest, Break Costs and other amounts payable in relation thereto under the Finance Documents (including without limitation, any amounts contemplated by sub-paragraph (iii) above).

- (b) The replacement of a Lender pursuant to this Clause 34.6 shall be subject to the following conditions:
 - (i) the Borrower shall have no right to replace the Agent;

- (ii) neither the Agent nor the Lender shall have any obligation to the Borrower to find a Replacement Lender;
- (iii) in the event of a replacement of a Lender such replacement must take place no later than 60 days after the date on which that Lender is deemed a Lender;
- (iv) in no event shall the Lender replaced under this Clause 34.6 be required to pay or surrender to such Lender any of the fees received by such Lender pursuant to the Finance Documents; and
- (v) the Lender shall only be obliged to transfer its rights and obligations pursuant to paragraph (a) above once it is satisfied that it has complied with all necessary "know your customer" or other similar checks under all applicable laws and regulations in relation to that transfer.
- (c) A Lender shall perform the checks described in paragraph (b)(v) above as soon as reasonably practicable following delivery of a notice referred to in paragraph (a) above and shall notify the Agent and the Borrower when it is satisfied that it has complied with those checks.
- (d) In the event that:
 - (i) the Borrower or the Agent (at the request of the Borrower) has requested the Lenders to give a consent in relation to, or to agree to a waiver or amendment of, any provisions of the Finance Documents;
 - (ii) the consent, waiver or amendment in question requires the approval of all the Lenders; and
 - (iii) Lenders whose Commitments aggregate in the case of a consent, waiver or amendment requiring the approval of all the Lenders, more than 66²/₃ per cent. of the Total Commitments (or, if the Total Commitments have been reduced to zero, aggregated more than 66²/₃ per cent. of the Total Commitments prior to that reduction) have consented or agreed to such waiver or amendment,

then any Lender who does not and continues not to consent or agree to such waiver or amendment shall be deemed a "**Non-Consenting Lender**".

34.7 Disenfranchisement of Excluded Lender

(a) In ascertaining whether the agreement of Lender(s) holding any given percentage (including unanimity) of the Total Commitments or the Commitments (in respect of any or all of the Facilities) has been obtained to approve any request from the Group, the Commitment and the participations in the Utilisations (in respect of the Facilities) of any Excluded Lender (unless it has ceased to be an Excluded Lender in accordance with the definition thereof) shall be deemed to be zero in calculating the Total Commitments, the Commitments and the participations in the Utilisations, in each case, of the

relevant threshold or class of Lenders required to approve such request and its status as a Lender shall be ignored (and it shall be deemed not to be a Lender).

- (b) For the purposes of this Clause 34.7, the Agent may assume that the following Lenders are Defaulting Lenders:
 - (i) any Lender which has notified the Agent that it has become a Defaulting Lender; or
 - (ii) any Lender in relation to which it is aware that any of the events or circumstances referred to in paragraph(a) or (c) of the definition of Defaulting Lender has occurred,

unless it has received notice to the contrary from the Lender concerned (together with any supporting evidence reasonably requested by the Agent) or the Agent is otherwise aware that the Lender has ceased to be a Defaulting Lender.

34.8 **Replacement of a Defaulting Lender**

- (a) The Borrower may, at any time a Lender has become and continues to be a Defaulting Lender, by giving five Business Days' prior written notice to the Agent and such Lender:
 - (i) replace such Lender by requiring such Lender to (and, to the extent permitted by law, such Lender shall) transfer pursuant to Clause 23 (*Changes to the Lenders*) all (and not part only) of its rights and obligations under this Agreement;
 - (ii) require such Lender to (and, to the extent permitted by law, such Lender shall) transfer pursuant to Clause 23 (*Changes to the Lenders*) all (and not part only) of the undrawn Commitment of the Lender; or
 - (iii) require such Lender to (and, to the extent permitted by law, such Lender shall) transfer pursuant to Clause 23 (*Changes to the Lenders*) all (and not part only) of its rights and obligations in respect of the Facilities,

to an Eligible Institution (a "**Replacement Lender**") which is acceptable to the Borrower and which confirms its willingness to assume and does assume all the obligations, or all the relevant obligations, of the transferring Lender in accordance with Clause 23 (*Changes to the Lenders*) for a purchase price in cash payable at the time of transfer which is either:

(i) in an amount equal to the outstanding principal amount of such Lender's participation in the outstanding Loans and all accrued interest, Break Costs and other amounts payable in relation thereto under the Finance Documents; or

- (ii) in an amount agreed between that Defaulting Lender, the Replacement Lender and the Borrower and which does not exceed the amount described in paragraph (i) above.
- (b) Any transfer of rights and obligations of a Defaulting Lender pursuant to this Clause 34.8 shall be subject to the following conditions:
 - (i) the Borrower shall have no right to replace the Agent;
 - (ii) neither the Agent nor the Defaulting Lender shall have any obligation to the Borrower to find a Replacement Lender;
 - (iii) the transfer must take place no later than 60 days after the notice referred to in paragraph (a) above;
 - (iv) in no event shall the Defaulting Lender be required to pay or surrender to the Replacement Lender any of the fees received by the Defaulting Lender pursuant to the Finance Documents; and
 - (v) the Defaulting Lender shall only be obliged to transfer its rights and obligations pursuant to paragraph (a) above once it is satisfied that it has complied with all necessary "know your customer" or other similar checks under all applicable laws and regulations in relation to that transfer to the Replacement Lender.
- (c) The Defaulting Lender shall perform the checks described in paragraph (b)(v) above as soon as reasonably practicable following delivery of a notice referred to in paragraph (a) above and shall notify the Agent and the Borrower when it is satisfied that it has complied with those checks.

34.9 Excluded Commitments

If any Lender fails to respond to such a request or such a vote within 21 days of that request being made (unless the Borrower and the Agent agree to a longer time period in relation to any request):

- (a) its Commitment(s) shall not be included for the purpose of calculating the Total Commitments under the relevant Facility/ies when ascertaining whether any relevant percentage (including, for the avoidance of doubt, unanimity) of Total Commitments has been obtained to approve that request; and
- (b) its status as a Lender shall be disregarded for the purpose of ascertaining whether the agreement of any specified group of Lenders has been obtained to approve that request.

35. CONFIDENTIAL INFORMATION

35.1 **Confidentiality**

Each Finance Party agrees to keep all Confidential Information confidential and not to disclose it to anyone, save to the extent permitted by Clause 35.2 (*Disclosure of*

Confidential Information) and Clause 35.3 (*Disclosure to numbering service providers*), and to ensure that all Confidential Information is protected with security measures and a degree of care that would apply to its own confidential information.

35.2 Disclosure of Confidential Information

Any Finance Party may disclose:

- (a) to any of its Affiliates and Related Funds, head office, other branches, regional offices and any of its or their officers, directors, employees, professional advisers, auditors, insurers, insurance brokers, reinsurers, other service providers, partners and Representatives such Confidential Information as that Finance Party shall consider appropriate if any person to whom the Confidential Information is to be given pursuant to this paragraph (a) is informed in writing of its confidential nature and that some or all of such Confidential Information may be price-sensitive information except that there shall be no such requirement to so inform if the recipient is subject to professional obligations to maintain the confidentiality of the information or is otherwise bound by requirements of confidentiality in relation to the Confidential Information;
- (b) to any person:
 - to (or through) whom it assigns or transfers (or may potentially assign or transfer) all or any of its rights and/or obligations under one or more Finance Documents or which succeeds (or which may potentially succeed) it as Agent, and, in each case, to any of that person's Affiliates, Related Funds, Representatives and professional advisers;
 - (ii) with (or through) whom it enters into (or may potentially enter into), whether directly or indirectly, any subparticipation in relation to, or any other transaction under which payments are to be made or may be made by reference to, one or more Finance Documents and/or one or more Obligors and to any of that person's Affiliates, Related Funds, Representatives and professional advisers;
 - (iii) appointed by any Finance Party or by a person to whom paragraph (b)(i) or (ii) above applies to receive communications, notices, information or documents delivered pursuant to the Finance Documents on its behalf (including, without limitation, any person appointed under paragraph (b) of Clause 25.14 (*Relationship with the Lenders*));
 - (iv) who invests in or otherwise finances (or may potentially invest in or otherwise finance), directly or indirectly, any transaction referred to in paragraph (b)(i) or (b)(ii) above;
 - (v) to whom information is required or requested to be disclosed by any court of competent jurisdiction or any governmental, banking, taxation

or other regulatory authority or similar body, the rules of any relevant stock exchange or pursuant to any applicable law or regulation;

- (vi) to whom information is required to be disclosed in connection with, and for the purposes of, any litigation, arbitration, administrative or other investigations, proceedings or disputes;
- (vii) to whom or for whose benefit that Finance Party charges, assigns or otherwise creates Security (or may do so) pursuant to Clause 23.9 (*Security over Lenders' rights*);
- (viii) who is a Party; or
- (ix) with the consent of the Borrower;

in each case, such Confidential Information as that Finance Party shall consider appropriate if:

- (A) in relation to paragraphs (b)(i), (b)(ii) and (b)(iii) above, the person to whom the Confidential Information is to be given has entered into a Confidentiality Undertaking except that there shall be no requirement for a Confidentiality Undertaking if the recipient is a professional adviser and is subject to professional obligations to maintain the confidentiality of the Confidential Information;
- (B) in relation to paragraph (b)(iv) above, the person to whom the Confidential Information is to be given has entered into a Confidentiality Undertaking or is otherwise bound by requirements of confidentiality in relation to the Confidential Information they receive and is informed that some or all of such Confidential Information may be price-sensitive information;
- (C) in relation to paragraphs (b)(v), (b)(vi) and (b)(vii) above, the person to whom the Confidential Information is to be given is informed of its confidential nature and that some or all of such Confidential Information may be price-sensitive information except that there shall be no requirement to so inform if, in the opinion of that Finance Party, it is not practicable so to do in the circumstances;
- (c) to any person appointed by that Finance Party or by a person to whom paragraph (b)(i) or (b)(ii) above applies to provide administration or settlement services in respect of one or more of the Finance Documents including without limitation, in relation to the trading of participations in respect of the Finance Documents, such Confidential Information as may be required to be disclosed to enable such service provider to provide any of the services referred to in this paragraph (c) if the service provider to whom the Confidential Information is to be given has entered into a confidentiality agreement

substantially in the form of the LMA Master Confidentiality Undertaking for Use With Administration/Settlement Service Providers or such other form of confidentiality undertaking agreed between the Borrower and the relevant Finance Party; and

(d) to any rating agency (including its professional advisers) such Confidential Information as may be required to be disclosed to enable such rating agency to carry out its normal rating activities in relation to the Finance Documents and/or the Obligors if the rating agency to whom the Confidential Information is to be given is informed of its confidential nature and that some or all of such Confidential Information may be price-sensitive information.

35.3 Disclosure to numbering service providers

- (a) Any Finance Party may disclose to any national or international numbering service provider appointed by that Finance Party to provide identification numbering services in respect of this Agreement, the Facilities (or any of them) and/or one or more Obligors the following information:
 - (i) names of Obligors;
 - (ii) country of domicile of Obligors;
 - (iii) place of incorporation of Obligors;
 - (iv) date of this Agreement;
 - (v) Clause 43 (Governing Law);
 - (vi) the names of the Agent and the Arranger;
 - (vii) date of each amendment and restatement of this Agreement;
 - (viii) amounts of, and names of, the Facilities (and any tranches);
 - (ix) amount of Total Commitments;
 - (x) currencies of the Facilities;
 - (xi) type of Facilities;
 - (xii) ranking of Facilities;
 - (xiii) Termination Date for the Facilities;
 - (xiv) changes to any of the information previously supplied pursuant to paragraphs (i) to (xiii) above; and
 - (xv) such other information agreed between such Finance Party and the Borrower,

to enable such numbering service provider to provide its usual syndicated loan numbering identification services.

- (b) The Parties acknowledge and agree that each identification number assigned to this Agreement, the Facilities (or any of them) and/or one or more Obligors by a numbering service provider and the information associated with each such number may be disclosed to users of its services in accordance with the standard terms and conditions of that numbering service provider.
- (c) Each Obligor represents that none of the information set out in paragraphs (i) to (xv) of paragraph (a) above is, nor will at any time be, unpublished price-sensitive information.
- (d) The Agent shall notify the Borrower and the other Finance Parties of:
 - (i) the name of any numbering service provider appointed by the Agent in respect of this Agreement, the Facilities (or any of them) and/or one or more Obligors; and
 - (ii) the number or, as the case may be, numbers assigned to this Agreement, the Facilities (or any of them) and/or one or more Obligors by such numbering service provider.

35.4 Entire agreement

This Clause 35 constitutes the entire agreement between the Parties in relation to the obligations of the Finance Parties under the Finance Documents regarding Confidential Information and supersedes any previous agreement, whether express or implied, regarding Confidential Information.

35.5 **Inside information**

Each of the Finance Parties acknowledges that some or all of the Confidential Information is or may be price-sensitive information and that the use of such information may be regulated or prohibited by applicable legislation including securities law relating to insider dealing and market abuse and each of the Finance Parties undertakes not to use any Confidential Information for any unlawful purpose.

35.6 Notification of disclosure

Each of the Finance Parties agrees (to the extent permitted by law and regulation) to inform the Borrower:

(a) of the circumstances of any disclosure of Confidential Information made pursuant to paragraph (b)(v) of Clause 35.2 (*Disclosure of Confidential Information*) except where such disclosure is made to any of the persons referred to in that paragraph during the ordinary course of its supervisory or regulatory function; and

(b) upon becoming aware that Confidential Information has been disclosed in breach of this Clause 35.

35.7 Continuing obligations

The obligations in this Clause 35 are continuing and, in particular, shall survive and remain binding on each Finance Party for a period of twelve months from the earlier of:

- (a) the date on which all amounts payable by the Obligors under or in connection with this Agreement have been paid in full and all Commitments have been cancelled or otherwise cease to be available; and
- (b) the date on which such Finance Party otherwise ceases to be a Finance Party.

36. CONFIDENTIALITY OF FUNDING RATES AND REFERENCE BANK QUOTATIONS

36.1 Confidentiality and disclosure

- (a) The Agent and each Obligor agree to keep each Funding Rate (and, in the case of the Agent, each Reference Bank Quotation) confidential and not to disclose it to anyone, save to the extent permitted by paragraphs (b), (c) and (d) below.
- (b) The Agent may disclose:
 - (i) any Funding Rate (but not, for the avoidance of doubt, any Reference Bank Quotation) to the Borrower pursuant to Clause 8.4 (*Notification of rates of interest*); and
 - (ii) any Funding Rate or any Reference Bank Quotation to any person appointed by it to provide administration services in respect of one or more of the Finance Documents to the extent necessary to enable such service provider to provide those services if the service provider to whom that information is to be given has entered into a confidentiality agreement substantially in the form of the LMA Master Confidentiality Undertaking for Use With Administration/Settlement Service Providers or such other form of confidentiality undertaking agreed between the Agent and the relevant Lender or Reference Bank, as the case may be.
- (c) The Agent may disclose any Funding Rate or any Reference Bank Quotation, and each Obligor may disclose any Funding Rate, to:
 - (i) any of its Affiliates and any of its or their officers, directors, employees, professional advisers, auditors, insurers, insurance brokers, other service providers, partners and Representatives if any person to whom that Funding Rate or Reference Bank Quotation is to be given pursuant to this paragraph (i) is informed in writing of its confidential nature and that it may be price-sensitive information except that there

shall be no such requirement to so inform if the recipient is subject to professional obligations to maintain the confidentiality of that Funding Rate or Reference Bank Quotation or is otherwise bound by requirements of confidentiality in relation to it;

- (ii) any person to whom information is required or requested to be disclosed by any court of competent jurisdiction or any governmental, banking, taxation or other regulatory authority or similar body, the rules of any relevant stock exchange or pursuant to any applicable law or regulation if the person to whom that Funding Rate or Reference Bank Quotation is to be given is informed in writing of its confidential nature and that it may be price-sensitive information except that there shall be no requirement to so inform if, in the opinion of the Agent or the relevant Obligor, as the case may be, it is not practicable to do so in the circumstances;
- (iii) any person to whom information is required to be disclosed in connection with, and for the purposes of, any litigation, arbitration, administrative or other investigations, proceedings or disputes if the person to whom that Funding Rate or Reference Bank Quotation is to be given is informed in writing of its confidential nature and that it may be price-sensitive information except that there shall be no requirement to so inform if, in the opinion of the Agent or the relevant Obligor, as the case may be, it is not practicable to do so in the circumstances;
- (iv) any person to the extent that such information has been publicly disclosed other than in breach of this Clause 36;
- (v) to any rating agency or direct or indirect provider of credit protection to such Finance Party or any of its Affiliates; and
- (vi) any person with the consent of the relevant Lender or Reference Bank, as the case may be.
- (d) The Agent's obligations in this Clause 36 relating to Reference Bank Quotations are without prejudice to its obligations to make notifications under Clause 8.4 (*Notification of rates of interest*) **provided that** (other than pursuant to paragraph (b)(i) above) the Agent shall not include the details of any individual Reference Bank Quotation as part of any such notification.

36.2 Related obligations

(a) The Agent and each Obligor acknowledge that each Funding Rate (and, in the case of the Agent, each Reference Bank Quotation) is or may be price-sensitive information and that its use may be regulated or prohibited by applicable legislation including securities law relating to insider dealing and market abuse and the Agent and each Obligor undertake not to use any Funding Rate or, in the case of the Agent, any Reference Bank Quotation for any unlawful purpose.

- (b) The Agent and each Obligor agree (to the extent permitted by law and regulation) to inform the relevant Lender or Reference Bank, as the case may be:
 - (i) of the circumstances of any disclosure made pursuant to paragraph (c)(ii) of Clause 36.1 (*Confidentiality and disclosure*) except where such disclosure is made to any of the persons referred to in that paragraph during the ordinary course of its supervisory or regulatory function; and
 - (ii) upon becoming aware that any information has been disclosed in breach of this Clause 36.

36.3 No Event of Default

No Event of Default will occur under Clause 22.3 (*Other obligations*) by reason only of an Obligor's failure to comply with this Clause 36.

37. CONTRACTUAL RECOGNITION OF BAIL-IN

- (a) Notwithstanding any other term of any Finance Document or any other agreement, arrangement or understanding between the Parties, each Party acknowledges and accepts that any liability of any Party to any other Party under or in connection with the Finance Documents may be subject to Bail-In Action by the relevant Resolution Authority and acknowledges and accepts to be bound by the effect of:
 - (i) any Bail-In Action in relation to any such liability, including (without limitation):
 - (A) a reduction, in full or in part, in the principal amount, or outstanding amount due (including any accrued but unpaid interest) in respect of any such liability;
 - (B) a conversion of all, or part of, any such liability into shares or other instruments of ownership that may be issued to, or conferred on, it; and
 - (C) a cancellation of any such liability; and
 - (ii) a variation of any term of any Finance Document to the extent necessary to give effect to any Bail-In Action in relation to any such liability.

(b) In this Clause 37:

"**Article 55 BRRD**" means Article 55 of Directive 2014/59/EU establishing a framework for the recovery and resolution of credit institutions and investment firms.

"Bail-In Action" means the exercise of any Write-down and Conversion Powers.

"Bail-In Legislation" means:

- (i) in relation to an EEA Member Country which has implemented, or which at any time implements, Article 55 BRRD, the relevant implementing law or regulation as described in the EU Bail-In Legislation Schedule from time to time;
- (ii) in relation to the United Kingdom, the UK Bail-In Legislation; and
- (iii) in relation to any state other than such an EEA Member Country and the United Kingdom, any analogous law or regulation from time to time which requires contractual recognition of any Write-down and Conversion Powers contained in that law or regulation.

"EEA Member Country" means any member state of the European Union, Iceland, Liechtenstein and Norway.

"**EU Bail-In Legislation Schedule**" means the document described as such and published by the Loan Market Association (or any successor person) from time to time.

"**Resolution Authority**" means any body which has authority to exercise any Write-down and Conversion Powers.

"**UK Bail-In Legislation**" means Part I of the United Kingdom Banking Act 2009 and any other law or regulation applicable in the United Kingdom relating to the resolution of unsound or failing banks, investment firms or other financial institutions or their affiliates (otherwise than through liquidation, administration or other insolvency proceedings).

"Write-down and Conversion Powers" means:

- (i) in relation to any Bail-In Legislation described in the EU Bail-In Legislation Schedule from time to time, the powers described as such in relation to that Bail-In Legislation in the EU Bail-In Legislation Schedule;
- (ii) in relation to the UK Bail-In Legislation, any powers under that UK Bail-In Legislation to cancel, transfer or dilute shares issued by a person that is a bank or investment firm or other financial institution or affiliate of a bank, investment firm or other financial institution, to

cancel, reduce, modify or change the form of a liability of such a person or any contract or instrument under which that liability arises, to convert all or part of that liability into shares, securities or obligations of that person or any other person, to provide that any such contract or instrument is to have effect as if a right had been exercised under it or to suspend any obligation in respect of that liability or any of the powers under that UK Bail-In Legislation that are related to or ancillary to any of those powers; and

- (iii) in relation to any other applicable Bail-In Legislation:
 - (A) any powers under that Bail-In Legislation to cancel, transfer or dilute shares issued by a person that is a bank or investment firm or other financial institution or affiliate of a bank, investment firm or other financial institution, to cancel, reduce, modify or change the form of a liability of such a person or any contract or instrument under which that liability arises, to convert all or part of that liability into shares, securities or obligations of that person or any other person, to provide that any such contract or instrument is to have effect as if a right had been exercised under it or to suspend any obligation in respect of that liability or any of the powers under that Bail-In Legislation that are related to or ancillary to any of those powers; and
 - (B) any similar or analogous powers under that Bail-In Legislation.

38. NON-RECOURSE LIABILITY

Notwithstanding any provision in the Finance Documents to the contrary no Operative shall be personally liable for payments due hereunder or under any of the Finance Documents or for the performance of any obligation hereunder or thereunder, save, in relation to any Operative, pursuant to any Finance Document to which such Operative is party. The sole recourse of the Finance Parties for satisfaction of any of the obligations of any of the Obligors hereunder and under the other Finance Documents shall be against the Obligors, and not against any assets or property of any Operative save to the extent such Operative is party to a Finance Document and is expressed to be liable for such obligation thereunder.

39. COUNTERPARTS

Each Finance Document may be executed in any number of counterparts, and this has the same effect as if the signatures on the counterparts were on a single copy of the Finance Document.

SECTION 12 GOVERNING LAW AND ENFORCEMENT

40. USA PATRIOT ACT

Each Lender hereby notifies each Obligor that pursuant to the requirements of the USA PATRIOT Act, such Lender is required to obtain, verify and record information that identifies each Obligor, which information includes the name and address of each Obligor and other information that will allow such Lender to identify each Obligor in accordance with the USA PATRIOT Act.

41. ACKNOWLEDGEMENT REGARDING ANY SUPPORTED QFCS

To the extent that the Finance Documents provide support, through a guarantee or otherwise, for any hedge agreement or any other agreement or instrument that is a QFC (such support, "**QFC Credit Support**", and each such QFC, a "**Supported QFC**"), the parties acknowledge and agree as follows with respect to the resolution power of the Federal Deposit Insurance Corporation under the U.S. Federal Deposit Insurance Act and Title II of the Dodd-Frank Wall Street Reform and Consumer Protection Act (together with the regulations promulgated thereunder, the "**U.S. Special Resolution Regimes**") in respect of such Supported QFC and QFC Credit Support:

(a) In the event a Covered Entity that is party to a Supported QFC (each, a "Covered Party") becomes subject to a proceeding under a U.S. Special Resolution Regime, the transfer of such Supported QFC and the benefit of such QFC Credit Support (and any interest and obligation in or under such Supported QFC and such QFC Credit Support, and any rights in property securing such Supported QFC or such QFC Credit Support) from such Covered Party will be effective to the same extent as the transfer would be effective under the U.S. Special Resolution Regime if the Supported QFC and such QFC Credit Support (and any such interest, obligation and rights in property) were governed by the laws of the United States or a state of the United States. In the event a Covered Party or a BHC Act Affiliate of a Covered Party becomes subject to a proceeding under a U.S. Special Resolution Regime, Default Rights under the Finance Documents that might otherwise apply to such Supported QFC or any QFC Credit Support that may be exercised against such Covered Party are permitted to be exercised to no greater extent than such Default Rights could be exercised under the U.S. Special Resolution Regime if the Supported QFC and the Finance Documents were governed by the laws of the United States or a state of the United States or a state of the Supported QFC and the Finance Documents were governed by the laws of the United States or a state of the Supported QFC and the Finance Documents were governed by the laws of the United States or a state of the United States or a state of the Supported QFC and the Finance Documents were governed by the laws of the United States or a state of the United States. Without limitation of the foregoing, it is understood and agreed that rights and remedies of the parties with respect to a Defaulting Lender shall in no event

affect the rights of any Covered Party with respect to a Supported QFC or any QFC Credit Support.

(b) As used in this Clause 41, the following terms have the following meanings:

"**BHC** Act Affiliate" of a party means an "affiliate" (as such term is defined under, and interpreted in accordance with, 12 U.S.C.1841(k)) of such party.

"**Covered Entity**" means any of the following: (i) a "covered entity" as that term is defined in, and interpreted in accordance with, 12C.F.R. § 252.82(b); (ii) a "covered bank" as that term is defined in, and interpreted in accordance with, 12 C.F.R. § 47.3(b); or (iii) a "covered FSI" as that term is defined in, and interpreted in accordance with, 12 C.F.R. § 382.2(b).

"**Default Right**" has the meaning assigned to that term in, and shall be interpreted in accordance with, 12 C.F.R. §§ 252.81, 47.2 or 382.1, as applicable.

"**QFC**" has the meaning assigned to the term "qualified financial contract" in, and shall be interpreted in accordance with, 12 U.S.C.5390(c)(8)(D).

42. BENEFICIAL OWNERSHIP

(a) In this Clause 42:

"**Beneficial Ownership Certification**" means a certification regarding beneficial ownership as required by the U.S. Beneficial Ownership Regulation; and

"Beneficial Ownership Regulation" means 31 C.F.R. § 1010.230.

- (b) At least five days prior to the first Utilisation Date (and no earlier than the date of this Agreement), if the Borrower qualifies as a "legal entity customer" under the Beneficial Ownership Regulation or if the Agent (for itself or on behalf of any Lender) at any time so requests a Beneficial Ownership Certification, it shall deliver a Beneficial Ownership Certification in relation to it within 30 days of the date of this Agreement or the date of such request (as the case may be).
- (c) The Borrower will promptly notify the Agent and each Lender of any change in the information provided in the Beneficial Ownership Certification that would result in a change to the list of beneficial owners identified in such certification.

43. GOVERNING LAW

This Agreement and any non-contractual obligations arising out of or in connection with it are governed by English law.

44. ENFORCEMENT

44.1 Jurisdiction

- (a) The courts of England have exclusive jurisdiction to settle any dispute arising out of or in connection with this Agreement (including a dispute relating to the existence, validity or termination of this Agreement or any non-contractual obligation arising out of or in connection with this Agreement) (a "**Dispute**").
- (b) The Parties agree that the courts of England are the most appropriate and convenient courts to settle Disputes and accordingly no Party will argue to the contrary.
- (c) Notwithstanding paragraphs (a) and (b) above, no Finance Party shall be prevented from taking proceedings relating to a Dispute in any other courts with jurisdiction. To the extent allowed by law, the Finance Parties may take concurrent proceedings in any number of jurisdictions.

44.2 Service of process

Without prejudice to any other mode of service allowed under any relevant law, each Obligor (other than an Obligor incorporated in England and Wales):

- (a) irrevocably appoints Law Debenture Corporate Services Limited as its agent for service of process in relation to any proceedings before the English courts in connection with any Finance Document; and
- (b) agrees that failure by a process agent to notify the relevant Obligor of the process will not invalidate the proceedings concerned.

44.3 Waiver of immunity

Each Obligor irrevocably and unconditionally:

- (a) agrees not to claim any immunity from proceedings brought by a Finance Party against it in relation to a Finance Document and to ensure that no such claim is made on its behalf;
- (b) consents generally to the giving of any relief or the issue of any process in connection with those proceedings; and
- (c) waives all rights of immunity in respect of it or its assets.

44.4 Waiver of trial by jury

EACH PARTY WAIVES ANY RIGHT IT MAY HAVE TO A JURY TRIAL OF ANY CLAIM OR CAUSE OF ACTION IN CONNECTION WITH ANY FINANCE DOCUMENT OR ANY TRANSACTION CONTEMPLATED BY ANY FINANCE DOCUMENT. THIS AGREEMENT MAY BE FILED AS A WRITTEN CONSENT TO TRIAL BY COURT.

This Agreement has been entered into on the date stated at the beginning of this Agreement.

SCHEDULE 1 THE ORIGINAL LENDERS

Name of Original Lender	Facility A Commitment (USD)	Facility B Commitment (HKD)
Bank of China Limited, Macau Branch	NIL	3,510,000,000
Industrial and Commercial Bank of China (Macau) Limited	NIL	1,950,000,000
United Overseas Bank Limited	250,000,000	NIL
Bank of Communications Co., Ltd. Macau Branch	NIL	1,170,000,000
Bank of America, N.A.	NIL	702,000,000
The Bank of Nova Scotia	NIL	499,200,000
Sumitomo Mitsui Banking Corporation	62,500,000	NIL
Banco Nacional Ultramarino, S.A.	NIL	429,000,000
Deutsche Bank AG, Singapore Branch	NIL	417,300,000
BNP Paribas, acting through its Hong Kong branch	NIL	390,000,000
Banco Comercial de Macau, S.A.	NIL	195,000,000

SCHEDULE 2 CONDITIONS PRECEDENT

1. Original Obligors and other entities

- (a) A copy of the constitutional documents and statutory registers of each Obligor, WML Corp, WML Finance and Wynn Resorts Macau.
- (b) A copy of a resolution of the board of directors of each Obligor (or, in respect of the Company, an extract of its board resolutions), WML Corp, WML Finance and Wynn Resorts Macau:
 - (i) approving the terms of, and the transactions contemplated by, the Finance Documents to which it is a party and resolving that it execute the Finance Documents to which it is a party;
 - (ii) authorising a specified person or persons to execute the Finance Documents to which it is a party on its behalf; and
 - (iii) authorising a specified person or persons, on its behalf, to sign and/or despatch all documents and notices (including, if relevant, any Utilisation Request) to be signed and/or despatched by it under or in connection with the Finance Documents to which it is a party.
- (c) A specimen of the signature of each person authorised by the resolution referred to in paragraph (b) above.
- (d) A certificate of each Obligor (signed by a director) confirming that borrowing or guaranteeing, as appropriate, the Total Commitments would not cause any borrowing, guaranteeing or similar limit binding on any Obligor to be exceeded.
- (e) A certificate of an authorised signatory of the relevant Obligor, WML Corp, WML Finance and Wynn Resorts Macau certifying that each copy document relating to it specified in this Schedule 2 is correct, complete and in full force and effect as at a date no earlier than the date of this Agreement.
- (f) In respect of each Obligor, WML Corp and WML Finance, a copy of a certificate of good standing issued by the Registrar of Companies in the Cayman Islands dated no earlier than one month of the date of this Agreement.

2. Legal opinions

- (a) A legal opinion of Clifford Chance, legal advisers to the Arranger and the Agent in England, substantially in the form distributed to the Original Lenders prior to signing this Agreement.
- (b) A legal opinion of Walkers (Hong Kong), legal advisers to the Arranger and the Agent in the Cayman Islands, substantially in the form distributed to the Original Lenders prior to signing this Agreement.

(c) A legal opinion of Mr Henrique Saldanha, Macanese legal adviser to the Arranger and the Agent in Macau, substantially in the form distributed to the Original Lenders prior to signing this Agreement.

3. Other documents and evidence

- (a) This Agreement, the Subordination Agreement and any Fee Letter, duly executed by the parties thereto.
- (b) Evidence that any process agent referred to in Clause 44.2 (*Service of process*) has accepted its appointment.
- (c) A copy of a prepayment consent letter in respect of the Existing Facilities substantially in the form circulated to the Agent prior to the date of this Agreement and countersigned by the intercreditor agent under the Existing Facilities (the "**Prepayment Consent Letter**"), together with copies of a Prepayment Notice and a Confirmation Notice.
- (d) The Funds Flow Statement in a form agreed between the Agent and the Borrower detailing the proposed movements of funds on the date on which the Existing Facilities are to be refinanced in full.
- (e) Evidence that the proceeds of the first Utilisation in respect of each Facility, together with other cash sources available to the Group, will be sufficient to discharge the Existing Facilities in full in the manner contemplated by the Prepayment Consent Letter.
- (f) Evidence that the fees, costs and expenses then due from the Borrower pursuant to Clause 11 (*Fees*) and Clause 16 (*Costs and Expenses*) have been paid or will be paid by the first Utilisation Date.

SCHEDULE 3 UTILISATION REQUEST

From: WM Cayman Holdings Limited II as Borrower

To: [Agent]

Dated:

Dear Sirs

USD1,500,000,000 Facility Agreement dated [] (the "Agreement")

1. We refer to the Agreement. This is a Utilisation Request. Terms defined in the Agreement have the same meaning in this Utilisation Request unless given a different meaning in this Utilisation Request.

2. We wish to borrow a Loan on the following terms:

Proposed Utilisation Date:	[] (or, if that is not a Business Day, the next Business Day)
Aggregate Amount of Loan (in USD, the Fixed HKD/USD Exchange Rate):	at [] or, if less, the aggregate of the Available Facilities	
Amount of Facility A Loan	USD[] or, if less, the Available Facility	
Amount of Facility B Loan	HKD	[] or, if less, the Available Facility	
Interest Period:	[]	

- 3. We confirm that each condition specified in Clause 4.2 (*Further conditions precedent*) of the Agreement is satisfied on the date of this Utilisation Request.
- 4. [This Loan is to be made in [whole]/[part] for the purpose of refinancing [*identify maturing Loan*]/[The proceeds of this Loan should be credited to [*account*]]

5. This Utilisation Request is irrevocable.

Yours faithfully

authorised signatory for WM Cayman Holdings Limited II

SCHEDULE 4 FORM OF TRANSFER CERTIFICATE

To: [] as Agent

From: [*The Existing Lender*] (the "Existing Lender") and [*The New Lender*] (the "New Lender")

Dated:

WM Cayman Holdings Limited II – USD1,500,000,000 Facility Agreement dated [] (the "Agreement")

- 1. We refer to the Agreement. This is a Transfer Certificate. Terms defined in the Agreement have the same meaning in this Transfer Certificate unless given a different meaning in this Transfer Certificate.
- 2. We refer to Clause 23.6 (*Procedure for transfer*) of the Agreement:
 - (a) The Existing Lender and the New Lender agree to the Existing Lender transferring to the New Lender by novation, and in accordance with Clause 23.6 (*Procedure for transfer*) of the Agreement, all of the Existing Lender's rights and obligations under the Agreement and the other Finance Documents which relate to that portion of the Existing Lender's Commitment and participations in Loans under the Agreement as specified in the Schedule.
 - (b) The proposed Transfer Date is [].
 - (c) The Facility Office and address, fax number and attention details for notices of the New Lender for the purposes of Clause 30.2 (*Addresses*) of the Agreement are set out in the Schedule.
- 3. The New Lender expressly acknowledges the limitations on the Existing Lender's obligations set out in paragraph (c) of Clause 23.5 (*Limitation of responsibility of Existing Lenders*) of the Agreement.
- 4. This Transfer Certificate may be executed in any number of counterparts and this has the same effect as if the signatures on the counterparts were on a single copy of this Transfer Certificate.
- 5. This Transfer Certificate and any non-contractual obligations arising out of or in connection with it are governed by English law.
- 6. This Transfer Certificate has been entered into on the date stated at the beginning of this Transfer Certificate.

THE SCHEDULE

Commitment/rights and obligations to be transferred

[Insert relevant details] [Facility Office address, fax number and attention details for notices and account details for payments,]

[Existing Lender] By: [New Lender] By:

This Transfer Certificate is accepted by the Agent and the Transfer Date is confirmed as [].

[Agent]

By:

SCHEDULE 5 FORM OF ASSIGNMENT AGREEMENT

To: [] as Agent and WM Cayman Holdings Limited II as Borrower, for and on behalf of each Obligor

From: [the Existing Lender] (the "Existing Lender") and [the New Lender] (the "New Lender")

Dated:

WM Cayman Holdings Limited II – USD1,500,000,000 Facility Agreement dated [] (the "Agreement")

- 1. We refer to the Agreement. This is an Assignment Agreement. Terms defined in the Agreement have the same meaning in this Assignment Agreement unless given a different meaning in this Assignment Agreement.
- 2. We refer to Clause 23.7 (*Procedure for assignment*) of the Agreement:
 - (a) The Existing Lender assigns absolutely to the New Lender all the rights of the Existing Lender under the Agreement and the other Finance Documents which relate to that portion of the Existing Lender's Commitment and participations in Loans under the Agreement as specified in the Schedule.
 - (b) The Existing Lender is released from all the obligations of the Existing Lender which correspond to that portion of the Existing Lender's Commitment and participations in Loans under the Agreement specified in the Schedule.
 - (c) The New Lender becomes a Party as a Lender and is bound by obligations equivalent to those from which the Existing Lender is released under paragraph (b) above.
- 3. The proposed Transfer Date is [].
- 4. On the Transfer Date the New Lender becomes Party to the Finance Documents as a Lender.
- 5. The Facility Office and address, fax number and attention details for notices of the New Lender for the purposes of Clause 30.2 (*Addresses*) of the Agreement are set out in the Schedule.
- 6. The New Lender expressly acknowledges the limitations on the Existing Lender's obligations set out in paragraph (c) of Clause 23.5 (*Limitation of responsibility of Existing Lenders*) of the Agreement.
- 7. This Assignment Agreement acts as notice to the Agent (on behalf of each Finance Party) and, upon delivery in accordance with Clause 23.8 (*Copy of Transfer Certificate, Assignment Agreement or Increase Confirmation to Borrower*) of the

Agreement, to the Borrower (on behalf of each Obligor) of the assignment referred to in this Assignment Agreement.

- 8. This Assignment Agreement may be executed in any number of counterparts and this has the same effect as if the signatures on the counterparts were on a single copy of this Assignment Agreement.
- 9. This Assignment Agreement and any non-contractual obligations arising out of or in connection with it are governed by English law.
- 10. This Assignment Agreement has been entered into on the date stated at the beginning of this Assignment Agreement.

THE SCHEDULE

Rights to be assigned and obligations to be released and undertaken

[Insert relevant details] [Facility Office address, fax number and attention details for notices and account details for payments]

[Existing Lender] By: [New Lender] By:

This Assignment Agreement is accepted by the Agent and the Transfer Date is confirmed as [].

Signature of this Assignment Agreement by the Agent constitutes confirmation by the Agent of receipt of notice of the assignment referred to herein, which notice the Agent receives on behalf of each Finance Party.

[Agent]

By:

SCHEDULE 6 FORM OF COMPLIANCE CERTIFICATE

To: [] as Agent

From: WM Cayman Holdings Limited II as Borrower

Dated:

Dear Sirs

WM Cayman Holdings Limited II – USD1,500,000,000 Facility Agreement dated [] (the "Agreement")

- 1. We refer to the Agreement. This is a Compliance Certificate. Terms defined in the Agreement have the same meaning when used in this Compliance Certificate unless given a different meaning in this Compliance Certificate.
- 2. We confirm that: [Insert details of covenants to be certified]
- 3. [We confirm that no Default is continuing.]¹

Signed:

Responsible Officer of WM Cayman Holdings Limited II

¹ If this statement cannot be made, the Compliance Certificate should identify any Default that is continuing and the steps, if any, being taken to remedy it.

SCHEDULE 7 TIMETABLES

Loans in US dollars

Loans in HK dollars

Delivery of a duly completed Utilisation Request	t U – 5, 5:00pm (or, in the case of the U – 5; 5:00pm (or, in the case of the U – 5; 5:00pm (or, in the case of the U – 5; 5:00pm (or, in the case of the U – 5; 5:00pm (or, in the case of the U – 5; 5:00pm (or, in the case of the U – 5; 5:00pm (or, in the case of the U – 5; 5:00pm (or, in the case of the U – 5; 5:00pm (or, in the case of the U – 5; 5:00pm (or, in the case of the U – 5; 5:00pm (or, in the case of the U – 5; 5:00pm (or, in the case of the U – 5; 5:00pm (or, in the case of the U – 5; 5:00pm (or, in the case of the U – 5; 5:00pm (or, in the case of the U – 5; 5:00pm (or, in the cas
(Clause 5.1 (Delivery of a Utilisation Request))	first Utilisation Date, any later date the first Utilisation Date, any
	and time agreed between the later date and time agree
	Borrower and the Agent) between the Borrower and th
	Agent)
Agent notifies the Lenders of the Loan in accordance	U = 4, 5:00pm (or, in the case of the U = 4, 5:00pm (or, in the case of
with Clause 5.4 (Lender's participation)	first Utilisation Date, any later date the first Utilisation Date, any
	and time agreed between the later date and time agree
	Borrower and the Agent) between the Borrower and th
	Agent)
LIBOR or HIBOR is fixed	Quotation Date, 11:00 a.m. Quotation Date, 11:00 a.m.
	(London time) (Hong Kong time)

"U - X" refers to the number of Business Days before the relevant Utilisation Date/the first day of the relevant Interest Period.

SCHEDULE 8 FORM OF INCREASE CONFIRMATION

To: [] as Agent and WM Cayman Holdings Limited II as Borrower, for and on behalf of each Obligor

From: [the Increase Lender] (the "Increase Lender")

Dated:

WM Cayman Holdings Limited II – USD1,500,000,000 Facility Agreement dated [] (the "Agreement")

- 1. We refer to the Agreement. This is an Increase Confirmation. Terms defined in the Agreement have the same meaning in this Increase Confirmation unless given a different meaning in this Increase Confirmation.
- 2. We refer to Clause 2.2 (Increase) of the Agreement.
- 3. The Increase Lender agrees to assume and will assume all of the obligations corresponding to the Commitment specified in the Schedule (the "**Relevant Commitment**") as if it had been an Original Lender under the Agreement in respect of the Relevant Commitment.
- 4. The proposed date on which the increase in relation to the Increase Lender and the Relevant Commitment is to take effect (the "**Increase Date**") is [].
- 5. On the Increase Date, the Increase Lender becomes party to the Finance Documents as a Lender.
- 6. The Facility Office and address, fax number and attention details for notices to the Increase Lender for the purposes of Clause 30.2 (*Addresses*) of the Agreement are set out in the Schedule.
- 7. The Increase Lender expressly acknowledges the limitations on the Lenders' obligations referred to in paragraph (i) of Clause 2.2 (*Increase*) of the Agreement.
- 8. This Increase Confirmation may be executed in any number of counterparts and this has the same effect as if the signatures on the counterparts were on a single copy of this Increase Confirmation.
- 9. This Increase Confirmation and any non-contractual obligations arising out of or in connection with it are governed by English law.
- 10. This Increase Confirmation has been entered into on the date stated at the beginning of this Increase Confirmation.

THE SCHEDULE

Relevant Commitment/rights and obligations to be assumed by the Increase Lender

[Insert relevant details]

[Facility Office address, fax number and attention details for notices and account details for payments]

[Increase Lender]

By:

This Increase Confirmation is accepted by the Agent and the Increase Date is confirmed as [].

Agent

By:

SIGNATURE PAGES

The Company

WYNN MACAU, LIMITED

By: <u>/s/ Matthew Maddox</u> Name: Matthew Maddox Title: Chief Executive Officer and Executive Director

Address: Wynn Macau, Rua Cidade de Sintra, NAPE Macau Tel: +853 2888 9966 Fax: +853 2832 9966

Attention: Chief Financial Officer

With a copy to:

Wynn Resorts (Macau) S.A.

Address: Wynn Macau, Rua Cidade de Sintra, NAPE Macau Tel: +853 2888 9966 Fax: +853 2832 9966

Attention: Legal Department of Wynn Resorts (Macau) S.A.

The Borrower

WM CAYMAN HOLDINGS LIMITED II

By: <u>/s/ Matthew Maddox</u> Name: Matthew Maddox Title: Director

Address: Wynn Macau, Rua Cidade de Sintra, NAPE Macau Tel: +853 2888 9966 Fax: +853 2832 9966

Attention: Chief Financial Officer

With a copy to:

Wynn Resorts (Macau) S.A.

Address: Wynn Macau, Rua Cidade de Sintra, NAPE Macau Tel: +853 2888 9966 Fax: +853 2832 9966

Attention: Legal Department of Wynn Resorts (Macau) S.A.

The Arrangers

BANK OF CHINA LIMITED, MACAU BRANCH

By: <u>/s/ Lei Kam Ho</u> Name: Lei Kam Ho Title: Deputy General Manager of Credit Administration Department

Address: 20/F Bank of China Building

Avenida Doutor Mario Soares, Macau

- Tel: +853 8792 1639 / 8792 1661
- Fax: +853 8792 1659 / 8792 0308
- Attention: Mr. James Wong / Ms.Jade Gan

With a copy to:

Address: 17/F Bank of China Building

Avenida Doutor Mario Soares, Macau

- Tel: +853 8792 1705 / 8792 1373
- Fax: +853 8792 1677
- Attention: Ms. Venus Huang / Ms. Grace Su

The Original Lenders

By:

BANK OF CHINA LIMITED, MACAU BRANCH

<u>/s/ Lei Kam Ho</u> Name: Lei Kam Ho Title: Deputy General Manager of Credit Administration Department

Address: 20/F Bank of China Building

Avenida Doutor Mario Soares, Macau

- Tel: +853 8792 1639 / 8792 1661
- Fax: +853 8792 1659 / 8792 0308
- Attention: Mr. James Wong / Ms.Jade Gan

With a copy to:

Address: 17/F Bank of China Building

Avenida Doutor Mario Soares, Macau

- Tel: +853 8792 1705 / 8792 1373
- Fax: +853 8792 1677
- Attention: Ms. Venus Huang / Ms. Grace Su

INDUSTRIAL AND COMMERCIAL BANK OF CHINA (MACAU) LIMITED

By: <u>/s/ Chan Kam Lun</u> Name: Chan Kam Lun Title: Deputy CEO

By: <u>/s/ Zheng Zhiguo</u> Name: Zheng Zhiguo Title: Chief Officer

Address: 18/F.,ICBC Tower,Macau Landmark,555 Avenida da Amizade,Macau

- Tel: +853 8398 2723/ +853 8398 2296/ +853 8398 2158/ +853 8398 2452/ +853 8398 2224/ +853 8398 2499/ +853 8398 2491/ +853 8398 7447
- Fax: +853 8398 2160/ +853 2858 4496

Attention: Kevin Kuok / Nick Wu / Janet Lei / Linda Chan / Lillian Hong / Selena Ren / Mandy Lai / Vanessa Chao

UNITED OVERSEAS BANK LIMITED

(incorporated in Singapore with limited liability)

By: <u>/s/ Eric Poon</u> Name: Eric Poon Title: Executive Director, Head, Corporate Banking, China Corporates Wholesale Banking

Address: 28/F, Champion Tower, 3 Garden Road, Central, Hong Kong

- Tel: +852 2820 9195 / +852 2820 9190 / +852 2820 9161 / +852 2820 9178 / +852 2820 9176 / +852 2820 9141 / +852 2820 9135 / +852 2820 9180 / +852 2820 9177 / +852 2820 9114 / +852 2820 9171 / +852 2820 9157
- Fax: +852 2596 0113 / +852 2820 9117 / +852 2501 5738
- Attention: Anthony Tse / Christina Lui / Ann Wong / Kelvin Wong (Corporate Banking HK) CC: Karie Leung / Amy Wong / Cyrus Chau (Credit Administration & Control) Dennis Yue / Percy Ning / Terry Wong / Tommy Fung / Wenda Ng (Wholesale Loan Operations)

BANK OF COMMUNICATIONS CO., LTD. MACAU BRANCH

By: <u>/s/ Huang Bin</u> Name: Huang Bin Title: President

Address: 16/F, AIA Tower, 251A-301 Avenida Comercial De Macau

- Tel: (853) 28286611
- Fax: (853) 28286686
- Attention: Wynn Macau Syndication 2021

BANK OF AMERICA, NATIONAL ASSOCIATION, HONG KONG BRANCH

By: <u>/s/ Joyce Chan</u> Name: Joyce Chan Title: MD, Head of Greater China, CBK-S

Address: 52/F, Cheung Kong Center, 2 Queen's Road Central, Hong Kong

- Tel: +852-3508-4071 / +852-3508-6442
- Fax: +852-3009-0073
- Attention: Jeffrey Tam / Alan Wong

THE BANK OF NOVA SCOTIA

(A chartered bank incorporated under the laws of Canada, whose shareholders' liability is limited, acting out of its Hong Kong Branch with registered address at Suite 2401, Central Tower, 28 Queen's Road Central, Hong Kong)

By: <u>/s/ Philip Ng</u>

Name: Philip Ng Title: Director, Corporate Banking

Address: Suite 2401, Central Tower, 28 Queen's Road Central, Hong Kong

Tel: 2861 4100

Fax: 2527 2526

Attention: Winton So / Philip Ng / Kenneth Ho

SUMITOMO MITSUI BANKING CORPORATION

By: <u>/s/ Jane Pedreira</u> Name: Jane Pedreira Title: Director

For Credit Matters:

Address: 277 Park Avenue, New York, NY 10172

Tel: 212-224-4735

Fax:

Attention: John Corrigan

For Operational Matters:

Address: 277 Park Avenue, New York, NY 10172

Tel: 212-256-7371

Fax: 212-224-4391

Attention: Keshia Octelene

BANCO NACIONAL ULTRAMARINO, S.A.

- By: <u>/s/ Cheong Hong Chon (Teren Cheong)</u> Name: Cheong Hong Chon (Teren Cheong) Title: General Manager
- By: <u>/s/ Chiu Po Lin Violet (Violet Choi)</u> Name: Chiu Po Lin Violet (Violet Choi) Title: General Manager
- Address: No. 22, Avenida de Almeida Ribeiro, Macau
- Tel: (853) 83989146 / 83989106
- Fax: (853) 28331206 / 28356867
- Attention: Teren Cheong / Violet Choi

DEUTSCHE BANK AG, SINGAPORE BRANCH

By: <u>/s/ Sreenivasan Iyer</u> Name: Sreenivasan Iyer Title: Managing Director, Investment Banking Coverage

By: <u>/s/ Choo Ping Quek</u> Name: Choo Ping Quek Title: Vice President, Strategic Corporate Lending

Address: One Raffles Quay, South Tower #14-00, Singapore 048583

Tel: +65 6423 5990 / +65 6423 6802

Fax: +65 6221 2306 / +65 6536 1328

Attention: Xuan-Ren Chen / Adeline Chua

BNP PARIBAS

(a public limited company (société anonyme) incorporated in the Republic of France with the liability of its members being limited and having a branch at 63/F Two International Finance Centre, 8 Finance Street, Hong Kong)

By: <u>/s/ Mary Hse</u> Name: Mary Hse Title: Co-Head of Large Corporates, Hong Kong, Managing Director, Global Banking APAC By: <u>/s/ Charmaine Lo</u> Name: Charmaine Lo Title: Director, Loans & Specialized Finance, Capital Markets Group

Address: 63/F, Two International Finance Centre, 8 Finance Street, Central, Hong Kong

Tel: +852 2909 8770/+852 2909 8790/+852 2909 8510

Fax: +852 2970 0296

Attention: Charmaine Lo/ Chris Lau/ Kathleen Cheung

BANCO COMERCIAL DE MACAU, S.A.

By: <u>/s/ Leung Wai Yip Vincent</u> Name: Leung Wai Yip Vincent Title: Chief Risk Officer

By: <u>/s/ Chong Sou Keong</u> Name: Chong Sou Keong Title: Head of Commercial Banking

Address: Avenida da Praia Grande, No.572, Macau

Tel: (853) 87910-233 / 251 / 255 / 259 / 861 / 273 / 280 / 293 / 296

Fax: (853) 28715-521 / 87910-276

Attention: Mr. Sam Lee / Mr. Simon Chong / Mr. Alex Chou / Ms. Victoria Kuang / Mr. Vincent Leung / Mr. Winson Wong / Ms. Wing Ku / Ms. Natalie Cheng / Ms. Camilla Chow

The Agent

BANK OF CHINA LIMITED, MACAU BRANCH

By: <u>/s/ Lei Kam Ho</u> Name: Lei Kam Ho Title: Deputy General Manager of Credit Administration Department

Address: 20/F Bank of China Building

Avenida Doutor Mario Soares, Macau

Tel: +853 8792 1639 / 8792 1661

Fax: +853 8792 1659 / 8792 0308

Attention: Mr. James Wong / Ms.Jade Gan

With a copy to:

Address: 17/F Bank of China Building

Avenida Doutor Mario Soares, Macau

Tel: +853 8792 1705 / 8792 1373

Fax: +853 8792 1677

Attention: Ms. Venus Huang / Ms. Grace Su

TRANSITION AGREEMENT

This **Transition Agreement** (this "**Agreement**") is entered into as of November 9, 2021 (the "**Effective Date**"), by and between Wynn Resorts, Limited ("**Employer**") and Matt Maddox ("**Employee**"). Capitalized terms that are not defined herein will have the meanings ascribed to them in the Agreement (as defined below).

RECITALS

WHEREAS Employer is a limited liability company duly organized and existing under the laws of the State of Nevada, and which maintains its principal place of business at 3131 Las Vegas Boulevard South, Las Vegas, Nevada 89109;

WHEREAS, Employee is an adult individual residing at [Address Redacted]:

WHEREAS, Employer and Employee have entered into that certain Amended and Restated Employment Agreement, effective as of December 16, 2019 (the "Employment Agreement");

WHEREAS, effective January 31, 2022, Employee will cease serving as the Chief Executive Officer of Employer and as a member of the Board of Directors (the "**Board**") of Employer;

WHEREAS, Employer and Employee have agreed to terminate the Employment Agreement;

WHEREAS, Employer confirms that all acts or omissions by Employee prior to the Effective Date in connection with his employment were in good faith and in the best interests of the Employer;

WHEREAS, Employee and Employee confirm that Employee and Employee have mutually agreed to terminate Employment Agreement;

NOW, THEREFORE, in consideration of the foregoing and the respective representations, warranties, covenants and agreements set forth in this **Agreement**, the parties hereto agree as follows:

1. <u>Base Agreement</u>.

(a) Effective as of the close of business on January 31, 2022 (the "**Separation Date**"), Employee's service with Employer and its affiliates in all capacities will terminate and Employee will cease to serve as Chief Executive Officer of Employer and as a member of the Board. Effective as of the Separation Date, Employee will commence service as Senior Advisor to Employer to serve at the direction of the Board to assist in facilitating the successful transition of Employer's new Chief Executive Officer. Employee shall serve as Senior Advisor without additional compensation, except as provided in paragraph 1(d) herein, to Employer through the one-year anniversary of the Separation Date, unless Employer no longer requires such services and terminates the Senior Advisor services prior to such date. As Senior Advisor, Employee shall be reasonably available to the Chief Executive Officer on a non-exclusive basis not to exceed twenty (20) hours per calendar month and may perform the duties set forth in this Section 1 of this Agreement remotely, from a location of Employee agrees to respond to email and other inquiries within a reasonable timeframe. Employee shall remain on the Board of Wynn Interactive, Ltd. and Wynn Macau, Limited as a member through December 31, 2022, but effective as of the Separation Date, shall otherwise be deemed to have resigned from all other positions (including board of director memberships) that Employee may have held immediately prior to the Separation Date.

(b) Employee agrees to reasonably cooperate with Employer, Employer's counsel, and any federal, state, or local governmental agency or regulatory body regarding any outstanding matters that involved Employee during the time and scope of his employment with Employer. Employer will reimburse Employee for all reasonable fees and expenses incurred through Employee's participation in such cooperation; provided that Employee must obtain Employer's approval before incurring fees or expenses in excess of \$10,000.

(c) The Indemnification Agreement dated October 21, 2002, by and between Employee, on the one side, and Employer or any of Employer's Affiliates, on the other side, shall remain in effect.

(d) In exchange for Employee's consulting advice and guidance in connection with Employee's transition and Employee's cooperation under Sections 1(a) and (b) above and contingent upon Employee's execution and non-revocation of this Agreement and the Subsequent Release (as defined below), Employee and Employer agree that:

- i. Employer will pay Employee a single, lump-sum cash payment in the amount equal to (A) \$3,000,000 (representing 18 months of Employee's current Base Salary of \$2,000,000), plus (B) 11/12th of the greater of (x) the bonus payable to Employee pursuant to <u>Section 1(d)(ii)</u> below and (y) the 2020 annual bonus of \$2,200,000 paid to Employee, less all applicable taxes and withholding, with such payment to be made within 15 days of the date the Subsequent Release becomes binding and irrevocable;
- ii. Employer will pay Employee the annual bonus earned for 2021 based on actual performance against the applicable performance metrics and not to be less than the percentage of target received by any other executive of Employer within the Executive Bonus program;
- iii. Employee will remain eligible to participate in Employer's senior executive health program through the first anniversary of the Separation Date on the same terms and conditions as an active employee. For the avoidance of doubt, such health benefits coverage will be paid for by Employer to the same extent as if Employee were still employed by Employer as a senior executive, and Employee will be required to make such payments as Employee would be required to make if Employee were still employed by Employer as a senior executive; and
- iv. On the date the Subsequent Release becomes binding and irrevocable following the Separation Date, Executive will vest in 82,238 shares of restricted Employer Common Stock (the "**Restricted Stock**").

(e) Employee understands and agrees that no other wages, vacation, sick pay, benefits, stock, or other compensation is due to Employee other than for accrued base salary and unused vacation pay or as set forth in this Agreement. The parties acknowledge and agree that all options granted under the Non-Qualified Stock Option Agreement dated May 6, 2009 have vested and those options will remain subject to the terms and conditions of such agreement.

- 2. <u>Continuing Obligations</u>. Employee acknowledges and agrees that the following provisions of the Employment Agreement remain in full force and effect:
 - (a) Section 9, relating to confidentiality.

(b) Section 10 of the Employment Agreement relating to non-competition and non-solicitation will remain in force and effect in accordance with its terms until December 31, 2022.

(c) Section 11 relating to remedies for violations of restrictive covenants applicable to Employee.

(d) Section 21 of the Employment Agreement relating to arbitration will remain in full force and effect and this Agreement will be covered by such arbitration provision.

3. <u>Waiver and Releases</u>. As a condition to the payments and benefits set forth in Section 1, Employee must execute and not revoke (i) this Agreement in the manner set forth in Section 4 below and (ii) a separate waiver and release agreement described in this Section 3 (the "Subsequent Release"). The Subsequent Release will release claims from the date of this Agreement through the date of its execution and will otherwise be identical in all material respects to Sections 3, 4, 5, and 6 hereof (with such changes Employer determines in good faith are reasonably necessary to reflect the intent of this provision) and will not contain other restrictions or limitations on Employee that are not otherwise set forth herein. Employer will deliver an executed version of the Subsequent Release to Employee promptly after the Separation Date and Employee will be required to execute and not revoke such Subsequent Release during the time frames contemplated by this Section 3 below, as measured from the date such Subsequent Release is delivered to the Employee.

(a) Except as provided in this Agreement, Employee, for Employee, Employee's spouse, children, heirs, executors, administrators, successors and assigns (hereinafter "Releasors"), to the extent permitted by applicable law, hereby fully and forever releases, acquits, discharges and promises not to sue Employer and its past, present and future parent and/or subsidiary entities, divisions, affiliates and any of their past, present or future partners, owners, joint venturers, stockholders, predecessors, successors, officers, directors, administrators, employees, agents, representatives, attorneys, heirs, executors, assigns, insurers, benefit and retirement plans and/or their trustees and any other person, firm or corporation with whom any of them is now or may hereafter be affiliated, and any of their successors and assigns (the persons and/or entities released hereunder, the "Releasees"), over any and all claims, counterclaims, agreements, debts, promises, grievances, complaints, demands, obligations, losses, causes of action, suits, controversies, actions, cross-claims, counter-claims, costs, expenses, attorney's fees, demands, compensatory damages, liquidated damages, punitive or exemplary damages, any other damages, liabilities and indemnities of any nature whatsoever, at law or in equity, whether negligent or intentional, whether now known or unknown, discovered now or in the future, (a) from the beginning of time through the date upon which Employee signs this Agreement; (b) arising from or in any way related to Employee's employment or termination of employment with any of the Releasees; (c) arising from or in any way related to any agreement with any of the Releasees, including but not limited to the Employment Agreement; and/or (d) arising from or in any way related to awards, policies, plans, programs or practices of any of the Releasees that may apply to Employee or in which Employee may participate, in each case, including, but not limited to, under any federal, state or local law, act, statute, code, order, judgment, injunction, ruling, decree, writ, ordinance or regulation, including, but not limited to, any claims based on race, age, disability, national origin, religion, gender, sexual orientation, marital status, veteran status, protected activity, compensation and benefits from employment, including stock, stock options, stock option agreements and retirement plans, whether based on contract, tort, statute or other legal or equitable theory of recovery, whether mature or to mature in the future.

(b) Without limiting the foregoing, this Agreement applies to any and all matters that have been or which could have been asserted against any Release in a lawsuit or in any state or federal judicial or administrative forum, up to the Effective Date under the Nevada Fair Employment Practices Act, the Equal Pay Act, the Family and Medical Leave Act of 1993, as amended, the Genetic Information Nondiscrimination Act of 2008, the National Labor Relations Act, as amended, Title VII of the Civil Rights Act of 1964, as amended, the Post-Civil War Reconstruction Acts, as amended (42 U.S.C. §§ 1981-1988), the Age Discrimination in Employment Act of 1967, as amended, the Americans with Disabilities Act of 1990, as amended, the Rehabilitation Act of 1973, as amended, the Employee Retirement Income Security Act of 1974, as amended, the Civil Rights Act of 1991, the Pregnancy Discrimination Act, any other federal statute, any state civil rights act, any state statutory wage claim such as those contained in Chapter 608 of the Nevada Revised Statutes, any other statutory claim, any claim of wrongful discharge, any claim in tort or contract (including but not limited to the Employment Agreement), any claim seeking declaratory, injunctive, or equitable relief, or any other claim of any type whatsoever arising out of the common law of any state. Notwithstanding the above, this Agreement does not apply to any rights, obligations or claims governed by Chapter 612 of the Nevada Revised Statutes pertaining to unemployment compensation.

(c) This Agreement is not intended to bar or affect any claims that may not be waived by private agreement under applicable law, or any claims that may be brought under the Indemnification Agreement. This Agreement also does not limit any party's right to file an administrative charge or participate in an investigative proceeding of any federal, state, or local government agency tasked with enforcing employment-related laws, such as the U.S. Equal Employment Opportunity Commission, the Department of Labor, or the National Labor Relations Board, or their state level equivalents, but does operate as a waiver of any personal recovery if related to the claims released herein. For example, and for the avoidance of doubt, this Agreement will not bar any claims for indemnification or advancement for any claims arising out of or in any way relating to the fact that Employer and Employee have entered into this Agreement, including those that may be brought by any shareholder directly or in the name of Employer.

(d) Employer and its past, present and future parent and/or subsidiary entities, divisions, affiliates and any of their past, present or future partners, owners, joint venturers, stockholders, predecessors, successors, officers, directors, administrators, employees, agents, representatives, attorneys, heirs, executors, assigns, insurers, benefit and retirement plans and/or their trustees and any other person, firm or corporation with whom any of them is now or may hereafter be affiliated, and any of their successors and assigns (hereinafter "**Employer Releasors**"), to the extent permitted by applicable law, hereby fully and forever releases, acquits, discharges and promises not to sue Employee, Employee's spouse, children, heirs, executors, affiliated entities, administrators, successors and assigns (hereinafter "**Employee Releasees**"), over any and all claims, counterclaims, agreements, debts, promises, grievances, complaints, demands, obligations, losses, causes of action, suits, controversies, actions, cross-claims, counter-claims, costs, expenses, attorney's fees, demands, compensatory damages, liquidated damages, punitive or exemplary damages, any other damages, liabilities and indemnities of any nature whatsoever, at law or in equity, whether negligent or intentional, whether now known or unknown, discovered now or in the future, (i) from the beginning of time through the Effective Date; (ii) arising from or in any way related to Employee's employment or termination of employment with any of the Employee Releasors (or any acts or omissions of Employee during the course and scope of Employee's employment with any of the Employer Releasors); and/or (iii) arising from or in any way related to any agreement with

any of the Employer Releasors, including but not limited to the Employment Agreement. Such release does not apply to any claims related to Employee's (a) criminal activity, or (b) intentional and material breach of his fiduciary duties to Employer that causes material financial harm to Employer. The foregoing release does not waive or release Employee's right to receive the consideration provided for by this Agreement.

(e) EMPLOYEE HEREBY ACKNOWLEDGES THAT BY EXECUTING THIS AGREEMENT EMPLOYEE IS AGREEING TO WAIVE ANY AND ALL RIGHTS OR CLAIMS UNDER THE AGE DISCRIMINATION IN EMPLOYMENT ACT OF 1967 (29 U.S.C. § 626 et. seq.). EMPLOYEE IS ADVISED TO CONSULT WITH AN ATTORNEY PRIOR TO EXECUTING THIS AGREEMENT. IN ADDITION, EMPLOYEE ACKNOWLEDGES THAT UPON RECEIPT OF THIS AGREEMENT, EMPLOYEE HAS A PERIOD OF TWENTY-ONE (21) DAYS WITHIN WHICH TO CONSIDER THIS AGREEMENT BEFORE SIGNING IT.

(f) EMPLOYEE FURTHER UNDERSTANDS THAT FOR A PERIOD OF SEVEN (7) DAYS FOLLOWING EMPLOYEE'S EXECUTION OF THIS AGREEMENT, EMPLOYEE MAY REVOKE EMPLOYEE'S WAIVER OF ANY POTENTIAL AGE DISCRIMINATION CLAIM AND THIS AGREEMENT SHALL NOT BECOME EFFECTIVE OR ENFORCEABLE AS TO ANY SUCH WAIVER OF AN AGE DISCRIMINATION CLAIM UNTIL THE REVOCATION PERIOD HAS EXPIRED. HOWEVER, ALL OTHER ASPECTS OF THIS AGREEMENT, EXCEPT FOR EMPLOYEE WAIVER OF ANY POTENTIAL AGE DISCRIMINATION CLAIM, BECOMES EFFECTIVE AT THE TIME EMPLOYEE EXECUTES THIS AGREEMENT.

(g) The parties agree that the twenty-one (21) day consideration period shall start on the date upon which this Agreement is presented to Employee or Employee's counsel and shall expire at midnight twenty-one (21) calendar days later. The parties further agree that the seven (7) day revocation period shall start on the date upon which the Employee executes this Agreement and shall expire at midnight seven (7) calendar days later. If Employee elects to sign this Agreement prior to the end of the twenty-one (21) day consideration period, the mandatory seven (7) day revocation period will commence immediately the day after the date of execution.

(h) The parties hereby agree that any modifications to the proposed Agreement originally forwarded to Employee or Employee's counsel, whether considered or deemed to be material or nonmaterial, shall not restart the twenty-one (21) day consideration period.

(i) Employee may sign this Agreement prior to the end of the twenty-one (21) day consideration period, thereby commencing the mandatory seven (7) day revocation period. If the Employee does sign this Agreement before the end of the twenty-one (21) day consideration period, Employee affirms that the waiver of the twenty-one (21) day consideration period is hereby effective on the date of Employee's execution.

4. <u>Notices</u>. Any and all notices required by this Agreement will be either hand-delivered or mailed, via certified mail, return receipt requested, addressed to:

<u>TO EMPLOYER</u>: Wynn Resorts, Limited 3131 Las Vegas Boulevard South Las Vegas, Nevada 89109 Attn: Legal Department TO EMPLOYEE: Matt Maddox [Address Redacted]

All notices hand-delivered will be deemed delivered as of the date actually delivered to the addressee. All notices mailed will be deemed delivered as of three (3) business days after the date postmarked. Any changes in any of the addresses listed herein will be made by notice as provided in this Section 4.

- 5. <u>Governing Law</u>; <u>Arbitration</u>; <u>Jury Waiver</u>. This Agreement will be governed by and construed in accordance with the laws of the State of Nevada, without regard to conflict of laws principles. Any and all claims or disputes arising out of or in connection with this Agreement will be submitted to binding arbitration in accordance Section 21 of the Employment Agreement, except for those provisions expressly authorizing injunctive relief. EMPLOYEE EXPRESSLY WAIVES THE RIGHT TO TRIAL BY JURY IN ANY ACTION OR PROCEEDING RELATING TO OR ARISING IN ANY WAY FROM THIS AGREEMENT OR THE MATTERS CONTEMPLATED.
- 6. Employer and Employee shall jointly issue a public announcement of Employee's separation from Employer in the Form of <u>Exhibit A</u> to this Agreement.
- 7. Upon request of Employee, Employer shall provide to potential future employers of Employee a positive recommendation of Employee that will be consistent with the language set forth in <u>Exhibit A</u> attached hereto.

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed and effective as of the date first written above.

WYNN RESORTS, LIMITED

<u>/s/ Ellen F. Whittemore</u> Ellen F. Whittemore Executive Vice President, General Counsel & Secretary <u>/s/ Matt Maddox</u> Matt Maddox

Employee

Statement from Wynn Resorts Regarding Chief Executive Officer Transition

LAS VEGAS (November 9, 2021) — Matt Maddox, CEO of Wynn Resorts, announced today that he will be leaving the Company January 31, 2022. Following a unanimous request by the Board of Directors, Mr. Maddox agreed to remain on the Wynn Macau Limited and Wynn Interactive Limited Boards through the end of 2022. Craig Billings, CEO of Wynn Interactive and who has served as President and CFO of Wynn Resorts, was unanimously selected by the Board to become CEO of Wynn Resorts and a member of the Board effective February 1, 2022.

Philip G. Satre, Chairman of the Board for Wynn Resorts said, "Matt has done an incredible job as CEO since his appointment four years ago. He very successfully led the Company through what could have been one of the most disruptive business transitions in decades. He ensured a swift refreshment and remaking of the Board of Directors and senior management, resulting in increased diversity and Wynn being recognized as having one of the most diverse Boards of Directors in the country.

"Matt's strategy to confront the pandemic and its economic impact on the Company proved to be the correct one: make decisions focused on the welfare of employees, guests and communities, and invariably the business and its shareholders will reap the rewards. He was a leader in the recovery of the Nevada hospitality industry, and his leadership in this area was recognized nationally as well.

"Our culture and operations have never been stronger. Matt led the successful licensure and opening of Encore Boston Harbor, which is now a highperforming part of the Wynn portfolio and a major source of economic growth for the greater Boston area. In addition to our assets in North America setting new records, our position in Macau as a premium operator is unrivaled.

Mr. Satre continued, "The entire board of directors is grateful for Matt's dedication and leadership of the company and thanks him for all that he has done. We are very fortunate to have a talented executive in Craig Billings who will assume the role of CEO and join the Board in February 2022. The Board and I have worked with Craig for five years, as CFO and President of the company. He has our confidence and trust as he begins his new role."

Mr. Maddox has been with Wynn Resorts for two decades and has been its CEO since February 2018, as well as a member of the Board of Directors. He led Wynn Resorts through its most difficult period, following the resignation of the Company's founder in 2018 and then the global pandemic in 2020. During Mr. Maddox's tenure as CEO, he successfully rebuilt the company's culture and reestablished Wynn Resorts as one of the world's leading luxury brands. Wynn Resorts was recently recognized by Forbes Travel Guide as being the highest ranked luxury hotel brand in the world. The Las Vegas resort won the "Best Place to Work" accreditation.

Mr. Maddox commented, "This has not been an easy decision. I am leaving a company that I love and that's full of people I admire. But I believe now is the right time for me and for the business.

"The last four years have been challenging but extremely rewarding, and I am incredibly proud that we accomplished so much. We quickly resolved litigation that had distracted the company for seven years. We reorganized the company's leadership team and focused on our 28,000 employees, rebuilding and strengthening our powerful culture to focus on a common purpose: create and deliver the best guest experiences.

"We led the industry with our response to the pandemic; many other industries looked to us for advice, guidance, and leadership as a result. I knew our biggest asset is our people, which is why we paid them throughout the shut down and why our culture has never been stronger. That investment is now paying off as Wynn Las Vegas and Encore Boston Harbor are generating more profit and market share since we opened our resorts, all while maintaining five-star service levels. I also look forward to serving on the Wynn Macau Board throughout 2022 to assist with the concession renewal and help solidify our place as the premium operator in the region. "I am grateful to the Board for their faith in me and the support and insight they have offered me as CEO. I am very pleased with the Board's decision and know I am leaving Wynn Resorts in great hands with Craig, as well as the entire management team," Mr. Maddox said.

Mr. Billings joined Wynn Resorts as Chief Financial Officer in March 2017, also served as the President of the company, and most recently has served as the CEO of Wynn Interactive. Mr. Billings has been a board member of Wynn Macau Limited since August 2018. Mr. Billings has a history of leadership and innovation in the gaming industry, both domestically and internationally. He has held executive and board positions at Goldman Sachs, where he covered the industry globally, Aristocrat Leisure Limited, NYX Gaming Group, and International Game Technology.

Prior to joining Wynn Resorts, Mr. Billings honed a global perspective with senior executive positions in both Australia and the United Kingdom during his time as Chief Digital Officer and Managing Director of Strategy and Business Development at Aristocrat Leisure. He is also the Lead Independent Director and Audit Committee Chair for Applovin Corporation (NASDAQ: APP).

Mr. Billings commented, "Having worked so closely with Matt and the Board, I appreciate the confidence they have placed in me and look forward to building upon the legacy of excellence we've established."

EMPLOYMENT AGREEMENT

THIS EMPLOYMENT AGREEMENT ("**Agreement**") is made and entered into as of the 9th day of November 2021, by and between **WYNN RESORTS, LIMITED** ("**Employer**") and **CRAIG BILLINGS** ("**Employee**").

<u>WITNESSETH</u>:

WHEREAS Employer is a corporation duly organized and existing under the laws of the State of Nevada, maintains its principal place of business at 3131 Las Vegas Boulevard South, Las Vegas, Nevada 89109, and is engaged in the business of developing, owning and operating casino resorts;

WHEREAS, in furtherance of its business, Employer has need of qualified, experienced personnel;

WHEREAS, Employee is an adult individual residing at [Address Redacted];

WHEREAS, Employer and Employee have entered into that certain Employment Agreement, effective as of March 1, 2017, as amended on April 17, 2018, May 29, 2019, December 31, 2020, May 24, 2021 (the "**2017 Agreement**");

WHEREAS, the 2017 Agreement terminates by its terms as of March 1, 2023, and Employee and Employer desire to enter into this Agreement to ensure the continued employment of Employee by Employer;

WHEREAS, Employee has represented and warranted to Employer that Employee possesses sufficient qualifications and expertise to fulfill the terms of the employment stated in this Agreement; and

WHEREAS, Employer is willing to employ Employee, and Employee is desirous of accepting employment from Employer under the terms and pursuant to the conditions set forth herein.

NOW, THEREFORE, for and in consideration of the foregoing recitals, and in consideration of the mutual covenants, agreements, understandings, undertakings, representations, warranties and promises hereinafter set forth, and intending to be legally bound thereby, Employer and Employee do hereby covenant and agree as follows:

1. **DEFINITIONS.** As used in this Agreement, the words and terms hereinafter defined have the respective meanings ascribed to them, unless a different meaning clearly appears from the context:

(a) "<u>Affiliate</u>" means with respect to a specified Person, any other Person who or which is (i) directly or indirectly controlling, controlled by or under common control with the specified Person, or (ii) any member, director, officer, or manager of the specified Person. For purposes of this definition only, "control", "controlling" and "controlled" mean the right to exercise, directly or indirectly, more than fifty percent (50%) of the voting power of the stockholders, members, or owners and, with respect to any individual, partnership, trust or other entity or association, the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of the controlled entity. For purposes hereof, "Person" shall mean an individual,

partnership, corporation, limited liability company, business trust, joint stock company, trust, unincorporated association, joint venture, or other entity of whatever nature.

(b) "<u>Anniversary</u>" means each annual anniversary date of the Effective Date during the Term (as defined in Section 5 hereof).

(c) "Cause" means

i. Employee's inability or failure to secure and/or maintain any licenses or permits required by government agencies with jurisdiction over the business of Employer or its Affiliate;

ii. the willful destruction by Employee of the property of Employer or its Affiliate having a material value to Employer or such Affiliate;

iii. fraud, embezzlement or theft committed by Employee (excluding acts involving a de minimis dollar value and not related in any manner whatsoever to Employer or its Affiliate or their business);

iv. Employee's conviction of or entering a plea of guilty or nolo contendere to any crime constituting a felony;

v. Employee's material breach of this Agreement;

vi. Employee's neglect, refusal, or knowing failure to discharge Employee's duties (other than due to physical or mental illness) commensurate with Employee's title and function, or Employee's failure to comply with a lawful direction of Employer or its board of directors;

vii. a knowing material misrepresentation to Employer or its board of directors;

viii. a failure to follow a material policy or procedure of Employer or its Affiliate that causes material financial harm to Employer;

ix. A violation of the Employer's Preventing Harassment and Discrimination Policy that has been substantiated by an independent investigation by external counsel, which the Board of Directors, in good faith, determines requires termination as opposed to other disciplinary action; or

x. Employee's material breach of a statutory or common law duty of loyalty or fiduciary duty to Employer or its Affiliate, including Employer's conflict of interest policy;

provided, *however*, that Employee's Complete Disability due to illness or accident or any other mental or physical incapacity shall not constitute "Cause" as defined herein.

(d) "<u>Change of Control</u>" means the occurrence, after the Effective Date, of any of the following events:

(i) any "Person" or "Group" (as such terms are defined in Section 13(d) of the Securities Exchange Act of 1934 (the "Exchange Act") and the rules and regulations promulgated thereunder) is or becomes the "Beneficial Owner" (within the meaning of Rule 13d-3 promulgated under the Exchange Act), directly or indirectly, of securities of Wynn Resorts, Limited ("WRL"), or of any entity resulting from a merger or consolidation involving WRL, representing more than fifty percent (50%) of the combined voting power of the then outstanding securities of WRL or such entity;

(ii) the individuals who, as of the Effective Date, are members of WRL's Board of Directors (the "Existing Directors") cease, for any reason, to constitute more than fifty percent (50%) of the number of authorized directors of WRL as determined in the manner prescribed in WRL's Articles of Incorporation and Bylaws; provided, however, that if the election, or nomination for election, by WRL's stockholders of any new director was

approved by a vote of at least fifty percent (50%) of the Existing Directors, such new director shall be considered an Existing Director; provided further, however, that no individual shall be considered an Existing Director if such individual initially assumed office as a result of either an actual or threatened "Election Contest" (as described in Rule 14a-11 promulgated under the Exchange Act) or other actual or threatened solicitation of proxies by or on behalf of anyone other than the Board (a "Proxy Contest"), including by reason of any agreement intended to avoid or settle any Election Contest or Proxy Contest; or

(iii) the consummation of (x) a merger, consolidation or reorganization to which WRL is a party, whether or not WRL is the Person surviving or resulting therefrom, or (y) a sale, assignment, lease, conveyance or other disposition of all or substantially all of the assets of Employer or WRL, in one transaction or a series of related transactions, to any Person other than WRL or an Affiliate, where any such transaction or series of related transactions as is referred to in clause (x) or clause (y) above in this subparagraph (iii) (singly or collectively, a "Transaction") does not otherwise result in a "Change in Control" pursuant to subparagraph (i) of this definition of "Change in Control"; provided, however, that no such Transaction shall constitute a "Change in Control" under this subparagraph (iii) if the Persons who were the members or stockholders of Employer or WRL immediately before the consummation of such Transaction are the Beneficial Owners, immediately following the consummation of such Transaction, of fifty percent (50%) or more of the combined voting power of the then outstanding membership interests or voting securities of the Person surviving or resulting from any merger, consolidation or reorganization referred to in clause (x) above in this subparagraph (iii) or the Person to whom the assets of Employer or WRL are sold, assigned, leased, conveyed or disposed of in any transaction or series of related transactions referred in clause (y) above in this subparagraph (iii), in substantially the same proportions in which such Beneficial Owners held membership interests or voting stock in Employer or WRL immediately before such Transaction.

(e) "<u>Complete Disability</u>" means the total inability of Employee, due to illness or accident or other mental or physical incapacity, to perform Employee's obligations under this Agreement for a period as defined under Employer's local disability plan or plans.

(f) "Confidential Information" means any information that is possessed or developed by or for Employer or its Affiliate and which relates to the Employer's or Affiliate's existing or potential business or technology, which is not generally known to the public or to persons engaged in business similar to that conducted or contemplated by Employer or Affiliate, or which Employer or Affiliate seeks to protect from disclosure to its existing or potential competitors or others, and includes without limitation know how, business and technical plans, strategies, existing and proposed bids, costs, technical developments, purchasing history, existing and proposed research projects, copyrights, inventions, patents, intellectual property, data, process, process parameters, methods, practices, products, product design information, research and development data, financial records, operational manuals, pricing and price lists, computer programs and information stored or developed for use in or with computers, customer information, customer lists, supplier lists, marketing plans, financial information, financial or business projections, and all other compilations of information which relate to the business of Employer or Affiliate, and any other proprietary material of Employer or Affiliate, which have not been released to the general public. Confidential Information also includes information received by Employer or any of its Affiliates from others that the Employer or Affiliate has an obligation to treat as confidential. No materials or information shall be considered Confidential Information if Employee can prove that the materials or information are: (1) already known to Employee at the time that they are disclosed; or (2) publicly known at the time of the disclosure to Employee. Additionally, the confidential obligations herein will cease as to particular

information that: (1) has become publicly known through no fault of Employee; (2) is received by Employee properly and lawfully from a third party without restriction on disclosure and without knowledge or reasonable suspicion that the third party's disclosure is in breach of any obligations to Employer or its Affiliate; (3) has been developed by Employee completely independent of the delivery of Confidential Information hereunder; or (4) has been approved for public release by written authorization of Employer or its Affiliate.

(g) "Effective Date" means February 1, 2022.

(h) "<u>Foreign Government Official</u>" is defined to include officers, office holders, and employees, full or part time, regardless of rank, of local governments, national governments, companies partially owned or controlled by a government, and public international organizations, such as the United Nations or World Bank. "Foreign Government Official" also includes political parties, party officials, candidates for public office, and family members of Foreign Government Officials.

(i) "<u>Good Reason</u>" means the occurrence, on or after the occurrence of a Change in Control, of any of the following (except with Employee's written consent or resulting from an isolated, insubstantial, and inadvertent action not taken in bad faith and which is remedied by Employer or its Affiliate promptly after receipt of notice thereof from Employee):

(i) Employer or an Affiliate reduces Employee's Base Salary (as defined in Subparagraph 7(a) below);

(ii) Employer discontinues its bonus plan in which Employee participates as in effect immediately before the Change in Control without immediately replacing such bonus plan with a plan that is the substantial economic equivalent of such bonus plan, or amends such bonus plan so as to materially reduce Employee's potential bonus at any given level of economic performance of Employer or its successor entity;

(iii) Employer materially reduces the aggregate benefits and perquisites to Employee from those being provided immediately before the Change in Control;

(iv) Employer or any of its Affiliates requires Employee to change the location of Employee's job or office, so that Employee will be based at a location more than 25 miles from the location of Employee's job or office immediately before the Change in Control;

(v) Employer or any of its Affiliates reduces Employee's responsibilities or directs Employee to report to a person of lower rank or responsibilities than the person to whom Employee reported immediately before the Change in Control; or

(vi) the successor to Employer fails or refuses expressly to assume in writing the obligations of Employer under this Agreement.

For purposes of this Agreement, a determination by Employee that Employee has "Good Reason" shall be final and binding on Employer and Employee absent a showing of bad faith on Employee's part.

(j) "<u>**Restricted Period**</u>" means the period the Employer employs or compensates Employee, and (x) in the event that Employee is entitled to the Separation Payment, the greater of 18 months following the termination of Employee's employment or the number of months remaining in the Term but for such termination of employment or (y) in the event that Employee

is not entitled to the Separation Payment, one year following the termination of Employee's employment.

(k) **Separation Payment**" means a lump sum payment equal to (A) Employee's Base Salary for the remainder of the Term (but not less than 18 months) (as defined in Subparagraph 7(a) of this Agreement), plus (B) an amount equal to the greater of (x) the bonus that was paid to Employee under Subparagraph 7(b) for the immediately preceding bonus period, projected over the remainder of the Term (which for avoidance of doubt may be less than 18 months) or (y) the bonus that was paid to Employee under Subparagraph 7(b) for the bonus period referred to in (x), projected over the remainder of the Term (which for avoidance of doubt may be less than 18 months) or (y) the bonus that was paid to Employee under Subparagraph 7(b) for the bonus period referred to in (x), projected over the remainder of the Term (which for avoidance of doubt may be less than 18 months), plus (C) any accrued but unpaid vacation pay.

(l) "<u>Trade Secrets</u>" as used in this Agreement, shall be given its broadest possible interpretation under applicable law and shall mean all forms and types of financial, business, scientific, technical, economic, or engineering information, including patterns, plans, compilations, program devices, formulas, designs, prototypes, methods, techniques, processes, procedures, programs, or codes, whether tangible or intangible, and whether or how stored, compiled, or memorialized physically, electronically, graphically, photographically, or in writing that (1) Employer has taken reasonable measures to keep secret, and that (2) derives independent economic value, actual or potential, from not being generally known to, and not being readily ascertainable through proper means by, another person who can obtain economic value from the disclosure or use of the information.

(m) "<u>Work of Authorship</u>" means any computer program, code or system as well as any literary, pictorial, sculptural, graphic or audio visual work, whether published or unpublished, and whether copyrightable or not, in whatever form and jointly with others that (i) relates to any of Employer's or its Affiliate's existing or potential products, practices, processes, formulations, manufacturing, engineering, research, equipment, applications or other business or technical activities or investigations; or (ii) relates to ideas, work or investigations conceived or carried on by Employer or its Affiliate or by Employee in connection with or because of performing services for Employer or its Affiliate.

2. <u>BASIC EMPLOYMENT AGREEMENT</u>. Subject to the terms and pursuant to the conditions hereinafter set forth, Employer hereby employs Employee during the Term hereinafter specified to serve in a capacity, under a title, and with such duties not inconsistent with those set forth in Section 3 of this Agreement, as the same may be modified and/or assigned to Employee by Employer from time to time; provided, however, that no change in Employee's duties shall be permitted if it would result in a material reduction in the level of Employee's duties as in effect prior to the change, it being understood, however, that a change in Employee's reporting responsibilities is not, itself, a basis for finding a material reduction in the level of duties. Notwithstanding anything to the contrary contained herein, nothing in this Agreement shall be interpreted so as to permit Employer to require Employee to relocate his primary residence or his primary office outside of Las Vegas, Nevada metropolitan area; provided however, that Employee acknowledges and agrees that Employee's duties may require Employee to occasionally travel to locations where Employer has operations or is investigating development opportunities.

As of the Effective Date, this Agreement supersedes and replaces any and all prior employment agreements (including the 2017 Agreement), change in control agreements and severance plans or agreements, whether written or oral, by and between Employee, on the one side, and Employer or any of Employer's Affiliates, on the other side, or under which Employee is a participant, with the exception of any agreement pertaining to the issuance of restricted stock to Employee by Employer or any of its Affiliates, or any agreement providing for a retention or long term incentive bonus. From and after the

Effective Date, Employee shall be employed by Employer under the terms and pursuant to the conditions set forth in this Agreement.

3. <u>DUTIES OF EMPLOYEE</u>. Employee shall perform such duties assigned to Employee by Employer as are generally associated with the duties of **Chief Executive Officer** for Employer or such similar duties as may be assigned to Employee by Employer as Employer may determine. Employee's duties shall include: (i) the efficient and continuous operation of Employer and its Affiliates; (ii) the preparation of relevant budgets and allocation of relevant funds; (iii) the selection and delegation of duties and responsibilities of subordinates; (iv) the direction, review and oversight of all programs under Employee's supervision; (v) adherence to the policies and procedures of Employer and its Affiliates as they may be amended from time to time without prior notice to Employee (unless such policies and procedures conflict with this Agreement, in which case this Agreement takes precedence) and for which Employee assumes responsibility for review and understanding; and (vi) such other and related duties as may be assigned by Employer to Employee from time to time. The foregoing notwithstanding, Employee shall devote such time to Employer, or its Affiliates as may be required by Employer, provided such duties are not inconsistent with Employee's primary duties to Employer hereunder.

4. <u>ACCEPTANCE OF EMPLOYMENT</u>. Employee hereby unconditionally accepts the employment set forth hereunder, under the terms and pursuant to the conditions set forth in this Agreement. Employee hereby covenants and agrees that, except upon Employer's prior express written authorization, during the Term, Employee will devote the whole of Employee's normal and customary working time and best efforts solely to the performance of Employee's duties under this Agreement, and Employee shall not perform any services for any casino, hotel/casino or other similar gaming or gambling operation not owned by Employer or any of Employer's Affiliates.

Employee represents and warrants to Employer that the execution and delivery of this Agreement and the performance of the Employee's duties hereunder shall not violate the terms or conditions of any employment agreement or arrangement or any other agreement to which Employee is a party.

5. <u>TERM</u>. Unless sooner terminated as provided in this Agreement, the term of this Agreement (the "Term") shall commence on the Effective Date of this Agreement and terminate on February 15, 2025, at which time the terms of this Agreement shall expire and shall not apply to any continued employment of Employee by Employer, except for those obligations under Sections 9, 10, 11 and 21. Following the Term, unless the parties enter into a new written contract of employment, (a) any continued employment of Employee shall be at-will, (b) any or all of the other terms and conditions of Employee's employment may be changed by Employer at its discretion, with or without notice, and (c) the employment relationship may be terminated at any time by either party, with or without cause or notice.

Concurrent with Employee's resignation from Employer or upon the termination of Employee's employment with Employer, Employee agrees to resign, and shall be deemed to have resigned, all other positions (including board of director memberships) that Employee may have held immediately prior to Employee's resignation or termination.

6. SPECIAL TERMINATION PROVISIONS.

(a) Notwithstanding the provisions of Section 5, this Agreement shall terminate upon the occurrence of any of the following events:

(i) the death of Employee;

(ii) the giving of written notice from Employer to Employee of the termination of this Agreement upon the Complete Disability of Employee;

(iii) the giving of written notice by Employer to Employee of the termination of this Agreement upon the discharge of Employee for Cause (Employer's right to terminate for Cause (as defined in Section 1(c) shall survive the expiration of this Agreement)). It is expressly acknowledged and agreed that the decision as to whether "Cause" exists for termination of the employment relationship by Employer is delegated to the Employer's board of directors. If Employee disagrees with the decision reached by Employer's board of directors, any dispute as to the "Cause" determination will be limited to whether Employer's board of directors reached its decision in good faith, based upon facts reasonably believed by Employer's board of directors to be true, and not for any arbitrary, capricious, or illegal reason. This shall be the standard applied by any fact finder, and Employee shall bear the burden to prove that "Cause," under this standard, did not exist;

(iv) the giving of written notice by Employer to Employee of the termination of this Agreement following a disapproval of this Agreement or the denial, suspension, limitation or revocation of Employee's License (as defined in Section 8(b) of this Agreement);

(v) the giving of written notice by Employee to Employer upon a material breach of this Agreement by Employer, which material breach remains uncured for a period of thirty (30) days after the giving of such notice. "Material breach" under this Section 6(a)(v) is defined as Employer's failure to pay Employee's Base Salary when due, Employer's implementation of a material reduction in the scope of duties or responsibilities of Employee such that Employee's remaining duties and responsibilities are materially inconsistent with the duties and responsibilities generally associated with Employee's position within Employer's organization, or if such position is the only position with Employer, or its Affiliates, irrespective of the title of the position, or a material reduction in Employee's Base Salary; provided, however, that "material breach" shall not be construed to include any change in reporting structure alone with no material change to title, duties and responsibilities, any changes to Employee's duties pursuant to Section 6(a)(vi), any changes to Employee's duties and responsibilities as a result of a request by the Authorities under Section 8, or the temporary suspension of the Employee from duty, pursuant to Employer's policy, pending investigation by Employer of any incident or occurrence that could give rise to discipline or termination of employment. Termination of employment pursuant to this Section 6(a)(v) does not relieve Employee of his duties and responsibilities under Sections 9, 10, 11, and 21 of this Agreement;

(vi) the giving of written ninety (90) day notice by Employer to Employee of Employer's intention to terminate this Agreement without Cause for any reason deemed sufficient by Employer to be effective at the end of such ninety (90) day period. During such ninety (90) day notice period, Employer shall be permitted to reduce Employee's responsibilities and time commitment to Employer; provided however, Employer may not reduce Employee's salary or benefits during such ninety (90) day period. At the end of such ninety (90) day period, Employee shall cease to be an employee of the Employer and this Agreement shall automatically terminate. Upon receipt of such notice, Employee shall have the option to resign Employee's employment effective as of the date of the notice, rather than remain employed through such ninety (90) day period. If Employee elects to resign in lieu of termination, Employee must exercise this option in writing within 72 hours of receipt of the Employer's notice of intention to terminate this Agreement without Cause. Employee's written resignation in lieu of termination must be transmitted to Employer by email or hand delivery;

(vii) at Employee's sole election in writing as provided in Paragraph 17 of this Agreement, after both a Change of Control and as a result of Good Reason, provided, however, that, within thirty (30) calendar days after Employer's receipt of Employee's written election, Employer must tender the Separation Payment to Employee; or

(viii) at Employee's sole election in writing as provided in Paragraph 17 of the Agreement upon ninety (90) day notice that Employee wishes to resign his position.

(b) Consequences of Termination.

In the event Employee resigns pursuant to Section 6(a)(v), 6(a)(vi), or 6(a)(vii) but not pursuant to Section 6(a)(viii), Employer's sole liability to Employee shall be payment of the Separation Payment; provided that Employee shall not be entitled to payment of the Separation Payment unless and until Employee first executes a written release-severance agreement, prepared and presented by Employer, that fully releases Employer, Affiliates, and their respective officers, directors, agents and employees, from any and all claims or causes of action, whether based upon statute, contract (including without limitation breach or construction of this Agreement), or common law, that have arisen as of the date of such execution, irrespective of whether Employee has knowledge of the existence of such claim; and provides for the confidentiality of both the terms of the releaseseverance agreement and the compensation paid. In the event Employee fails or refuses to execute such release-severance agreement, Employer shall have no further obligation to Employee other than payment of all accrued but unpaid Base Salary through the date Employee last performs services for Employer, vacation pay accrued but unpaid and expenses incurred but not reimbursed through the termination date; specifically, in such event, Employee shall not be entitled to any benefits pursuant to any severance plan in effect by Employer or any of its Affiliates. Employee will also be entitled to receive health benefits coverage for Employee and Employee's dependents, at Employee's election, under 7(c) or under the same plan(s) or arrangement(s) under which Employee was covered immediately before Employee's termination, or plan(s) established, or arrangement(s) provided by Employer or any of its Affiliates thereafter. Such health benefits coverage shall be paid for by Employer to the same extent as if Employee were still employed by Employer, and Employee will be required to make such payments as Employee would be required to make if Employee were still employed by Employer. The health benefits provided under this Paragraph 6 shall continue until the earlier of (x) the expiration of the period for which the Separation Payment is paid], (y) the date Employee becomes covered under any other group health plan not maintained by Employer or any of its Affiliates; provided, however, that if such other group health plan excludes any pre-existing condition that Employee or Employee's dependents may have when coverage under such group health plan would otherwise begin, coverage under this Paragraph 6 shall continue (but not beyond the period described in clause (x) of this sentence) with respect to such pre-existing condition until such exclusion under such other group health plan lapses or expires. In the event Employee is required to make an election under Sections 601 through 607 of the Employee Retirement Income Security Act of 1974, as amended (commonly known as COBRA) to qualify for the health benefits described in this Paragraph 6, the obligations of Employer and its Affiliates under this Paragraph 6 shall be conditioned upon Employee's timely making such an election. In the event of a termination of this Agreement pursuant to any of the provisions of this Paragraph 6, Employee shall not be entitled to any benefits pursuant to any severance plan in effect by Employer or any of Employer's Affiliates.

(ii) In addition to the provisions set forth in Section 6(b)(i) above, in the event of a termination of this Agreement pursuant to Section 6(a)(v) or 6(a)(vi), prior to the final vesting date, a pro-rated portion of the stock awards granted to Employee pursuant to Section 7(d) below equal to the number of full calendar months elapsed between the grant date and the date of such termination of employment divided by the number of full calendar months between the grant date and the final vesting date shall vest, less those shares that have already vested or have been forfeited, and become payable within 30 days following such termination of employment.

(iii) In addition to the provisions set forth in Section 6(b)(i) above, in the event of a termination of this Agreement pursuant to Section 6(a)(vii), any unvested shares of restricted stock of Wynn Resorts, Limited granted to Employee pursuant to Section 7(d) below shall immediately vest upon the termination date.

(iv) In the event of a termination of this Agreement pursuant to Section 6(a)(i), 6(a)(ii), 6(a)(iv), or 6(a)(viii) Employer shall not be required to make any payments to Employee other than payment of Base Salary, vacation pay accrued but unpaid and expenses incurred but not reimbursed through the termination date; specifically, in such event, Employee shall not be entitled to any benefits pursuant to any severance plan in effect by Employer or any of its Affiliates.

(v) In the event of a termination of this Agreement pursuant to Section 6(a)(iii), Employer shall not be required to make any payments to Employee other than payment of Base Salary and expenses incurred but not reimbursed through the termination date; specifically, in such event, Employee shall not be entitled to any benefits pursuant to any severance plan in effect by Employer or any of its Affiliates.

(vi) In the event of a termination of this Agreement for any reason, the termination provisions set forth in any stock agreements (other than in respect of the equity awards granted pursuant to Section 7(d) below) shall control with regard to the vesting of the applicable award.

7. <u>COMPENSATION TO EMPLOYEE</u>. For and in complete consideration of Employee's full and faithful performance of Employee's duties under this Agreement, Employer hereby covenants and agrees to pay to Employee, and Employee hereby covenants and agrees to accept from Employer, the following items of compensation:

(a) **Base Salary**. Employer hereby covenants and agrees to pay to Employee, and Employee hereby covenants and agrees to accept from Employer, a base salary at the rate of One Million Eight Hundred Thousand Dollars (\$1,800,000.00) per annum, payable in such installments as shall be convenient to Employer (the "Base Salary"). Employee shall be subject to performance reviews and the Base Salary may be increased but not decreased as a result of any such review. Such Base Salary shall be exclusive of and in addition to any other benefits which Employer, in its sole discretion, may make available to Employee, including any discretionary bonus, profit sharing plan, pension plan, retirement plan, disability or life insurance plan, medical and/or hospitalization plan, or any and all other benefit plans which may be in effect during the Term.

(b) **Bonus Compensation.** Employee will participate in Employer's Amended and Restated Annual Performance Based Incentive Plan for Executive Officers (the "Annual Bonus Plan") with an annual target bonus of no less than 200% of the Base Salary. Any annual bonus Employee may be eligible to earn under the Annual Bonus Plan shall be conditioned on

Employee's continued employment through the applicable bonus performance period and may be subject to forfeiture in whole or in part upon Employee's knowing and material violation of the Employer's Code of Business Conduct and Ethics or the Preventing Harassment and Discrimination Policy. Employee shall also be eligible to receive a bonus at such times and in such amounts as Employer in its sole and exclusive discretion may determine. Employer retains the discretion to adopt, amend or terminate any bonus plan at any time prior to a Change of Control.

(c) <u>Employee Benefit Plans</u>. Employer hereby covenants and agrees that it shall include Employee, if otherwise eligible, in any profit-sharing plan, executive stock option plan, pension plan, retirement plan, disability or life insurance plan, Executive Medical Plan and/or hospitalization plan, and any other benefit plan which may be placed in effect by Employer or any of its Affiliates and on the same terms and conditions available to Employer's executives during the Term. All issues as to eligibility for specific benefits and payment of benefits shall be as set forth in the applicable insurance policies or plan documents. Nothing in this Agreement shall limit Employer's or any of its Affiliates' ability to exercise the discretion provided to it under any employee benefit plan, or to adopt, amend or terminate any benefit plan at any time prior to a Change of Control.

Employee shall also participate in the senior executive health program at all times while employed by Employer and for the twelvemonth period subsequent to (i) the Term or (ii) the termination of the Agreement pursuant to Sections 6(a)(v), 6(a)(v) or 6(a)(vi).

(d) **Equity Grant.** Employee was granted 30,000 shares of restricted stock of Wynn Resorts, Limited common stock pursuant to the Wynn Resorts, Limited 2014 Omnibus Incentive Plan. Employee and Employer entered into a separate restricted stock agreement, dated March 1, 2017, and amended on April 17, 2018, incorporating the terms and conditions of the grant, including the grant date, vesting schedule, and termination provisions.

Employee was granted 25,000 shares of restricted stock of Wynn Resorts, Limited common stock pursuant to the Wynn Resorts, Limited 2014 Omnibus Incentive Plan. Employee and Employer entered into a separate restricted stock agreement, dated April 17, 2018, incorporating the terms and conditions of the grant, including the grant date, vesting schedule, and termination provisions.

As stated in the Fourth Amendment to the 2017 Agreement, subject to and effective upon the approval of the Compensation Committee of Wynn Resorts, Limited, and upon closing of the merger between Wynn Interactive Limited and Austerlitz Acquisition Corporation I, Employee shall receive 10,000 shares of stock of Wynn Resorts Limited common stock pursuant to the Wynn Resorts Limited 2014 Omnibus Incentive Plan. Such shares shall be immediately vested and available to Employee without restriction.

Employee shall be granted 52,715 shares of performance-vesting restricted stock of Wynn Resorts, Limited common stock pursuant to the Wynn Resorts, Limited 2014 Omnibus Incentive Plan and pursuant to an award agreement to be entered into by and between Employee and Employee, which award agreement will set forth the terms and conditions of the grant, including the conditions for grants, grant date, vesting schedule, and termination-related provisions

(e) <u>Annual Equity Grant</u>. Commencing on the Effective Date, Employee shall be eligible to receive an annual restricted share grant of Wynn Resorts, Limited common stock with a target value equivalent to 375% of the annual Base Salary for Employee in effect at the end of the applicable year, with vesting requirements consistent with comparable positions in the

Company. Employee and Employer will enter into a separate restricted stock agreement incorporating the terms and conditions of the grant, including the grant date, vesting schedule, and termination provisions. Should Employee's annual equity award for the 2022 fiscal year (the "2022 Annual Equity Award") be granted prior to the Effective Date, and should such 2022 Annual Equity Award have a grant date fair value of less than \$6,800,000, then, within 10 days after the Effective Date, Employer shall provide Employee with an additional one-time equity award grant that has a grant date fair value equal to the difference between \$6,800,000 and the grant date fair value of the 2022 Annual Equity Award (the "2022 Top up Award"). The terms of the 2022 Top Up Award shall be the same as those of the 2022 Annual Equity Award.

(f) **Expense Reimbursement.** During the Term and provided the same are authorized in advance by Employer, Employer shall either pay directly or reimburse Employee for Employee's reasonable expenses incurred for the benefit of Employer in accordance with Employer's general policy regarding expense reimbursement, as the same may be modified from time to time. Prior to such payment or reimbursement, Employee shall provide Employer with sufficient detailed invoices of such expenses as may be required by Employer's policy.

(g) <u>Vacations and Holidays</u>. Commencing as of the Effective Date, Employee shall be entitled to (i) annual paid vacation leave in accordance with Employer's standard policy, but in no event less than four (4) weeks each year of the Term, to be taken at such times as selected by Employee and approved by Employer, and (ii) paid holidays (or, at Employer's option, an equivalent number of paid days off) in accordance with Employer's standard policy.

(h) <u>Section 409A Provision</u>. Notwithstanding any provision of the Agreement to the contrary, if, at the time of Employee's termination of employment with the Employer, he or she is a "specified employee" as defined in Section 409A of the Internal Revenue Code (the "Code"), and one or more of the payments or benefits received or to be received by Employee pursuant to the Agreement would constitute deferred compensation subject to Section 409A, no such payment or benefit will be provided under the Agreement until the earlier of: (a) the date that is six (6) months following Employee's termination of employment with the Employer or (b) the Employee's death. The provisions of this Section shall only apply to the extent required to avoid Employee's incurrence of any penalty tax or interest under Section 409A of the Code or any regulations or Treasury guidance promulgated thereunder. In addition, if any provision of the Agreement would cause Employee to incur any penalty tax or interest under Section 409A of the code or any regulations or Treasury guidance promulgated thereunder, the Employer may reform such provision to maintain the maximum extent practicable the original intent of the applicable provision without violating the provisions of Section 409A of the Code.

(i) <u>Withholdings</u>. All compensation provided to Employee by Employer under this Section 7 shall be subject to applicable federal, state, or local employment-related withholdings.

(j) **<u>Original Hire Date</u>**. Employee's original hire date of March 1, 2017 shall be used for determining all other benefits.

8. LICENSING REQUIREMENTS.

(a) Employee and Employee hereby covenant and agree that this Agreement and/or Employee's employment may be subject to the approval of one or more gaming regulatory authorities (the "**Authorities**") pursuant to the provisions of the relevant gaming regulatory statutes (the "**Gaming Acts**") and the regulations promulgated thereunder (the "Gaming Regulations"). Employer and Employee hereby covenant and agree to use their best efforts to obtain any and all approvals required by the Gaming Acts and/or Gaming Regulations. In the

event that (i) an approval of this Agreement or Employee's employment by the Authorities is required for Employee to carry out Employee's duties and responsibilities set forth in Section 3 of this Agreement, (ii) Employer and Employee have used their best efforts to obtain such approval, and (iii) this Agreement or Employee's employment is not so approved by the Authorities, then this Agreement shall immediately terminate and shall be null and void, thus extinguishing any and all obligations of either party, subject to any surviving obligations of Employee under Sections 9, 10 and 21.

(b) If applicable, Employer and Employee hereby covenant and agree that, in order for Employee to discharge the duties required under this Agreement, Employee must apply for or hold a license, registration, permit or other approval (the "**License**") as issued by the Authorities pursuant to the terms of the relevant Gaming Act and as otherwise required by this Agreement. In the event Employee fails to apply for and secure, or the Authorities refuse to issue or renew Employee's License, Employee, at Employer's sole cost and expense, shall promptly defend such action and shall take such reasonable steps as may be required to either remove the objections or secure or reinstate the Authorities' approval, respectively. The foregoing notwithstanding, if the source of the objections or the Authorities' refusal to renew or maintain Employee's License arise as a result of any of the acts, omissions or events described in Section 1(c) of this Agreement, then Employer's obligations under this Section 8 also shall not be operative and Employee shall promptly reimburse Employer upon demand for any expenses incurred by Employer pursuant to this Section 8.

(c) Employer and Employee hereby covenant and agree that the provisions of this Section 8 shall apply in the event Employee's duties require that Employee also be licensed by governmental agencies other than the Authorities.

9. <u>CONFIDENTIALITY</u>.

(a) Employee hereby warrants, covenants, and agrees that:

(i) Employee shall not directly or indirectly use or disclose any Confidential Information, Trade Secrets, or Works of Authorship, whether in written, verbal, electronic, or model form, at any time or in any manner, except as required in the conduct of Employer's business or as expressly authorized by Employer in writing. Employee shall take all necessary and available precautions to protect against the unauthorized disclosure of Confidential Information, Trade Secrets, or Works of Authorship. Employee acknowledges and agrees that such Confidential Information, Trade Secrets, or Works of Authorship are the sole and exclusive property of Employer or its Affiliates.

(ii) Employee shall not remove from Employer's premises any Confidential Information, Trade Secrets, Works of Authorship, or any other documents pertaining to Employer's or its Affiliates' business, unless expressly authorized by Employer in writing. Furthermore, Employee specifically covenants and agrees not to make any duplicates, copies, or reconstructions of such materials and that, if any such duplicates, copies, or reconstructions are made, they shall become the property of Employer or its Affiliates upon their creation.

(iii) Upon termination of Employee's employment with Employer for any reason, Employee shall return to Employer the originals and all copies of any and all papers, documents and things, including information stored for use in or with computers and software, all files, Rolodex cards, phone books, notes, price lists, customer contracts, bids, customer lists, notebooks, books, memoranda, drawings, computer disks or drives, or other

documents: (i) made, compiled by, or delivered to Employee concerning any customer served by Employer or any of its Affiliates or any product, apparatus, or process manufactured, used, developed or investigated by Employer or any of its Affiliates; (ii) containing any Confidential Information, Trade Secret or Work of Authorship; or (iii) otherwise relating to Employee's performance of duties under this Agreement. Employee further acknowledges and agrees that all such documents are the sole and exclusive property of Employer or its Affiliates.

(b) Employee hereby warrants, covenants and agrees that Employee shall not disclose to Employer, or any Affiliate, officer, director, employee or agent of Employer, or use in the course of performing Employee's duties and responsibilities for Employer any proprietary or confidential information or property, including any trade secret, formula, pattern, compilation, program, device, method, technique or process, which Employee is prohibited by contract, or otherwise, to disclose to Employer (the "**Restricted Information**"). In the event Employer requests Restricted Information from Employee, Employee shall advise Employer that the information requested is Restricted Information and may not be disclosed by Employee.

(c) Notwithstanding any provision of this Agreement prohibiting the disclosure of Trade Secrets or other Confidential Information, Employee understands that Employee may not be held criminally or civilly liable under any federal or state trade secret law for the disclosure of a trade secret that (i) is made (A) in confidence to a federal, state or local government official, either directly or indirectly, or to an attorney, and (B) solely for the purpose of reporting or investigating a suspected violation of law, or (ii) is made in a complaint or other document filed in a lawsuit or other proceeding, if such filing is made under seal. In addition, if Employee files a lawsuit or other court proceeding against the Employer for retaliating against Employee for reporting a suspected violation of law, Employee may disclose the trade secret to the attorney representing Employee and use the trade secret in the court proceeding, if Employee files any document containing the trade secret under seal and does not disclose the trade secret, except pursuant to court order.

(d) Notwithstanding anything to the contrary contained herein, no provision of this Agreement will be interpreted so as to impede Employee (or any other individual) from (a) making any disclosure of relevant and necessary information or documents in any action, investigation or proceeding relating to this Agreement, or as required by law or legal process, including with respect to possible violations of law, (b) participating, cooperating or testifying in any action, investigation or proceeding with, or providing information to, any governmental agency, including any gaming regulatory agency, legislative body or any self-regulatory organization, including, but not limited to, the Department of Justice, the Securities and Exchange Commission, the Congress and any agency Inspector General, (c) accepting any U.S. Securities and Exchange Commission awards or (d) making other disclosures under the whistleblower provisions of federal law or regulation. In addition, nothing in this Agreement or any other agreement or Employer policy prohibits or restricts Employee from initiating communications with, or responding to any inquiry from, any administrative, governmental, regulatory, or supervisory authority regarding any good faith concerns about possible violations of law or regulation. Employee does not need the prior authorization of Employer to make any such reports or disclosures, and Employee will not be required to notify Employer that such reports or disclosures have been made.

(e) The obligations of this Section 9 are continuing and shall survive the termination of Employee's employment with Employer for any reason.

10. <u>RESTRICTIVE COVENANT/NO SOLICITATION.</u>

(a) Employee hereby covenants and agrees that during the Restricted Period, Employee shall not, directly or indirectly, either as a principal, agent, employee, employer, consultant, partner, member of a limited liability company, shareholder of a closely held corporation, or shareholder in excess of two percent (2%) of a publicly traded corporation, corporate officer or director, manager, or in any other individual or representative capacity, engage or otherwise participate in any manner or fashion in any business that is in competition in any manner whatsoever with the principal business activity of Employer or its Affiliates, in or about any market in which Employer or its Affiliates currently operate or have announced, publicly or otherwise, a plan to have hotel or gaming operations, including any hotel, casino, restaurant, lounge, nightclub, day club, beach club, or sports wagering or online gaming operations.

(b) Employee hereby further covenants and agrees that during the Restricted Period, Employee shall not take any actions, whether directly or indirectly, including by way of a third-party intermediary, to solicit, encourage or otherwise cause any employee of Employer or its Affiliates with or on behalf of any business that is in competition in any manner whatsoever with the principal business activity of Employer or its Affiliates, in or about any market in which Employer or its Affiliates currently operate or have announced, publicly or otherwise, a plan to have hotel or gaming, nightclub beach club, or sports wagering or online gaming operations. The parties agree that the terms "solicit, encourage or otherwise cause" include Employee's participation in the recruitment, applicant assessment or review, and employee selection. The parties further agree that this Section 10(b) applies even if the then-Employer's or Affiliate's employee makes the initial contact seeking employment with Employee or competitor as defined above.

(c) Employee hereby further covenants and agrees that the restrictive covenants contained in this Section 10 are reasonable as to duration, terms, and geographical area and that they protect the legitimate interests of Employer, impose no undue hardship on Employee, and are not injurious to the public. In the event that any of the restrictions and limitations contained in this Section 10 are deemed to exceed the time, geographic or other limitations permitted by Nevada law, the parties agree that a court of competent jurisdiction shall revise any offending provisions so as to bring this Section 10 within the maximum time, geographical or other limitations permitted by Nevada law.

(d) Employee hereby agrees that any subsequent material change or changes in Employee's title, duties, salary, or compensation will not affect the validity or scope of this Section 10 or invalidate this Section 10 in any way.

11. <u>**REMEDIES</u>**. Employee acknowledges that Employer has and will continue to deliver, provide, and expose Employee to certain knowledge, information, practices, and procedures possessed or developed by or for Employer at a considerable investment of time and expense, which are protected as confidential, and which are essential for carrying out Employer's business in a highly competitive market. Employee also acknowledges that Employee will be exposed to Confidential Information, Trade Secrets, Works of Authorship, inventions, and business relationships possessed or developed by or for Employer or its Affiliates, and that Employer or its Affiliates would be irreparably harmed if Employee were to improperly use or disclose such items to competitors, potential competitors, or other parties. Employee further acknowledges that the protection of Employer's and its Affiliates' customers and businesses is essential and understands and agrees that Employer's and its Affiliates' relationships with its customers and its employees are special and unique and have required a considerable investment of time and funds to develop, and that any loss of or damage to any such relationship will result in irreparable harm. Consequently, Employee covenants and agrees that any violation by Employee of Section 9 or 10 shall</u>

entitle Employer to immediate injunctive relief in a court of competent jurisdiction. Employee further agrees that no cause of action for recovery of materials or for breach of any of Employee's representations, warranties or covenants shall accrue until Employer or its Affiliate has actual notice of such breach. Employee further agrees that the time period covered by the covenants of this Agreement will not include and shall be extended by any period(s) of time required for litigation brought by the Company to enforce any covenant or in which Employee is in violation of his/her promises contained in Section 10.

12. <u>**BEST EVIDENCE.</u>** This Agreement shall be executed in original and "Xerox" or photostatic copies and each copy bearing original signatures in ink shall be deemed an original.</u>

13. <u>SUCCESSION</u>. This Agreement shall be binding upon and inure to the benefit of the parties hereto and their heirs, executors, administrators, personal representatives, successors and permitted assigns.

14. <u>ASSIGNMENT</u>. Employee shall not assign this Agreement or delegate Employee's duties hereunder without the express written prior consent of Employer thereto. Any purported assignment by Employee in violation of this Section 14 shall be null and void and of no force or effect. Employer shall have the right to assign this Agreement freely, including Employee's obligations under Section 10, and Employee hereby acknowledges receipt of consideration in exchange for Employee's consent to the assignability of Employee's obligations under Section 10 that is additional to and separate from the consideration provided to Employee in exchange for the other covenants in this Agreement.

15. <u>AMENDMENT OR MODIFICATION</u>. This Agreement may not be amended, modified, changed, or altered except by a writing signed by both Employer and Employee.

16. *GOVERNING LAW*. This Agreement shall be governed by and construed in accordance with the laws of the State of Nevada, without regard to conflict of laws principles.

17. <u>NOTICES</u>. All notices required under this Agreement shall be in writing and shall be either hand-delivered or mailed, certified mail, return receipt requested, addressed to:

TO EMPLOYER: Wynn Resorts, Limited 3131 Las Vegas Boulevard South Las Vegas, Nevada 89109 Attn: Legal Department

TO EMPLOYEE: Craig Billings [Address Redacted]

All notices hand-delivered shall be deemed delivered as of the date actually delivered. All notices mailed shall be deemed delivered as of three (3) business days after the date postmarked. Any changes in any of the addresses listed herein shall be made by notice as provided in this Section 17.

18. <u>INTERPRETATION</u>. The preamble recitals to this Agreement are incorporated into and made a part of this Agreement; titles of sections and paragraphs are for convenience only and are not to be considered a part of this Agreement. Whenever the words "include," "includes" or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation."

19. <u>SEVERABILITY</u>. In the event any one or more provisions of this Agreement is declared judicially void or otherwise unenforceable, the remainder of this Agreement shall survive, and such provision(s) shall be deemed modified or amended so as to fulfill the intent of the parties hereto.

20. <u>*WAIVER*</u>. None of the terms of this Agreement, including this Section 20, or any term, right or remedy hereunder, shall be deemed waived unless such waiver is in writing and signed by the party to be charged therewith and in no event by reason of any failure to assert or delay in asserting any such term, right or remedy or similar term, right or remedy hereunder.

21. <u>DISPUTE RESOLUTION.</u> Except for a claim by either Employee or Employer for injunctive relief where such would be otherwise authorized by law to enforce Sections 9, 10 and/or 11 of this Agreement, any controversy or claim arising out of or relating to this Agreement, the breach hereof, or Employee's employment by Employer, including any claim involving the interpretation or application of this Agreement, or claims for wrongful termination, discrimination, or other claims based upon statutory or common law, shall be submitted to binding arbitration in accordance with the employment arbitration rules then in effect of the American Arbitration Association ("AAA"), to the extent not inconsistent with this Section as set forth below, and the Federal Arbitration Act, 9 U.S.C. § 1, et seq. and the Uniform Arbitration Act as adopted in Nevada Revised Statutes 38.015, et seq. This Section 21 applies to any claim Employee might have against any officer, director, employee, or agent of Employer or its Affiliate, and all successors and assigns of any of them. These arbitration provisions shall survive the termination of Employee's employment with Employer and the expiration of this Agreement.

(a) **Coverage of Arbitration Agreement:** The promises by Employer and Employee to arbitrate differences, rather than litigate them before courts or other bodies, provide consideration for each other, in addition to other consideration provided under this Agreement. The parties contemplate by this Section 21 arbitration of all claims against each of them to the fullest extent permitted by law except as specifically excluded by this Agreement. Only claims that are justiciable or arguably justiciable under applicable federal, state, or local law are covered by this Section, and include any and all alleged violations of any federal, state, or local law whether common law, statutory, arising under regulation or ordinance, or any other law, brought by any current or former employee. Such claims may include claims for: wages or other compensation; breach of contract; torts; work-related injury claims not covered under workers' compensation laws; wrongful discharge; and any and all unlawful employment discrimination and/or harassment claims. Employee and Employer agree to pursue any and all covered claims individually and waive any rights they may have to pursue said claims as part of any class action. In that regard, Employee and Employer agree that the arbitrator shall have no authority or jurisdiction to hear class or collective claims.

This Section 21 excludes claims under state workers' compensation or unemployment compensation statutes; claims pertaining to any of Employer's employee welfare, insurance, benefit, and pension plans, with respect to which are applicable the filing and appeal procedures of such plans shall apply to any denial of benefits; claims for injunctive or equitable relief for violations of non-competition and/or confidentiality covenants contained in Sections 9, 10 and 11; or any claims that are prohibited as a matter of law from being covered by this Section 21.

(b) <u>Waiver of Rights to Pursue Claims in Court and to Jury Trial</u>: This Section 21 does not in any manner waive any rights or remedies available under applicable statutes or common law but does waive Employer's and Employee's rights to pursue those rights and remedies in a judicial forum and waive any right to trial by jury of any claims covered by Section 21(a). By signing this Agreement, the parties voluntarily agree to arbitrate any covered claims against each other. In the event of any administrative or judicial action by any agency or third

party to adjudicate, on behalf of Employee, a claim subject to arbitration, Employee hereby waives the right to participate in any monetary or other recovery obtained by such agency or third party in any such action, and Employee's sole remedy with respect to any such claim will be any award decreed by an arbitrator pursuant to the provisions of this Agreement.

(c) **Initiation of Arbitration:** To commence arbitration of a claim subject to this Section 21, the aggrieved party must, within the time frame provided in Section 21(d) below, make written demand for arbitration and provide written notice of that demand to the other party. If a claim is brought by Employee against Employer, such notice shall be given to Employer's Legal Department. Such written notice must identify and describe the nature of the claim, the supporting facts, and the relief or remedy sought. In the event that either party files an action in any court to pursue any of the claims covered by this Section 21, the complaint, petition, or other initial pleading commencing such court action shall be considered the demand for arbitration. In such event, the other party may move that court to compel arbitration.

(d) <u>Time Limit to Initiate Arbitration</u>: To ensure timely resolution of disputes, Employee and Employer must initiate arbitration within the statute of limitations (deadline for filing) provided by applicable law pertaining to the claim, or one year, whichever is shorter, except that the statute of limitations imposed by relevant law will solely apply in circumstances where such statute of limitations cannot legally be shortened by private agreement. The failure to initiate arbitration within this time limit will bar any such claim. The parties understand that Employer and Employee are waiving any longer statutes of limitations that would otherwise apply, and any aggrieved party is encouraged to give written notice of any claim as soon as possible after the event(s) in dispute so that arbitration of any differences may take place promptly.

(e) **Arbitrator Selection:** The parties contemplate that, except as specifically set forth in this Section 21, selection of one (1) arbitrator shall take place pursuant to the then-current rules of the AAA applicable to employment disputes. The arbitrator must be either a retired judge or an attorney experienced in employment law. The parties will select one arbitrator from among a list of qualified neutral arbitrators provided by AAA. If the parties are unable to agree on the arbitrator, the parties will select an arbitrator by alternatively striking names from a list of qualified arbitrators provided by AAA. AAA will flip a coin to determine which party has the final strike (that is, when the list has been narrowed by striking to two arbitrators). The remaining named arbitrator will be selected.

(f) <u>Arbitration Rights and Procedures</u>: Employee may be represented by an attorney of his/her choice at his/her own expense. Any arbitration hearing or proceeding will take place in private, not open to the public, in Clark County, Nevada. The arbitrator shall apply the substantive law (and the law of remedies, if applicable) of Nevada (without regard to its choice of law provisions) and/or federal law as applicable to the claim(s) for relief asserted. The arbitrator is without power or jurisdiction to apply any different substantive law or law of remedies or to modify any term or condition of this Agreement. The arbitrator will have no power or authority to award non-economic damages or punitive damages except where such relief is specifically authorized by an applicable federal, state, or local statute or ordinance, or common law. In such a situation, the arbitrator shall specify in the award the specific statute or other basis under which such relief is granted. The applicable law with respect to privilege, including attorney-client privilege, work product, and offers to compromise must be followed. The parties will have the right to conduct reasonable discovery, including written and oral (deposition) discovery and to subpoena and/or request copies of records, documents, and other relevant discoverable information consistent with the procedural rules of AAA. The arbitrator will decide disputes regarding the scope of discovery and will have authority to regulate the conduct of any hearing.

The arbitrator will have the right to entertain a motion or request to dismiss, for summary judgment, or for other summary disposition, permitting a motion, a brief in opposition, and a reply brief by the movant. The parties will exchange witness lists at least 30 days prior to the hearing. The arbitrator will have subpoen power so that either Employee or Employer may summon witnesses. The arbitrator will use the Federal Rules of Evidence in connection with the admission of all evidence at the hearing. Both parties shall have the right to file post-hearing briefs. Any party, at its own expense, may arrange for and pay the cost of a court reporter to provide a stenographic record of the proceedings.

(g) **Arbitrator's Award:** The arbitrator will issue a written decision containing a statement as to the specific claims and issues raised by the parties, the specific findings of fact, and the specific conclusions of law. The award will be rendered promptly, typically within 30 days after conclusion of the arbitration hearing, or after the submission of post-hearing briefs if requested. The arbitrator shall have no power or authority to award any relief or remedy in excess of what a court could grant under applicable law. The arbitrator's decision shall be final and binding on both parties. Judgment upon an award rendered by the arbitrator may be entered in any court having competent jurisdiction.

(h) **Fees and Expenses:** Unless the law requires otherwise for a particular claim or claims, the party demanding arbitration bears the responsibility for payment of the fee to file with AAA and the fees and expenses of the arbitrator shall be allocated by the AAA under its rules and procedures. Employee and Employer shall each pay his/her/its own expenses for presentation of their cases, including attorney's fees, costs, and fees for witnesses, photocopying and other preparation expenses. If any party prevails on a statutory claim that affords the prevailing party attorney's fees and costs, the arbitrator may award reasonable attorney's fees and/or costs to the prevailing party, applying the same standards a court would apply under the law applicable to the claim.

(i) <u>Severability and Waiver of Trial by Jury</u>: Employee and Employer further agree that, if a court of competent jurisdiction finds any term or condition of this dispute resolution process is not in compliance with the law, that court shall sever or revise ("blue pencil") any offending provision(s) of this dispute resolution process so as to bring it within legal compliance. Should such a court of competent jurisdiction decline to sever or revise this dispute resolution process to render it enforceable as to all covered claims asserted in any particular dispute and instead voids the application of this dispute resolution process as to one or more covered claims and/or refuses to enforce the parties' waiver of class action/collective release, Employee and Employer agree to mutually waive their respective rights to a trial by jury in a court of competent jurisdiction in which an action is filed to resolve any such covered claims.

Employee and Employer agree to sign below to specifically authorize and affirmatively agree to utilize the provisions of Section 21 of this Agreement.

22. <u>PAROL</u>. This Agreement constitutes the entire agreement between Employer and Employee, and supersedes any prior understandings, agreements, undertakings, or severance policies or plans by and between Employer or its Affiliates, on the one side, and Employee, on the other side, with respect to its subject matter or Employee's employment with Employer or its Affiliates (including the 2017 Agreement). As of the Effective Date, this Agreement supersedes and replaces any and all prior employment agreements (including the 2017 Agreement), change in control agreements and severance plans or agreements, whether written or oral, by and between Employee, on the one side, and Employer or any of Employer's Affiliates, on the other side, or under which Employee is a participant. From and after the Effective Date, Employee shall be employed by Employer under the terms and pursuant to the conditions set forth in this Agreement.

23. <u>FCPA COMPLIANCE</u>. Employer advises Employee that the United States Foreign Corrupt Practices Act ("FCPA") prohibits offering, providing, or promising anything of value (including money, gifts, preferential treatment, and any other sort of advantage), either directly or indirectly, by a United States company, or any of its employees, subsidiaries, affiliates, or agents, to a Foreign Government Official for the purposes of influencing an act or decision in that individual's official capacity, or inducing the official to use his or her influence with the foreign government to assist the United States company, its subsidiaries or affiliates, or anyone else, in obtaining or retaining business or securing an improper advantage.

Employee understands that Employee may not directly or indirectly offer, promise, grant, or authorize the giving of money or anything else of value to a Foreign Government Official to influence official action, obtain or retain business, or secure an improper advantage. Employee understands that these legal restrictions apply fully to Employee with regard to Employee's activities in the course of or in relation to Employee's employment with Employer, regardless of Employee's physical location. Employee represents and warrants that Employee fully understands and will act in accordance with all applicable laws regarding anti-corruption, including the FCPA, the U.K. Bribery Act, and any other applicable state, federal, and international laws related to anti-corruption. Employee agrees that he or she will not take any action which would cause Employer to be in violation of the FCPA or any other applicable anti-corruption law, regulation, or Company policy or procedure. Employee further represents and warrants that Employee will know and understand, and act in accordance with, all Company policies and procedures related to anti-corruption and business conduct. Employee agrees to attend mandatory compliance training. Employee undertakes to duly notify Employer if Employee becomes aware of any such violation of Company policies or procedures, or any other violation of law, committed by Employee or any other person or entity, and to indemnify Employer for any losses, damages, fines, and/or penalties which Employee may suffer or incur arising out of or incidental to any such violation committed by Employee.

Employee also represents and warrants that Employee will disclose to the Employer if Employee or any member of Employee's family is an official of a foreign government or foreign political party or is a candidate for foreign political office.

In case of breach of this provision, the Employer may suspend or terminate this Agreement at any time without notice or indemnity.

24. <u>**REVIEW BY PARTIES AND THEIR LEGAL COUNSEL.</u>** The parties represent that they have read this Agreement and acknowledge that they have discussed its contents with their respective legal counsel or have been afforded the opportunity to avail themselves of the opportunity to the extent they each wished to do so.</u>

**Employee and Employer have read and understand that Section 21 (Dispute Resolution) of this Agreement contains provisions requiring the Employee, as well as the Employer, to submit certain covered disputes between Employee and Employer to arbitration. By signing below, Employee and Employer, specifically authorize and affirmatively agree to utilize the provisions of Section 21 of this Agreement.

WYNN RESORTS, LIMITED

<u>/s/ Ellen F. Whittemore</u> Ellen F. Whittemore Executive Vice President, General Counsel & Secretary

Employee

<u>/s/ Craig S. Billings</u> Craig S. Billings

IN WITNESS WHEREOF AND INTENDING TO BE LEGALLY BOUND THEREBY, the parties hereto have executed and delivered this Agreement as of the year and date first above written.

WYNN RESORTS, LIMITED

<u>/s/ Ellen F. Whittemore</u> Ellen F. Whittemore, Executive Vice President, General Counsel & Secretary <u>/s/ Craig S. Billings</u> Craig S. Billings

Employee

Certification of the Chief Executive Officer

Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002

I, Matt Maddox, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Wynn Resorts, Limited;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 9, 2021

/s/ Matt Maddox

Matt Maddox Director, Chief Executive Officer (Principal Executive Officer)

Certification of the Chief Executive Officer

Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002

I, Craig S. Billings, certify that:

- 1. I have reviewed this Annual Report on Form 10-K of Wynn Resorts, Limited;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 9, 2021

/s/ Craig S. Billings Craig S. Billings Chief Financial Officer (Principal Financial and Accounting Officer)

Certification of CEO and CFO Pursuant to

18 U.S.C. Section 1350, as Adopted Pursuant to

Section 906 of the Sarbanes-Oxley Act of 2002

In connection with the Quarterly Report on Form 10-Q of Wynn Resorts, Limited (the "Company") for the quarter ended September 30, 2021 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), Matt Maddox, as Chief Executive Officer of the Company, and Craig S. Billings, as Chief Financial Officer of the Company, each hereby certifies, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that, to the best of their knowledge:

- 1. The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- 2. The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

/s/ Matt Maddox Matt Maddox Name: Title: Director, Chief Executive Officer (Principal Executive Officer) November 9, 2021 Date: /s/ Craig S. Billings Name: Craig S. Billings Title: Chief Financial Officer (Principal Financial and Accounting Officer) Date: November 9, 2021

A signed original of this written statement required by Section 906, or other document authenticating, acknowledging, or otherwise adopting the signature that appears in typed form within the electronic version of this written statement required by Section 906, has been provided to Wynn Resorts, Limited and will be retained by Wynn Resorts, Limited and furnished to the Securities and Exchange Commission or its staff upon request.